

Lived experiences of space in secure mental healthcare environments

Katharine Elizabeth Harding

https://orcid.org/0009-0000-0432-9215

A thesis submitted in partial fulfilment of the requirements of London South Bank University for the degree of Doctor of Philosophy

August 2023

Acknowledgements

Firstly, my thanks go to everyone who participated in the research and generously shared their experiences, alongside all the people who kindly helped facilitate the project. Special thanks to my supervisors Professor Paula Reavey and Dr Zoë Boden-Stuart for your invaluable guidance, insight and encouragement throughout. Also to my family, friends and colleagues, with much appreciation for your kindness and support.

Abstract

Focusing on the lived experiences of patients¹ and staff within low and medium secure mental healthcare facilities in the UK, the research examines experiences of space through an exploration of occupants' affective experiences in these settings.

Whilst a body of research examining healthcare settings exists in the fields of health and environmental psychology, the literature commonly focuses on measurable effects concerning physiological and cognitive responses to environments, with limited focus on embodied experience and meaning making in relation to space. Research examining mental healthcare is also frequently approached from predominantly clinical perspectives and psychosocial research exploring patient perspectives and experiences of inpatient settings is more limited, particularly within secure mental healthcare facilities where occupants may typically be hard to reach.

In the context of limited empirical exploration in relation to everyday experiences in secure mental healthcare environments, the primary aim of the study is to expand understanding of how patients and staff make sense of their affective experiences within low and medium secure inpatient settings and the ways in which these experiences might be spatially mediated. Through investigating how patients and staff inhabit the organisational spaces of institutional care, the research also aims to explore how these institutions can constitute supposedly 'private' spaces within their own 'public' spaces.

The theoretical grounding to the research integrates a psychosocial process account of experience with a narrative-psychological approach to enable exploration of relationships between space and affective experience in secure mental healthcare

¹ The term 'patient' is used for technical accuracy in this study to describe participants detained in secure mental healthcare services under a section of the UK Mental Health Act, however, the difficulties associated with the use of this term are acknowledged and 'people who use services' or 'people who live with distress' may be used in other contexts (see Cromby et al., 2013).

services. The research objectives are supported by the chosen qualitative research methodology, which combines a narrative-based approach with visual research methods to access participants' stories about their experiences and the settings in which they take place.

Nineteen interviews in total were carried out with patients and staff from low and medium secure wards within a large mental healthcare facility in the UK. Semistructured interviewing was used alongside the photographs produced by participants to illustrate the spaces or objects that were relevant to their everyday experiences of the environment and narratives were constructed together with the researcher during the interview. Through this approach, the study explored how participants made sense of their affective experiences within secure mental healthcare environments using narratives that were spatially grounded by their photographs. Following a narrativebased approach to analysis, participants' stories were interpreted using a conceptual framework of narrative levels of analysis alongside narrative analytic tools to identify spatially situated storylines and significant aspects of the accounts.

Approaching the analysis through the transdisciplinary concept of liminality as a theoretical perspective enabled exploration of how tensions relating to the conflicting priorities and paradoxical functionality of secure settings as both places of carceral containment and therapeutic spaces were experienced through participants' interaction with the institutional environment. Ongoing relationships between space and experience were observed, alongside engagement with 'liminal affective technologies' (Stenner, 2017, 2021) by both patients and staff to manage their affective experiences and distress, including the discomfort of feeling 'stuck' in liminal situations. Similarities observed in patient and staff accounts also included experiences of 'public' spaces as typically uncomfortable and a wish for more 'homely' environments to assist in enabling positive social interaction.

Table of Contents

Preface	8
Chapter 1: Space and mental health	11
1.1 Asylum spaces and mental healthcare	11
1.2 Secure mental healthcare services	12
1.3 Environmental tensions in psychiatric settings	15
1.3.1 Balancing risk and rehabilitation	16
1.3.2 Privacy, choice and control	18
1.3.3 Hybrid environments	20
1.4 Therapeutic environments	22
1.4.1 Creating connections to nature	26
1.4.2 The sensory environment	29
1.4.3 Hospital soundscapes	32
1.5 Expert experience and mental healthcare research	34
Chapter 2: Psychosocial experience, process and space	37
2.1 Exploring experience through a psychosocial process approach	38
2.1.1 Experience and process	39
2.1.2 Transdisciplinarity	39
2.1.3 Relational process ontology	41
2.2 Experience and liminality	44
2.2.1 Rites of passage and liminal experience	44
2.2.2 Liminality and process	45
2.2.3 Affectivity and liminality	47
2.2.4 Imagination and liminal experience	51
2.2.5 Enduring liminality and health	52
2.2.6 Liminal space	54
2.3 A relational understanding of experience and space	55
2.3.1 Topology and life space	55
2.4 Concluding comments	58
Chapter 3: Methodology	60
3.1 Qualitative research and narrative psychology	60
3.1.1 Narratives and identity	62
3.1.2 Visual storytelling	63
3.2 Research methods	66
3.2.1 Ethics	67
3.2.2 Participants and recruitment	71
3.2.3 Data collection	74
3.2.4 Transcription	77
3.3 Narrative analysis	78
3.3.1 Narrative levels of analysis	79
3.3.2 Analysing and interpreting the data	83
Chapter 4: Negotiating uncomfortable spaces	94
4.1 Leon	96

4.2 Carl		109
4.3 Wendy		107
4.4 Concluding	g comments	133
	naging difficult experiences using objects and space	138
5.1 Tom	inging unitent experiences using objects and space	138
5.2 Ingrid		153
5.3 Patricia		164
5.4 Concluding	z comments	178
	rigating relationships in secure spaces	182
6.1 Mike		182
6.2 Nathan		194
6.3 Bradley		209
6.4 Estelle		216
6.5 Concluding	g comments	225
	cussion and conclusions	233
7.1 Summary of		234
7.2 Key insigh		236
• •	iscomfort of feeling 'stuck'	236
	ng comfort and becoming 'unstuck'	239
	rating worlds	243
7.3 Implication	ns	246
7.4 Reflections	s on the research process	251
7.4.1 Resea	rch in secure settings	253
7.5 Future dire	ections and concluding comments	258
References		266
Appendices		312
Appendix A:	Service user participant information sheet (page 1 of 2)	313
Appendix A:	Service user participant information sheet (page 2 of 2)	314
Appendix B:	Staff participant information sheet (page 1 of 2)	315
Appendix B:	Staff participant information sheet (page 2 of 2)	316
Appendix C:	Consent form	317
Appendix D:	Service user interview schedule (Study one - group one)	318
Appendix E:	Service user interview schedule (Study one - Group two)	319
Appendix F:	Staff interview schedule	320
Appendix G:	Service user participant debrief	321
Appendix H:	Staff participant debrief	322
Appendix I:	Transcription conventions	323
Appendix J:	Alice	324
Appendix K:	Darren	325
Appendix L:	Felicity	326
Appendix M:	Gavin	327
Appendix N:	Јо	328
Appendix O:	Karen	329

Appendix P:	Sal	330
Appendix Q:	Vicky	331
Appendix R:	Sample participant storyline summary	332

List of Tables

 Table 1
 Research participants

List of Figures

Figure 1	Environmental tensions in secure mental healthcare settings	21
Figure 2	Analytical approach using integrated levels of analysis	82
Figure 3	Summary of participants' main storylines	95
Figure 4	Leon's photograph of the bedroom window	99
Figure 5	Leon's photograph of the bathroom	106
Figure 6	Wendy's photograph of door signage on the ward	131
Figure 7	Tom's photograph from a window on the ward towards the grounds	143
Figure 8	Tom's photograph of keys and radio on staff member's belt	147
Figure 9	Tom's photograph of the airlock traffic light system	148
Figure 1	0 Ingrid's photograph of the ward courtyard taken on a snowy day	159
Figure 1	${f 1}$ Nathan's photograph of the communal lounge and dining area	198
Figure 1	2 Nathan's photograph of the side room	206
Figure 1	${f 3}$ Estelle's photograph of the servery hatch (with the roller shutter down)	220
Figure 1	4 Estelle's photograph of the kitchen servery equipment	220
Figure 1	5 Estelle's photograph of the clinic room on the ward	223

Preface

By instinctively drawing on personal human experience to inform how spaces might be shaped or understood, designers and architects can be seen to presuppose associations between environmental conditions and the experiential qualities of people's everyday lives. My own early appreciation of the built environment, materiality and the sensorial aspects of everyday interactions with spaces and objects evolved through a career in design. Later, alongside my ongoing work as a designer within an architecture and interior design practice, a longstanding interest in human experience, space and mental health developed into a decision to undertake a postgraduate conversion course in psychology. These studies at London South Bank University included inspiring teaching in the field of mental health and distress from scholars whose research expertise and interests include exploring associations between psychosocial processes, relational experience and environments.

Building on an interest in the relationships between space and mental health, a small empirical study undertaken in my final year was concerned with exploring women's affective experiences in domestic environments, with a focus on experiences of low mood and depression. Participants' photographs of home settings grounded their stories in particular spatial contexts and revealed how these everyday domestic spaces or the objects inside them were closely intertwined with their accounts of embodied and psychosocial experiences. Alongside studies in psychology and my continuing work as a designer, a voluntary placement undertaken in a high-support housing setting for people using forensic mental health services in the community contributed to furthering this interest in exploring relationships between residential spaces and mental health.

The current research was accordingly developed with the aim of exploring occupants' lived experiences in the very particular context of forensic mental healthcare settings, where 'home' spaces are typically closely monitored and highly

restrictive environments that are co-inhabited by members of staff. In this context the research aims to investigate the shaping of patients' perceptions of private space by institutional care and explore how co-managing a relationship to 'home' spaces may be implicated in the joint constitution of 'recovery' as a personal and organisational goal.

A body of research examining occupants' experiences of healthcare spaces exists in the fields of environmental and health psychology, however, the literature is typically focused on measurable effects in terms of occupants' physiological and psychological responses to the environment. Whilst research examining the perspectives of people using services and the experiential, psychosocial or symbolic² aspects of healthcare spaces is more limited, particularly within forensic mental health, research exploring the experiences of patients and staff in relation to the environment has been undertaken in secure psychiatric settings (see Donald et al., 2015; McGrath, Brown, et al., 2021; Olausson et al., 2019, 2021; Reavey et al., 2019). Building on this and focusing on the lived experiences of patients and staff, the current research explores how both groups experience low and medium secure mental healthcare environments and make sense of their affective experiences in these settings. The thesis is organised around the following chapters:

Chapter One outlines the scope and focus of the research and provides an empirical background to the project through a review of the literature examining associations between space and mental health, with a focus on healthcare settings. The potentially therapeutic or untherapeutic role of environments in healthcare and psychiatric services specifically is explored.

Chapter Two presents the theoretical grounding of the thesis within an embodied psychosocial approach, through which a relational understanding of space,

² The words 'symbolic' or 'symbolism' occur throughout this thesis in relation to meaning making and the significance of lived experiences and not in reference to psychoanalytical understandings of these terms.

experience and mental health is explored, with particular attention to affect, subjectivity and the concept of liminality.

Chapter Three provides a rationale for the methodological and analytical approaches chosen for the research. The empirical study is outlined to include accounts of the ethical considerations and processes involved, alongside descriptions of the data collection methods which comprised visual methods and semi-structured interviews. The data set was analysed using a narrative-based approach and the analytical process is also described.

Chapter Four is the first analytical chapter and proposes that participants' accounts of negotiating uncomfortable spaces in the hospital environment can be understood in terms of the experience of being caught in an enduring transitional situation, or 'liminal hotspot'. The discomfort experienced by staff and patients is elaborated in relation to the ongoing indeterminacy of being suspended in liminal circumstances between contrasting worlds.

Chapter Five explores how patients and staff engage with spaces and objects to mediate discomfort or distress experienced within the everyday environment. Drawing further on the theoretical perspective of liminality, participants' processes of managing difficult experiences are explored in relation to the self-production of liminal affective experiences that can enable transition between affective states.

Chapter Six examines the role of space and objects in shaping relational dynamics between and amongst patients and staff. This involves exploring how interpersonal relationships are spatially distributed in the environment and examining the borders that material aspects of the hospital can be seen to create between contrasting worlds of patient and staff activity.

Chapter Seven contains a discussion of the overall findings of the research, including reflections on the research process. The key insights and implications of the study and considerations for future research are also discussed.

Chapter 1: Space and mental health

This introductory chapter will initially situate secure psychiatric environments within an overall context of inpatient mental healthcare and provide a background orientation to secure services in the UK. An examination of the literature exploring patient and staff experiences of the environment, atmosphere and social milieu in psychiatric inpatient settings will be drawn on to evidence an argument that a series of inherent tensions are at play in these spaces. This includes an exploration of the contrasting operational objectives and priorities within psychiatric services, such as the requirement to balance risk management with the provision of a therapeutic environment. The associated contradictions that have been identified in mental healthcare spaces and secure psychiatric settings specifically are understood to be mediating factors in the everyday lived experiences of patients and staff.

Taking account of these contradictions, the chapter will turn to the literature concerned with the potentially therapeutic role of the environment itself within general and mental healthcare settings. Theoretical understandings of how aspects of the environment might contribute to the perceived therapeutic value of healthcare spaces will be examined. Alongside this, the chapter will explore the literature studying the use of theoretical frameworks and evidence-based design to optimise the therapeutic potential of spaces through environmental interventions. This includes research examining the role that exposure to aspects of nature and multi-sensorial experiences might play in mediating health and well-being in healthcare environments.

1.1 Asylum spaces and mental healthcare

Mental healthcare services in the UK can be traced back to the foundation of the monastic Priory of St Mary of Bethlehem in London in 1247 (Killaspy, 2006; Sendula-Jengić et al., 2011). The Bethlem Hospital, known also as 'Bedlam', which was established at the priory in 1330 and later evolved in a series of subsequent buildings, is recognised as being the first psychiatric hospital in Europe (Sendula-Jengić et al., 2011). Asylums were systematically developed in the UK during the 19th century and The County Asylums Act of 1808 permitted the establishment of public asylums to provide care for pauper or criminal 'lunatics' outside the workhouse or prison (Unsworth, 1993). Following the Lunacy Act of 1845, the construction of county asylums became compulsory and the numbers of people detained within asylums proliferated. Such institutions formed the primary context for mental healthcare treatment until the second half of the 20th century when the shift towards de-institutionalisation and community care brought about the closure of most large psychiatric hospitals in the late 1980s and early '90s (T. Turner, 2004).

1.2 Secure mental healthcare services

This study focuses on the contemporary context of medium and low secure mental healthcare inpatient facilities in the UK³, as environments in which patients with mental health difficulties may be detained for assessment or treatment under a section of the Mental Health Act. Commonly described as 'forensic'⁴ wards or units, secure services exist within the 'post-asylum' landscape of de-institutionalised mental healthcare as a remaining form of institutional inpatient setting in which patients may be detained, often for significant periods of time (Hare Duke et al., 2018; Holley et al., 2020). Patients are typically detained due to the potential risk that they may present to themselves or others and depending on perceived levels of risk, patients may be

³ Undertaking a second parallel study to explore the lived experiences of staff and people using forensic mental health services within supported accommodation in the community was also planned and pursued originally. Due to protracted approval processes, however, participant recruitment and data collection were not progressed in community settings as this was no longer deemed practicable within the overall project programme.

⁴ Whilst the term 'forensic' is commonly used to describe secure mental healthcare services and users of these services, not all people detained in 'forensic' wards have had offended or had contact with the criminal justice system. The settings where this research was undertaken are generally described here as 'secure wards/facilities/units', or 'secure services'.

admitted to inpatient services in healthcare settings that are categorised as high, medium or low secure environments (Kennedy, 2002).

Secure services intersect the mental healthcare and criminal justice systems and work predominantly with people detained on what are commonly referred to as 'forensic sections' under Part III of the Mental Health Act. These patients have had involvement with the criminal justice system and might have been transferred to hospital direct from prison, or may be subject to a hospital order imposed by the courts (Rutherford & Duggan, 2008). Whilst the majority of patients are detained in secure environments under forensic sections (Coid, Kahtan, Gault, Cook, & Jarman, 2001), patients who have had no involvement with the criminal justice system may also be detained in secure services on what are typically known as 'civil sections' under Part II of the Mental Health Act (Galappathie et al., 2017). Individuals who are receiving care in secure environments under civil sections are typically there due to concern about significant risk that they may pose to themselves.

Amidst widespread de-institutionalisation of general mental healthcare provision (Thornicroft & Bebbington, 1989; T. Turner, 2004), secure units exist amongst a reduced number of institutional inpatient psychiatric environments (Shen & Snowden, 2014). Whilst the number of beds in general psychiatric settings diminished as a result of community-based care, capacity in secure services has increased significantly in recent decades (Chow & Priebe, 2016; Jansman-Hart et al., 2011; O'Donahoo & Simmonds, 2016; T. Turner, 2004) and there is increasing need for new secure facilities (Mclaughlan et al., 2021). This process of 're-institutionalisation' and growing demand for secure services can be seen to reflect unintended consequences of de-institutionalisation, including increased visibility within the community of problematic behaviour expressed by people with serious mental health difficulties and the use of the criminal justice system to manage this (Jansman-Hart et al., 2011; O'Donahoo & Simmonds, 2016; T. Turner, 2004). Alongside this, there has been a

decrease in sites of community mental health service support, such as day centres, that may help to prevent more serious problems from arising (McGrath & Reavey, 2016; Pilgrim & Ramon, 2009).

Patients may be detained in secure facilities for indefinite periods of time and whilst length of stay may typically be between 2 to 5 years, more than a quarter of people in high or medium secure settings may spend upwards of 10 years within secure services (Rutherford & Duggan, 2008; Shah et al., 2011). During their pathway through secure services patients may typically transition between levels of security or between units in different geographical locations. Facilities are commonly located at significant distances from a patient's family, home or community (Durcan et al., 2011) and this physical distance can in turn contribute to patients' perceptions of disconnection from social networks (Coffey, 2012a).

Movement within secure facilities is restricted and to leave the ward or visit places including the hospital grounds or wider community, patients require permissions from a Responsible Clinician and where relevant, the Ministry of Justice. Subject to levels of risk, patients may require staff escorts when accessing spaces off the ward and closed-circuit television (CCTV) is generally used throughout the environment. Staff, patients or visitors enter and exit the facilities via airlock door systems and posters displayed in reception highlight stringent rules about items that are banned or restricted within the units. Visitors may be provided with a personal alarm for use inside the facility and any prohibited personal possessions, including mobile phones, must typically be stored in lockers before entering.

By their nature secure psychiatric services typically constitute isolated and highly restrictive environments where visiting is regulated and patients' contact with the wider world, including internet access is limited (Tomlin et al., 2020). Because patients may be involuntarily detained in secure services for prolonged periods of time, such restrictions can also be in place for many years (Hare Duke et al., 2018). Secure

psychiatric settings can consequently be understood to be less 'permeable' or connected to the world outside hospital than other contemporary forms of inpatient mental healthcare environment, where patients may be admitted voluntarily and length of stay may typically be shorter (Quirk et al., 2006).

1.3 Environmental tensions in psychiatric settings

Whilst length of stay may mean that inpatient mental healthcare units represent 'home' for some patients, these residential spaces also comprise workplace environments that are continuously co-inhabited by staff. Environmental or atmospheric tensions may then exist in psychiatric wards, where both workplace and residential functions are combined in a hybrid setting that can be seen to represent both 'hospital' and 'home'.

Studies exploring patient and staff views about these shared spaces have highlighted how the ward environment and atmosphere can be experienced in different ways by both groups (Brunt & Rask, 2005; Rossberg & Friis, 2004; Shattell et al., 2008). For example, Rossberg and Friis' (2004) examination of staff and patient assessments of a psychiatric facility found that perceptions of the ward atmosphere were more important in relation to satisfaction with the ward for patients than for staff. In addition, whilst the working environment was strongly linked to the satisfaction of staff members, it appeared to be unrelated to patient satisfaction. Rossberg and Friis (2004) also point out how disparities in the everyday experiences of both groups can give, "[...] the impression that patients and staff live in "different worlds," even though they share the same physical and social environment" (p.798).

Research examining staff and patient viewpoints has also exposed ways in which the priorities and expectations of both groups in relation to inpatient psychiatric settings may overlap or differ on aspects of the environment including the therapeutic milieu (Livingston et al., 2012; Shattell et al., 2008) and the ward design (Csipke et al., 2016; Curtis et al., 2007). Shattell and colleagues' (2008) exploration of the therapeutic milieu in an acute psychiatric setting found that although staff and patients reported

parallel perceptions of feeling constricted within a prison-like environment, they had greatly contrasting experiences regarding the passing of time. Within this environment, time was felt to be rushing by for the busy staff, yet it stood still for patients, who were typically bored and also reported that not having enough to do to occupy their minds promoted rumination and a sense of anxiety. Such differences in staff and patient experiences within the same spatial context have also been described by researchers as a "split milieu" (Nicholls et al., 2015 p.292).

In a post-occupancy study of a new build psychiatric facility, Curtis et al. (2007) highlight how an ongoing discussion about the potential inclusion of an aquarium on the ward provides an example of the disparity between patient and staff perspectives in relation to the built environment. Whilst it was staff members' opinion that the aquarium glass would pose a safety risk in the environment, patients disagreed with this view. This debate also highlights a fundamental tension regarding the design of psychiatric wards, which will be elaborated in the following section, whereby the creation of therapeutic or home-like spaces to facilitate 'recovery' or rehabilitation must be balanced with predominant concerns about safety, security and risk management (Curtis et al., 2013).

1.3.1 Balancing risk and rehabilitation

Despite growing aspiration for person-centred approaches to care in mental healthcare settings, existing design models for psychiatric environments typically focus on risk reduction protocols and creating spaces that enable efficient patient management and observation by staff (Golembiewski, 2015). The inherent environmental tension between managing risk and creating therapeutic spaces is especially pronounced in secure psychiatric settings that are situated between the criminal justice and mental healthcare systems, whilst not being fully located in either (McGrath, Brown, et al., 2021). Livingston et al. (2012) highlight how, from a healthcare perspective, people detained in secure facilities are 'patients', for whom the system is there to provide care and support in a process of rehabilitation. Yet, conversely, for those involved in the criminal justice system, from this perspective they are also regarded as 'offenders' for whom the system and environment are concerned with containment in order to reduce the risk of recidivism or control potentially dangerous behaviour.

Paradoxically, secure psychiatric services are thus simultaneously required to provide robust levels of security in a high-risk setting to maintain the safety of patients and the wider community, whilst also creating a therapeutic context within the least restrictive environment possible (Seppänen et al., 2018; Shepley et al., 2016). Reflecting the challenges presented by this paradox, Curtis et al. (2013) point out how the Department of Health (2011a) design guidelines stress that the balance in secure psychiatric settings is, "[...] between maintaining the safety and security of patients and staff, providing an effective, beneficial therapeutic environment, and protecting the public" (p.7), yet provide limited guidance on how to achieve this. As a consequence of these tensions and the predominant focus on risk mitigation, studies examining patient and staff perspectives have found the environmental attributes of psychiatric inpatient spaces to be commonly perceived as non-homely or clinical, with limited therapeutic quality (Csipke et al., 2016; Donald et al., 2015).

Csipke et al. (2016) employed participatory methods in a psychiatric ward to explore staff and patient views about the physical environment and created a questionnaire with patient-generated measures to assess the ward design. In addition, patient participants were invited to take two photographs to capture their perspectives on the best and worst physical aspects of the environment. Both groups described the overall environment as being "institutional" and "bland" and participants' accounts suggested that a visual "brightening up" of the ward environment, including the integration of artwork, would help improve perceptions of well-being (p.118). Patients' appraisals of the therapeutic value of a psychiatric inpatient unit examined in research by Donald et al. (2015), similarly included perceptions of the environment being,

"sterile" and requiring a greater sense of comfort, or "confused" and "weird" in terms of lacking a clear spatial identity (p.65). The authors sum up the awkwardness of the indeterminate and paradoxically 'doubled' environmental characteristics by highlighting how participants found the ward to be, "[...] somewhere between a hospital and a home, without being either" (p.65).

1.3.2 Privacy, choice and control

As environments designed to accommodate detained patients in a residential setting, yet simultaneously comprising a workplace context for non-resident staff, psychiatric wards constitute hybrid spaces with both 'private' and 'public' operational objectives and priorities. Within these settings and especially so in the highly restrictive and monitored spaces of secure mental healthcare services, patients may commonly experience a loss of autonomy or control and an associated loss of privacy (Koller & Hantikainen, 2002). Patients' loss of privacy can be seen to relate to environmental experiences in secure psychiatric settings, including sharing a residential space with many other people and being continually observed by staff. Due to primary operational concerns about safety and risk management, patients are closely monitored in communal areas of psychiatric wards and are routinely checked during the day and night in the more 'private' spaces of their bedrooms. In instances where there is significant concern about risk, patients may also be observed by staff when using the bathroom. CCTV cameras are used throughout the hospital environment and provide a technical form of surveillance over the behaviour of both patients and staff members (Curtis et al., 2013).

Perceptions of privacy in psychiatric settings are also linked to spatial or social density and comparisons between contemporary mental healthcare facilities and historic asylum buildings suggest that patients may be confined in modern wards that occupy around 70% less space than their Victorian equivalents (Johnstone, 2004). Correspondingly, issues concerning crowding and privacy are prevalent themes across

the literature in studies exploring relationships between environmental design and the experience of mental healthcare spaces (Connellan et al., 2013; Kumar & Ng, 2001). In a review of the literature concerned with violence and crowding in psychiatric settings, Kumar and Ng (2001) highlight how density, privacy and control are three interrelated variables that can contribute to crowding on wards being a risk factor for safety issues or aggression. The authors argue that social density can increase stress and frustration by reducing patients' sense of privacy and control over their immediate environment. Accordingly, crowding on an acute psychiatric unit, as studied by Ng et al. (2001), was found to be significantly associated with the occurrence of aggression, particularly incidents of verbal aggression. Aggression related to crowding stress in psychiatric wards is hence understood to reflect attributes of the physical environment that restrict patients' access to privacy and ability to regulate imposed interaction with others, or avoid environmental stressors, including noise (Lundin, 2021; Ulrich et al., 2018).

Congruently, perceiving a lack of control over the environment in hospitals has been found to promote anxiety or stress responses in patients and a key principle in Ulrich's (1991) theory of supportive design centres on enabling patients to experience a sense of control over the physical-social surroundings. Choice, control and tailored services for hospital patients are also promoted in the personalisation agenda set out within the UK health system (Department of Health, 2011b). However, research findings suggest that opportunities for patients to exert control over their everyday environments in mental healthcare settings are typically very limited (Lawson et al., 2003; Papoulias et al., 2014). Karlin and Zeiss (2006) highlight how perceptions of choice and control are entwined with social experiences in psychiatric wards and argue that design interventions, including zoned seating in communal spaces, can offer choice and flexibility to allow patients greater control over levels of social contact.

Privacy is also significant to how staff, patients and visitors manage interpersonal relationships within psychiatric wards, including patients' romantic or intimate relationships with partners, or other patients and expressions of sexuality that may take place alone. For reasons including lack of privacy, patients have limited opportunity for sexual expression within inpatient psychiatric settings (McCann, 2010), yet it is acknowledged that sexual activity takes place, despite policies prohibiting sex commonly being in place on psychiatric wards (Warner et al., 2004). Facilities do not allow for patients to have intimate relationships with a partner or other patients and there is a lack of formal policy concerning patient sexual relationships across UK secure services (Dein & Williams, 2008).

Many complex considerations around sexual expression and relationships, including patients' health, safety, ability to consent, or offending behaviour where relevant, present dilemmas to providers of secure services and staff are often reticent about discussing sexuality with patients (S. D. Brown et al., 2014). Whilst acknowledging the concerns and challenges associated with facilitating patient relationships in secure services, staff and patients also recognise how closeness, companionship or intimacy might be beneficial and help to promote patient health and well-being in these environments (Quinn & Happell, 2015a). Accordingly, accounts of patient and staff perspectives towards patients' experiences within secure services suggest that both groups recognise a need to provide private and dignified spaces for sexual activity (Quinn & Happell, 2015b). However, the complex issues around patient sexuality are commonly neglected in the guidelines or policies within psychiatric institutions and further research on patient sexual behaviour is required to assist with clear and inclusive policy making (Anex et al., 2022; Reavey et al., 2022).

1.3.3 Hybrid environments

The environmental tensions that have been explored thus far reflect some of the contrasting and typically conflicting operational objectives and priorities around which

mental healthcare and secure psychiatric services in particular are organised. As indicated diagrammatically in Figure 1, it has been argued that these dualities present a sense of paradox that brings 'both/and' and 'neither/nor' qualities to the physical and psychosocial environment that can mediate the everyday experiences of patients and staff within the spaces of secure psychiatric care.

Figure 1

Environmental tensions in secure mental healthcare settings



Secure hospital spaces function as busy workplaces for staff members and are also accessed by external visitors, yet these same spaces simultaneously constitute residential settings where patients live out the day-to-day experiences of domestic life. As custodial environments in which patients are involuntarily detained, secure psychiatric spaces have dual functionality as highly restrictive places of containment, yet also existing to provide supportive, caring environments in which to facilitate processes of rehabilitation and community re-integration for patients. Resulting from these hybrid attributes, secure psychiatric services occupy an indeterminate zone between different functional priorities, in spaces that can be seen to be fully fulfilling neither one set of priorities, nor the other. In light of these tensions, the following sections explore the literature that is concerned with examining how the therapeutic potential of the environment might be optimised through design within healthcare contexts. This includes reference to the increasing awareness of the role that exposure to natural environments and multisensorial experiences can play in mediating health and well-being.

1.4 Therapeutic environments

Before exploring contemporary thinking about therapeutic environments in mental healthcare, it is helpful to return to the historical asylum context and specifically, the York Retreat, a small private asylum where the 'moral treatment' approach was pioneered at the end of the 18th century. In a turn against prevailing punitive attitudes in traditional places of confinement, the moral treatment philosophy was based on compassionate principles and aimed to afford comfort and dignified, humanitarian care in a carefully designed environment (Edginton, 1997, 2003). The physical context and design of the built environment was accordingly a key concern and the concept that an institution in itself could possess therapeutic potential was integral to the approach (du Plessis, 2012). Examples of how consideration of the therapeutic environment was built into the architectural design of the Retreat include the discreet concealment of windows bars within the window frames and door locks encased in leather to muffle the disconcerting sound of bolts (Fennelly, 2014; Laffey, 2003). Moral treatment principles had a widespread influence on 19th century asylum architecture, alongside the external landscape design, in which the active role of the natural environment in treatment was prioritised (du Plessis, 2012; Edginton, 1997, 2003; Hickman, 2009, 2014).

From the second half of the 20th century, the 'post-asylum' era saw a reduced focus on the therapeutic role of the physical environment in mental healthcare, with significant contributing factors including the development of psychopharmacology, deinstitutionalisation and the introduction of community care (T. Turner, 2004). Recent

decades have, however, seen renewed interest in and increasing recognition of the potentially therapeutic role of the environment within healthcare contexts. The notion of 'therapeutic landscapes' (Gesler, 1992, 2005) provides a conceptual framework for examining how the physical, social, or symbolic attributes of environments might interact and contribute to experiences of physical and mental health and well-being in different places (Curtis et al., 2007). Applying this perspective to healthcare environment design, Gesler and colleagues (2004) thus argue that, "the therapeutic value of hospitals is related to their physical, social and symbolic design" (p.117). Correspondingly, research examining the therapeutic value of inpatient psychiatric environments suggests that perceptions of the physical, atmospheric, social, cultural and symbolic aspects of these spaces are intertwined and can be equally significant to staff and patient experiences (Connellan et al., 2013; Curtis et al., 2013; Donald et al., 2015; McGrath & Reavey, 2019). It can hence be argued that assessments of how mental healthcare spaces might be considered to be healing should consider not only physical attributes, but also social and spatial affordances of environments, such as the extent of free movement or social interaction that occupants may be offered (Simonsen & Duff, 2020).

Drawing on the combination of psychosocial and spatial qualities that may contribute to positive environmental experiences, healthcare organisations have also generated approaches towards developing therapeutic services and environments. Stichler (2008) outlines the relationship-based philosophy of the international, nonprofit organisation, Planetree, which aims to create therapeutic healthcare spaces based on nine key principles: "human interaction; consumer and patient education; healing partnerships with patients' family and friends; food and nutritional nurturance; spirituality; human touch; healing arts and visual therapy; integration of complementary therapies; healing environments created in the architecture and design of the healthcare setting" (p. 506). Staff culture and attitudes are highlighted as being

integral to the organisation's relationship-based approach and having an association with both user and staff satisfaction.

The 'Enabling Environments' (EE) initiative developed by the UK Royal College of Psychiatrists similarly draws on relational factors that are understood to contribute to positive psychosocial environments and centres around ten core values: belonging, boundaries, communication, development, involvement, safety, structure, empowerment, leadership and openness (Royal College of Psychiatrists, 2019). The EE approach is applicable to a range of different sectors, including healthcare and criminal justice environments and offers a set of standards and criteria to provide guidance on how the 'enabling' values might be achieved and implemented in practice. As an offshoot of EE, the associated concept of 'Psychologically-informed Environments' (PIEs) forms a flexible framework to promote shared reflective practice within services and encourage tailored, therapeutic environments informed by psychosocial perspectives (R. Johnson & Haigh, 2011). The parallel concept of 'Psychologically-informed Planned Environments' (PIPEs) draws on similar principles to PIEs, whilst also acknowledging the robust security considerations required for approaches in high-risk settings, including prisons and secure environments (Haigh et al., 2012).

Researchers examining the geography of health have drawn on the notion of therapeutic landscapes to explore relationships between the characteristics of mental healthcare environments and the experiences or well-being of occupants (Curtis et al., 2007; Gesler et al., 2004; Wood et al., 2013b). Curtis et al. (2007) undertook a postoccupancy evaluation of a new psychiatric facility based on interviews and discussion groups with staff and former patients. Whilst participants made a range of positive and negative appraisals of physical aspects of the environment, including air quality, light, materiality and food quality, the symbolic and social characteristics of the environment were referred to with equal frequency. Participants' reflections about social features included issues of privacy and the need to empower patients to have greater control

over decisions regarding their treatment or environment. Accounts of the symbolic environment included references to specific characteristics, such as the hospital's location on a waste site adjacent to a busy road and the tall fencing around the secure unit which was perceived to have prison-like associations.

Similarly, Payne and May (2009) explored the relationships between the environment and occupants' perceptions of well-being in an evaluation of a psychiatric intensive care unit refurbishment undertaken as part of the 'Enhancing the Healing Environment' initiative in the UK supported by the King's Fund (Department of Health, 2008). Patients' perceptions of homeliness in the new ward were associated with a combination of positive physical, social and symbolic features, including the overall quality and cleanliness of the environment compared with the original ward, private spaces for visitors, comfortable furniture, natural light, openable windows, high quality food and staff attitudes. Staff described the new environment as being calmer than the original ward, in addition to having a greater sense of light and feeling more open. The average length of stay for patients reduced by 20% following the refurbishment and a significant reduction in physical assaults on staff and other patients was also reported. Such findings can be seen to offer support for the argument that the environmental context in itself can be considered a key health intervention in mental healthcare settings (Golembiewski, 2013).

In the specific context of secure psychiatric services, Lawson et al. (2003) compared an existing facility with a new build medium secure mental health unit to explore the relationships between patient behaviour or health outcomes and the environmental conditions. Rates of verbal or physical aggression remained the same in both sites, however, in the new facility the severity of incidents reduced, alongside a two thirds reduction in patient self-harm. Seclusion use reported in the new unit also reduced by 70% and there was a 14% reduction in patient length of stay. Patient participants' comments suggested that greater provision of tactility and texture in the

material qualities of the hospital and variation in lighting and use of colour would afford an increased sense of homeliness in the overall environment. As Lawson and colleagues (2003) point out, these suggestions are situated in direct contrast to the conventional appearance of hospital wards, where smooth, bland or clinical finishes and bright, uniform lighting are typical.

Overall, research specifically examining relationships between the design of psychiatric wards and patient outcomes or experiences is limited (Liddicoat et al., 2020; Papoulias et al., 2014). In general healthcare, however, an increasing body of literature has examined ways in which environmental design based on research evidence may influence the impact of healthcare provision, including patient outcomes, length of stay and quality of experience (Devlin & Andrade, 2017; Devlin & Arneill, 2003; Iyendo, 2016; Reavey, Harding, et al., 2017; Ulrich et al., 2008). The therapeutic value of exposure to nature is frequently cited within this literature (Reavey & Harding, 2019; Ulrich et al., 2006, 2008) and is pertinent to the current exploration of secure psychiatric spaces, where patients typically have limited access to natural spaces.

1.4.1 Creating connections to nature

The associations between natural environments and mental health are long established and research findings suggest that exposure to aspects of nature can afford restorative and protective health benefits by mediating psychological and physiological stress (Berto, 2014). From an evolutionary perspective, the principles of Ulrich's (1983) Stress Recovery Theory (SRT) assume an inherent human predisposition towards natural environments. Accordingly, within the SRT framework, Ulrich argues that nonthreatening natural environments can produce positive feelings and enhance perceptions of well-being by reducing stress and physiological arousal. Similarly, the biophilia hypothesis proposed by Wilson (1984) contends that humans have an innate need for contact with nature and other forms of life. Biophilic principles thus presuppose associations between human affinity with nature and an evolutionary

dependence on the natural environment to provide essential resources for survival, such as sunlight, shelter, food and water (Heerwagen, 2009).

Amidst the context of rising global urbanisation, the amount of time that people typically spend in natural environments is reducing (W. R. Turner et al., 2004). Whilst decreased contact with nature may be associated with increased mental health difficulties (Bratman et al., 2012), research findings also suggest that access to natural or environments or 'green space' can help promote psychological well-being (Ulrich et al., 2008; Ward Thompson, 2011; Ward Thompson et al., 2012). The therapeutic potential of exposure to water has also been explored in the emerging area of research concerned with experiences of 'blue space' (Foley & Kistemann, 2015; Gascon et al., 2017; Völker & Kistemann, 2011). Exposure to bodies of water has been linked to lower levels of psychological distress (Nutsford et al., 2016) and the presence of water has been found to enhance the positive influence of natural spaces on mood and levels of self-esteem for people with lived experience of mental distress (Barton & Pretty, 2010). Natural landscapes, are accordingly argued to be vital public health resources that can provide accessible, affordable and effective means of health promotion in general society (Maller et al., 2006).

Healthcare environments are commonly felt to be detached from nature and a body of literature has examined how exposure to aspects of nature can be restorative or promote patient, staff and visitor well-being within general hospital settings (Iyendo et al., 2016; Joseph, 2006; Ulrich et al., 2008). Ulrich's (1984) influential study found that hospital window views containing natural scenery were associated with reductions in patient post-operative recovery time and use of pain-killing medication, when compared to hospital rooms with windows overlooking a brick building. Views of nature have also been associated with health benefits including stress reduction (Ulrich et al., 2006) and an increased sense of connection with life beyond hospital for inpatients (Douglas & Douglas, 2005; Lawson et al., 2003).

Studies have also examined the influence of exposure to natural light in healthcare settings and in comparison to dull rooms, bedrooms with natural sunlight have been found to reduce length of stay in psychiatric hospital settings for patients experiencing low mood (Beauchemin & Hays, 1996; Benedetti et al., 2001). Similarly, Lewy et al. (1998) found that the effect of exposure to morning bright light treatment was at least twice as effective as evening light in reducing measures of depression for people with a diagnosis of seasonal affective disorder. Correspondingly, the design recommendations proposed by Joseph (2006) drawn from research examining associations between light quality and health outcomes in hospital settings include orienting windows to optimise exposure to early morning sunlight, alongside making provision to facilitate temperature and glare control in patient bedrooms.

Although research examining the use of interior finishes in healthcare environments is limited, an experimental study by Zhang et al. (2017) compared participants' experiences of inhabiting rooms with plain white walls, to rooms with varying extents of timber panelling. Measures of participants' mean blood pressure and heart rate variability levels were lower in rooms with timber panels, suggesting that less stress and tension were experienced in these spaces where timber was visible. Similarly, Nyrud et al. (2014) compared hospital staff views towards images of the same patient room depicted with either no timber in the design, or differing amounts of timber finishes on the floor, walls and ceiling. The rooms at both ends of the 'no timber' to 'fully timber' continuum were the least preferred, while the most favoured design contained an intermediate amount of timber, applied only to the floor, loose furniture and a single feature wall. These findings can be seen to be supportive of arguments for facilitating exposure to potentially stress-reducing and restorative natural elements or materials within mental healthcare spaces, which in turn might contribute to reduced aggression by helping to minimise environmental stress (Ulrich et al., 2012).

In the context of designing mental healthcare environments, Shepley et al. (2016) draw attention to the significance of providing access to therapeutic outdoor spaces with multiple functions, such as vegetable gardens, sports and recreation facilities, as key environmental features that may benefit patients and staff. Likewise, Dvoskin et al. (2002) outline the approach to designing a new build secure mental health facility and recommend providing patients with freely accessible outdoor space. The day areas that were designed in the project are accessible to patients at any time and directly connected to outdoor spaces to form indoor-outdoor day rooms that are fully visible to staff from indoors. Studies have also examined the therapeutic value of active physical engagement with nature and a review of research evaluating gardeningbased interventions in mental healthcare found that positive benefits were associated with interventions in all the studies reviewed, including significant reductions in symptoms of depression and anxiety (Clatworthy et al., 2013).

1.4.2 The sensory environment

Recent years have also seen increasing interest in trauma-informed approaches to care and design, including the use of sensory environments in psychiatric settings (LeBel et al., 2010; Procter et al., 2017). Trauma-informed practice is grounded in an awareness of trauma and lived experience and seeks to actively empower patients in approaches to care and avoid re-traumatisation through a set of guiding principles that prioritise safety, trust, choice, collaboration, empowerment and cultural consideration (Office for Health Improvement and Disparities, 2022; Procter et al., 2017). Difficult or traumatic experience can have lasting neurological, biological, psychological and social effects (LeBel et al., 2010; Procter et al., 2017; van der Kolk, 2015) and exposure to adverse life experiences or trauma is common amongst people receiving treatment for severe mental health problems (LeBel et al., 2010; Rosenberg et al., 2001). Research also suggests that people experiencing mental distress may be hyper or hypo sensitive to sensory stimuli including noise, touch, light and vestibular input relating to movement

and balance (Abernethy, 2010; LeBel et al., 2010; Lloyd et al., 2014; Sutton & Nicholson, 2011). Whilst sensory processing difficulties may contribute to symptoms of distress, including anxiety or the negative symptoms experienced by some people with a diagnosis of schizophrenia, problems in regulating sensory input are understood to be a corollary of mental health problems (Sutton & Nicholson, 2011) and traumatic experience (van der Kolk, 2015). Sutton and Nicholson (2011) also highlight how sensory modulation issues are intensified by the response of the sympathetic nervous system which is frequently overly reactive in people experiencing acute distress.

Sensory rooms in healthcare settings are environments that can be tailored to suit individual needs and typically include therapeutic elements to engage different senses, such as optic lamps, scenic pictures, weighted blankets, music and aromas that are used to promote sensory modulation and emotional regulation or soothing (Champagne & Stromberg, 2004; Costa et al., 2006; Sutton et al., 2013). Within an acute mental healthcare environment, Sutton and Nicholson (2011) found that a variety of multi-sensory stimuli were felt to be effective in reducing distress and promoting calm for patients with a broad range of clinical diagnoses and lived experiences, including anxiety, voice hearing and paranoia. In the same study, sensory environments were also typically perceived to afford safe, quiet spaces away from ward activity and sensory approaches enabled patients to develop tools to self-manage distress and reduce the need for *pro re nata* (PRN) treatment (medication which is dispensed as needed).

Scanlan and Novak's (2015) scoping review examining the emerging field of sensory approaches within mental healthcare found that patients and staff in the majority of studies perceived sensory rooms to have a positive effect on the overall ward environment. The authors highlight how studies have predominantly examined sensory interventions in relation to reductions in either the use of seclusion and restraint, or levels of patient distress and the review found that sensory interventions

were generally associated with reduced levels of distress for patients. Whilst some evidence suggests that sensory approaches can contribute to the reduced use of seclusion and restraint in psychiatric settings (Champagne & Stromberg, 2004; Cummings et al., 2010; Lloyd et al., 2014) findings across the literature are mixed and further research is required to test initial findings (Scanlan & Novak, 2015).

Alongside increasing interest in the use of designated sensory rooms within healthcare settings, there is growing awareness of the potentially therapeutic influence of interior design and sensory interventions in the hospital environment more widely (Connellan et al., 2013; Iyendo et al., 2016; Schweitzer et al., 2004). However, guidance or information available to assist with developing the design of the visual environment in hospitals, including psychiatric settings has generally been very limited (Dalke et al., 2006). In a report focusing on the use of colour and lighting in hospitals, Dalke and colleagues (2004) argue that enclosed spaces with intense colours can be threatening or overly stimulating to people experiencing distress. Accordingly, it is recommended that colours and lighting are used to allow spaces to appear as light or open as possible and suggested that palettes of muted colours, which are 'greyed-off' to include a percentage of black, can aid relaxation and reduce stress. Karlin and Zeiss (2006) similarly report that bright colours may be over-stimulating to people experiencing agitation or distress and suggest that colours which are closely matched in terms of intensity and tone can have a calming quality.

Mazuch and Stephen (2007) argue that touch is a significant sense for patients with mental health difficulties as it can help individuals to re-engage with the materiality of the world around them. The authors also highlight the importance of avoiding perceptual confusion in psychiatric environments, such as using timber grain finishes on metal doors which would be surprisingly heavy and colder to the touch than timber doors. The tactile experience of materials is explored in Sakuragawa and colleagues' (2008) study examining participants' physiological and subjective

responses to touching samples of timber, aluminium or acrylic stored at different temperatures. The study found that the sensation of touching timber stored at room temperature and cooled timber was perceived to be 'natural' and neither condition prompted an increase in blood pressure. In contrast, contact with cooled acrylic or aluminium at room temperature induced sensations described as "artificial", "flat", "dangerous" or "uncomfortable" (p.110) and was associated with a significant increase in blood pressure, suggesting a close connection between the subjective evaluation of materials and physiological responses. Despite a lack of research examining experiences of touch in relation to the built environment and materiality in psychiatric settings, these findings suggest that the tactile qualities of architectural finishes could be as meaningful as their visual properties.

Regarding the sense of smell, commonalities across the literature examining patient experiences and sensory responses to general healthcare environments suggest that pleasant aromas may contribute to reducing blood pressure, lowering pain perception levels and slowing respiration, whilst unpleasant smells may prompt stressful responses (Connellan et al., 2013). Mazuch and Stephen (2007) accordingly emphasise that air quality and movement is especially important in secure settings where windows are generally of a small size and are kept closed.

1.4.3 Hospital soundscapes

Psychiatric hospitals are typically noisy environments and sound levels recorded in mental health wards have matched or exceeded the levels shown to affect cardiovascular or cognitive functioning in workplace and community settings (Holmberg & Coon, 1999). Multiple mechanical and human sound sources contribute to high hospital noise levels and research in general medical settings suggests that noise can have effects on patient health, including sleep disruption and raised blood pressure levels, in addition to increased length of stay and increased probability of re-admission (Hsu et al., 2012; Joseph & Ulrich, 2007). Despite the adverse effects of hospital noise

pollution on both patient and staff health (Hsu et al., 2012; Ryherd et al., 2012), Lawson et al. (2003) point out that contemporary healthcare design has typically focused on visual or spatial aspects of environments, with limited emphasis on sound control and overall acoustic quality.

As people experiencing mental distress may have increased sensitivity to noise (Stansfeld, 1992; Sutton & Nicholson, 2011), patients' experiences of the acoustic environment are particularly significant in mental healthcare settings. Sound control is also relevant to patients' sleep quality in psychiatric wards, since people with mental health difficulties may commonly experience disturbed sleep (Abad & Guilleminault, 2005; de Niet et al., 2008; Kamphuis et al., 2013), or have an increased need for sleep (Southwell & Wistow, 1995). Despite this, reports of disturbed sleep associated with noise on the ward and disruption from nursing observations at night time are widespread amongst patients in psychiatric care (Veale, 2019). Studies also suggest that patients' vulnerability to the emotional dysregulating effects of sleep disturbance could be a potential risk factor for aggression and violence in secure psychiatric settings (Kamphuis et al., 2012). Accordingly, Kamphuis et al. (2014) found that higher levels of insomnia and lower levels of sleep quality, as reported by patients in secure settings, were associated with higher levels of self-rated aggression or impulsivity and higher levels of clinician-rated hostility.

Rice (2003) studied the soundscape in a general hospital environment and in reference to 'panopticism' as a theorisation of visual surveillance by authorities (Foucault, 1975/1991), Rice uses the term 'panauralism' to allude to the monitoring function of listening for hospital staff. Similarly, research undertaken with patients and staff in a newly built medium secure psychiatric unit highlights how such listening practices are typically used by staff to attune to the atmosphere on entering a ward and sound out unrest in advance of seeing it (S. D. Brown et al., 2019a, 2019b). Staff participants in Brown and colleagues' research reported, however, that high ceilings

with skylights, which were designed to optimise a sense of space and light, also contributed to sound reverberation that disrupted listening practices, as the reflected sounds were not easy to localise. Alongside this, the associated exacerbation of specific sound sources in these spaces, including jangling keys and slamming doors, was found to contribute to patients' perceptions of being incarcerated and created associations with punitive environments.

Acoustic quality and environmental conditions are also significant to issues surrounding privacy or confidentially in hospital environments (Joseph & Ulrich, 2007). Wood et al. (2013b) found that the absence of designated visiting rooms in a new psychiatric facility affected visitors' experiences and courtyard gardens afforded a more comfortable and relaxing place for visits than communal areas where the acoustic conditions generated unpleasantly high noise levels. Isobel, Foster and Edwards (2015) similarly highlight how the provision of designated family visiting rooms in psychiatric settings affords respite from high noise levels on the ward and acknowledges patients' needs for privacy and connection with friends and family, including children.

1.5 Expert experience and mental healthcare research

Policy makers and researchers are increasingly concerned to draw empirical knowledge directly from the lived experiences of people accessing healthcare services (Beresford, 2002). Valuing this expert experience can help moderate misalignments between user and clinician perspectives and qualitative research to facilitate accounts of lived experiences and service evaluation can also expose clinical outcomes which could not be easily articulated purely through quantitative measures (Naylor et al., 2008). Although user involvement in the planning and delivery of mental health services has been promoted over recent decades by the Department of Health in the UK (Rutter et al., 2004; E. L. Simpson & House, 2002), there remains a lack of recognition for user-led and survivor research in mental health as an entity in its own right (D. Rose et al., 2018). There is also limited existing research evidence about what is
important to inpatients within the experience of psychiatric services to help inform recovery-based approaches to care (Staniszewska et al., 2019).

According to the nature of being detained, patients within secure psychiatric services are not easily accessible and concerns in relation to access, ethics and security present particular challenges to undertaking research in secure environments (Spiers et al., 2005). Evidence relating to the efficacy and quality of secure services has typically been drawn from clinical viewpoints and studies exploring patients' narratives about their lived experiences within secure settings are underrepresented in the literature (Askola et al., 2016). The benefits of user involvement in research are widely recognised in the context of forensic mental health (Brooker et al., 2007), yet studies exploring users' experiences and views about the impact of forensic mental healthcare services are limited (Coffey, 2006). Furthermore, despite recognition of the importance of collaboration between designers, patients and staff in relation to the development of secure services (Dvoskin et al., 2002), there is only a limited evidence base available to inform the design of secure mental healthcare environments, for which there is increasing demand (Mclaughlan et al., 2021; Shepley et al., 2016, 2017).

Within this context, the current study aims to contribute to the literature concerned with exploring patients' accounts of lived experiences in secure psychiatric settings to inform approaches to care (Tapp et al., 2013) and extend understandings of the experiential significance of secure spaces for patients and staff (S. D. Brown et al., 2019b, 2019a; Simonsen & Duff, 2020; Tucker et al., 2019). The study integrates the expert knowledge and experience of patients and staff and the research specifically aims to help expand knowledge of how both groups experience secure psychiatric environments and make sense of their affective experiences within these settings. Accordingly, it is envisaged that insights gained from the research might help inform the management and development of secure psychiatric services and increase awareness of connections between the design of secure environments and how

patients and staff experience these settings day to day. The study therefore looks to address the following overarching research question:

• How do patients and staff experience everyday spaces in low and medium secure mental healthcare environments and make sense of their affective experiences within these settings?

The research is also concerned with exploring how patient and staff accounts of inhabiting the ward environments may expose disconnects or dilemmas in the ways that spaces are experienced by both groups. Alongside this, the question of how secure psychiatric institutions are able to constitute supposedly 'private' spaces within their own 'public' spaces and the formulation of subjectivities in relation to these spaces will be examined. The study consequently also aims to address the following secondary research question:

• How do patients and staff negotiate and manage 'public' and 'private' space within secure mental healthcare environments?

These questions will be considered from a theoretical perspective in the following chapter, while the study design and methodological approach that were developed to address them will be discussed in Chapter Three.

Chapter 2: Psychosocial experience, process and space

The introductory chapter examined relationships between space and experience and considered how environments can be understood not to simply provide a passive backdrop for events, but instead to have an active role to play in shaping everyday experience, including perceptions of health and well-being. Accordingly, the literature examining associations between space and health was explored, including empirical studies investigating relationships between the spatio-cultural milieu and occupants' embodied emotional experiences with a focus on healthcare spaces (e.g. Connellan et al., 2013; Devlin & Andrade, 2017; McGrath & Reavey, 2019; Ulrich et al., 2008). The built environments, spaces and atmospheres that constitute the institutional sites of inpatient mental healthcare are therefore regarded as being closely associated with both the shaping of occupants' lived experiences and the delivery of services in these settings (Reavey, Poole, et al., 2017).

Within the of de-institutionalised era of mental healthcare provision, secure psychiatric units represent an institutional form of inpatient care, where patients may be detained for extended periods of time in a highly restrictive environment (Hare Duke et al., 2018). Occupants' experiences of the atmospheric, spatial and material qualities of these settings are of interest within the current research, where a focus is placed on how relationships and interactions with these very particular environments may contribute to shaping affective experience and everyday management of mental health and distress (Reavey et al., 2019). The study is consequently situated within an area of psychosocial research concerned with examining relationships between the social, cultural or material aspects of human experience and the management of mental distress, within spatial contexts that include inpatient psychiatric environments (e.g. Fenton et al., 2014; Reavey et al., 2019; Simonsen & Duff, 2020; Tucker et al., 2019), domestic settings (e.g. Tucker, 2010a, 2010b; Tucker & Smith, 2014) and community or public spaces (e.g. McGrath & Reavey, 2013, 2015).

Within this chapter, a rationale will initially be presented for why a transdisciplinary psychosocial process approach to understanding experience (Stenner, 2017) is helpful to enabling exploration of the affective experiences of patients and staff who live and work in secure mental healthcare spaces. A theoretical background to this approach will be outlined, with a focus on examining how the concept of liminality can offer a processual account of psychosocial experience and affectivity (Stenner, 2013). Before drawing together conclusions from the overall chapter, a relational understanding of space and embodied experience will be discussed with respect to mental health and distress.

2.1 Exploring experience through a psychosocial process approach

Reflecting on the 19th century beginnings of psychology alongside the social sciences, Stenner and Taylor (2008) observe that the close link between the social and the psychological is fundamental to the issues studied in psychology and sociology, yet paradoxically, the division of these concerns into separate disciplines obscures this bond. Over recent years increased interest in the relationship between subjective experience and societal processes has given rise to psychosocial approaches to research that aim to avoid the abstraction of subjective experience from its social, cultural, material or historical context (Stenner, 2014a, 2017). Through engagement with the influential work of philosophical process thinkers, including Alfred N. Whitehead, William James and Henri Bergson, contemporary psychologists have accordingly contributed to the development of critical alternatives to mainstream experimental psychology, alongside critical approaches to social psychology (see S. D. Brown & Stenner, 2009; Stenner, 2008). Drawing on this work, a process-oriented theoretical framework will be explored in the following sections to support a transdisciplinary psychosocial account of experience, that enables thinking about the social and the psychological together and seeks to avoid separating psychological processes from their overall context (Stenner, 2014b, 2017).

2.1.1 Experience and process

The rationale for a process-oriented approach to thinking about psychosocial experience can be situated in relation to the 'crisis' concerning the nature of social psychology in the 1970s, that responded to the focus on objectivity and abstraction of human experience from its wider context (S. D. Brown & Stenner, 2009). The overshadowing of subjective experience by the study of objective behaviour within mainstream psychology forms what Brown and Stenner refer to as a 'foundational paradox' for the discipline, in which the exclusion of subjectivity from objective scientific study paradoxically precludes the objective study of subjectivity itself.

Recognising, however, that psychological experience exists everywhere and is not purely confined to experimental examination in laboratory-like conditions, Brown and Stenner accordingly propose that a 'second-order' psychology should instead focus on a process of continually exploring and following human experience within all its forms, across all disciplines and contexts. Therefore, rather than seeking fixed 'foundations' for psychology, the authors propose a 'reflexive' foundationalism, in which the 'foundations' can be seen to be constantly evolving through a process of construction and re-construction, whilst still maintaining an identity over time. This approach reflects the concept of transdisciplinarity which is elaborated in the following section and recognises that the psychological is present in all aspects of human activity, including realms of psychological enquiry that exist within disciplines beyond academic psychology.

2.1.2 Transdisciplinarity

As a theoretical and practical approach to addressing the limitations of disciplines and disciplinarity, the concept of 'transdisciplinarity' is associated with efforts to transcend the epistemological boundaries of recognised disciplines within which specialist forms of knowledge are typically organised (Stenner, 2017). Transdisciplinary approaches to research may also be concerned with rethinking relationships between theory and

practice to allow collaboration between scientists and lay people on 'real-world' problems, by moving beyond 'pure/applied' or 'lay/expert' borders in relation to knowledge (Stenner, 2014b). Accordingly, transdisciplinarity has been defined as, "[...] a concept that has been used in efforts to describe integrative activity, reflection, and practice that addresses, crosses, and goes through and beyond the limits of established disciplinary borders, in order to address complex problems that escape conventional definition and intervention" (Stenner, 2014b, p.1989).

Within a hypothetical spectrum between disciplinarity and transdisciplinarity, further distinctions may also be made between the associated concepts of 'multisdisciplinarity' and 'interdisciplinarity' (Stenner, 2014b). Multidisciplinarity can be understood as a co-ordinated approach to dealing with problems from a number of discipline-based perspectives and therefore within multisdisciplinary research collaborations, each discipline is focused on its own area of specialism or abstraction (Stenner & Taylor, 2008). As Stenner and Taylor (2008) point out, this bringing together of findings from multiple disciplines can enable enhanced understanding of problems, but without necessarily requiring a discipline to change or transcend its boundaries. By contrast, interdisciplinarity is concerned with collaborative approaches to problems that involve the transferral of concepts or methods between disciplines and interdisciplinary processes may also at times give rise to new disciplines (Stenner, 2014b).

Multi and interdisciplinary approaches can therefore both be understood as 'disciplinary' in the sense that the research objectives are discipline-based, while transdisciplinarity is concerned with what lies between disciplines and the opening up of new spaces of knowledge and practice (Stenner & Taylor, 2008). Whilst an interdisciplinary approach to psychosocial research may combine psychological and sociological findings to address a problem, transdisciplinary psychosocial studies are concerned with hybrid psychosocial spaces that are not currently dealt with

sufficiently by either discipline (Stenner & Taylor, 2008). Correspondingly, Stenner and Taylor (2008, p.431) suggest that, "[...] if interdisciplinarity were the careful setting up of trade-routes between pre-established disciplines, then transdisciplinarity would be the invention of new spaces of knowledge and practice that transform the existing territory by opening it up to the new". Accordingly, within a process-oriented perspective, the concept of disciplinarity can be understood to represent wellorganised patterned arrangements and by contrast, transdisciplinarity can be conceptualised as a process of *transformation* occurring prior to the *emergence* of new patterned arrangements (Stenner, 2017).

A theoretical framework grounded within a transdisciplinary psychosocial account of experience is drawn upon within the current study to explore the multiple perspectives of patients and staff in secure mental healthcare settings and examine relationships between spatial and affective experiences in these environments. Transdisciplinary concepts can also enable the assimilation of ideas that may be considered unrelated when addressed purely within the boundaries of individual disciplines (Stenner et al., 2017). The current study accordingly engages with varied areas of research and practice including, psychology, geography, architecture and design, health and organisational studies.

2.1.3 Relational process ontology

Interpreting the work of philosopher and mathematician Alfred N. Whitehead, Brown and Stenner (2009) and Stenner (2008, 2017) provide a rationale for a 'relational process ontology' that promotes the exploration of human psychological processes within their social and cultural contexts. Brown and Stenner (2009), highlight that Whitehead's process ontology includes two essential relational principles. Firstly, that things (which may be biological, physical, psychological or cultural), can be defined in terms of their relevance to other things, or how other things are relevant to them. Secondly, that the existence of these things is not independent of time, but instead all things are brought into being through processes, or in other words, through a series of specific situated encounters. In terms of its relevance to social psychology, this mode of ontological thinking emphasises the inherently relational nature of subjectivity and supports the view that human experience is interwoven with the wider material, social, cultural or historical context.

For Stenner (2008, 2017), engagement with a relational process ontology and Whitehead's processual perspective on subjectivity enables a 'deep' form of empiricism. This is presented in contrast to a 'shallow' form of empiricism that resembles traditional empiricist principles in which all knowledge is seen to derive primarily from sensory experience. Accordingly, Stenner argues that a 'shallow' empiricism implies a split within nature between a 'materialist aspect', composed of meaningless substance that excludes subjectivity and an 'idealist aspect' in which subjectivity is attributed to high-grade human mental processes. Within this perspective, reality is not seen to exist independently of the mind and a distinction is implied between the 'known' and 'knower', such that matter is viewed objectively and forms the objective focus of the human 'knower' or subject. Subjectivity is then attributed to the 'knower', who 'knows' through their sensory experience, however, subjectivity is also understood to be separate from and therefore excluded from matter.

Whilst such distinctions between subjectivity and objectivity maintain a dualistic approach to nature, by contrast, a 'deep' form of empiricism suggests that there is always an interaction between subject and object (Stenner, 2017). Accordingly, Stenner highlights how subject and object are thus intrinsically relational since the subject experiences an object and the object results from a process involving subjective experience. Within a deep empiricism, subject and object are also argued to be inherently processual, as they do not pre-exist experience, but instead are brought about through experiential processes.

Accordingly, within Whitehead's philosophy, (1933/1935 p.226, as cited in Stenner, 2008), "[..] the subject-object relation is the fundamental structure of experience". Correspondingly, in Whitehead's terms, an 'actual entity' or 'actual occasion' refers to the basic unit of what can be seen to really exist within the world. For Whitehead, an 'actual occasion' or 'actual entity' thus represents the 'concern' of a subject towards an object, such that the object is a component of subjectivity within the occasion and the subject *comes into being* through having concern with the object. The material can be understood to participate in the production of subjectivity and Whitehead's 'actual occasion' of experience can thus be defined as the synthesis of subject and object within an integrated event of experience (Stenner, 2008).

A 'deep' empiricism consequently shifts the perspective from 'known' and 'knower' towards a process ontology that extends the realms of subjectivity and avoids what Whitehead terms the 'bifurcation of nature' into 'matter' as the primary essence of reality and 'mind' as a secondary side made up of ethereal 'projections'. Accordingly, a 'deep' form of empiricism resists splitting the world into outer 'objective reality' and inner 'subjective experience' and instead enables an account of experience in which nature is not reduced to insignificant matter (S. D. Brown & Stenner, 2009; Stenner, 2008). Process thinking is thus grounded in the principle that psychosocial reality is made up of the assembly of multiple processes or events and following Whitehead's principles, process can be understood as 'the 'becoming' of 'actual occasions'. Correspondingly, reality can be conceptualised as a process made up of transient occurrences. Therefore any 'being' can be understood as a temporary realisation composed from multiple individual events that come together in a 'form' that can sustain its structure or pattern through continual construction and re-construction (Stenner & Zittoun, 2020).

2.2 Experience and liminality

Paul Stenner's development of process-theoretical thinking in relation to liminal experience is drawn upon in the following sections, with a focus on the proposition that liminality provides a useful transdisciplinary framework to support a relational process account of psychosocial experience (see Stenner, 2017). Liminality derives from the Latin word 'limen' which describes a threshold or border and the term 'liminal' was initially given meaning within anthropology in the early 20th century by Arnold van Gennep (1909/1960), to describe the middle phase within a rite of passage (Thomassen, 2009). The concept of liminality was later taken up and developed during the 1960s by the process anthropologist Victor Turner and more recently it has been further extended within the social sciences (e.g. Stenner, 2017; Szakolczai, 2009; Thomassen, 2016).

2.2.1 Rites of passage and liminal experience

Considering rituals and ceremonies to be significant processes of transition within societies, van Gennep (1909/1960) first used the term 'rites of passage' to describe such events and identified 'rites of separation', 'transition rites' and 'rites of incorporation' as three phases or processes that take place within a rite of passage. Accordingly, van Gennep argued that a pre-liminal phase occurs initially during a rite of passage and comprises a process of uncoupling from a preceding state of affairs or routine practices (*rites of separation*). This is followed by the middle phase, or liminal period, in which the preceding rules of play or usual limits are temporarily suspended (*transition rites*). The liminal period is concerned primarily with passage and transitional processes occur within this phase prior to a final post-liminal stage in which a new status is established and the rules or limits of a new state of being are integrated (*rites of incorporation*) (Stenner, 2013, 2017).

Through his anthropological fieldwork examining ritual practice and symbolism, Victor Turner developed van Gennep's work and placed particular focus on

the liminal phase within rites of passage to expound the transitional and paradoxical qualities of liminal rites (V. Turner, 1967). Turner's use of the term '*betwixt and between*' therefore captures the ambiguous threshold, or period of limbo in the liminal phase that is experienced during the transition to a new state of being. Correspondingly, Thomassen (2009) highlights that whilst liminality can be understood as a transitional threshold between different states or spaces, it also constitutes a passage through time, such that spatial and temporal dimensions are both relevant to liminal experience, at various possible scales of magnitude. Subjectivity represents a further core component of liminality and although liminal situations might apply to individuals, they may also be experienced collectively by groups of people or by wider societies (Thomassen, 2009).

2.2.2 Liminality and process

Following a processual understanding of experience, in which the world is understood to be a 'form-of-process' (Whitehead, 1929/1985, as cited in Stenner, 2017), the concept of liminality presupposes a taken-for-granted sense of stability in everyday flows of human experience, or in what might also be termed 'worlds' or 'circles of activity' (Stenner, 2017). Within this logic it is reasoned that disruptions within our everyday flows of experience may lead to liminal situations or transitions *between* different forms-of-process or circles of activity. Accordingly, Turner (1969/1996) provides a conceptualisation of liminality in process terms as an 'interstructural' situation that constitutes an unstable threshold where structural ruptures or voids occur through the breaking down of existing structure, prior to new structure being created. Turner accordingly refers to the fluid and indeterminate position of liminality as 'anti-structure', or a state where it is as though the usual conventions or limits are melted down before taking on a different form, with the emergence of a new order. Turner's (1969/1996) specific use of the Latin term *communitas* within anthropology further defines a component of anti-structure, to describe an unstructured and non-

hierarchical state of being. Within the transitional state of *communitas*, members of a community are understood to have become equal through the shared experience of liminality, typically by taking part in a rite of passage.

Liminal experience can hence be understood as a transformative process and can be seen to represent a 'process of becoming', or the experience of a 'world-within-a-world', which itself takes place between worlds (Stenner, 2021). By pointing to the 'doubled' nature of such a world-within-a-world, it is also argued that the suspension of limits within the liminal zone between worlds creates a form of paradox in which two worlds or states that might otherwise be considered 'either/or', become combined as, 'both/and' and yet simultaneously, 'neither/nor' (Kofoed & Stenner, 2017; Stenner, 2017). The liminal passage of adolescence represents an example of such a paradoxical state of being, whereby a young person going through this transitional period of life could be considered *both* child *and* adult, but also at the same time, *neither* a child *nor* an adult. Similarly, in the context of secure psychiatric environments, as the focus of the current study, secure services and the institution itself represent a transitional zone situated between patients' lives before and after hospital. Within these terms, a patient could then be understood to be in a transitional and 'in-between' position, in which they are no longer 'well', but not yet 'recovered'.

Focusing on the transformative nature of liminality, McGrath et al. (2021) accordingly theorise that secure psychiatric institutions could also be conceptualised as a form of 'rite of passage' as a means to explore how their organisational practices might promote or obstruct transformative processes for patients. McGrath and colleagues found that a therapeutic graffiti arts initiative undertaken by patients in a secure hospital setting gave rise to a transformative environment that was characterised by a sense of *communitas* (V. Turner, 1996). However, whilst this creative environment was found to facilitate therapeutic transformation, by contrast, it also exposed how the usual practices of secure psychiatric institutions can hamper

transformative processes. Although secure psychiatric institutions must accommodate the seemingly contradictory functions of maintaining order and stability whilst facilitating change, it is argued that the requirement for order dominates the necessity for change. Consequently, since the highly structured and hierarchical characteristics of secure settings do not typically afford states of anti-structure or *communitas*, these environments do not readily offer a liminal space to facilitate transitional processes. Hence, according to the authors, secure facilities could also be conceived as a form of failed rite of passage.

Drawing attention to both the spatial and temporal characteristics of liminality, Greco and Stenner (2017) highlight the analogy made by van Gennep (1909/1960) to compare societies with the structure of a house composed of separate rooms and corridors. Within this comparison, formal transitions made between the different rooms and progression through the house are likened to rites of passage undertaken within the traditional societies studied by anthropologists. Continuing the analogy, however, van Gennep suggests that modern societies, by contrast, can be conceptualised by houses that have a greater sense of interconnection, with wider doorways or thinner walls. Accordingly a sense of passage is perceived less distinctly within 'open plan' house layouts and as such is likened to societal shifts from passage constituting a highly structured, formal and temporary occurrence, towards liminal states becoming an increasing typical and central feature of modern society (Greco & Stenner, 2017).

2.2.3 Affectivity and liminality

The increased overall interest in affect, emotions and embodied experience across the social sciences in recent years is reflected by what has been termed a 'turn to affect' or 'affective turn' within the literature (Clough, 2008). Whilst social constructionism and a turn towards discourse and text emerged in response to dominant traditional empiricist and cognitivist perspectives within the social sciences (Gergen, 2004), the

'affective turn' is generally seen to follow on from and also critique the 'discursive turn' associated with social constructionist approaches. The 'affective turn' within the social sciences consequently reflects a critique of the overriding focus on structure that set limits for social constructionism and prompted a turn towards affect as a way to explore experience that cannot be readily expressed through discourse, or in other words, to examine what exists beyond structure (Stenner, 2013).

As Stenner (2013) points out, the transitional events occurring within liminal situations have affective qualities that can be appreciated through the 'affective turn' literature, such that the concept of liminality can also assist in clarifying the agenda and development of the turn towards affect. Alongside this, it is argued that integrating the transdisciplinary concepts of liminality and affectivity can enable examination of the intense affective experience that Stenner terms 'liminal affectivity' i.e. the interplay of emotional experiences within liminal situations created through processes of transformation. Furthermore, the integration of thinking about liminal and affective experience can facilitate exploration of the mechanisms used to produce 'liminal affectivity' and also the ways in which this experience is managed (Stenner, 2013, 2020).

Building on Turner's (1982) distinction between 'un-staged' and 'staged' liminal experience, Stenner (2017) makes a similar contrast between forms of liminal experience described as 'spontaneous' (i.e. events that fall upon us, which may be unanticipated and beyond our control) or 'devised' (i.e. contrived events that we perform on ourselves). According to this logic, rites of passage constitute a form of *devised* liminal experience and represent processes that have developed to assist with navigating the transitions brought about by disruptions to experience created through situations of *spontaneous* liminality. Rituals are processes designed to stimulate and intensify emotions within those taking part and Stenner hence employs the expression 'liminal affective technologies' to describe mechanisms, including rites of passage, that

are purposefully employed to generate devised liminality and arouse affective experience (Stenner, 2017; Stenner & Zittoun, 2020). Moving beyond ritual and the anthropological origins of liminality, however, it is also argued that liminal experience could encompass any form of passage during which a process of becoming or transformation takes place. As a result, such forms of 'ontological liminality', as Stenner describes it, can provide a transdisciplinary extension of the concept across the sciences, within which anthropological liminality can be understood as a specific category (Stenner, 2020; Stenner & Moreno-Gabriel, 2013).

Consequently, whilst rites of passage comprise one of the earliest types of liminal affective technology, it is also reasoned that diverse cultural forms exist, including music, art, theatre or play, that can be employed to promote liminal affective experience and assist with psychosocial transformation (Stenner, 2017, 2021). Such technologies therefore provide a means to self-generate liminal affectivity and facilitate the process of becoming that occurs at the point when individuals transition from one state to another (Stenner, 2020). Similarly, from a cultural psychological perspective, Tania Zittoun's use of the term 'symbolic resources' describes symbolically mediated processes, such as engagement with art, film, music or literature that are made use of by individuals during what Zittoun terms 'life transitions' (Zittoun, 2007a). Drawing on such cultural and symbolic resources can hence be regarded as especially relevant for young people as a means to address and make sense of unfamiliar circumstances experienced during periods of uncertainty or biographical transition (Stenner & Zittoun, 2020; Zittoun, 2007b). To exemplify this in the context of the current research, patients' experiences of mental health crises, or being detained in a secure psychiatric hospital, represent disruptive and transitional life events that could be conceptualised as forms of 'spontaneous' or unplanned liminal experience. To negotiate or make sense of the disruption or unfamiliarity associated with this imposed transitional experience,

patients might then be seen to engage with liminal affective technologies, or symbolic resources, to self-generate 'devised' forms of liminal experience.

Stenner's (2020) interpretation of Baruch Spinoza's philosophical thinking (1677/1989, as cited in Stenner, 2020) also proposes that affect constitutes experience that is 'on the turn' and consequently points to liminality as a mechanism for conceptualising the turning point between 'raw' affective sensations and their meaningful articulation as emotions. Within this thinking, affect can be regarded as a liminal transition or the turning point between different orders of affective experience. In contrast to the discursive turn's concern with discourse, meaning and structure, the affective turn can be seen to emphasise 'event' over 'structure' and to be concerned predominantly with affect, intensity and event (Stenner, 2013). Stressing the inherent relationship between structure and event, however, Stenner proposes that a process-oriented account of experience allows the re-thinking of 'structure' as forms-of-process and the re-conceptualisation of 'event' as occurrences that result in transformation between forms-of-process. According to this logic, event can be seen as a turning point or a liminal period of transition that takes place between forms-of-process or circles of activity, such that event can be understood to occur *between* structure.

A process-oriented account of affect thus requires a focus on both structure and event and avoids a polarised split between affect and discourse, with the result that the affective turn does not necessarily need to be understood as a dismissal of the discursive turn, but is argued rather to represent its further development and deepening (Stenner, 2020). Integrating the transdisciplinary concepts of liminality and affectivity can hence be seen to offer a conceptual and relational understanding of experience within which thinking about 'structure' and 'event' are combined. Furthermore, in terms of exploring psychosocial experience, the integration of affectivity and liminality enables an understanding of affective experience as a phenomenon of liminal transition (Stenner, 2013).

2.2.4 Imagination and liminal experience

As discussed in the previous section, in liminal terms, the experience of mental distress or hospital admission can be understood to represent significant transitional life events that may cause disruption to an individual's everyday 'flow' of experience or sense of self. The ways in which liminal affective technologies or symbolic resources, including creative or imaginative processes might be used to mediate such transitional experiences are accordingly pertinent concerns for the current study (Stenner & Zittoun, 2020). These processes may include engagement with physical or imagined environments and consequently participants' use of imaginative processes involving space or objects to facilitate liminal experience is also of interest.

Imagination is a vital aspect of human experience and drawing on Vygotsky's thinking (e.g. 2004), Zittoun and Cerchia (2013) propose that imagination can be understood as an expansion and enrichment of psychological experience that is essentially social and cultural in its nature. This 'expansion' model of imagination is presented in contrast to alternative 'deficit' perspectives (e.g. Pelaprat & Cole, 2011), which view imagination as a response to 'gaps' in our fragmented experience of the world that must be filled to provide a sense of stability. Zittoun and Cerchia instead contend that imagination is prompted by a temporary disruption within a stream of thinking, brought about by a rupture in the given flow of one's embodied experience of the world. Imagination is therefore conceptualised as a 'loop' of thinking processes that expand human experience before returning to 'reality' and re-joining an ongoing flow of thinking. From this perspective, imagination is described accordingly as, "[...] a process unfolding in time: in a person's current apprehension of reality, something triggers imagination, imagination develops on its own, and eventually the person comes back to reality, usually having gained something from that excursion" (Zittoun & Cerchia, 2013, p.313).

Imagination can therefore be understood as a process of becoming, or a liminal situation characterised by the creation of a transitional zone of experience where pragmatic everyday rules and reality are temporarily suspended (Stenner & Zittoun, 2020). Correspondingly, Stenner and Zittoun also highlight how imaginative processes can be seen to possess a similar tripartite framework to van Gennep's (1909/1960) rites of passage, whereby following 'separation' from a flow of experience, a liminal phase of imaginative transition occurs, prior to a stage of 'reincorporation' when a typical flow of thinking is re-joined.

Liminal occasions typically evoke a form of emotion, whether positive or negative (Szakolczai, 2017) and the temporary suspension of limits within liminal situations may also give rise to novelty or creativity (Stenner, 2021). Correspondingly, Stenner and Zittoun (2020) suggest that unplanned liminal situations can also prompt self-generated liminal experience, creating a rich environment where creativity and expression become more vital and imagination may be used to enable processes of liminal transition.

2.2.5 Enduring liminality and health

Whilst liminal occasions have been defined in process terms as the transitional phase between forms-of-process or circles of activity, the associated concepts of 'permanent liminality' and 'liminal hotspots' may also be used to offer an account of extended or enduring liminal experience (see Greco & Stenner, 2017; Stenner et al., 2017; Szakolczai, 2017). The term 'liminal hotspot' thereby defines the experience of being caught within the intermediate space between forms-of-process, or in other words, an occasion of suspended transformative transition that results in an extended phase of liminality between circles of activity (Greco & Stenner, 2017).

Although liminal experience may be associated with imaginative processes which may engender novelty or creativity, Szakolczai (2017) points out that liminal situations can also be construed less positively and be characterised by a sense of

anguish and uncertainty. In the case of extended liminality, the experience of being caught in a suspended state of transition between forms-of-process, i.e. being suspended within a liminal hotspot, generates a paradox that disrupts flows of experience and in turn can bring about a sense of paralysis (Greco & Stenner, 2017; Stenner et al., 2017). Alongside this, Greco and Stenner highlight how liminal hotspots may also be characterised by 'polarisation' that occurs when individuals attempt to escape the paradoxical and indeterminate state between forms-of-process by finding a solution that best aligns with one of the worlds between which they are suspended. In instances where when a liminal paradox cannot be readily avoided, however, the paralysis experienced may also potentially give rise to 'pattern shift' or novelty, thus sparking the emergence of a new form-of-process that integrates the existing forms-ofprocess as a way to move forward (Greco & Stenner, 2017).

As it is argued that transitional situations bring about forms of affectivity, the concepts of liminality and liminal hotspots are pertinent to exploring lived experience in the field of health, whereby health challenges can be understood to represent transitional life events (Greco & Stenner, 2017; Stenner, 2013). Liminal experience is also typically associated with identity transformation and in relation to ritual processes, Beech (2011) defines liminality as, "[...] a reconstruction of identity (in which the sense of self is significantly disrupted) in such a way that the new identity is meaningful for the individual and their community" (p.287). In the context of mental healthcare, detention in secure psychiatric institutions may commonly imply labelling in respect to both criminal offending and mental health, such that the management of transitions to other 'non-deviant' identities can be an ongoing endeavour for patients upon discharge to the community (Coffey, 2012b). Coffey (2012a) accordingly highlights how patients released from hospital on conditional discharge might frequently perceive their sense of identity to be in a liminal state of *betwixt and*

between (V. Turner, 1967) when seeking to re-integrate into communities, whilst also being positioned as an 'offender'.

In relation to the current study, the concept of enduring liminality is argued to be significant to exploring patients' experiences of detention within secure mental healthcare environments, typically for indefinite periods of time. Experiences of being caught in an indeterminate state of transition for an extended period may then influence patients' perceptions of moving forward, that in turn may have implications for 'recovery' processes, including the construction of alternative identities.

2.2.6 Liminal space

As the representation of passage through a threshold or boundary, liminality possesses an intrinsically spatial dimension and liminal space can incorporate both physical space and psychological or imagined space. Within the context of organisational studies, Shortt (2015) draws attention to the spatial characteristics of liminality through an exploration of how everyday 'in-between spaces' or 'liminal spaces' are experienced and given meaning by the occupants of a workplace environment. Shortt makes reference to Dale and Burrell's (2008) definition of 'liminal space' as being, "at the boundary of two dominant spaces, which is not fully part of either" (p238), to highlight how the 'spatial turn' literature exploring workplaces has typically focused on 'dominant' physical environments and less so on 'in-between' spaces such as corridors, staircases or cupboards. Such seemingly inconsequential areas, however, were captured in photographs taken by hairdresser participants in Shortt's research and identified as significant aspects of their working environments.

These in-between spaces were found to offer a sense of respite from the expectations of professional or social 'performance', in both the front of house (salon spaces) and back of house areas (staffrooms). Following the terminology of Erving Goffman (1959, as cited in Shortt, 2015), these spaces can be seen to be positioned between the 'front-stage' and 'back-stage' and Shortt extends the concept of liminality

to consider the significance of such transitional spaces for those who dwell within them. According to Shortt, when occupants find meaning within 'in-between' environments, beyond their essential function, they can be seen to change from purely liminal spaces to what might be termed 'transitory dwelling places'. Correspondingly, 'in-between' places such as smoking areas within inpatient psychiatric settings have been found to possess psychological and social significance for patients and staff. Notwithstanding conflicts between smoking and physical health, such spaces are seen to hold meaning and thus may contribute to the creation of 'therapeutic landscapes' in psychiatric care (Wood et al., 2013a). In the context of secure psychiatric settings, the focus of the current study similarly involves an exploration of meaning making by patients and staff members in relation to experiences of everyday spaces in the hospital environment.

2.3 A relational understanding of experience and space

A relational understanding of psychological experience and space will be further examined here, including exploration of topological accounts of experience that have developed within social psychology. This exploration is grounded in topological theories of psychology previously proposed by the gestalt-influenced psychologist Kurt Lewin (1936), as interpreted by contemporary social psychologists (see S. D. Brown, 2012; S. D. Brown & Reavey, 2015; Goodings & Tucker, 2019; Tucker & Smith, 2014).

2.3.1 Topology and life space

The concept of topology involves the mathematical study of spatial properties that are conserved when geometric objects undergo deformation and Kurt Lewin engaged with the principles of topology to rethink relationships between person and environment within psychology (S. D. Brown, 2012). Kurt Lewin's *Principles of Topological Psychology* (1936) thus explores the interaction between psychological experience and the spaces that people inhabit to offer a framework which avoids the traditional

subject-object dualism that separates 'outer' environments from 'inner' cognitive processes. Lewin argues accordingly that behaviour is a function of a person in their environment and suggests that "every psychological event depends upon the state of the person and at the same time the state of the environment" (1936, p.12). This statement implies an intrinsic relationship between human experience and the environment and points to a need within psychological study to recognise both person and environment as integrated parts of a 'whole' situation.

Lewin's application of topology to psychology accordingly offers an alternative relational understanding of space and lived experience, in contrast to prevailing models of space as an external 'container' for activity (S. D. Brown & Reavey, 2015). Lewin (1936) expresses this concern with the removal of boundaries between people and the world through the topological concept he terms 'life space'. For Lewin, life space represents interactions between the environment and a person's sphere of action, such that psychological experience can be regarded as not simply occurring 'in space', but instead to be interwoven with the spatial distribution of everyday activities (hence the use of the term 'relational'). Within this perspective the mind is therefore understood to reach beyond the confines of the brain and be extended into the world through spatially distributed experience (S. D. Brown & Reavey, 2015). Life space is hence more concerned with connections and relationships between people and events than with measurable temporal and spatial distance, or the dimensional properties of space such as size or scale (Reavey, 2017).

As Brown and Reavey (2015) highlight, a relational understanding of space as expressed through the concept of life space consequently encompasses physical, external understandings of space, in addition to psychological space and internal processes such as memory and imagination. Within this perspective, it is the relationships between events that create meaning for people and these connections extend across divisions between physical and psychological to become relevant to

current experience at any point within an ongoing 'flow' of experience (S. D. Brown & Reavey, 2015). Life space therefore comprises multiple connections that link a person's current actions with other people or events across time and space. As such, the relations and context that are outside the current 'scene' of activity can have influence and effect on a person's feelings or behaviour within the present moment. According to this perspective, whilst events or relations might be both temporally and spatially distanced, they can be understood to be 'acting' in the present moment (Reavey, 2017). For Lewin, such actions from a distance comprise 'quasi-physical facts' and his associated term 'quasi-conceptual facts' refers to the topological expansions that have potential to bring about future possibilities or influence the future. Within this logic, life space can be seen to constitute a topological manifold of possible and non-possible events (S. D. Brown, 2012).

Through the emphasis on the spatial distribution of experience, including memory and anticipation of the future, topology and life space have been employed to expand understanding about the management of mental health and distress by patients in secure psychiatric settings (Reavey et al., 2019) and in the home spaces of people who use mental health services in the community (Tucker & Smith, 2014). In a newbuild medium secure psychiatric facility Reavey et al. (2019), draw on the concept of life space in an exploration of how the ward environment might contribute to shaping transformative processes of 'recovery' through mediating patients' experiences of social interaction and agency. Accordingly, rather than being conceptualised as a set of 'inner' cognitive processes, patients' subjective experience is examined in relational terms, both with regards to other people and the social or material context. Agency was seen to be associated with the expansion or contraction of patients' psychological life space, as mediated by staff-patient relations and the affordances of the environment, such as enabling patients to experience a sense of control. Life space was found to be compressed through experiences including a lack of regular social activity, or patients

perceiving they were ignored by staff and thereby feeling relationally distanced, despite being in a newly built space designed to bring people together physically. In contrast, the bedroom spaces were found to be relationally richer than the shared areas of the ward and the extension of patients' life space was made possible through engagement with activities including listening to music or the radio. Although patients' experiences in secure psychiatric settings might be seen to be bounded by the physicality of the institutional environment, a topological understanding of experience as life space recognises that relationships transcend physical barriers, for example via means including photographs, letters, phone calls and music (S. D. Brown & Reavey, 2014).

2.4 Concluding comments

A transdisciplinary psychosocial process account of experience has been explored in this chapter as a framework that enables an understanding of human experience as the embodied convergence of social, psychological and spatial processes. Accordingly, it has been argued that a relational process approach to the psychosocial provides an account of subjective experience that is intrinsically interwoven with the wider social, material, historical and cultural context. The concept of transdisciplinarity is concerned with the exploration of hybrid spaces located between disciplinary borders, or beyond the boundaries of individual knowledge and experience. Correspondingly, transdisciplinary approaches can help integrate diverse knowledge or the experience of different stakeholders. In the current study this includes exploring the perspectives of patients and staff who occupy shared environments within secure psychiatric settings and for whom the same spaces may have differing functions or meanings.

Liminality has been examined alongside this as a conceptual mechanism for exploring affective and embodied experience in process terms. Integrating the transdisciplinary concepts of liminality and affectivity can accordingly enable exploration of the affective experiences generated by processes of transformation. It

has consequently been argued that the concept of liminality provides a theoretical framework that assists with exploring the less tangible aspects of experience, including perceptions of health or identity. Liminality also provides a theoretical perspective in relation to imaginative processes and the role of imagination in facilitating selfproduced liminal affectivity that can assist in mediating liminal situations.

Furthermore, the notion of 'liminal hotspots' offers a way of conceptualising the processes at play when liminal occasions become extended or sustained and it has been argued that the concept of enduring liminality is pertinent to exploring experiences of mental distress and detention in secure environments. The spatial dimension of liminality is also relevant to the study of 'in-between' and seemingly insignificant or mundane spaces which may play a part in mediating subjective experience and facilitating affective transformation. Alongside this, a relational understanding of space and lived experience has been explored, in which boundaries between people and the environment are removed and psychological experience is understood to be intertwined with the spatial distribution of everyday activity.

Chapter 3: Methodology

The methodological approach to the research will be outlined within this chapter, alongside an exploration of how the methodology and data collection methods employed are situated within the theoretical framework that has so far been discussed.

The first section of the chapter sets out the theoretical orientation of the study and methodological approach. This includes a rationale for grounding the study within a qualitative research paradigm that was informed by and developed in conjunction with the research questions. The narrative psychological orientation of the study and the empirical significance of examining narrative accounts of lived experiences within secure hospital settings are also discussed.

The second section outlines the research design and provides a detailed account of the work stages and activities involved throughout the study. This includes information relating to processes and considerations regarding research ethics, participant recruitment, data collection, audio data transcription. Reflexivity will be discussed in the final chapter of the thesis.

The interpretative process and analytical concerns relating to the data are detailed in the third section of this chapter. The conceptual structure supporting the chosen narrative-based analytical approach is outlined here and the processes involved in interpreting the data are described. This final section concludes by detailing the process of developing and collating the analysis to determine the content of the three subsequent empirical chapters, within which the findings that were found to best address the research questions are presented and discussed.

3.1 Qualitative research and narrative psychology

A qualitative research paradigm was chosen for this research due to the emphasis placed on exploring participants' perspectives alongside the texture and quality of their lived experiences. Through its concern with meaning, qualitative research seeks to expand understanding of 'what it is like' in relation to participants' experiences of specific situations or conditions (Willig, 2013). Accordingly, qualitative research methods are used to gain in-depth understanding about how people make sense of the world and their experiences of events within specific naturally occurring contexts. Qualitative studies can thus offer insights into psychosocial phenomena alongside the social and cultural contexts in which these may be grounded. This is significant to the study of patient and staff experiences of secure psychiatric environments which, as discussed in the introductory chapter, are highly complex and particular environments. In relation to exploring experiences of hospital settings, qualitative research can also help integrate different stakeholder perspectives to further understandings about the therapeutic potential of the built environment and design in healthcare (Curtis et al., 2007). The remainder of this section will outline narrative psychology as a qualitative framework for the research and examine how narratives function to enable people to make sense of their lives and experiences.

Interest in narrative theory emerged from the general turn to language in the social sciences during the 1980s and the study of narrative has become increasingly widespread across all disciplines (Murray, 2003, 2008). Murray (2008) highlights how Sarbin's (1986, as cited in Murray, 2008) edited volume entitled, *Narrative Psychology: The storied nature of human conduct* was an influential text in marking the specific turn to narrative in psychology. Within this work, Sarbin presents narrative as an alternative metaphor for psychology, in contrast to the dominant mechanistic, context-free and laboratory-situated metaphors perceived to underpin the majority of mainstream psychology (Kirkman, 2002; Murray, 2008). Through this perspective Sarbin suggests that narratives have an ontological status, since it is argued that people actively construct the world through stories and live through the stories they tell. Accordingly, narrative psychology is concerned not only with research methods, but also with wider ontological issues and is focused on the content, structure and function

of the stories that people tell in social interaction (Murray, 2003). Informed by a social constructionist epistemological orientation, narrative psychology thus aims to produce knowledge about how people create meaningful stories from their experiences (Silver, 2013). Correspondingly, Mattingly and Garro (2000 p.1) note that, "[n]arrative is a fundamental human way of giving meaning to experience. In both telling and interpreting experiences, narrative mediates between an inner world of thought-feeling and an outer world of observable actions and states of affairs". As narratives play a central role in helping people structure or understand experiences, a narrative psychological approach can thus contextualise human action.

3.1.1 Narratives and identity

Ricoeur (1979) argues that the primary function of narrativity is to bring a sense of order and temporality to human understandings of a world which is in constant flux. Storytelling is hence a means through which human activity is organised and narratives are thereby used to construct a sense of self (Crossley, 2000a, 2000b; McAdams, 1993). Alongside this, Polkinghorne (1991) argues that thinking about 'self' in terms of a narrative, rather than as a substance or thing, emphasises the temporal and relational dimensions of human existence. This perspective also underscores the constructive and interpretative nature of 'self', such that self-concept can be understood as a temporal process of becoming. Polkinghorne accordingly suggests that "self-concept is a storied concept, and our identity is the drama we are unfolding" (p.149). Social life is consequently argued to be produced through stories and narrative is seen to constitute an "ontological condition of social life" (Somers, 1994 p.614). Somers correspondingly highlights how this perspective suggests that people construct identities (which may be multiple or changing) within diverse intersecting relational storylines amongst which they may find or locate themselves.

Whilst narratives enable individuals to draw together distinct life events to create a meaningful whole, self-narratives may become fragmented under stressful

circumstances, resulting in a loss of meaning and a break-down in the unity and integrity of a person's self-concept (Polkinghorne, 1991). For example, health challenges can disrupt a person's prior expectations about how the story of their life might unfold and accordingly, the effects of chronic illness on self-narrative and selfconcept have been described as "narrative wreckage" (Frank, 1995 p.53) or "biographical disruption" (Bury, 1982 p.167). As serious health issues can cause significant disruption to the flow of a person's everyday life, storytelling may then assist in making sense of ever-changing personal experiences (Crossley, 2000b) and perform a therapeutic role (Frank, 1995). Murray (2000) correspondingly suggests that storytelling might also be considered an act of self-care or a natural response to health challenges. Crossley (2000b) similarly argues that, "[...] when disorder and incoherence prevail, as in the case of trauma, narratives are used to rebuild the individual's shattered sense of identity and meaning" (p.527). This is significant to exploring patients' lived experience in forensic populations, within which exposure to trauma, including neglect and childhood abuse is prevalent (Adshead, 2012; Coid, 1992). In the context of secure psychiatric services, the generation of alternative selfnarratives, that can enable patients move beyond 'psychiatric patient' or 'offender' identities is also understood to play a significant part in processes of 'recovery' (Adshead, 2011, 2012; Gardner-Elahi & Zamiri, 2015).

3.1.2 Visual storytelling

As a form of visual storytelling, images can contribute to narrative inquiry (Stephens, 2011) and the use of visual materials in research can enable participants to convey stories through non-verbal means (Keats, 2009). Visual research methods include a diverse range of techniques that involve engagement with visual materials in the process of generating empirical data (Reavey, 2011; Reavey & Johnson, 2017). Materials may comprise a variety of media, including drawings, photographs, film or objects and artifacts that are either made or collected by participants. The strengths of visual research methods include facilitating access to topics or experiences which may be unnoticed or difficult to express in purely verbal interviews (Reavey, 2011). These experiences might include sensorial or physical phenomena that are less easily articulated solely through words (Boden & Eatough, 2014), or reflections on the more ordinary aspects of life that may easily be taken for granted (G. Rose, 2014). Visual materials can hence help people to access and articulate the complex emotions and details grounded in everyday experience that might otherwise be overlooked (Reavey, 2011).

Visual research methods have consequently been found to help elicit rich narrative accounts within research interviews (Reavey & Johnson, 2017). Active engagement with imagery can also encourage greater narrative contribution from participants who may be verbally reserved, or may have limited capacity to maintain concentration for reasons such as medication use (Reavey, 2011). This is significant to the secure mental healthcare context of the current research, in which patients may commonly experience side-effects from psychotropic medication, including drowsiness or mental clouding (Seale et al., 2007). Visual methods have accordingly enabled indepth examination of health and well-being related narratives (Frith & Harcourt, 2007) and exploration of the everyday life experiences of people who use mental healthcare services (K. Johnson, 2011; McGrath et al., 2008; McGrath & Reavey, 2013, 2015).

As spatial and psychosocial experiences are inherently interwoven, the physical environment is a fundamental aspect of everyday lived experiences. Environments play an integral role in storytelling as the contexts within which events and emotions are experienced and visual research methods can assist in exploring the spatial dimensions of experience. The combination of visual and verbal methods can thus facilitate narrative accounts that are grounded in very specific spatial contexts (McGrath & Reavey, 2013; Reavey, 2011), in contrast to narratives produced through verbal

research methods alone, which tend to be organised within more temporal structures (Reavey & Johnson, 2017).

The medium of photography can facilitate creative expression and the use of photo-production as a visual research method, whereby participants create photographs for discussion in an interview, has been shown to offer a meaningful and creative activity for participants (Reavey, 2011). These aspects of photo-production are particularly significant for the current research exploring experiences in secure psychiatric settings, where patients do not typically have access to cameras and may commonly spend long periods of unstructured time on the wards between activities.

Engaging with photographs can enable access to the embodied sensations evoked by environments or atmospheres that may be beyond immediate conscious awareness or discursive accessibility and facilitate their transformation into emotions or narratives to articulate experience. In process terms, the use of photographs within qualitative research could then also be conceptualised as a form of 'liminal affective technology' (Stenner, 2017, 2021). Photo-production processes may accordingly help research participants to explore the more 'intangible' aspects of their experiential worlds, including perceptions of identity (Reavey, 2011).

Studies involving photo-production techniques also demonstrate how participants can shape the research process by guiding how the images are used in interviews (Radley & Taylor, 2003b). This in turn can adjust the power dynamics of traditional research interviews to allow participants a greater sense of perceived control (Radley & Taylor, 2003b; Ray & Smith, 2012) and foreground participants' voices in the research process (S. Warren, 2005). This is pertinent to the restrictive context of secure psychiatric settings, where patients may typically have a limited sense of choice or control and whose lived experiences are underrepresented in the literature (Coffey, 2006). Furthermore, photo-production techniques can also afford wider visual exposure to 'marginalised' environments, such as the spaces within secure mental healthcare settings, which are typically accessible only to limited groups of people (Kanyeredzi et al., 2019; Reavey et al., 2019).

3.2 Research methods

The multi-modal approach chosen for the study combined visual research methods with semi-structured interviews as a means to generate verbal data for transcription and interpretation using a narrative-based analysis approach (Silver & Reavey, 2010). The visual research methods used in the research comprised photo-production (Del Busso, 2011; Radley & Taylor, 2003a, 2003b), whereby participants produced photographic data which were later discussed in semi-structured interviews.

Photo-production was chosen as a method that could enable participants to share the everyday spatial contexts of the research settings, including the spaces or objects which may not be readily accessible to visitors in a secure setting. The images produced were not treated as visual data for analysis, but typically formed prompts during the interviews which helped elicit participants' accounts of their experiences within specific spatial contexts (Reavey and Prosser, 2012).

Radley and Taylor (2003a) demonstrated how photo-production techniques enabled patients to explore the hospital environment as an immediate emotional landscape in which to construct 'recovery'. Similarly, photo-production was used in the present study to facilitate the active exploration of secure psychiatric spaces as an emotional terrain in which occupants' spatial experiences can be understood to be fully interwoven with psychological experiences (Tucker & Smith, 2014). The use of photographs enabled participants to have control over creating the image content and in turn, greater choice and influence in how narrative accounts of spatial experiences were constructed during the interview (Radley & Taylor, 2003a).

The combination of photo-production with semi-structured interviews in the research was designed to support the narrative-based approach to the data analysis as the images encouraged storytelling and became entwined with each participant's

narrative during the interview. This approach reflects a broadly social constructionist perspective and recognises the role of social interaction and language to the construction of thinking and experience (Burr, 2003; Willig, 2013). The processes involved in designing and implementing the research are summarised in the following sections and include detailed accounts of the ethical considerations, participant recruitment, data collection and transcription. Copies of the supporting documents that were used in the study are attached in Appendices A-H.

3.2.1 Ethics

Ethical concerns were a central and ongoing focus throughout all stages from the inception of the project and involved the following key considerations and processes.

i) Ethical approach and approvals

The research was designed and undertaken in line with ethical guidance included in the British Psychological Society Code of Human Research Ethics and the London South Bank University Code of Practice for Research with Human Participants, in addition to guidance provided by the research setting and NHS Research Ethics Committee. Before commencing participant recruitment, ethical consent for the project was confirmed with the NHS Research Ethics Committee (reference number: 16/NW/0114) and the Research Ethics Committee at London South Bank University. To enable contact with patients in mental healthcare facilities, the necessary disclosure and barring service clearance (DBS) certificate was also obtained prior to commencing data collection. Whilst the approvals and coordination processes required to facilitate participant access and data collection took considerably longer than originally envisaged, following written confirmation of all necessary approvals, the participant recruitment and data collection for the study took place between May 2016 and July 2018.

ii) Use of cameras

For reasons associated with confidentiality and patient protection, camera use is generally prohibited in secure psychiatric settings and cameras typically constitute a form of 'contraband' that cannot be brought onto a ward. Consequently, it was also necessary to gain permission from the research setting to allow the use of a digital camera for research purposes on the participating low and medium secure psychiatric wards. Digital cameras were generally supplied by each ward for use in the research, however, permission was also obtained for camera equipment to be supplied for research purposes in some instances where a ward camera was not available. As required by the research setting, patient camera use was risk assessed on an individual basis by the nurse in charge on each ward and patient participants were supported by staff throughout when using cameras. It was explained to all participants that they should avoid taking photographs which included people and that any photographs produced which show people who could be recognised would be deleted in the interests of protecting the anonymity of others. Participants were also advised to use the camera very sensitively when around other people.

iii) Participant and researcher well-being

Through the process of meeting with potential participants to introduce the research and subsequently meeting again to discuss the project in further detail or confirm consent, time was typically spent together on at least two occasions before the interview. These individual encounters with participants benefitted the process by enabling initial familiarity, such that a sense of rapport could be developed together as far as possible before meeting again for the interview. The possibility of participants becoming upset whilst reflecting on difficult experiences was a key consideration and due attention was given to ways in which participant distress might be mitigated throughout the research. Before the interviews, it was made clear to patient participants that no direct questions would be asked about their mental health or the

reasons why they were receiving care in a secure psychiatric setting. Care was also taken to consider potential sensitivities around exploring participants' experiences of their immediate working or living environments. Whilst for some patient participants these spaces may also constitute 'home ', it was recognised that for some people the contexts of home environments might also be associated with difficult experiences. It was explained to participants that they could refuse to answer any questions or to request that any responses provided be removed from the interview transcript.

In practice it was not necessary to end or pause an interview due to participant distress, however, in the event of a person becoming upset, the recording would have been paused, subject to the participant's agreement, or discontinued entirely as appropriate. Members of the multi-disciplinary team (MDT) were also aware that patients had participated in a research study and were available to provide support in the event of distress occurring during or after the research process and interview. Participants were asked afterwards how they had found the interview experience and the debrief sheet that was provided and talked through directly after the interview included details of independent organisations that could offer confidential support if required (see Appendices G & H for copies of the debrief sheets for patients and staff respectively). Following the debrief, a little time was also typically spent talking more generally with each participant before concluding the meeting.

Interviews with patients and some staff participants typically took place within side rooms or meeting rooms on the wards in which CCTV was installed. Some staff interviews were also undertaken in meeting rooms within staff office areas that were located outside the wards. When undertaking interviews with patient participants, a personal alarm was typically provided by the research setting and staff were on hand nearby to provide support in the event of any emergency. Any distress experienced personally during the process of undertaking the research was generally discussed with the academic supervisory team in the first instance. Informal reflexive notes were

also used at times as a tool through which to mitigate personal distress by promoting self-awareness as a first step to seeking support from the supervisory team, or other support as and when required.

iv) Confidentiality and data protection

All potential participants were made aware that the parameters regarding confidentiality were such that information provided would remain confidential, unless criminal activity was disclosed, or a participant suggested that future harm may occur to themselves or others. In such cases, disclosures would have been reported as appropriate to staff within the research setting in the first instance. Participants were made aware that pseudonyms would be applied by the researcher during transcription. It was also advised that potentially identifying features, such as place names, would be redacted in the transcribed data. Similarly, potentially identifying features were also redacted or cropped from the visual data that were included in the reported findings, as part of efforts to support participant anonymity.

The digital data collected were encrypted and stored on password protected equipment. Any hard copies of data produced, such as printed photographs, or project information were locked in a filing cabinet within an access-controlled office at the university. Only the researcher listened to the audio recordings and transcribed the interviews. The files created were coded numerically without reference to participants' real name and the data from each interview was given a successive participant number. The files were organised such that data could easily be retrieved if participants subsequently wished for any data to be removed from the study. Participants were made aware that the data would be retained for a period of 10 years after completion of the study before being destroyed.
3.2.2 Participants and recruitment

Contact with potential participants and the process of recruitment involved the following key stages and commenced after all ethical approvals and permissions had been confirmed in writing.

i) Initial contact with wards and potential participants

Initially contact was established with the wards that the research setting had identified as being appropriate to the study. Meetings were held with members of the multidisciplinary team (MDT) as required on each ward to discuss the project in further detail and confirm that participant recruitment could take place. Following guidance from the research setting, the study was then typically introduced initially to patients and staff by the researcher during a community meeting⁵ on each ward. During the meeting a short verbal introduction was given to outline the study and patients and staff were invited to ask questions, during or after the meeting, such that they could consider whether they might be interested in taking part. It was made clear at this stage that there was no obligation to participate and that there was no financial or commercial benefit offered for taking part.

Patients and staff were invited to confirm interest in the study, or to ask any questions individually with the researcher after the meeting. A copy of the relevant participant information sheet containing further detailed information about the research was given to those who expressed interest in taking part (for copies of the patient and staff information sheets, see Appendices A & B respectively). At times, separate introductory meetings were also arranged with staff, for example, to enable

⁵ The community meetings provide a forum for patients and staff members to discuss any issues or concerns relating to the ward. These weekly meetings are generally chaired by a patient and where possible, attended by all patients on the ward and available staff members.

the research to be introduced to staff members who were not on shift when the community meeting took place.

ii) Informed consent

Potential participants were allowed at least 24 hours after expressing interest in the research to decide if they would like to take part and typically several days were allowed before meeting again for the process of confirming written consent to participate. During the intervening period, liaison took place with the relevant Responsible Clinician (RC) and MDT members to confirm that any patients who had expressed interest were deemed to have capacity to provide informed consent and it was considered appropriate for them to participate, should they decide to.

Individual meetings with potential participants were then scheduled and typically took place privately in a side room on each ward. Whilst it was possible to communicate directly and arrange meetings with staff participants, it was necessary to liaise with staff members on each ward to coordinate the meetings with potential patient participants. During these individual meetings with staff or patients, any questions raised about the study were answered and the participant information sheet was read through together to assist with clarity. If a potential participant decided to proceed at this point, the consent form was also read through and signed together (see Appendix C for a copy of the consent form). As part of this process, it was made clear that signing the consent form did not bind participants into taking part and that they had the right to withdraw completely from the study without giving a reason. Photocopies of the signed consent form were made such that both parties retained a copy and in the case of patient participants, a copy was also retained by staff for the patients' medical files. Consent was revisited verbally when meeting with participants throughout the research process and participants were reminded that there was no obligation to take part.

iii) Participants

A purposive sampling strategy was employed to enable access to a mix of patient and staff experiences across medium and low secure inpatient mental healthcare wards within a large psychiatric hospital in the United Kingdom. In total⁶ 19 patients and staff participated in the study. The inclusion criteria required participants to be patients or staff within a secure forensic mental health inpatient setting and to be over 18 years old. Exclusion criteria included potential patient participants being deemed by the care team to be lacking capacity to provide informed consent, or if participation was considered inappropriate, or presented a potential risk to the well-being of the individual or others.

The study was introduced to medium and low secure wards identified by the research setting and ultimately the participants were recruited from six individual same-sex secure wards located in several different buildings within the overall facility⁷. These six wards comprised three medium secure and three low secure settings. In total, nine patients participated in the study, of whom three were women and six were men. The participating patients were recruited on two medium secure and three low secure wards. Ten staff participated in the study in total and were recruited on three medium secure and three medium secure and three new women and three medium secure wards. The staff participants comprised seven women and three men within the multidisciplinary team, whose professions included nurses, healthcare assistants, occupational therapists and psychologists.

Although direct questions were not asked to elicit specific demographic information, some participants provided personal details during the course of discussions, which included their age, ethnicity and sexual identity, or where relevant,

⁶ Due to a technical anomaly, data for one of the interviews were not available.

⁷ On the wards where participants were recruited and on other wards where the study was introduced but no participants were recruited, a small number of additional patients and staff expressed interest in the study but were unable to participate due to aspects of the inclusion/exclusion criteria or subsequent changes in their personal circumstances.

details of their length of stay in hospital and mental health diagnoses. However, as part of the efforts to protect participant anonymity in the reported findings of the study, these data were not systematically collected. A summary of the research participants in terms of their gender and ward level of security is provided in Table 1.

Table 1

Research participants

	Medium Secure		Low Secure		
	Patients	Staff	Patients	Staff	Total
	Patricia	Vicky	Alice	Jo	
Women		Felicity	Hazel	Karen	10
women		Ingrid		Estelle	
				Wendy	
	Sal	Gavin	Leon	Nathan	
Men	Darren	Mike	Bradley		9
men	Tom				9
	Carl				
Total	5	5	4	5	19

Note: All names listed are pseudonyms

3.2.3 Data collection

Data collection took place after consent had been confirmed with each participant. A photo-production process was undertaken individually by participants and followed by a semi-structured research interview. On average each interview typically lasted between 40-60 minutes approximately. The photo-production and interview processes are outlined in the following sections.

i) Photo-production process

Participants were invited to take up to ten digital photographs of their everyday environment for discussion in a research interview. It was explained that the images produced should include spaces or objects that they might encounter during a typical day on the ward or other accessible parts of the overall facility. Participants were asked to produce images that could help to capture a range of the emotions that they might experience in relation to spaces or objects in the everyday environment. To maintain the anonymity of other people within the research setting, participants were advised to avoid taking photographs that included people. Appropriate timescales were agreed with each participant as convenient to allow the photographs to be produced and a date was scheduled accordingly for the interview.

Patient camera use was facilitated and supervised by staff members and the period of time required for the photographs to be produced varied depending on staff and patient schedules. For logistic and organisational reasons relating to staff and patient availability, it was not always possible to hold the interview directly after the photographs were taken, but the interviews typically took place as soon as possible afterwards. In advance of the interview, the photographs taken by participants were printed where possible, usually via printer equipment on the ward, where this was available. As it was generally necessary for the photographs to be printed on ward equipment, the physical copies of the images were typically black and white prints on A4 paper.

ii) Interview process

The interviews were undertaken individually and privately in a room with each participant. The interviews were recorded with the participants' consent and prior permission was obtained from the research setting to allow a small digital audio recorder to be brought onto each ward for research purposes. Before each interview started, participants were asked to arrange all available prints as they wished on a table in the room, such that the images could be easily reviewed individually or collectively as required. In some instances, where it had not been possible for images to be downloaded from the ward camera for printing, it was necessary for participants' images to be reviewed on the camera screen during the interview.

Taking each image in turn, participants were invited to discuss why they had chosen to take the photographs and describe their feelings and range of emotions

experienced in relation to the spaces and objects captured in the images. An interview schedule was also prepared and questions drawn from this were threaded into the course of the dialogue where appropriate. In contrast to fully structured interviews, in which the agenda is prescribed through specific questions posed by the researcher, the interviews were semi-structured to encourage storytelling and afford participants greater power and control in shaping the interview content (Murray, 2003). The parts of the interview in which patients and staff talked through their photographs had especially strong narrative characteristics, as the participants took the lead on what was discussed. Following this, remaining questions from the schedule were also asked by the researcher as appropriate to elicit further reflections about participants' psychosocial and spatial experiences.

The questions on the interview schedule were grouped around different aspects of psychosocial experience (e.g. 'well-being' or 'privacy') and aimed to help address the research questions by prompting participants to talk about their experiences and affective responses to the everyday environment. Whilst the schedules for staff and patients were similarly constructed, the questions were tailored accordingly for each group to reflect their different occupational experiences e.g. in reference to the ward as a workplace or as a residential space. The same interview schedule was used with patient participants on both medium and low secure wards, although for organisational purposes it was necessary to label the schedule separately as Group One (medium secure) and Group Two (low secure). Copies of the patient interview schedule for Groups One and Two are therefore included in Appendices D and E respectively and the schedule for staff participants is attached in Appendix F.

In several instances, a pause in recording was required due to different types of interruption that occurred during the interview. These included staff participants responding to enquiries from colleagues and events on the ward, or the interview being relocated because the room was needed for another purpose. After the interview,

each participant was thanked and fully debriefed about the rationale for the study and a hard copy of the debrief sheet was provided and read through together. Where available and with each participant's permission, the photographs discussed were retained for reference following the interview. In all instances participants in the study gave permission for their images to be retained and reproduced in reported findings of the research, however, in the event of this not being the case, the relevant images would have been removed from the data set.

Throughout the research process, in addition to time spent in the interviews, a significant amount of time was spent more generally on the wards in spaces such as the ward office. Naturalistic observation of staff and patient interaction and inhabitation of the ward spaces therefore took place during the course of visits to the hospital and notes were also made at times as an informal resource.

3.2.4 Transcription

Transcription of the audio data was undertaken by the researcher and took place throughout the data collection process. The audio data were transcribed verbatim and each line of transcribed speech was numbered for reference. The transcription did not follow the detailed conventions of approaches such as conversation analysis, however, additional information was included in the transcripts to assist the contextualisation and understanding of discussions. For example, any non-word sounds such as laughter or pauses were noted in the transcripts and any emphasis placed on particular words was highlighted. A summary of the conventions that were used during transcription is included in Appendix I.

The first few interviews carried out with patients and staff were transcribed initially and preliminary readings of this initial data were undertaken and discussed during academic supervision. This process was helpful in assisting personal reflection on the experience and evolving process of carrying out the interviews. It also enabled

supervisory feedback to be discussed, based on initial assessment of the data generated thus far. Each subsequent interview was then also transcribed in the same way.

During the transcription process the digital files were played back via specialist software installed on university computer equipment, using an associated foot pedal to control and adjust the speed of audio playback. Throughout the process the transcribed data were checked back against the audio record for accuracy. It was generally possible to clarify most instances of ambiguous or unclear speech through the process of slowing down and replaying sections of the recording, although in a number of cases background noises such as slamming doors or loud voices on the wards rendered some words or short sections of speech incomprehensible.

The process of repeated listening to the audio recordings and reviewing participants' photographs, where available, whilst personally transcribing the interviews provided a helpful initial familiarisation with the data. This process also enabled a nuanced appreciation of the interview context, including any interruptions or background sounds, alongside providing an awareness of the subtleties in how the dialogue was spoken by both parties within each interview. The processes of data transcription and narrative analysis are not easily distinguished (Riessman, 2008) and reflecting on recollections of each interview when listening to the recordings and notemaking during transcription formed a valuable part of the preliminary analytical and interpretative process (Esin, 2011).

3.3 Narrative analysis

Narrative analysis represents a theoretical approach to interpreting stories and this overarching term covers a diverse range of perspectives through which to conduct an analysis (Esin, 2011; Stephens, 2011). Narrative-based approaches to analysis enable exploration of how individuals story their experiences and can allow researchers to gain understanding about the intricacies of social relations (Esin et al., 2014; Sharp et al., 2018). Whilst there is no single or definitive form, narrative-based analytical

approaches typically focus on the different features of narratives as a theoretical perspective for interpreting stories. Accordingly, different approaches to narrative analysis may focus on aspects such as the content (e.g. Riessman, 2008) or structure of narratives (e.g. Labov, 1972, as cited in Frost, 2009), or the performative function of stories (e.g. Langellier & Peterson, 2004) and social processes involved in storytelling (e.g. Clark & Mishler, 1992). Pluralistic approaches may also apply a number of different theoretical models of narrative analysis to the same data in order to generate a multi-dimensional interpretation (e.g. Frost, 2009). Within the widely ranging versions of narrative analysis, a commonality in approach is the typical focus on interpreting narratives as whole entities as opposed to breaking down stories into individual discourses (Murray, 2003; Riessman, 2008). Consequently, in narrative-based approaches to analysis, stories are understood to represent the unit of analysis (Esin, 2011) and the analysis is concerned with both the 'what' in terms of a story's content and the 'how' in relation to how stories are constructed (B. Smith, 2016).

As storytelling can be a way to make sense of the disruption to life associated with health challenges, it is also argued that narrative-based approaches are well suited to social and health-related psychological research (Murray, 2008; Stephens, 2011). Alongside providing understandings of people's lived experiences of health and healthcare, narrative analysis may also help to illuminate the social contexts around health, including social justice and inequalities (Stephens, 2011).

3.3.1 Narrative levels of analysis

Murray (2000) builds on concepts developed by Doise (1986) to outline an approach to narrative data interpretation in health psychology research using a conceptual framework of narrative levels of analysis. Within this approach, four levels of analysis which are defined as, 'personal', 'interpersonal', 'positional' and 'ideological' are proposed as analytical perspectives through which to consider narrative data. Stephens (2011) highlights how this framework of narrative levels also maps onto approaches

occurring elsewhere in the literature, including Somers' (1994) approach to understanding how identity may be constructed through narrativity and Langellier and Peterson's (2004) examination of the performative functions of narratives.

Within the conceptual framework of narrative levels put forward by Murray (2000), the 'personal' level of narrative analysis considers individual stories as an immediate focus and offers a phenomenological perspective to understanding accounts of lived experiences. Consideration is given to the functions served by telling stories, including the use of narratives to bring a sense of order to personal experiences (Ricoeur, 1979). Such narrative functions may also include mitigating perceptions of disruption or incoherence which might be felt by people when experiencing challenges to health (Frank, 1993; Radley & Taylor, 2003b). This 'personal' level of analysis may hence be concerned with the ways in which narratives are used to construct a sense of individual identity or self (Crossley, 2000b; McAdams, 1993). Similarly, Somers (1994) describes such personal stories as 'ontological narratives' and highlights how individuals may use stories to construct narrative identities as a means to make sense of their lives and actions.

Working with narrative data at Murray's (2000) 'interpersonal' level of analysis acknowledges the co-created nature of research interviews, wherein the stories produced are constructed as part of an evolving conversation between participant and researcher (Silver, 2013). At this level of interpretation attention is given to how the interpersonal processes within these social contexts and potential audiences may influence how and why stories are told. In acknowledging the researcher to be an active participant in a dialogue and not simply an onlooker within an interview, the analytical process is also concerned with the role played by the researcher in shaping how narratives may evolve in this context.

As an extension of the 'interpersonal' level, the 'positional' level in Murray's framework is concerned with social characteristics and the relative positioning of

researcher and participant within the interview context. Narrative interpretation at this level considers how commonalities or differences between social characteristics of both parties, including health status (Radley & Billig, 1996), gender (Phibbs, 2008), age, or moral position might be relevant to social interaction and the production of narrative accounts.

Finally, interpreting narratives at the 'ideological' level proposed by Murray (2000) focuses on the ways in which all social interactions are situated within wider sociocultural contexts that permeate and shape the structure of everyday stories. Interpretation at this level is concerned with how narrative accounts may draw on or disrupt these broader assumptions and belief systems, which Somers (1994) also refers to as the 'public narratives' shared within societies and cultures.

Building on Murray's (2000) conceptual framework, Stephens and Breheny (2013) describe a similar approach to narrative analysis in psychological research based on integrating narrative levels of analysis. Stephens and Breheny's (2013) approach includes 'personal' and 'interpersonal' narrative levels, alongside a third 'public' level of narrative analysis, representing a combination of Murray's (2000) 'positional' and 'ideological' levels. This 'public' level of analysis is thus concerned with openly available social narratives and how these may shape shared understandings or disconnects within societies. Within the analytical process, it is therefore of interest to consider how such shared social narratives might be used to position moral or social identities, alongside how these wider public narratives may be drawn upon or challenged in social interaction during the co-construction of interviews (Breheny & Stephens, 2011; Murray, 2000; Stephens, 2011; Stephens & Breheny, 2013; Wong & Breheny, 2018).

With reference to Doise (1986), Murray (2000) emphasises that these four outlined levels of narrative analysis do not constitute four different versions of reality, but instead comprise a conceptual framework within which to organise the different

levels considered by researchers across the various differing narrative-based approaches to analysis. As stories may typically be considered from the perspective of several narrative levels simultaneously, Murray (2000) highlights how integrating levels is of particular interest in the data analysis process. This integrated approach to narrative levels of analysis is summarised diagrammatically in Figure 2. Accordingly, Murray (2000) argues that the challenge for researchers is to be able to articulate connections in the telling of narratives across the different levels.

Figure 2

Analytical approach using integrated levels of analysis



Individual lived experiences
Phenomenological perspective
What the story's function is

e.g. constructing identity or
creating a sense of order

INTERPERSONAL

- Co-creation of stories
- Participant/researcher dialogue
 Interpersonal processes
- How and why stories are told
- Potential audiences

POSITIONAL

- Extension of interpersonal level • Participant/researcher positions
- and social characteristics • How commonalities + differences
 - affect narrative production

IDEOLOGICAL

- Sociocultural contexts
- Broader belief systems
 Social narratives
- How stories draw on or
- disrupt wider assumptions

Note:

Diagram based on Murray's (2000) Levels of Narrative Analysis in Health Psychology

3.3.2 Analysing and interpreting the data

The transcribed data collected for the current study were subjected to rigorous qualitative data analysis, using a narrative-based approach. This interpretive approach focused primarily on Murray's (2000) framework of narrative levels of analysis, with reference to variants of this conceptual model as described by other researchers (e.g. Breheny & Stephens, 2011; Stephens, 2011; Stephens & Breheny, 2013; Wong & Breheny, 2018). Alongside analysing the data through an integrated framework of narrative levels, a series of analytic devices was also engaged with to assist in building an analysis from the data (Crossley, 2000a; Wong & Breheny, 2018). The remainder of the chapter provides a detailed account of these processes to explain how the analysis was undertaken and describe the activities involved throughout. Whilst the approach to the data analysis is described as a series of stages, the process was flexible and iterative (B. Smith, 2016; Wong & Breheny, 2018). Building the analysis involved moving fluidly back and forth between processes as required and writing the analysis was an integral part of the analytical process.

i) Familiarisation

Initially the data were reviewed several times both during and after the transcription process to gain in-depth familiarity with the content of each interview. Whilst the images generated by participants were used primarily as visual aids during the interviews and were not formally analysed independently as visual data, where available, the images were reviewed again for reference during the transcription process and when reading the corresponding transcripts. Each transcript was read through several times to increase familiarity and any initial observations or responses to the data were noted down for reference or elaboration and refinement during the further stages of interpretation that followed. The narratives and visual data produced by staff and patient participants were reviewed as a single overall data set.

ii) Identifying narratives

Whilst stories may often be based around a developing plot with clearly defined characters, scenes and sequences of events, it was recognised that accounts may also be unstructured and diverging or incomplete (Wong & Breheny, 2018). Therefore, a comprehensive approach was taken to identifying different stories told within each transcript. In line with a narrative-based focus on examining the use of storytelling to make sense of lived experiences, each story was considered as a whole entity rather than being broken down into individual discourses (Murray, 2003; Riessman, 2008). As argued by Wong and Breheny (2018), a flexible approach to analysis based on integrating narrative levels, as used in this study, also enables the same interpretative process to be applied both to complex narratives and to the more mundane or unstructured stories that people may tell. Stories within each transcript were identified by looking for new beginnings marked by a noticeable shift in the content of the talk and endings signalled by declarations or concluding events (B. Smith, 2016). The transcripts were also reviewed as a whole to try to get a feel for any stories that were in development across the overall interview. Storylines drawn from the narratives identified within each transcript were summarised and grouped into a set of overarching storylines for each participant. Each overarching storyline was given a name that captured the combined essence of the individual storylines. For reference the storyline synopsis for Wendy, one of the participants, is included as an example in Appendix R. The overarching storylines for each participant were also collated into main storylines across the data set and a diagram summarising this is included as Figure 3 within chapter four. The storylines for each of the participants were reviewed across the data set and those considered best able to address the research questions were identified, alongside corresponding extracts from the data. During this process attention was paid to common storylines and prominent features within the narratives,

in addition to considering stories which were significant or striking for reasons that included seemingly not fitting with other narratives across the data set.

iii) Interpreting narratives

As the levels of narrative analysis are intertwined within stories, the integrated approach to interpretation involved coding the data in reference to the different levels simultaneously and the process was repeated across the narratives. When interpreting the data at a personal level of analysis, the stories told were perceived as an expression of each participant's lived experiences, with consideration given to how narratives might also be used to outline a sense of identity to others (Somers, 1994). Riessman (2008) argues that narratives do not necessarily reveal an 'essential' self and attention was accordingly given to how participants' stories may be used to position a sense of self. This can be contextualised via the extract below in which Carl, a patient participant, is invited to give some information about himself to the researcher at the start of the interview.

"R: Great. Well thanks ever so much for, um, taking the time to meet and um, I just wondered if you could possibly just start by just telling me a little bit about yourself, you could include, um, you know, any information that you'd like. C: (*Little laugh / outbreath*). Er, my name's [*name*] I've been on [*name of ward*] since [*name of month*] so that's been about [*number*] months is it? I'm [*age*] and I've been in the service, er, as a whole since [*name of year*], so nearly [*number*] years. But luckily, I've come here and they've realised I don't need to be in the service and I should be moving on very quickly. [R: Mm] But it's really not been easy."

Carl stories his entire journey through the healthcare service in this narrative which although brief, can be seen to have a beginning, middle and end. This short story includes characters (i.e. the authorities or staff within the institution) who play a significant part in the narrative and it also contains a form of resolution (i.e. that Carl will now be moving on quickly from the hospital). Through highlighting the length of time that he has been detained on the current ward and the number of years spent within the service overall, Carl positions his experiences of moving through psychiatric services as a lengthy process. In presenting himself as someone who does not need to be detained in secure services, Carl can also be understood to be storying an account of how he would like to be seen by others. Whilst the narrative conveys some optimism and alludes to a sense of his changing fortunes, Carl's concluding comment stating that, "it's really not been easy", succinctly summarises to the listener that his journey so far has been a testing process of endurance and a difficult experience.

When interpreting the data at an interpersonal level, the analytical process focused on relational aspects of the narratives and considered how the stories were cocreated through dialogue and social interaction between the participant and researcher within the interviews (Silver, 2013). This included consideration of how the characteristics of the researcher and participant and the relationship between them could influence both the content of the stories produced and how they were told. Attention was given to how questions were presented and the ways in which the researcher's responses, including pauses or silences, may have contributed to how participants' stories were produced (Esin, 2011; Murray, 2003). Additionally, narrative devices employed by participants and the researcher were also examined, including the use of humour or phrases repeated for narrative effect (Breheny & Stephens, 2011; Wong & Breheny, 2018). An example of the interpersonal level of analysis can be illustrated via the following extract in which Darren, a patient participant, responds to the researcher's question about his experiences of enjoyment within the environment.

1.	R:	[] Alright, just, we've touched on it really, some of these points while you've
2.		been describing the photos, but I just wondered whether you could perhaps
3.		tell me about any, um, whether you might experience any sort of sense of
4.		enjoyment or pleasure, um, in the physical qualities of um, the, the spaces here,
5.		or from any of the objects within it at all? If there are any spaces that you
6.		(slight pause)
7.	D:	Er.
8.	R:	enjoy particularly?
9.	D:	What, is this inside the building or outside the building?
10	. R:	It could be either, yeah.

- 11. **D**: Um, I quite enjoy going to the gate. Because I can smoke there. 12. **R**: To the where sorry? 13. **D**: To the gate. 14. **R**: The gate, OK. 15. **D**: Yeah, so. 16. **R**: Yeah. The entrance gate? 17. **D**: Yeah, yeah. 18. **R**: Yeah, yeah. 19. **D**: 'Cause we're allowed to smoke there. 20. **R**: Ah. OK. 21. **D**: So it's, 22. **R**: Yeah. 23. **D**: you know, it's relaxing. You know, you can become quite tranquil you know. 24. **R**: Mm, mm. Er, do, why do you think that is, is that part of being outdoors, um? 25. **D**: A little bit, plus, it's just, I don't know if you smoke yourself, but it can be quite 26. relaxing you know. 27. R: Yes, yeah, yeah. And would and you can go there ev-, when-, several? 28. **D**: Yeah, not all the time, 29. **R**: Mm. 30. **D**: you know they do try to get us out as much as they can. 31. R: Mm, yeah, yeah. How do you find being outdoors, is that, is that something 32. that's positive? 33. **D**: I, I, I love being outside the building. 34. **R**: Mm. Yeah, are there particular (*slight pause*) places that you would go to out-, 35. outside? 36. **D**: Yeah, er, [name of café on site]. 37. **R**: Yeah. 38. D: [Name of café on site] café. I don't know if you've been there? 39. R: I have yes, yeah, that's nice. Yeah. 40. **D**: I quite like it down there. 41. **R**: Mm. 42. D: Yeah, it's nice down there. It's, plus it's a public place, you know, it's not [name 43. of hospital] controlled environment if you like. 44. **R**: Mm. Yeah, yeah. Does that and that (*slight pause*) has an impact on how you 45. feel, 46. **D**: Yeah. 47. **R**: when you're in that space?
- 48. **D:** Makes you feel normal.

In this extract, the researcher is initially confused about the reference Darren makes to his enjoyment of visiting the "gate". Despite Darren's explanation from the outset that the reason for his enjoyment is because he can smoke there, it requires several stages of clarification before the researcher understands the story. Eventually the researcher grasps that patients are able to smoke by the hospital entrance gate and by implication, also realises that smoking is not allowed within the hospital grounds. This exchange between Darren and the researcher highlights how the contrasts in their individual characteristics as patient and visitor influence the dialogue and points to differences in their respective experiences and operational understanding of the hospital environment. Additionally, as the relational dynamics in interviews imply that participants' experiences are the primary focus of interest, researchers may not necessarily anticipate receiving questions from participants during an interview. Darren's query here about whether the researcher smokes could also be interpreted as an attempt to create a greater sense of balance within the interpersonal dynamics of the meeting and prompt dialogue that is reflective of a more typical and naturalistic conversation. However, by asking the question rhetorically using the pre-fix, "I don't know if", Darren poses the question to the researcher indirectly, without necessarily expecting or requiring an answer. Following on from this, Darren similarly frames another question, "I don't know if you've been there?", in relation to the on-site café. In asking this question he is again opening up a more conversational form of dialogue and is seen to be inviting the researcher into a more equal exchange of information.

When analysing narratives at the positional level of analysis, consideration was given to the social characteristics of the participants and researcher, in addition to their relative positioning in the context of the interview. Differences or commonalities in their positions were identified, including the relative liberty of both groups. The analysis was accordingly concerned with how such characteristics were relevant to social interaction within the interview and the narratives produced. In the above extract, Darren concludes by stating how he feels "normal" when spending time in the public café on the hospital site and this story can also be used to illustrate the positional narrative level of analysis, as an extension of the interpersonal level. In contrast to spending time amongst members of the public in the café setting, his narrative implies that he does not typically feel "normal" in other spaces (i.e. in the areas he describes as hospital-controlled environments). Through the narrative Darren is highlighting perceived contrasts between his position and self-identity as a detained patient and the positions of members of the wider public, including the researcher. In addition to the example from Darren's narrative, the following extract from the interview with Karen, a staff participant, provides a further illustration of the positional narratives that may be used to shape individual stories.

"R: [...] I was also just wondering, er, if you, if you could describe any ways in which, um, you might think that having a sense of choice is important to you when you're considering the sort of details of your working environment at all? K: Um, (*slight pause*) for me, I dunno, I've never been considered in that decision (*little laugh*), I'm a HCA. Um, I'd like to see the walls, I'd like to see more, um, pictures and things on the walls in the lounge, but again with it being a secure building it's difficult to get them, I mean, it's, it is a phone call to maintenance for them to put things up on the wall, but um, I don't know whether that comes down to budgetary constraints or not, um, we have a project at the moment where we're, we do-, which we're doing ourselves, by painting an inspiration tree."

Karen's pause and her initial response, combined with her short laugh that follows this suggest that she is slightly perplexed by the researcher's question about experiencing a sense of choice within the environment. Her next comment, "I'm a HCA" (healthcare assistant), is presented to the researcher as a self-explanatory statement and is used to qualify the preceding remark that she has not typically been involved in decisions regarding the hospital environment. By implicitly positioning her role at the lower end of a decision-making hierarchy Karen can also be understood to be making a wider statement about how the position of HCA might be regarded or valued by the

institution. Karen presents herself as having opinions about ways to improve the environment aesthetically, yet the story positions her sphere of influence over decision-making or effecting change to be very limited. In contrast, however, Karen's conclusion to the story restores a sense of agency to her role by highlighting to the researcher how a ward-driven initiative to enhance the environment is currently being implemented. Karen's positioning of herself in relation to the researcher is also evident at the end of the interview when she is invited to share her thoughts about taking part in the study.

"R: That's great and I just wondered if I could just finally ask you if, if you could just tell me something about why you were interested in taking part and also how you found the interview? **K:** Um, honestly, 'cause no one had me to take part in anything before, ever, really (*laughs*), I've done sort of ward-based surveys and things that have gone round on a bit of paper to help people out, but, um, not something on this sort of scale and um (*slight pause*), yeah it was just nice, it's nice to be involved in something like this and um, hopefully (*slight pause*), hopefully, um, what I've said is important and is relevant and yeah."

In this short account, Karen builds on her earlier allusion to how the institution might value her opinions about the environment, or how input from staff members in the role of HCA may be regarded. Whilst this story can be seen to reinforce Karen's perceptions of a lack of institutional interest in her perspectives and experience, by contrast, taking part in the research study is presented as a meaningful activity through which she feels valued. The researcher is therefore situated through the narrative in a contrasting position to the institution and is understood to be interested in hearing about Karen's experiences and what she has to say.

Approaching the data analysis at an ideological or societal level of analysis took into account the shared public narratives, or 'master narratives' that can inform and position individual accounts at broader social and cultural levels (Murray, 2000; Stephens, 2011). Accordingly, consideration was given to the ways in which participants' stories might reveal social processes and reflect or challenge dominant

master narratives (Murray, 2000). The exchange between Darren and the researcher in the earlier extract can also be used as an example to contextualise how the data could be interpreted at the ideological level of analysis. Here, Darren's account of feeling "normal" when spending time in a public place rather than inside the hospital draws on broader social or moral narratives about the types of people or circumstances that are considered to constitute 'normality' within society. Alongside this example, a further illustration of the ideological level is provided via the following story in which Leon, another patient participant, describes his perceptions of how the ward furniture is associated with how people might feel in that environment.

"L: [...] so a lot of stuff we get here, is like the chairs we're sitting on now, are clinical, they're heavy, they're weighted. There's <u>nothing</u> homely about them. I mean like a threepiece sofa with like, nice, like, footstools, just normal stuff. [**R**: Mm, mm.] And to be fair, it's like, especially this is like a low secure, or even a medium secure, I've, we had the same problem there. It was just like, I understand there's risk, but if you make a place more homely, yeah, even that, we are all not family, we're not blood, or anything [**R**: Mm.] if you feel at home [**R**: Mm.] doesn't matter who's there, you're gonna be naturally more relaxed, the ward's gonna be more relaxed, the, the atmosphere's gonna be calmer. [**R**: Yeah.] There'd be less situations. But where everything's sort of clinical it's just feels so <u>tense</u>."

Leon draws on social and cultural understandings of homeliness and domesticity for narrative effect in this story to present a jarring contrast between wider societal expectations of residential spaces and his own experiences of the ward environment. Accordingly, his narrative builds on shared cultural beliefs that a physically or atmospherically comfortable 'home' environment can facilitate rest and support relaxation. By situating the "clinical" qualities of the ward environment within this social narrative, Leon disrupts the notion of home comfort to position the ward as nonhomely and actively uncomfortable. The narrative image of utilitarian and clunky ward furniture presented in his account is accordingly contrasted with imagery of traditional domestic sofas and the added sense of comfort implied by "footstools". Leon uses this story to propose an intrinsic relationship between the ward environment and the behaviour of occupants, such that creating a "calmer" atmosphere and a more homely space might lead directly to a reduction in relational friction. Drawing on wider understandings of "clinical" spaces as being cold, stark or sterile, Leon's final sentence concludes this story by suggesting that these pervasive and undesirable characteristics of the ward are closely connected to an underlying sense of atmospheric and relational tension experienced within the environment.

Alongside approaching the data analysis through these narrative levels, several analytical devices were also used throughout to assist in highlighting social processes, or the performative and functional qualities of the stories told (Crossley, 2000a; Wong & Breheny, 2018). These analytical mechanisms included paying attention to the stage and setting, i.e. the places where stories are set and the spaces in which the stories were told. In addition to this, significant details in the stories, such as characters, places, objects or events were highlighted to enable examination of their role within the narratives. Participants' narrative tone and the imagery generated through their accounts were also examined. Further analytical devices included observing tensions within the narratives, examining the function of humour and exploring how aspects of language or refrains were used to create narrative effect.

iv) Building the analysis

Analysing the data and writing interpretations of participants' narratives across the data set was an integrated and ongoing process to build the narrative analysis (B. Smith, 2016). The process was approached inductively, in that it was grounded within the data, but the interpretation was also influenced by theory (Murray, 2008). The analysis drew upon literature identified in the first stages, with a focus on liminality (Stenner, 2017) as a theoretical concept within the overall perspective of a relational process ontology (S. D. Brown & Stenner, 2009). When building the analysis, the coding applied through the process of analysis using the narrative levels of analysis, analytical

devices and theoretical perspectives described above was elaborated into an interpretation of the participants' narratives, focusing on selected extracts. Consistent with a narrative-based approach, the data analysis is presented through several individual participant narratives. As part of this process a pen portrait was created for each participant, including a summary of their main storylines and the images discussed in the interview. The portraits for participants who are not featured individually within the empirical chapters are attached in Appendices J-Q. The analysis was organised around three key narrative interpretations of the data, exemplified through extracts from the individual participants' accounts which are presented and discussed within the following empirical analysis chapters.

Chapter 4: Negotiating uncomfortable spaces

As discussed in the preceding chapters, a psychosocial process account of experience has been integrated with a narrative-psychological approach in this research to explore relationships between space and affective experience in secure mental healthcare environments. The visual-qualitative study design combined verbal and visual research methods to explore staff and patient stories about their experiences in secure settings and the spaces where events take place. Participants produced photographs to reflect their experiences of the everyday environment for discussion during a semi-structured interview and a narrative-based method of analysis was employed to interpret the data, drawing on Murray's (2000) narrative levels of analysis. Participants' main storylines were collated across the overall data set, as summarised diagrammatically in Figure 3 in terms of 'physical', 'social' or 'symbolic' aspects of environmental experience. Whilst the storylines typically reflected a combination of all three aspects, this illustrative visual overview reflects the prominent focus of each storyline.

The analysis will be presented across three chapters and each is organised around a key interpretation of the data. Each chapter focuses on a set of narratives that are explored as individual 'case studies'. Participants' individual storylines commonly overlapped across the overall data set and the choice of extracts selected for inclusion in the analytical chapters represent examples of those that were most illustrative and able to address the research questions. The narratives were also selected to include a mix of staff and patient perspectives within each chapter. This chapter focuses on stories told by participants about negotiating uncomfortable spaces within the environment. The analysis presented in Chapter Five will focus on participants' accounts of engaging with spaces or objects to create strategies for coping with difficult affective experiences within secure hospital environments. Chapter Six is concerned with how spaces and objects in the everyday environment might affect social interaction and relationships between and amongst staff and patients.

Figure 3 Summary of participants' main storylines



The analysis presented in this first empirical chapter explores stories told by participants about experiences of physical and psychosocial discomfort in relation to spaces or objects in the everyday environment. The analysis focuses on the narratives of three participants, Leon, Carl and Wendy. Carl is a patient on a medium secure ward, whilst Leon and Wendy are based in low secure settings, as a patient and staff member respectively. The participants' accounts also illustrate how experiences of negotiating uncomfortable spaces in the environment may be shared by both patients and staff.

4.1 Leon

Leon is a young man who at the time of meeting was detained in a low secure ward from which he had not been granted leave for several months. Leon had recently moved to the current ward from another low secure environment in the hospital and prior to this had also spent time in medium secure and prison settings. He describes having had very difficult experiences in early life, referring to himself as someone who had grown up fighting for his life. Leon mentions he grew up in a different town to the place where the hospital is located and describes having moved many times throughout his life.

From the outset of the interview Leon describes being actively concerned about patients' lived experiences in hospital and highlights how he has regularly contributed to initiatives within the organisation to promote patient perspectives. The narrative conveys a sense of how Leon's ongoing concern with seeking to improve patient experiences is reflective of his own determination and strength of character. However, the general narrative tone throughout suggests that he has become disheartened and exasperated to an extent, perceiving that his voice as a patient is not being valued or heard, specifically in relation to perspectives about the ward environment. Leon's overarching narrative is thus imbued with a tone of frustration relating to how he experiences his everyday surroundings and a sense of increasing desperation about his

current personal situation, having been detained on the ward for a significant time without any leave.

Whilst Leon's narratives about his experiences are intertwined, his main storylines focus around: (1) negotiating uncomfortable spaces, (2) perceiving a lack of choice or control over the environment, (3) having emotional relationships with possessions and objects, (4) needing connections with outdoor or natural spaces, (5) perceiving spaces as clinical or institutional and (6) negotiating relationships with the hospital system, staff and patients.

In the interview Leon reflected on photographs of: (a) his bedroom, (b) bedroom window, (c) en suite bathroom and (d) outdoor courtyard for the ward.

The interpretation of Leon's narrative is focused on the first, second, fourth and fifth storylines, as interwoven narratives which recur throughout the interview. Through these storylines Leon describes negotiating difficult spaces within an institutional environment over which he perceives very limited control and an acute sense of being separated from the wider external environment and nature. In the first extract that follows, Leon considers the affective responses associated with everyday experiences of his bedroom environment, which is presented as a significant emotive terrain within the overall context of the ward.

Leon: Extract 1

49. R :	[] Thanks very much and thanks ever so much for taking the, um, time to take
50.	the photos, um, here and I just wondered if you could just start perhaps, just
51.	taking me through them one by one, taking your time and just to tell me
52.	something about why you chose to take those
53. L:	Yeah.
54. R:	photos and what, what meanings or feelings, um, are attached to, to those
55.	spaces?
56. L:	Um, what I'll probably do is I'll start with the ones that have caused the, what
57.	have more of an emotional response for me than the ones that, less likely to.
58. R:	Great.
59. L:	But, um, for me, I would go with the window spaces, as you can see <u>here</u>

60. R :	Mm.
61. L:	is very (<i>laugh</i>), very depressing. It's basically, I like, if you like, a good view like
62.	surroundings, trees, nature, that type of stuff.
63. R:	Mm.
64. L:	And these windows on [name of previous ward] used to open, the ones here
65.	<u>don't</u> , so you don't get fresh air, it's constant aircon, it's just <u>not</u> as nice.
66.	Especially when you wake up in the morning, you always wake up groggy
67. R :	Mm.
68. L:	'cause the aircon. But um, it's like, having a wood panel to look at right, soon as
69.	you wake up just (<i>slight pause</i>), so I've got to that point now, where a lot of the
70.	time I just cover it up.
71. R :	Mm. What is it that you're looking at there?
72. L:	Um, at the minute this is like, basically, just a general layout of the room and
73.	that's where the windows are.
74. R :	Uhum.
75. L:	So I thought it would be easier to have the general layout, then, when like these
76.	type of pictures, will fit in these here.
77. R:	The detail, yes. No exactly. And what's, what's the view from the window?
78.	What is it?
79. L:	It's a wooden panel. Basically it feels like you're buried underground, it's easy
80.	way to describe it. Um, it's about 35-degree angle going up, and all it is, it's
81.	literally dirt in between these two panels, you can see there.
82. R:	Mm. So, it's like a retaining, er,
83. L:	Yeah.
84. R :	wall for the, for the, the grounds?
85. L:	Yeah. Well, it's not nice.
86. R:	Mm. Yeah and is, are all the rooms, um?
87. L:	On this ward, yes.
88. R:	Mm.
89. L:	A lot of us, keep our like blinds down and we cover our windows 'cause it's
90.	just, it's even, having a dark room's like, can be depressing for some people, but
91.	it's less depressing than looking at <u>that</u> . To be fair, when we came over that's
92.	probably the main thing that annoyed a <u>lot</u> of us.

The window is a dominant aspect of the bedroom space in the images produced by Leon and its significance is reflected in his choice to highlight this at the start of his narrative journey through the photographs. Leon's account draws on wider societal understandings of windows as commonplace architectural elements that are typically perceived positively and generally provide a sense of connection between internal and external spaces. Windows are widely understood to afford outward views from an interior, or to provide access to natural light and Leon's narrative builds on commonly held beliefs that, "a good view" from a building may contain greenery and natural elements, including trees⁸ (R. Kaplan, 2001). These assumptions are disrupted by the image of timber slats and soil directly adjacent to the bedroom window and Leon's description of having this view as, "soon as you wake up", suggests this to be an especially poignant experience within the space where he begins each day. In contrast to wider cultural understandings of opening up bedroom curtains in the morning representing daylight and the optimism of a fresh start to the day, the imagery used by Leon instead positions the window content as "depressing" through its dark and imposing influence over the space.

Figure 4



Leon's photograph of the bedroom window

⁸ Whilst they are not visible through Leon's window, there are many mature trees situated within the hospital's grounds.

Within this extract, the researcher's repeated questioning to clarify the content of the window view conveys a sense of it presenting an unfamiliar and unexpected image that is not readily identified or associated with the bedroom context.

The metaphor used to suggest that Leon's embodied experiences are like being "buried underground" creates a disturbing image of the external landscape as an enveloping and smothering force. This imagery thus evokes a sense of threat being associated with Leon's dislike for the window view (Ulrich, 1983) and highlights his perceived lack of control over the ward's physical environment (Lawson et al., 2003). Through the horrifying narrative image of potential suffocation being inflicted by the built environment, Leon can also be seen to be making a powerful statement about his interpretation of the hospital as an overpowering institutional force and his own relationship to it as a detained patient.

Leon's narrative draws on broader societal understandings about light and health, including widespread perceptions that environments with poor light quality can have the potential to influence mood or be "depressing". Alongside the benefits of sunlight and fresh air for controlling infection in hospital environments (Hobday & Dancer, 2013), sunny bedrooms in psychiatric inpatient settings have also been found to reduce length of stay for patients with low mood, when compared with dull rooms (Beauchemin & Hays, 1996; Benedetti et al., 2001). Daylight exposure regulates the circadian rhythms that can improve patients' quality of sleep and accordingly, design recommendations drawn from studies examining light quality in healthcare facilities include positioning windows to maximise exposure to early morning sunlight (Joseph, 2006). Mental health problems are typically associated with disturbances in sleepwake cycles (Foster & Wulff, 2005) and the poor sleep commonly experienced by patients in psychiatric inpatient settings can also be attributed to the sensory qualities of the ward environment, including lighting conditions (Novak et al., 2020; Veale, 2019). Light plays a significant role in regulating sleep-wake cycles and evening

exposure to blue light from artificial light sources in mental healthcare settings can disrupt the circadian system and affect patients' quality of sleep (Scott et al., 2021). Accordingly, trials of dynamic lighting systems that include blue-depleted light sources in the evening have been found to improve sleep duration within psychiatric ward environments (Scott et al., 2021; Vethe et al., 2021).

By stressing a preference for minimal levels of natural light in the bedroom with the blind pulled down, rather than face the window view, Leon underscores the strength of his negative feeling towards the window and its contents. This depth of feeling is further emphasised by Leon's use of the expression, "I've got to that point now" which recurs throughout the interview and suggests that his exasperation with the environment has been heightened by being detained on the ward without leave for some time. The description of typically choosing to keep the blind down during the daytime could be interpreted as a strategy for adapting the bedroom that Leon and other patients have used to cope with and gain control over undesirable environmental attributes. In this action the materiality of the blind facilitates a sense of separation between the oppressive external environment and Leon's personal space. Control over the blind also provides Leon with some opportunity for choice, albeit a limited choice between what he expresses as two unsatisfactory options afforded by the space.

Notwithstanding the troubling content of the view, the bedroom window affords Leon a visual connection to the external environment, yet at the same time, the window represents an impenetrable barrier that prevents access to multisensorial interaction with the natural world. Previous research has similarly highlighted how the use of glass within the built environment can present barriers that 'speak' to the occupants of mental healthcare facilities and convey potentially ambiguous messages (Connellan et al., 2015). In a new build psychiatric unit, Connellan et al. (2011) observed that full-height glazing afforded internal-external connections in communal ward areas overlooking garden spaces, yet windows on a secure ward presented

patients with views of outdoor spaces that were inaccessible to them. Similarly, in another study within a refurbished psychiatric intensive care unit where patients did not have direct access to outdoor space, staff expressed concern about the view of an inaccessible rooftop garden and perceived that this offered patients the experience of, "you can look but you can't touch" (Payne & May, 2009, p. 82). The potential for windows to both offer up the natural environment to patients, whilst at the same time creating a barrier to access, is therefore argued to require careful ethical design consideration in mental healthcare settings (Connellan et al., 2011).

As a threshold between the internal world of the ward and the external world beyond, the sealed window unit thus presents a paradoxical situation for Leon, who is both connected to and yet simultaneously separated from the outside world and nature by the window. Following Stenner, Greco and Motzkau (2017), these circumstances can be elaborated in relation to the concept of a 'liminal hotspot', or a situation in which Leon is caught in a suspended state between these two worlds. By keeping the blind down and obscuring the outside world, including natural light, Leon's actions could be interpreted as a form of 'polarisation' that can characterise attempts to avoid the paradox and sense of paralysis typically experienced in a situation of ongoing liminality (Greco & Stenner, 2017). By shutting out connections to nature and the external world as an attempt to avoid an indeterminate liminal state, Leon can be understood to be forcing a polarised solution that aligns most closely with the conditions of the internal world. Leon's account provides an image of intense affectivity associated with his experience of enduring liminality and his actions can be interpreted as an approach to alleviate the intensity of his affective discomfort in the space.

The disconnect Leon perceives between internal and external spaces is conveyed throughout the interview as a multisensorial experience. The description of non-openable windows and poor air quality evokes an impression of the ward as being a sealed, almost airtight environment within which a heady affective atmosphere can

build up. Leon's account of waking up feeling "groggy" contributes to the narrative imagery of how his embodied experiences may be associated with the built environment and underlines his perceptions of the air quality as a devitalising and pervasive force within the ward. Previous research in hospital settings has similarly identified close associations between health outcomes and characteristics of the sensory environment, including air quality (Schweitzer et al., 2004; Ulrich et al., 2008). The provision of adequate ventilation in psychiatric settings can therefore optimise health by affording fresh air and also controlling unpleasant odours that may generate negative emotional responses (Karlin & Zeiss, 2006; Mazuch & Stephen, 2007), including stress, fear and anxiety (Connellan et al., 2013; Schweitzer et al., 2004). The need for fresh air, as highlighted by Leon, could also be understood to be especially pronounced in a mental healthcare setting, where patients commonly experience mental clouding or sedation associated with the use of psychotropic medication (Seale et al., 2007).

Within this extract and throughout the interview, Leon's repeated use of the phrase, "to be fair" positions the evaluation of his experiences and contribution to the research interview as measured and rational. Leon's concluding observation that the windows were "probably the main thing that annoyed a <u>lot</u> of us", conveys his depth of feeling and offers a source of validation to the researcher in suggesting that his experiences are widely shared by the other patients on the ward.

Later in the interview, Leon elaborates further about his perceptions of the bedroom and his emotional relationship with the space. The following account implies that his adaptive processes, which include manipulating the blind and rearranging possessions within the space, can mediate his affective responses to the environment by attempts to conceal its institutional qualities.

Leon: Extract 2

1.	L:	But it's still sometimes, it's like, uncomfortable. The window view's, is very
2.		like, claustrophobic. I literally have to like, I black out my windows, just to
3.		really, just let the room dark, so I don't really see all the clinical things.
4.	R:	Mm.
5.	L:	But, after a while that gets a bit boring. You have to find something else to like,
6.		spicen your room up a bit.
7.	R:	Mm.
8.	L:	I rearrange my room about 2, 3 times a week. Just because, literally, it's just so
9.		clinical (laugh). It's like this whole shelving system and those cupboards
10.		up there, it's sort of, it's what I used to have in prison.

In contrast to Leon's earlier account of his appreciation for the natural qualities of the external world, the narrative here presents an image of the ward interior being imbued with an undesired tone that Leon perceives as both clinical and carceral. The narrative reiterates Leon's perceptions of the bedroom window view as an oppressive force that can generate psychological discomfort and distress. Earlier Leon describes using the blind to cut off the troubling connection to the external world outside the window and it is noteworthy here that the associated reduction in natural light is also used to diminish the clinical qualities of the internal bedroom environment. This account of keeping the room dark to ameliorate the institutional tone that pervades the space emphasises Leon's strength of negative feeling and conveys a sense of his need to protect himself from the unhelpful cues he perceives in the built environment. The functional design of the robust joinery fittings is also understood to evoke a punitive environment that serves as a reminder to Leon of his time previously spent in prison.

His account therefore suggests that the process of re-ordering possessions and updating his bedroom is a further strategy employed in resistance to the draining forces of the institutional and carceral qualities that he perceives within the environment. Elsewhere in the interview, Leon similarly describes feeling like he is, "in prison again" when walking on the homogenous flooring material in the ward's open-

air courtyard. In contrast, he expresses a desire to experience the sensation of natural materials such as, "real grass or real turf" within that outdoor space. His accounts therefore suggest that authentic, natural materials are perceived to be psychologically supportive, whilst artificial materials and utilitarian design contribute to his perceptions of an unhelpful institutional tone being ingrained within the built environment.

In the following extract Leon expands upon the storylines already discussed and further explores the ways in which he perceives that a clinical and institutional ambience is pervasive within the ward environment.

Leon: Extract 3

1.	L:	Um, generally, we are trying to make this ward better, as, as patients we are
2.		asking for more stuff, but generally, the problem is the <u>clinical</u> and I mean like,
3.		bog standard sink and that's it
4.	R:	Mm.
5.	L:	type looks and feel to things is depressing. It annoys people. It doesn't, even
6.		though it's such a minor and I realise sub-consciously we pick this stuff up,
7.		that, it's like, if you're pissed off or you're angry, you wanna sit down in a nice
8.		comfy chair and think about things <u>carefully</u> , but most fresh, most like
9.		oxygenated room, is enjoy that fresh air, not aircon. 'Cause aircon's not helping
10.		anything.
11.	R:	Mm.
12.	L:	Normal fresh air, the smell of rain after it's rained. Just <u>simple</u> pleasures in life
13.	R:	Mm.
14.	L:	that we are cut out, we're cut off from, because this place was designed poorly.
15.	R:	Mm.
16.	L:	'Cause like [name of previous ward] is an older building, it was designed before
17.		[name of another building on the hospital site]. We had windows that opened,
18.		even though it had like a grille.
19.	R:	Mm.
20.	L:	Now, I preferred to have that little grille, than what we've got here.
21.	R:	Mm.
22.	L:	Just because, the, the smell of the rain when you wake up in the morning, it
23.		just, it wakes you up, it makes you feel <u>happy</u> . You know it's gonna be a good
24.		day.
25.		You know it's going to be good like by the weather.

- 26. R: Yeah, yeah.
- 27. L: It is, it's just, as you can see from this room, there's none of that.
- 28. R: Mm. I think we've got the same view [the orientation of the window in the room
- 29. used for the interview is similar to the bedroom, although the content is largely
- 30. *concealed by a window covering*]. Yeah.
- 31. L: Yeah. They even tried putting bird boxes up at some point just to try and entice
- 32. animals, so make it look more inviting, but, even the birds don't come, so that
- 33. explains everything. (*Laughs*). The animals aren't willing to come to this part.

Following on from the previous extract, this narrative continues to build up an impression of the overall ward space being permeated with a sense of medical starkness. Whilst Leon comments that patients are, "asking for more stuff" to enrich the ward, his perception of a "clinical" quality being inherent in the material and symbolic fabric of the space repeats throughout the interview and is presented as being problematic and restricting the potential for environmental improvement. This story suggests that the physical attributes and aesthetic components of the environment including furniture and fixtures, convey meaning to occupants and combine with atmospheric conditions to set the institutional tone that Leon perceives throughout.

Figure 5

Leon's photograph of the bathroom


Leon's account of the "bog standard sink" in his bathroom evokes the sense of a utilitarian or cost-driven approach to the specification of fixtures, which in turn implies a lack of care or consideration for the end user and the prioritisation of functionality over aesthetics. Leon's observations hence suggest that his bathroom environment and the design of the sanitary fittings within it are communicating with him more widely about how patients might be valued or regarded within the mental healthcare system.

Within this narrative, Leon's description of desiring a "comfy chair" evokes imagery of domesticity and ease. His account suggests that perceptions of comfort or homeliness within the material or atmospheric characteristics of an environment can be psychologically supportive and have the potential to assist with relaxation (Duque et al., 2019). In contrast, Leon's observation that the air conditioning is, "not helping anything" implies a sense of the environmental conditions and air quality being unsupportive and actively working against him by hindering the potential to experience comfort and in turn, his ability to relax.

Leon's account of appreciating the experience of "fresh air" and "the smell of the rain" through an open window expands the storyline of his desire for greater sensory connection between the ward interior and the external environment. Through these examples Leon illustrates his perceptions about the potential to gain pleasure or vitality from modest and ordinary sensory experiences. Whilst such everyday interactions with the natural world may commonly be taken for granted in other contexts, Leon highlights that their absence is especially significant when detained as a patient without leave within this ward environment. Leon's description of being "cut off" from these experiences emphasises the image of sensory isolation imposed by the architectural features of the building. Whilst the account suggests that window grilles were perceived as undesirable features of the previous ward architecture, Leon's statement of preference for grilles over the current non-openable windows underlines the extent of his wish to experience greater sensory connection with external spaces.

When considered in conjunction with his earlier account of the non-desired visual content of the bedroom window, Leon's narrative also underscores the significance of his wish to have sensory connections with the external environment that are beyond purely visual experiences. Leon reiterates his desire for openable windows to provide fresh air elsewhere in the interview and there he also describes how his room can be "very stuffy" and "like a sauna sometimes". The imagined multisensorial experiences which might be made available via an openable bedroom window are presented as being invigorating and Leon's account implies that being attuned with, or having exposure to the external environment and nature can enhance his mood and perceptions of vitality (Iyendo et al., 2016; Ulrich et al., 2008). The narrative also highlights the positional contrasts between the researcher, who is permitted to freely access such experiences and Leon, who has been detained without leave from the ward for a significant length of time and for whom the need for connection with external spaces is now especially pronounced.

Dry humour features throughout the interview and is used by Leon to emphasise key points within the narrative. Leon's wry conclusion that, "even the birds don't come" and "the animals aren't willing to come to this part" to make their homes, despite attempts of enticement, is presented as evidence to the researcher of how the environment may be more widely perceived as unnatural, inhospitable, or nondomestic. As with his earlier reflections about feeling "buried underground" by the oppressive window view, Leon can be understood to be making a statement here about the broader messages he receives from the built environment regarding his place as a patient within the institution. Accordingly, his observation that "even" animals feel uncomfortable in this space evokes an image of how the physical environment might potentially convey a message to patients that they are regarded as less than human.

4.2 Carl

At the time of meeting Carl was detained in a medium secure setting and had recently been granted leave from the ward for the first time in several months. Carl is a young man who has been a patient within mental health services for a number of years, however, he is hopeful that he will soon be discharged. The overall tone of his narrative conveys a sense of resilience and Carl describes having struggled with many aspects of being detained in hospital and his journey through the mental health system. Whilst the narrative is infused with a range of Carl's emotional responses to his experiences of being in hospital, including fear and frustration, it contains an underlying tone of optimism and a sense of looking forward to future possibilities. The hospital is located at a significant distance from Carl's family and his concerns and considerations about the future and life beyond hospital include the potential of moving in with his partner.

The level of nuance and detail included within Carl's accounts of the social, atmospheric and material qualities of the ward imply that he is a perceptive and sensitive person who is attuned to the subtleties of ward milieu. His stories of lived experiences within the hospital environment commonly draw on visual metaphors that are used to convey his emotional responses to everyday spaces. Using this narrative device throughout, Carl also builds up an impression of his affective ambivalence towards many of the spaces encountered within the hospital.

Throughout the narrative Carl presents himself as an imaginative and creative person who enjoys artistic activities, including designing and making hand crafts. Creativity is described as a form of self-expression and the narrative also suggests that Carl's creative activities have a therapeutic quality, in addition to facilitating a sense of connection with other people.

Carl's main storylines focus around: (1) negotiating uncomfortable spaces, (2) being attuned to the environment, (3) negotiating the hospital environment and

system, (4) using space as a distraction or escape, (5) connections between space and social relationships, (6) creativity and self-expression.

Carl reflected during the interview on photographs of⁹: (a) crafts room, (b) offward café, (c) off-ward café courtyard, (d) hospital corridor, (e) hospital grounds, (f) hospital grounds juxtaposed with the large hospital building, (g) occupational therapy kitchen, (h) day area skylight, (i) his bed with homemade blanket, (j) his desk, (l) his bedroom door, (m) ward telephone booth, (n) ward courtyard and (o) day area.

The interpretation of Carl's narrative will focus on the first four interwoven storylines, through which he expresses an awareness of associations between his emotional experiences and the everyday hospital environment and describes ways in which he negotiates these spaces and the hospital system overall.

Carl: Extract 1

1.	C:	Er, the next one is a shot of the, the skylight in the day area. Which is a really,
2.		looks lovely, lovely to have the daylight shining in, floods the room full of light,
3.		but then the next shot, is a z-, a close up of just how shabby it is at the edges
4.		and for me it sums up my entire experience of being, just, of being in the, in the
5.		environment. Everything looks great until you look a bit deeper. And it just, it
6.		looks like, for me, it sums up how much is available to us and how good the
7.		service can be, but then if you just zoom in and just take it a little bit closer, you
8.		can see, the edges are, aren't there, it's not refined, it's not right, it's just,
9.		shoddy and dirty. And that for me, that sort of, these two shots in parallel
10.		convey.
11.	R:	Mm. Yeah. And do you think that, that has an impact on um, how you feel
12.	C:	Yeah.
13.	R:	within, within the spaces?
14.	C:	Definitely, when you're sat, when you're sat there and you look and you think
15.		it's a lovely thing and you notice the extra details, it's just, it just bugs you, just
16.		something else to play on, you're sat there, to play on your mind.
17.	R:	Mm.
18.	C:	I'm a big day dreamer, I love sitting on the floor looking up at the ceiling, at the

⁹ The photographs taken by Carl were reviewed on the camera screen during the interview as they could not be printed. The images are not available for reference here as the digital files could not be transferred.

- 19. clouds when it's a nice day.
- 20. **R:** Yeah.
- 21. **C:** I just find it really relaxing. But then the little details, for me just ruin it. But it
- 22. depends on what you focus on, if you're on a good day, you see the whole nice
- 23. of it, you're having a bad day you focus on that little detail.

In this extract Carl's narrative is focused on his emotional responses to the skylight as a predominant architectural feature of the day area and the perceived qualities that this affords to the space. The account draws on wider public understandings about the vital properties of natural light and common perceptions of sunlight specifically as an enriching attribute for interior spaces, with the capacity to potentiate positive mood (G. W. Evans, 2003; Joseph, 2006; Ulrich et al., 2006, 2008). Carl's description of "daylight shining in" that "floods the room full of light" evokes a sense of openness and abundance, in which natural light pours in fluidly from the exterior to saturate the internal space. The imagery of reflectivity and brightness in the space also builds on wider understandings of associations between sparkling, polished surfaces and perceptions of cleanliness and newness. Through this imagery, Carl presents the bright, optimistic qualities afforded to the interior by daylight coming through the glazing, in stark contrast to his alternative perceptions of the skylight edge detail being "shabby", "shoddy" and "dirty".

Alongside his literal use of a camera lens to zoom in and out of the spaces discussed, Carl's reflection on how, "everything looks great until you look a bit deeper", also evokes the use of a metaphorical lens to shift between contrasting viewpoints in his appraisals of the overall institution. This motif of zooming between macro and micro perspectives recurs as a narrative device throughout the interview and is powerful in conveying Carl's sense of ambivalence towards the hospital environment. Spaces are positioned ambiguously as being neither completely one thing nor another, with the potential to be perceived both positively and negatively depending on the affective choice of lens and the specific spatial attributes that are brought into focus. Throughout the interview Carl highlights how several other areas, including the telephone booth and the view from his bedroom window, all possess a sense of affective ambivalence that he underscores with a recurring metaphor of his relationship with the hospital environment being like a "double-edged sword". Carl thus implies that the spaces he inhabits can be appreciated positively for their affordances, such as creating connections to aspects of nature or to family and friends, whilst also being perceived simultaneously as unpleasant, unattractive or lacking in quality. Carl's concluding statement that, "if you're on a good day, you see the whole nice of it, you're having a bad day you focus on that little detail", also proposes an association between aesthetic perceptions of the ward environment and mood states. The narrative therefore implies that broader, more optimistic perspectives and impressions of the environment and the institution as a whole may be available when Carl is feeling good, whilst low mood may magnify or encourage an emphasis on the perceived flaws.

Carl's accounts of his ambivalence towards the hospital environment suggest that through his relationships with different spaces he finds himself continually caught in an 'in-between' position between affective states. Spaces are valued for facilitating an enjoyable sense of connection and opening up to the wider world, but at the same time they are also perceived as uncomfortable or constraining. Whilst the everyday environments Carl inhabits can be supportive and pleasurable on one hand, they are also typically troubling and disquieting on the other. The narrative thus suggests that Carl is suspended in an ongoing state of liminal affectivity (Stenner, 2013), perpetuated by his interpretations of the hospital environment's material and symbolic characteristics. Carl's experiences of the 'both/and' and 'neither/nor' qualities within spaces on the ward and the enduring ambivalence associated with his affective responses are therefore understood to generate unhelpful tensions that can contribute to his distress and limit perceptions of progression and psychological comfort.

Carl's metaphor of the bright, shining environment representing opportunities and "how good" he perceives the mental health service can be, alongside his positioning of an unkempt skylight detail as being "not refined" and "not right" implies that appreciation of material detail is also associated with wider perceptions of quality and integrity. The flaws perceived within this architectural detail can therefore also be understood to be speaking more broadly to Carl about the shortcomings and limitations of the overall institution. The narrative suggests that perceptions of aesthetic purity and refinement within physical surroundings can be gratifying and convey value, whilst details perceived as "shoddy" or "dirty" are symbolically associated with a sense of imperfection and lack of value or care. Correspondingly, through his appraisal of the skylight detail as lacking in quality or aesthetic consideration, Carl can also be seen to be receiving a wider message from the built environment about how patients might be valued or regarded by the institution. His accounts of defective or unclean aspects of the physical environment also present an image of aesthetic contagion that can be elaborated in reference to the sociologist Erving Goffman's (1961/1991) pioneering ethnographic research within psychiatric settings. Goffman argues that patients may experience a sense of "contaminative exposure" (p.31) in relation to the violation of boundaries between the self and the environment following admission to an institution. Physical forms of contamination might therefore be associated with the lack of privacy and constant scrutiny experienced by psychiatric patients in institutional settings, or with their exposure to unclean objects or dirty communal facilities. Drawing on Goffman's thinking, the awareness of environmental impurity described here by Carl and elsewhere in the interview, including when he describes how, "some of the corridors can smell of urine" can thus be interpreted as forms of contaminative exposure imposed by the built environment.

Within this extract, Carl's account of how his awareness of defects within the physical environment creates psychological discomfort accordingly highlights how negative perceptions of spaces and objects on the ward can be unhelpful and unsupportive to occupants' mental health. The description of a perceived environmental flaw as being something that, "just bugs you" suggests a sense of pervasive irritation that Carl does not have the option to resolve for himself and emphasises how inpatients typically experience very limited control over the ward environment (Csipke et al., 2016). Carl's account of his perceptions of the space playing on his mind while he is sitting within it positions the material environment as an actively intrusive force due to the psychological friction it can induce. Furthermore, the narrative also underscores how details within the physical environment are especially significant in secure settings, where patients have restricted access to varied visual or multisensorial landscapes and regularly spend long periods of unstructured time within the same spaces on a ward (Farnworth et al., 2004; O'Connell et al., 2010). In contrast to his experience of affective discomfort in relation to the imperfections identified within the built environment, Carl's appreciation of cloud-gazing as a "really relaxing" activity suggests there is a soothing quality and lack of friction within this natural scene that contributes to a positive affective state (S. Kaplan, 1995). Through the description of sitting on the day room floor to gaze up at the clouds, Carl presents an image of behaviour more commonly undertaken outdoors, whereby sitting or lying on the ground is typically associated with leisure activity or relaxation. Interpreting Carl's choice to sit on the floor in an internal space whilst looking up at the sky could suggest that having this perspective and connection to the ground provides an embodied association with the experience of relaxation in outdoor spaces and may assist his ability to relax in this setting (Berto, 2014).

Throughout the interview, Carl talks about ways in which he experiences a sense of physical and psychological discomfort within spaces on the ward, including a

lack of privacy. In the following extract, when reflecting on his perceptions of privacy in the hospital environment, Carl identifies his bedroom as one of the spaces that affords him the greatest sense of being private. However, he is also prompted to tell a story of feeling threatened and distressed within this space by an incident taking place outside his bedroom door.

Carl: Extract 2

1.	R:	[] Um, just thinking about the environment here, um, again and just thinking
2.		about the spaces, I was just wondering if there were any spaces here that you
3.		would sort of consider to be private at all?
4.	C:	Um, my room's probably one of the most private environments. It seems a bit
5.		daft, but sometimes if I'm in my room and I get really scared, like the other
6.		week when the door was being kicked in for a while, I kind of locked myself in
7.		my bathroom because it's one extra door (<i>laugh</i>) away from everyone and it's,
8.		I like that it's smaller. And for me I like, I do like smaller environments, I like,
9.		but not <u>too</u> small.
10.	R:	Mm.
11.	C:	So like, the phone room I find really small, but the bathroom here is just
12.		smaller than my room, with an extra layer of protection of another door.
13.	R:	Yeah, yeah.
14.	C:	But, I don't feel that many of the rooms are private in any degree, 'cause people
15.		are always popping in and out and not knowing what's going on and
16.	R:	Mm.
17.	C:	I'd rather just not be in here to talk to anyone privately.
18.	R:	Mm.
19.	C	
	L:	I know that at any moment someone could just ping that door open. But I know
20.	C:	I know that at any moment someone could just ping that door open. But I know they're not gonna know anything, I know they're not gonna hear anything, in
20. 21.	ι:	
		they're not gonna know anything, I know they're not gonna hear anything, in
21.	R:	they're not gonna know anything, I know they're not gonna hear anything, in that moment, but there's always the potential.
21. 22.	R: C:	they're not gonna know anything, I know they're not gonna hear anything, in that moment, but there's always the potential. Mm. Yeah. You mentioned about the size of the space,
21. 22. 23.	R: C:	they're not gonna know anything, I know they're not gonna hear anything, in that moment, but there's always the potential. Mm. Yeah. You mentioned about the size of the space, Mm.
 21. 22. 23. 24. 	R: C: R:	they're not gonna know anything, I know they're not gonna hear anything, in that moment, but there's always the potential.Mm. Yeah. You mentioned about the size of the space,Mm.could you tell me a little bit more about, um, the sort of, the feelings that you
 21. 22. 23. 24. 25. 	R: C: R: C:	 they're not gonna know anything, I know they're not gonna hear anything, in that moment, but there's always the potential. Mm. Yeah. You mentioned about the size of the space, Mm. could you tell me a little bit more about, um, the sort of, the feelings that you get from those different sort of scales of environment?
 21. 22. 23. 24. 25. 26. 	R: C: R: C: R:	they're not gonna know anything, I know they're not gonna hear anything, in that moment, but there's always the potential. Mm. Yeah. You mentioned about the size of the space, Mm. could you tell me a little bit more about, um, the sort of, the feelings that you get from those different sort of scales of environment? For me, if the corri-, if
 21. 22. 23. 24. 25. 26. 27. 	R: C: R: C: R:	they're not gonna know anything, I know they're not gonna hear anything, in that moment, but there's always the potential. Mm. Yeah. You mentioned about the size of the space, Mm. could you tell me a little bit more about, um, the sort of, the feelings that you get from those different sort of scales of environment? For me, if the corri-, if (Coughs).

- 31. be a bit homely, a bit snugger, tend to be a bit too vast.
- 32. **R**: Mm.

33. C: But then I like the freedom in the courtyard's vast and open, feeling like you're34. not trapped in anymore. But um, the day area can almost feel like you've got an

- 35. ant in a large dome. It's free to go wherever it wants within the parameters of
- 36. the dome. But the courtyard feels different because it's outside. It's hard to
- 37. explain.

Carl's account draws on widespread perceptions of doors as dynamic architectural features that are operated to facilitate connection or separation between spaces. Whilst doors may commonly be used to control levels of social interaction or privacy, Carl's narrative positioning of the doorways on the ward as sites of potential intrusion, both from patients and staff members, accentuates the limitations of his perceived control over the environment and the actions of others. Although the person outside his door cannot physically enter the bedroom, the account highlights how Carl is unable to control the sounds or vibrations transmitted into the space via the materiality of the door and his associated fear responses (Stansfeld & Matheson, 2003).

Carl's story of retreating to the bathroom emphasises the degree of his affective response and the significance of the situation as an undesired intrusion on his personal space. The narrative creates an image of the hospital as a series of nested spaces ranging from public to private and within this continuum, Carl perceives the en suite bathroom, as one of the smaller and most deeply nested rooms, to be one of the more private and in turn, one of the safest spaces. Closing the door to complete the bathroom volume as an enclosed room-within-a-room allows Carl to perceive reassurance from the "extra layer of protection" represented by the physicality of the door. Carl's account evokes imagery of doors as forms of physical and psychological defence, yet at the same time it positions doorways as points of vulnerability and potential exposure, depending on who has control of the door. When expressing concern that, "at any moment someone could just ping that door open", Carl presents an image of how power relations between staff and patients can be directly expressed through the materiality of the ward. This comment also conveys a sense of his awareness towards potential threat within the ward environment which recurs throughout the interview.

Carl's account therefore highlights how intense affective experiences can be associated with doorways on the ward that signify physical and symbolic thresholds between different spheres of activity that occur either side. Doors can represent a border between different worlds, including the contrasting worlds of patients and staff activity or between environments and circles of activity that might be understood to be 'public' or 'private'. Whilst a closed door can create separation between different worlds, an open door can instantly bring two worlds together. Having control of a door can both enable or limit transition across a boundary and accordingly, the narrative implies that Carl's perceptions of having a lack of privacy within a secure psychiatric setting are associated with his experiences of lacking a sense of environmental control. It is also noteworthy that earlier in the interview the conversation with Carl was interrupted by a staff member opening the meeting room door to speak with him. At this point the interview recording was briefly paused, however, it is significant that he was not free from unpredictable intrusion even when participating in a research interview.

In his further reflections on associations between affective experiences and environmental scale, Carl's narrative imagery of a tiny ant moving within a big dome indirectly positions his own experiences of the day area as being like that of the ant. The dominant scale of the dome in his account also underscores his perceptions of the day room space as being "vast" and oversized. By positioning the restrictive dome as a metaphor for the building, Carl expresses his perceptions of power relationships between patients and the mental health system being symbolically embedded in the material structure of the hospital. Drawing on Foucault's (1975/1991) characterisation of psychiatric institutions as the embodiment of discipline and control over social deviancy, the power of the institution is understood to be being communicated to Carl

through the physicality, scale and proportions of the building. This symbolic link between environmental scale and perceptions of power is similarly expressed earlier in the interview in relation to Carl's awareness of the high walls around the courtyard. His account of experiencing the walls as dominant and overbearing highlights how the scale of the institutional architecture is able to position him and other patients as what he describes perceiving to be, "just a small part" of the overall healthcare system.

Such references to scale and proportion within the narrative suggest that an appropriate match of scale to function can contribute to perceptions of psychological comfort within spaces. The perceived oversizing and therefore non-domestic scaling of areas such as the day room and hospital corridors is positioned as a spatial attribute which contributes to the sense of unease that Carl experiences within those spaces. Whilst Carl's reflection that the communal areas should be "a bit snugger" evokes imagery of potential cosiness and a homely environment, the language also conveys his perceptions that the sizes of these spaces are not a comfortable fit and could be better tailored to suit the function and the users of the spaces.

This sense of congruence between scale and function contributing to perceptions of physical and psychological comfort within spaces is also conveyed through Carl's stories of his embodied experiences within smaller scale spaces on the ward. Whilst his account of the bathroom suggests that a sense of reassurance and protection can be afforded by small spaces, Carl's observation that some spaces can be "too small" presents an image of the relationships between scale, function and environmental conditions creating perceived thresholds of comfort within spaces. Carl references the telephone room as one of the smallest spaces encountered and earlier in the interview, his reflections on a photograph of that space further expound his sense of affective ambivalence in response to different spaces on the ward. Although Carl appreciates the room positively as a place to connect with friends and family, his account of its very small scale, combined with intrusive noise from the lounge and

conditions he perceives as "really hot and dirty", convey his parallel experience of embodied discomfort within the space.

Whilst vastness is presented as a problematic attribute within some of the internal spaces, including the day area, it is a quality that Carl appreciates perceiving in the external courtyard. The glazing in the day area offers a sky view, yet Carl's account presents the open-topped courtyard as able to afford a sense of liberation which is not available through the glazed internal skylight. Carl values the way in which the skylight in the day area creates connections between the ward interior and natural elements, including sunlight and clouds in the sky, yet the glazing also represents a symbolic overhead barrier that contributes to his perceptions of containment. Conversely, despite the high enclosing walls around the courtyard volume, there is no physical barrier within this outdoor space between Carl and the sky far above. By contrasting the image of the internal day area as an impenetrable dome with his perceptions of "freedom" and sense of not being "trapped" in the external courtyard, the account also implies that access to fresh air and direct sensory connections to nature are significant to Carl's perceptions of psychological comfort.

In the following extract Carl further elaborates on his perceptions of environmental scale and atmosphere in relation to his embodied experiences within the communal day area.

Carl: Extract 3

1.	C:	[] It just never feels nice out there, it just feels eerie. Just too vast, too open,
2.		not home, not homely.
3.	R:	Mm. Could you say a little bit more about what features of it, um, might
4.		contribute to those, those feelings?
5.	C:	It's a large space, where you're constantly being watched from every angle,
6.		with the different rooms looking on, into, it's almost like an atrium, it's not
7.		almost like a room.
8.	R:	Mm.
9.	C:	Everything feeds off and you've, it's, you've got people coming and going

- 10. in and out, which isn't good to just sit and watch something and concentrate
- 11. on the TV or in a conversation. You're always worried about what's behind
- 12. you if that door slammed and what's this, what's that.

Carl's account of the day area as "eerie", "open" and "vast" presents an image of a strange and discomforting environment and suggests that scale and atmosphere contribute to his perceptions of it as a "not homely" space. Carl's description of being "out there" in the day area, "with different rooms looking on", evokes a sense of exposure and being out on display in the space, which underscores his perceptions of being subject to continual surveillance, "from every angle". Through his account of the day area as a space in which, "everything feeds off", Carl presents an image of the environment as a busy thoroughfare without clearly defined boundaries. The narrative implies that the lack of clear spatial definition also contributes to Carl's perceptions of the day area as a liminal space without a clear identity. In contrast to his references to "home" and "homely", which prompt images of domestic living spaces, Carl's account of the day area as, "almost like an atrium" evokes imagery of public lobbies and large-scale corporate architecture. The narrative imagery therefore positions the day area as an unbounded space in which it is difficult for Carl to relax and where he remains vigilant towards potential threat from all around.

Carl reiterates an account of the day area's oversized scale elsewhere in the interview and the narrative implies that his perceptions of the space as "vast" and "not homely" are also associated with its poor acoustic qualities and a non-domestic arrangement of furniture. Carl perceives that the uncomfortable noise in the day area is associated with, "how far everything can be away from one another, the chair at one end of the room and a chair at the other" and his account implies that furniture spacing affects communication, such that his experience is, "almost like having to shout across". Noise issues are common in hospitals (Busch-Vishniac et al., 2005; Holmberg & Coon, 1999) and poor acoustics in these environments are typically aggravated by the

prevalence of smooth, sound-reflective surfaces that can cause different sounds to echo, overlap or produce long reverberation times (Joseph & Ulrich, 2007). Associations identified between noise sensitivity and mental distress (Stansfeld, 1992; Sutton & Nicholson, 2011) suggest that occupants' experiences of sound and the overall acoustic environment are especially pertinent to psychiatric settings. Ambient noise levels are a significant stressor and incorporating acoustic materials into the design of secure wards can control sound and mitigate levels of aggression that may occur amidst consistently high levels of noise and raised voices (Dvoskin et al., 2002; Hunt & Sine, 2017).

Large, open day areas, as described by Carl, are commonly connected to long echoic corridors, however, since highly reverberant spaces may exacerbate the perceptual distortions experienced by some patients (Karlin & Zeiss, 2006), acoustic control is a particular consideration in psychiatric settings. Clinicians involved in the design and development of medium secure psychiatric units have also suggested that facilities should ideally possess, "a comfortable, domestic scale and ambience" (Watson, 1998 p.524) to optimise the health and well-being of occupants. Watson highlights how the rationale for this is not limited to the creation of physically comfortable or pleasing spaces, but additionally it aims to induce perceptions of intimacy and mutual trust that are vital to therapeutic relationships yet may typically be lacking in patients' lives.

4.3 Wendy

Wendy is a mental health nurse and at the time of the interview she was working in a low secure ward environment. Wendy has previous experience working with patients in acute inpatient mental healthcare facilities and she notes that it had taken her time to adjust to working with a new patient group and adapt to working within a secure setting. Throughout the narrative Wendy presents herself as a conscientious, thoughtful and empathetic person who is concerned about the welfare of patients and colleagues with whom she works.

Wendy's stories position her in a busy, challenging and frequently stressful role of care and responsibility. Alongside duties relating to her own ward, she is at times also responsible for coordinating staffing and incident responses on other wards across the hospital site. Wendy's narrative conveys an underlying tone of stoicism and professionalism, yet it is also mixed with a sense of some frustration and concern in relation to staffing levels and her perceptions of how limited staff resources can affect the everyday experiences of patients and other members of staff.

The overall narrative implies that Wendy is very busy managing the different aspects of her role at work and that her time is generally in high demand. This is further emphasised at one point in the interview when it is necessary to pause the recording for a few minutes when a colleague knocked on the door to consult with her. Wendy's storylines focus around: (1) managing professional challenges and stress, (2) negotiating uncomfortable spaces, (3) managing her own and colleagues' emotions within her professional role, (4) having emotional responses to spaces and objects, (5) navigating boundaries and relationships with patients.

In the interview Wendy discussed photographs of: (a) walkie-talkie unit and bleep, (b) seclusion room intercom, (c) seclusion room door, (d) poster with list of items that are prohibited on the ward, (e) clinic room door, (f) poster on exit door noting that, "all staff and visitors must notify safety nurse when they are leaving the ward" and (g) door signage noting that, "only staff are to enter this room".

The following interpretation of Wendy's narrative focuses on the first, second and fifth intertwined storylines, through which Wendy articulates the complexity of her role, alongside ways in which she navigates her own emotions and relationships with other people in an all-consuming environment. Through these narratives, Wendy highlights similarities between the affective responses of staff and patients to spaces on the ward, whilst also emphasising how the unequal status of staff and patients within the institution contributes to significant contrast and disparity in their experiences of

space. In the following story, Wendy conveys a sense of the everyday challenges of her role and the embodied experiences of discomfort associated with working inside a secure ward environment for long periods of time.

Wendy: Extract 1

1.	W:	Um, just that I can feel quite, um, sometimes if I've been here for a long, if I've
2.		been here, kind of, quite often I don't get a break (<i>laughs</i>). Um, so and I do
3.		twelve-hour shifts, so it gets to a point in the day where I have to get out.
4.	R:	Mm.
5.	W:	I have to go and get some fresh air, because I can feel quite, um, what's the
6.		word I'm thinking of? Anxious, um, like I need fresh air.
7.	R:	Mm.
8.	W:	Because, because we're downstairs, we don't, we're not getting any natural
9.		sunlight or anything like that.
10.	R:	Mm.
11.	W:	Um and I feel like I need to get off, just for kind of five, ten minutes. Sometimes
12.		it can feel quite intoxicating in here.
13.	R:	Mm, mm.
14.	W:	Especially when, because, out in the office, it is like a goldfish bowl. You know,
15.		you, patients know exactly what you're doing, yes, I understand it needs to be
16.		like that so we can see what's going on, um, but staff can also, patients can also
17.		be very aware of what you're doing, um, also if there comes, patients will
18.		constantly count how many staff there are in the office. Because that will then
19.		determine to them, this is what they think, if they can go out on leave for a
20.		walk. They don't seem to take into consideration that actually my other nurse
21.		is doing medication, she's about to do medication, I've, I'm coordinating, my
22.		Healthcare support workers are busy doing safety checks. They're not
23.		interested in things like that, so they're constantly harassing you to go out.
24.	R:	Mm.
25.	W:	Um and sometimes you feel like, I've just gotta go out for five minutes, gotta go
26.		outside, need some fresh air and kind of and that, that usually works fine. Once
27.		I've been outside for five, five minutes I can come back in.
28.	R:	Yeah.
29.	W:	Um, but I'm always very grateful to leave at the end of the day (<i>laughs</i>).

Through highlighting the length of her shifts at work, Wendy's narrative positions her everyday lived experiences on the ward as typically intense and demanding. Wendy

emphasises a sense of commitment to her role through her account of frequently being too busy to take a break and her laugh following this statement suggests that the lack of break during a long shift might potentially surprise the researcher. Wendy describes feeling a sense of anxiety after spending a length of time within the ward environment and her use of the phrase, "it gets to a point in the day where I have to get out", evokes the sense of a regular and somewhat inevitable daily build-up of embodied distress that requires a form of release to allow her to go on working. By stressing her need for "fresh air" and highlighting the lack of sunlight in the "downstairs" space, Wendy presents an image of the ward as having gloomy, enclosed and semi-subterranean qualities, from which she feels compelled to emerge, as if coming up from below the surface for air and light. Wendy's perception of the "intoxicating" environment evokes the sense of a potent atmosphere accumulating inside the ward that can pervade and influence her state of being. In contrast, her appreciation of the fresh air outside when she is able to leave the ward for a short time, presents an image of restorative qualities being available within the outdoor environment (Hartig et al., 2014), that enable her to regain a sense of equilibrium before returning to the ward.

Wendy's perceptions of discomfort in the ward environment are further expressed by her description of challenges associated with working within the "goldfish bowl" setting created by the glazed nursing office. In this account, the glazing is positioned as being problematic since it visually connects staff and patients, yet simultaneously separates them by presenting a communicative barrier that offers up a misleading picture regarding staff availability. Wendy's account suggests that through its capacity to engender miscommunication and emphasise a staff-patient divide, the glazing can induce frustration and distress on all sides by exposing misalignments in the priorities and expectations of both groups. The glass is therefore positioned as being unhelpful to staff-patient relations and Wendy's comment that patients are, "not interested" in staff administrative duties inside the office further stresses a distinction

between the different 'worlds' of patients and staff. The sense of division created by the nursing station glazing is similarly emphasised through her use of the pronouns "they" and "them" in reference to patients.

Correspondingly, previous research has found that glazing around nursing stations can accentuate the differing status positions of patients and staff in psychiatric settings and convey messages to patients that staff are occupied and inaccessible (Reavey et al., 2019; Schweitzer et al., 2004). Power imbalances can thus be reinforced by the glazing, since patients must typically knock on the windows to attract attention, whilst staff are able to choose when to engage (Andes & Shattell, 2006). For some patients detained in secure psychiatric settings, the use of glass or acrylic screens around nursing stations may also be associated symbolically with prison environments or invoke perceptions of punishment (Shattell et al., 2015). Open nursing stations without glazing may therefore benefit patient-staff relations and help to improve communication within mental healthcare wards (Edwards & Hults, 1970; Shattell et al., 2015).

Patient participants in the present study and previous research (e.g. McGrath & Reavey, 2013; Riordan & Humphreys, 2007) have reported their perceptions of being under continual scrutiny from staff within psychiatric facilities. Wendy's account, however, suggests that she and staff colleagues may in turn feel self-conscious and closely watched by patients, particularly when working inside a glazed nursing station (Simonsen & Duff, 2021). Secure ward layouts are typically informed by 'panopticon' principles conceived by the philosopher Jeremy Bentham in the late 18th century, such that the central position of nursing stations on contemporary wards affords staff sightlines from a single point to observe patients in the common areas. Devised as a form of institutional surveillance, the panopticon design allowed a watcher in a central tower to observe multiple detainees at once, yet those being observed could not know if they were being watched or not. As highlighted by Wendy's experience, however, an

inversion of panopticon surveillance also occurs in modern psychiatric units, whereby staff within the nursing station are in turn exposed to observation by patients (Scanlon & Adlam, 2011). Wendy's repetition of the word "constantly" in relation to patients counting and "harassing" staff also contributes to an image of staff receiving unremitting and undesired attention from patients when at work in the nursing office.

Goffman's (1961/1991) examination of the social situation within psychiatric institutions can be drawn upon again here to provide theoretical elaboration on Wendy's perceptions of being uncomfortably visible within the nursing station. Following Goffman's thinking, Wendy can be understood to be subject to a form of "contaminative exposure" (p.31) that patients might typically experience under the constant gaze of others in psychiatric institutions, with limited opportunity for private activity. Whilst Wendy mentions elsewhere in the interview that nurses have access to an office space outside the ward, she suggests that the lack of private working space on the ward makes it difficult for staff to carry out administrative work without interruption. Similarly, previous research findings suggest that staff and patients may both benefit from the provision of 'offstage' areas on the ward to allow nurses to retreat or carry out administrative work (D. J. Brown, 2009). Improved positive nursepatient interaction has also been found on wards where increased private space was made available for both groups (Tyson et al., 2002).

In addition to perceptions of mutual visibility on the ward, Wendy's narrative implies that patients and staff also share the common experience of wishing to spend time outside the ward environment. Wendy presents leaving the ward as a necessary strategy to manage her affective state, however, the account is also imbued with irony, since her distress is associated with perceptions of patients harassing staff due to the patients' own desire to leave the ward. Wendy's story emphasises how patients in secure settings commonly require one or more escorts in order to leave the ward, such that opportunities to access spaces outside the immediate ward environment are

closely linked to staffing levels and staff availability. Correspondingly, existing research examining experiences of psychiatric hospital admission has found that a lack of staff availability to provide escorted leave from the ward can induce distress and contribute to patients' perceptions of a lack of freedom in hospital (Gilburt et al., 2008).

Moments of laughter and humour feature throughout the interview and are typically used by Wendy to underscore the strength of feeling associated with her experiences. At the end of this story, Wendy's laugh emphasises her concluding comment that she, "is always very grateful to leave at the end of the day". Throughout the interview, Wendy's accounts highlight how similarities exist in staff and patient experiences of the ward environment, including perceptions of spaces as being uncomfortable or exposing and lacking connection to nature and the outside world. In line with experiences expressed by several patient participants, Wendy also articulates the strength of her need to spend time each day in the open air outside the confines of the hospital building. Whilst Wendy identifies key similarities in staff and patient experiences of space within the everyday environment, her concluding remark in this extract also emphasises the significant disparity in the relative liberties of staff and patients within secure settings and the positional contrast between both groups.

Alongside her accounts of how the material and spatial qualities of the ward can influence her own affective experiences, Wendy identifies further ways in which the environment might mediate everyday relationships between staff and patients. Within the following extract she elaborates on storylines of negotiating staff-patient interactions and explores how spatial experiences on and off the ward might be linked to her perceptions of creating therapeutic relationships with patients.

Wendy: Extract 2

1.	W :	Um, yeah. I mean, it, down here, yeah, it can be quite, um, difficult I think
2.		sometimes to kind of engage in a meaningful conversation, particularly with a
3.		patient, however, if you take them out for a walk and get them talking, within

4.		the grounds, I think you get a whole different experience with the patient.
5.	R:	(Coughs). Mm.
6.	W:	Um, here, I think, you know obviously I think they feel very confined, um,
7.		but when we're out in the grounds, I think you can get a lot more from the
8.		patient.
9.	R:	Mm.
10.	W:	Um. So yeah, I think it, it does have an impact, yeah.
11.	R:	Yeah. Are there any reasons why you think that might be that you get that
12.		slightly different experience?
13.	W:	Freedom I suppose. Bit of freedom. You know, you're not constantly, you
14.		know, locking doors, um, opening doors, um, they're not, you know, I suppose,
15.		as well when they're out as well, they feel like they're a normal member of
16.		society, people wouldn't look at them and think they're a patient at [name of
17.		<i>hospital</i>]. They perhaps feel a little bit normal, more normal, so they're able to
18.		be a bit more themselves
19.	R:	Mm.

20. **W:** when they're out.

Wendy's description of being "down here" in her account of the ward reiterates her earlier positioning of the environment as a lower ground floor space with limited natural light and she contrasts perceptions of staff-patient relations inside the building with open air experiences in the hospital grounds above. Wendy's observation that being outside with a patient can provide "a whole different experience" suggests that she perceives a significant distinction in the kind of interaction possible in a natural outdoor setting, when compared with dynamics inside the enclosed ward environment. Whilst suggesting that patients may typically feel physically "confined" on the ward and perceive a sense of "freedom" when outside, Wendy's account also conveys a sense of emotional constraint and limitation indoors. Accordingly, alongside other participants across the data set, she perceives that patients may be more inclined to open up and confide in staff within more open and fluid outdoor spaces.

Wendy's perception of the value in spending time outdoors with patients also complements a body of existing research that highlights how people with mental health problems may experience multiple health benefits from walking outdoors (Doughty,

2013; Richardson et al., 2005) and engaging in physical exercise and social interaction within natural settings (Barton et al., 2012; Barton & Pretty, 2010; Mind, 2013). For people with lived experience of distress, the psychological benefits associated with being active in nature and participating in what is also termed, 'green exercise', may include improvements in mood and levels of self-esteem (Barton et al., 2012; Barton & Pretty, 2010). Recent decades have also seen increasing interest in the emerging area of therapeutic 'walk and talk' practice, involving counselling and psychotherapeutic activity undertaken in outdoor spaces (Doucette, 2004; Revell & McLeod, 2017).

Revell and McLeod (2017) explored therapists' perceptions of outdoor therapy and found that participants perceived a positive adjustment in traditional therapeutic relationship dynamics when walking side by side with clients in an outdoor setting. In contrast to sedentary office-based interaction, the participants' accounts suggest that the altered physicality of working outdoors can enable a range of beneficial processes, including a greater sense of equality between client and therapist and increased possibilities for non-verbal communication. The combination of physical movement and therapeutic work outdoors is therefore understood to be helpful in enabling clients to integrate mind-body activity and physical movement may also help clients to shift perceptions of feeling 'stuck' and generate alternative psychological perspectives. Research findings also indicate that therapeutic walk and talk practices can assist in the development of connections to nature that can support mental health (Jordan, 2014) and may contribute to improved individual perceptions of overall well-being and self-efficacy (Doucette, 2004).

Whilst Wendy refers to informal outdoor interactions with patients, her account implies that moving and talking together in an outdoor setting can be potentially therapeutic and associated with a shift in the relational dynamics between staff and patients. Through this extract and the earlier account of her own distress associated with spending prolonged periods of time on the ward, Wendy presents the

process of moving between spaces as a transformative experience that can facilitate a turning point between affective states. Wendy's observations can hence be understood to support previous research which suggests that movement through space, including the particular drive to move from enclosed internal spaces into the open air, can be a significant mediator of experiences of distress (McGrath & Reavey, 2015).

The narrative also indicates that staff-patient status positions are reinforced by power differences expressed through physical interactions with aspects of the built environment, such as doors and locks. Wendy's account thus emphasises how staff control over doors on the unit presents a hierarchy of access to space and highlights how perceptions of doors as symbolic and material barriers can serve as a continual reminder to patients of their status and lack of liberty (Muir-Cochrane et al., 2012; Reavey et al., 2021). Earlier in the interview, Wendy highlights how disparities in staff and patients' access to space are similarly reinforced through other material aspects the ward, including the 'staff only' notice on a door to the room where patients' cigarettes and vaping equipment are stored. When discussing a photograph of this room, that staff members typically go in and out of frequently during the course of the day, Wendy observes that,

"Um, and it's just again, kind of, er, only staff are allowed, so it's kind of, the nurse patient divide, again, um, wh-, wh-, which can be quite, quite difficult on here. [**R**: Mm.] Um, kind of having to show your authority to patients, but then also we want them to tell us our deepest darkest secrets about thoughts and things and you, you try and build a relationship with somebody, um, but then you're kind of saying, oh actually no you can't come in here, you know, you're not allowed in here."

Wendy articulates a sense of discomfort experienced in her professional role, whereby she must navigate the paradox of attempting to establish both authoritarian and confidante relationships with patients. Wendy alludes to how perceptions of trust are essential for building therapeutic relationships, yet she is mindful that the sign on the door indicates a lack of trust. The door also marks a physical and symbolic boundary that underscores the differing status positions of staff and patients and divides their circles of activity. Wendy's account thus highlights how the built environment conveys messages that can affect relational dynamics. By emphasising a sense of division, disparity and mistrust between staff and patients, the door and signage are therefore seen to be contributing to experiences of discomfort that are shared by both groups.

Figure 6

Wendy's photograph of door signage on the ward



In contrast to an image of power relations being embedded within the ward environment, Wendy presents the natural space outside the building as a more neutral territory that can mediate relational differences and create a sense of parity between patients, staff and the wider community. The account therefore implies that an absence of physical reminders of detention and control within the natural outdoor environment is significant to the change in staff-patient dynamics that Wendy perceives outside the ward. Her observations also reflect findings of previous research in secure psychiatric settings where support staff perceived a positive shift in relationships with patients in off-ward environments where staff were not continually locking and unlocking doors (K. Evans et al., 2012).

Goffman's (1961/1991) exploration of patient experiences within the asylum highlights how boundaries between typical spheres of life, such as sleep, work and leisure are removed when patients enter a psychiatric institution, meaning that everyday activity is commonly compressed into a single environment. Wendy's account, however, suggests that being active within the natural environment outside the ward might help patients to recreate a sense of spatial distinction between spheres of activity that enables a more typical representation of everyday life. This is underscored by her repetition of "normal" to describe how patients might feel when spending time outside the building when compared to their experiences on the inside. By implication, Wendy's narrative consequently conveys a sense of the ward as an unnatural environment, such that patients' experiences inside the building are nonrepresentative of typical everyday life. The narrative thus presents a contrast between the external grounds, where perceptions of life outside the hospital are potentially more available to patients and the ward, which is positioned as a space where patients find it difficult to perceive a sense of ordinary life. Wendy's account implies that the secure ward environment is a contributing factor to detained psychiatric patients' perceptions of 'deviant' labelling (Coffey, 2012b), which in turn can be associated with experiences of being 'othered' by wider society (Tomlin et al., 2020).

Accordingly, movement from indoor to outdoor space is presented as a transitional process that can enable the possibility of an identity shift for patients – from the position of a detained psychiatric patient to what Wendy terms, "a normal member of society". The transformations associated with movement through space can thus be seen to be affording a shift away from the enduring conditions of a 'liminal hotspot' (Stenner et al., 2017) that may typically be experienced by detained patients in secure institutions. Wendy therefore suggests that patients may feel more at ease as individuals within normative off-ward spaces, where their status position is less reinforced by cues within the environment. Elsewhere in the interview Wendy also

notes that during her career she has at times felt, "a bit like a prison guard, not a mental health nurse". Her account here, however, implies here that being outside, where she is not "locking doors", can affect perceptions of her professional identity by reducing carceral associations to reinforce a role of care and underpin a more human-to-human relationship with patients that is more reciprocal and equitable.

4.4 Concluding comments

The narratives explored in this chapter have exemplified participants' storylines about physical and psychological discomfort associated with experiences of spaces or objects within the hospital environment. Corresponding stories that conveyed patient and staff accounts of disquieting spatial experiences in these settings were also expressed more widely by both groups across the overall data set. Using liminality as a theoretical perspective through which to consider these experiences in process terms, the discomfort and unease expressed through the narratives has been conceptualised in the context of participants' experiences of enduring liminality, or as a sense of being 'stuck' in a 'liminal hotspot' (Greco & Stenner, 2017; Stenner, 2013). The narratives thus highlight varied ways in which the occupants of secure settings can be seen to be uncomfortably suspended between different and typically contrasting 'worlds' or 'forms-of-process' when living or working in these environments (Stenner, 2017) and how spatial experiences might mediate transitions between worlds.

Participants expressed perceptions of their self-identity through the narratives (e.g. as detained patients or members of the wider society), or their differing positional status (e.g. as patients or members of staff). The stories explored here highlight how these self-positions can be mediated or sustained by the everyday spaces and objects encountered in secure settings. The barriers presented by physical features of the environment, including robust doors, locks and non-openable windows form symbolic reminders to patients about their lack of liberty, in addition to highlighting how staff are required to negotiate custodial responsibilities alongside their duties of care. For

some patients, the material and aesthetic attributes of architectural components and the utilitarian detailing of furniture items within the ward environment also present an unhelpful reminder of time spent in prison settings. Qualities of the built environment, including the material, aesthetic, atmospheric and acoustic attributes of spaces are thus argued to be reinforcing participants' perceptions of status or identity and defining the limits of their ability to transition between different and commonly conflicting selfpositions. Correspondingly, symbolic cues perceived in the everyday environment, including the materiality, scale and proportion of spaces are seen to be conveying messages about power or control and the contrasting status and liberty of patients and staff. Such distinctions between staff and patients are thereby understood to be being continually expressed through their daily interactions with architectural components and the built environment.

The narratives examined in this chapter and across the data set frequently describe the hospital spaces as 'clinical' or lacking in domestic ambience and reflect how institutional concerns about safety and risk management are visibly expressed within the material environment. In the interest of reducing risk, the spaces and surfaces within secure settings are seen to have been pared back to a functional minimum and are hence portrayed through the narratives as lacking the layers of textural detail and 'warmth' typically associated with domestic interiors. In addition to spatial sparseness, however, the narratives convey how the visible and explicitly safety-derived detailing of architectural touchpoints, such as doors and joinery elements, expresses the prioritisation of physical safety over psychological comfort or aesthetics. Accordingly, narratives constructed around wider societal understandings of domesticity and expectations of comfort within residential spaces are used to highlight intractable tensions in the environment. Participants' stories draw particular attention to how the contrasting aesthetic, atmospheric, social and functional

characteristics of both 'home' and 'hospital' spaces are paradoxically combined within the same overall environment.

Participants' awareness of both carceral and clinical symbolism being built into the material fabric of the environment in this way emphasises how secure wards are positioned between the criminal justice and healthcare systems, whilst not being fully situated in either. Furthermore, the narratives illustrate how staff members can be seen to be negotiating a position of enduring liminality when navigating the conflicting professional identities associated with their contrasting custodial and caregiving responsibilities. Storylines concerned with the aesthetic or material attributes and upkeep of interior spaces, including the quality of architectural details, are also associated with patient participants' wider perceptions of value and care, or appraisals of how their own position might be regarded within the overall system.

Accounts of participants' affective experiences in over-scaled spaces illustrate how overbearing architectural features or volumes contribute to spatial expressions of institutional power. By reinforcing a sense of power inequality that can contribute to detained patients' perceptions of having limited agency or influence, these spatial experiences can be seen to be compounding a sense of patients being 'stuck' in their present circumstances. Such narratives also highlight how congruence between the scale or atmospheric tone and the function of spaces is significant to the levels or limits of physical and psychological comfort afforded by those environments. Spaces are thus perceived to be disconcerting or uncomfortable in instances when an environment's scale, proportions or ambient qualities are not well matched to its ostensible function. Storylines also convey how loud ambient noise in the environment, including slamming doors and shouting constitutes a form of psychosocial intrusion that is typically uncomfortable and unsettling. As a form of the 'contaminative exposure' observed in psychiatric institutions by Goffman (1961/1991), unwanted sound in secure settings is

also seen to be compounding patients' perceptions of lacking control and privacy in an environment where occupants have little or no private space.

Throughout the narratives, the physical settings and material elements that constitute the hospital spaces, including doors, walls and windows, are understood to be defining the limits of different spaces. This is unsurprising as these are boundarymaking materials, however, these architectural features become significant within participants' stories when they are seen to have a role in shaping the boundaries between typically contrasting 'worlds' of activity. Participants' experiences of such boundaries are portrayed as being troublesome and discomforting when they present physical or symbolic barriers to communication between patients and staff or disconnect ward occupants from the valued and vitalising qualities of the natural world outside the hospital environment. The physical and symbolic attributes of restrictive internal ward environments are typically described by both patient and staff occupants as being increasingly constraining or distressing over time. Consequently, in addition to creating a valued physical connection to nature, the relative liberation of spending time in open-air spaces is understood to provide a welcome release from an uncomfortable build-up of affective 'pressure' contained within the secure boundaries of the built environment. Doors are presented as dynamic devices that can connect or separate contrasting worlds and commonly feature in storylines expressing participants' experiences of unease at the thresholds between 'public' and 'private' circles of activity, or between the differing status positions and activities of patients and staff.

In storylines explored here and across the data set, glass is specifically highlighted as a paradoxical material that can both connect and separate spaces simultaneously. Internal glazing is hence described as being able to generate discomforting ambiguity and influence communication by presenting misleading pictures to patients about staff intentions and availability. Furthermore, participants'

narratives highlight how the experience of being viewed through glass is typically unpleasant and exposing for patients and staff and contributes to the experience of psychological discomfort amongst both groups. By separating occupants from the multi-sensorial qualities of the natural world outside the hospital, or by 'offering up' inaccessible spaces, glazing may typically prompt frustration or remind patients about their status and lack of liberty.

Storylines conveying ambivalent attitudes towards the environment are seen to be helping participants make sense of their conflicting emotions when contrasting the positively perceived affordances of spaces with more challenging environmental attributes. Therefore, whilst particular spatial affordances (including features that facilitate social interaction or connections to people outside the hospital) are appreciated for being comforting or supportive, spaces can be simultaneously perceived to be uncomfortable, unpleasant and actively unhelpful in respect of their ambient or material qualities. The narratives convey how areas such as the communal lounge, which is portrayed as a thoroughfare that is insufficiently shaped to resemble a complete room, have a lack of spatial distinction that can result in the experience of troubling and ambiguous environments. By being open to other spaces all around, the day area thus has unbounded and indeterminate qualities which are perceived by occupants as exposing and uncomfortable. These 'both/and' and 'neither/nor' characteristics of the hospital environment can therefore be understood to be contributing to participants' perceptions of being suspended 'betwixt and between' in a liminal position between different worlds.

Chapter 5: Managing difficult experiences using objects and space

The first empirical analysis chapter explored patient and staff participants' stories that exemplified an overarching narrative concerning experiences of discomfort associated with spaces and objects in secure hospital environments. The analysis presented in this second empirical chapter explores a further set of stories that focus on participants' physical and psychosocial engagement with spaces and objects to negotiate difficult affective experiences and mediate discomfort experienced in the hospital environment. Tom and Patricia's accounts of their experiences as patients in medium secure settings will be explored in this chapter, alongside stories told by Ingrid, who is a staff member based on a medium secure ward.

The narratives contain participants' accounts of experiencing distress or troubling events in these settings and the ways in which patients and staff negotiate and manage difficult experiences day to day. The interpretation of the narratives focuses particularly on participants' accounts of how spaces and objects can mediate discomfort associated with spaces or objects encountered in the everyday environment and facilitate transition between affective states.

5.1 Tom

Tom is a young man and a patient on a medium secure ward. He has previously spent time in prison and describes leaving home at a young age before moving between different care home and hostel settings. Tom has not been permitted ground leave since his admission to the ward a few months before meeting, however, he talks about the enjoyment he experiences from spending time with other patients in the communal areas of the hospital building outside his own ward environment. Whilst Tom describes enjoying his own company, he also presents himself throughout the narrative as a sociable person with a cheerful sense of humour, who enjoys laughing with and being in the company of other people. Tom values the relationships he has formed in hospital

and his appreciation of perceiving a sense of community and mutual understanding between the patients on his ward repeats throughout the interview. Tom also keeps in touch with family and friends each day via the ward telephone booth and he expresses the importance to him of maintaining this regular contact alongside his appreciation of receiving monthly visits.

Throughout the narrative Tom positions himself as a person who enjoys having opportunities to contribute to the ward community and engage in meaningful activities. He also talks about experiencing a sense of pleasure and satisfaction through socialising and working alongside other patients, including time spent collaborating on creative initiatives designed to enhance and decorate the ward environment. Music has been a significant interest for Tom throughout his life and he describes how listening to music through headphones in his bedroom affords a valuable sense of retreat from the ward when needed and enables him to relax.

Although he refers to many challenges associated with being detained in a secure psychiatric unit, including his current lack of access to space outside the hospital building, Tom's overarching tone is determined and forward focused. Throughout the narrative he provides examples of using the less positive aspects of his experiences as incentives to enable him to remain focused on his goals and life beyond hospital. Tom is reflective about his past and present circumstances and he expresses a sense of pride in his perceptions of gaining new skills, including development in areas of his life that have previously been challenging, such as maintaining a tidy bedroom.

Tom's primary storylines focus around: (1) the significance of social connections and relationships and the spaces that facilitate these, (2) looking forward and using objects and space to motivate towards goals, (3) using spaces and objects as a form of distraction from difficult experiences, (4) valuing taking on responsibility and having a sense of contributing to the wider community, (5) negotiating relationships with staff

members and (6) perceiving connections between well-being and having a sense of control over the environment.

In the interview Tom discussed photographs of¹⁰: (a) a mural created by patients in the ward entrance corridor, (b) outline of his hand forming part of another collective patient mural installation, (c) a window on the ward containing a view of the grounds outside, (d) the on-site swimming pool, (e) belt worn by staff with radio and keys attached, (f) communal café courtyard located within the hospital building, (g) traffic light system associated with the airlock at the hospital's main entrance door, (h) dining room, (i) therapy kitchen, (j) telephone booth and (k) his bedroom.

The interpretation of Tom's narrative will focus on the first, second and third storylines, which intertwine throughout the interview and illustrate how psychosocial engagement with the environment is used to assist his transition between affective states. In the first extract, Tom's narrative is focused on his emotional responses to the windows on the ward and the views of the hospital grounds presented through them.

Tom: Extract 1

1.	T:	OK. So, the next photo is, um, of one of the windows on the ward.
2.	R:	Yeah.
3.	T:	Now the windows, um, just to give a brief description, they're probably like a
4.		foot by foot, panes of glass, with, um, you know sort of solid metal bars in
5.		between.
6.	R:	Mm.
7.	T:	Um, you know, they don't look prison-like, but, you know, I guess they
8.		illustrate that it is a medium secure and beyond the glass, is, is, I guess it
9.		shows what you could have had, sort of thing, you know, what I'm missing out
10.		on because, you know, due to the section I'm on I, I haven't got grounds leave
11.		yet so.
12.	R:	Mm.
13.	T:	I guess every time I see outside, it reminds me of the, you know, the life that I

¹⁰ As Tom's images could not be printed beforehand, his photographs were reviewed on the camera screen during the interview. The digital files could not be transferred, however, prints of some of his photographs were later available.

- 14. had before I came into hospital and the, the freedom.
- 15. **R:** Mm.
- 16. **T:** And I guess you know, it's, it's, it reminds me that, you know, I don't have
- 17. that, that choice to be able to just walk along that path that's in the photo
- 18. and enjoy the, you know, the scenery and trees and the and things like that.
- 19. **R:** Mm, mm.
- 20. **T:** So, it's, it's quite, you know it's quite negative photo, but I guess, in certain
- 21. senses it's what I want to work towards, it's an incentive.
- 22. **R:** Yeah, yeah. So you, you, when you're looking out, you're making a positive, um
- 23. **T:** From a negative.
- 24. **R:** Yeah.
- 25. **T:** Yeah, that's it.
- 26. **R**: Yeah, yeah.
- 27. **T:** You know while it doesn't feel great right now, you know, it's my, it's my,
- 28. um, my drive, my incentive to work hard towards my goals and I guess, you
- 29. know, stay on the right track.
- 30. **R**: Mm.
- 31. **T:** Because it's quite easy to sort of fall off sometimes and, you know, get
- 32. dragged down by the negatives, but, you know, that's, it's quite important for
- 33. me to stay on track and I think things like that help.
- 34. **R**: Yes, yeah, absolutely. And is there something about being outdoors as well that,
- 35. um, is a, has, um, a positive, um, feeling
- 36. **T:** Yeah.
- 37. **R:** for you?
- 38. T: That's it. Yeah, definitely. I, I, before I came to, to hospital I used to be very
- 39. much into my walking. Um, you know I got a lot fitter and I lost a lot of weight40. due to walking.
- 41. **R**: Mm.
- 42. **T:** I've just lost the camera again.
- 43. **R**: Don't worry, don't worry.
- 44. **T:** And um, you know, it, it's, it's quite a, it's been quite a big influence to me, so
- 45. I guess, coming here and not being able to just freely go for a walk and clear my
- 46. mind, is, is quite, quite a big change for me, but.
- 47. R: Mm. And how do you feel at times when you might want to go for a walk and48. clear your mind and you're not able?
- 49. **T:** It's, it's quite disheartening
- 50. **R**: Mm.
- 51. **T:** it is, you know, understandably, but, you know, I, I, understand why I'm here,
- 52. for the reasons I'm here for and, you know, it's for me, you know, it's not for
- 53. anybody else, it's to get me to where I want to be, to my, to my goal.

- 54. **R:** Mm, mm.
- 55. **T:** In my future. So you know it's just, ha-, being able to turn that round into
- 56. something positive is quite, quite important for me.
- 57. **R:** Yes, yeah.
- 58. **T:** It's quite hard to feel sometimes, but yeah.
- 59. **R:** Yeah, yeah.

Although Tom does not perceive the window design to be "prison-like", his description of "solid metal bars" located between small panes of glazing emphasises an image of robust window construction, which he perceives as a symbolic reminder of the medium secure context within which he is detained. In turn, the inaccessibility of the natural scenes and greenery framed by the windows presents Tom with a further reminder of his lack of liberty and his inability to leave the building due to the restrictions of his current section. Whilst everyday opportunities for walking outside might not be significant in other circumstances, Tom's characterisation of the hospital grounds as something that, "I'm missing out on" presents spending time within a natural outdoor setting as a longed-for activity and positions his lack of access to this external space as a poignant loss.

The view of the grounds therefore presents a continual reminder to Tom about his lack of "freedom" and his loss of access, not only to this external space, but also to the life he had before hospital and to the life that he hopes for beyond. The window represents a physical and symbolic barrier between confinement in hospital and relative liberation in life outside and the glazing is thus presented through the narrative as a material threshold between Tom's present circumstances and his past and future lives. Tom's wistful imagery of the window view representing what he "could have had" therefore not only expresses feelings of regret about his current circumstances, but also evokes a sense of grief associated with the loss of an alternative envisioned version of his life story.
Figure 7 *Tom's photograph from a window on the ward towards the grounds*



Tom positions the image of the window view, containing a path leading through the trees in the hospital grounds, as a "negative photo" that exemplifies what he has lost by being detained in hospital. Yet he also seeks to perceive the image positively by transforming it into an "incentive" that represents his ambitions and the future life he wishes to have. By visualising walking along the path through the grounds as a motivational image to aspire to and move forward towards, Tom is attempting to transform his affective experience from a negative state of loss to a position of optimism where he is focused instead on what there is to be gained. Interpreting these experiences as a liminal process (Stenner, 2017), Tom can be understood to be creating this imaginative imagery as a form of 'liminal affective technology' or 'symbolic resource' to self-generate liminal affective experience that enables him to move between affective states (Stenner, 2021; Stenner & Zittoun, 2020). That is to say that the aspirational images generated through Tom's imaginative processes are used to

assist with the production of emotional effects that can bring about psychological transformation.

Using imagination and visualisation to prompt such affective transformation can therefore also be interpreted as a process that is assisting Tom to alleviate the disconcerting experience of 'spontaneous' liminality (Stenner, 2017). In other words, this process of self-generating liminal affectivity can be understood to be helping Tom to mitigate the significant rupture in his everyday psychosocial experience brought about by hospital admission and experiences of psychological distress. Tom's depiction of walking outside to "clear my mind" implies that walking through open-air space is another previously used strategy to facilitate transition between affective states. Accordingly, Tom's use of this physical movement outdoors and integration of mindbody activity when walking can be interpreted as a further form of liminal affective technology that can facilitate psychosocial transformation. Tom hence feels dispirited by not being able to walk "freely" outside. His story suggests that, in addition to his loss of liberty, the loss of access to walking as a process for maintaining both his mental and physical health has been a significant adjustment following his admission to hospital.

Tom's storyline of staying "on the right track" extends the imagery of walking to express a sense of focus and purposeful movement forward in his life, however, it also conveys a moral tone, implying that other paths he might choose may not necessarily lead in the "right" direction. Tom's narrative imagery builds on wider cultural understandings of success and societal beliefs that dedication, focus and "hard work" are positive and respected qualities required to achieve a desired ambition. Through his account of having the potential to easily "fall off" the track and be "dragged down by the negatives", Tom presents an image of sinking into a low position, that is juxtaposed with the alternative uplifting imagery of him walking forwards towards his "goals" and the anticipated place he refers to as "my future". Tom's psychological processes are

presented as forces that have the potential to lower his mood and hinder progression, but which conversely, can also lift his spirits to support his onward "drive" forwards.

Tom's experiences can also be elaborated further in relation to Arthur Frank's (1995) book *The Wounded Storyteller*, in which Frank identifies 'restitution', 'chaos' and 'quest' narratives as three different narrative forms that individuals may typically create in response to experiencing challenges to health. Drawing on this thinking, Tom's overarching narrative can thus be interpreted as a form of 'quest' narrative, through which he is understood to be navigating his 'recovery' journey whilst focusing on the hopeful quest of pursuing his goals and imagined representations of his future self. Frank (1995) also describes a significant loss of health as being a loss of the, "destination and map" (p.1) that were used previously as a guide in that person's life. Correspondingly, it is suggested that producing narratives can have a therapeutic quality, such that telling stories might enable people experiencing a loss of health to recreate a 'map' to navigate their experiences and regain a sense of 'destination' in their lives.

Although Tom describes his journey as a difficult process that, "doesn't feel great right now", his telling of this story out loud can be understood as a self-motivating process that reinforces his intentions and underscores his commitment to the quest. Tom acknowledges that adjusting his affective state by attempts to transform difficult aspects of his experience into incentivising imagery is challenging and "quite hard to feel sometimes". However, his comment that, "it's not for anybody else, it's to get me to where I want to be, to my, to my goal", further reinforces his sense of destination and self-motivational discourse. Tom elaborates on this principal 'quest' narrative later in the interview when considering his thoughts about moving on from hospital. In this short reflection he is more explicit about the focus of his "goals" and the process of moving towards them when he notes that,

"I think, one the, one of the most important things for me to, to maintain my sight on, on

my goals and, and where I, you know, the direction I wanna go is, painting quite a detailed picture of what I, I want my life to look like in the future, you know, from things to like how much sp-, time spent with friends and family, career, you know, my house and my, where I live [**R**: Mm.] you know, it's, it's moving on from here is quite an important thing for me and, and planning quite thoroughly is, is really helpful for me to maintain that, that goal."

In this account of "painting quite a detailed picture" of his future life, Tom highlights how imaginative processes are enabling him to envision and inhabit a comprehensive future space that represents his "goal" of an idealised life beyond hospital. Tom thus emphasises how using his imagination to produce motivational imagery constitutes a form of transformative technology that is helping him move away from difficult affective states and in turn, assist in managing the overarching experience of being detained in hospital. Through focusing on the specific details of his anticipated future life, Tom can accordingly be seen to be attempting to alleviate perceptions of being indefinitely suspended in an indeterminate zone situated between his life before and after hospital. Furthermore, by describing how detailed planning and visualisation of the future is helpful in enabling him to "maintain" his goals, Tom also highlights how these are ongoing and iterative process that he continually returns to for support and reassurance.

Elsewhere in the interview, Tom tells similar stories of everyday interactions with other poignant objects and spaces on the ward, including the radios and keys that are attached to staff belts and the 'airlock' double door security system at the entrance to the building. Keys and radios are used to control the hospital environment and Tom highlights how the presence of these objects strengthens his awareness of a lack of freedom within a medium secure setting, alongside providing an unhelpful reminder of his time spent previously in prison. However, as with the window view, he also describes his attempts to use imagination as a transformative device to generate alternative affective responses to the troubling objects or spaces he encounters within the everyday environment.

Figure 8 *Tom's photograph of keys and radio on staff member's belt*



By transforming his perceptions of such spaces and objects into positive imagery that represents the envisioned future he is striving towards, Tom is seen to be transitioning from an anguished affective state to one that is more hopeful and supportive. Tom thus captures his attempts to generate alternative images when confronting these objects through his assertion that, "[...] one day we won't have to hear and look at the keys and radio and you know, it won't, it won't be a part of my life forever". Through the process of articulating this statement definitively to the researcher, Tom can also be understood to be reinforcing his own belief in its validity.

Tom further expands on the storyline of transforming troubling aspects of the environment into motivational imagery, when discussing his emotional responses to a photograph of the traffic light system associated with the 'airlock' doors at the main entrance to the building. The airlock forms a transitional zone between the internal and external environment and is a reminder to Tom of his detention in a secure institution and his inability to leave hospital or leave the building itself. Tom thus articulates ambivalent feelings towards the airlock when describing it as a place where he is both happy to greet visitors on arrival and also sad to say goodbye to family and friends at the end of a visit.

Figure 9

Tom's photograph of the airlock traffic light system



Through this story, Tom describes the significance of the airlock as a threshold between his present life as a detained patient and his life both before and beyond hospital. In the following quote, Tom articulates how his imaginative processes are again employed to help transform difficult and distressing encounters with this part of the everyday environment into self-motivational imagery.

"It, it's, you know, again, I guess it could be turned around in a positive way, one day it will be me walking out, you know, past the green light and, you know, whether it's just on grounds leave which is a more, you know, a closer goal, [**R**: Mm.] or leaving for good, you know, it's going, it's going to come around one day, so that's what I like to see it as. [**R**: Yeah.] You know, it will be me one day walking through there, you know." When he visualises his future passage through the airlock to leave the building, Tom draws again on imagery of walking and forward movement to represent progression and his personal journey through life. By transforming the implied imagery of the airlock as a barrier into uplifting images of movement that represent his short-term and long-term goals, Tom can once more be seen to be using imagination as a process to stimulate affective transformation as a way of coping with his circumstances. Through the process of narrating this story to the researcher, Tom is again understood to be reinforcing his own belief in the realisation of his goals as a form of selfreassurance.

Throughout the interview Tom also recounts stories about how times spent socially with other patients, particularly in areas off the ward, can help mitigate difficult experiences associated with being detained in hospital. In the following extract Tom reflects on his embodied experiences in the swimming pool and his affective responses to this environment.

Tom: Extract 2

1.	T:	The next one is the swimming pool.
2.	R:	Mm.
3.	T:	This one's a positive one for me because I've always liked swimming. I
4.		I really enjoyed it and we have two weekly sessions. Um, so we go down there
5.		for an hour and it, it, it's, we often go down with 6 or 8 different patients and
6.		just have fun. It's our time to, you know we nearly, we nearly forget where we
7.		are, we nearly forget the situation that we're in. You know, it's an escape. If the
8.		ward's unsettled, we can just sort of have fun.
9.	R:	Mm.
9. 10.		Mm. Be ourselves. Be, I guess, be silly, you know we don't really get much
10.	T:	Be ourselves. Be, I guess, be silly, you know we don't really get much
10. 11.	T: R:	Be ourselves. Be, I guess, be silly, you know we don't really get much opportunity to just have raw fun like that. You know, it, it is great.
10. 11. 12.	T: R:	Be ourselves. Be, I guess, be silly, you know we don't really get much opportunity to just have raw fun like that. You know, it, it is great. Yeah, yeah.
10. 11. 12. 13.	T: R: T:	Be ourselves. Be, I guess, be silly, you know we don't really get much opportunity to just have raw fun like that. You know, it, it is great. Yeah, yeah. It's good to go down and have like that social sort of setting where we all

- 17. **R**: Yes, yeah. And is there something, um, about being there with the, with other
- 18. people that, um, contributes to that, that sense of en-, enjoyment
- 19. **T:** Yeah.
- 20. **R**: that you have in that space?
- 21. T: I mean you know, we're all in the same situation on this ward, so there's like,
 22. you know, quite a strong sense of you know, community and sense of
 23. belonging between us, um, you know, we all, we all understand each other's
- 24. positions and we're all quite, you know, I guess, yeah, understanding towards
- 25. each other, so you know being, having that, you know, mutual interest between
- 26. everybody is is really nice, you know, in a different zone. Just, just, like I say,
- 27. just coming away from the ward, you know, having that escape, that, we all
- 28. enjoy it.
- 29. **R:** Mm, mm.
- 30. T: We do, so it's really nice having just, just to be with each other and be friends.31. It's lovely.
- 32. R: Mm. Yeah. And you mentioned being away from the ward, is that something
 33. that, um, you get in other, other spaces as well, that feeling of, um, almost
 34. escape from?
- 35. **T:** Yeah, that's it. You know, although the ward is generally quite a nice place to
- 36. be, it's the same, you know, everybody has to have a like, you know, variation
- 37. in their life and where they live and their environment and we don't get much
- 38. of it, so opportunities like this where we have a solid hour where we can
- 39. escape and you know, do something different is, is really nice and cherished.

In this story Tom positions the pool environment as a relaxing and enjoyable space that facilitates lively social interaction with other patients. By highlighting his long-standing enjoyment of swimming, Tom's narrative implies that spending time in the pool area also provides a sense of connection to his life before hospital and suggests that similar settings previously have typically been "happy" and enjoyable places. Tom's appreciation of the swimming activity as "our time" emphasises his perception of the regular pool sessions as a valued space and time for bonding socially as a group, where patients are able to enjoy each other's company, not only as peers, but also as "friends". His description of the environment as a "different zone" positions the physical location of the pool as a contrast to the ward context, but also evokes imagery of leisure activity

in the pool enabling an alternative social experience and the possibility of "escape" to another psychological space.

Although the swimming pool is situated inside the boundaries of the secure hospital building, Tom's narrative suggests that the pool is especially appreciated for being, "away from the ward" and as such, it affords perceptions of liberation, alongside a sense of retreat from the stresses of the ward. Through his account of valuing opportunities to experience "raw fun" and "be silly", Tom presents an image of untempered pleasure being afforded by spending time with others in the pool area, that he juxtaposes with a sense of the ward being at times a challenging and "unsettled" environment. Through describing the fun enjoyed by patients in this setting as "raw", Tom implies that this emotion is experienced in a natural and authentic state that contributes to its power. This image of authentic experience in turn suggests that patients may feel more natural and at ease together within this 'off-ward' space and thus more able to be light-hearted or experience pleasure there. The description of being silly similarly presents an image of how carefree play or childlike innocence might assist in patients' ability to gain respite from their everyday worries and concerns when socialising together in this space.

Accordingly, whilst the narrative implies that Tom associates being on the ward with a sense of restriction in an often volatile and stressful environment, he contrasts this with spending time in the swimming pool, which is positioned as a liberating, cheerful and almost unlimited space. In contrast to the confines of the ward, the absence of partitions in the pool area creates a large, open space. As a more generic and commonplace form of built environment, the pool setting also contains fewer physical cues, such as robust doors or locks and keys, to suggest that it is located within a secure unit. Correspondingly, Tom's repetition of "nearly" forgetting the realities of his personal circumstances and detention in a secure institution implies that spending time with other patients in the liberating pool environment can afford a closer

connection to typical life outside hospital than might generally be available on the ward. Tom's emphasis of "nearly", however, also highlights, that even when enjoying social activity in this more normative and less restricted space, he is not able to be fully transported away from his current situation. As a further illustration of these experiences, Tom talks later in the interview about how spending time and collaborating with patients in other 'off-ward' spaces, including the building's communal garden, can afford similar perceptions of escape and a sense of togetherness. He describes the shared garden accordingly as a "haven" that facilitates enjoyable social interaction and like the pool, the garden is expressed as a place where he can, "almost forget" that he is detained within a secure environment.

Tom's descriptions of appreciating his perceptions of a sense of "community", "belonging" and togetherness with other patients recur throughout the interview. His comments that as patients, "we all understand each other's positions" and "we're all in the same situation" draw on wider cultural understandings that shared experiences can create supportive bonds to connect individuals as groups. Tom's remarks also convey his appreciation of perceiving a sense of parity and a mutual lack of judgement between patients that he perceives to be particularly tangible in this setting. Detention in a secure environment, where patients can be understood to be 'suspended' in an indeterminate position between life before and life after hospital, is conveyed through the narrative as a shared and thereby equalising experience.

Tom's portrayal of his experiences can thus be interpreted in reference to Victor Turner's (1969/1996) specific use of the Latin term *communitas*, to define an unstructured state in which it is argued that individuals become equal through shared participation in a situation of liminal transition. Tom implies that his perception of *communitas* is especially pronounced during activity in the pool area, which by contrast to the ward itself, is positioned as an unstructured and less restrictive space. In this fluid environment where the typical 'rules' or limits of the ward are seen to be

broken down and temporarily suspended, the activity taking place in the swimming pool can be interpreted as an 'interstructural' situation (Turner, 1969/1996). Accordingly, activity in the pool is presented as a world-within-a-world, located between the restricted forms-of-process that comprise the ward environment and the more liberated spheres of activity associated with leisure spaces in the everyday world outside hospital (Stenner, 2017). Tom's perceptions of a strong sense of community and a mutual bond between patients specifically in the pool area suggests that the liminal fluidity of this specific environment also enables patients to have a greater awareness and enjoyment of equality and perceive a sense of *communitas*.

In contrast to the ward, where patients repeatedly encounter staff radios and keys and in which their movement through space is highly controlled and restricted, Tom presents an image of patients experiencing a smoother 'flow' through space within the pool environment. This picture of fluidity and a lack of friction is accentuated further by Tom's implied imagery of swimming and free movement through water as typical activity in a pool setting. Tom therefore suggests that the fluidity of this environment, where a collective identity as 'detained patients' can also be understood to have become melted down, is a contributing factor in enabling him and other patients to simply "be ourselves". Accordingly, Tom and the others can be seen to be enjoying the fluidity of this liminal situation, in which the temporary deconstruction of their collective patient identity enables closer connection to their individual identities and enhances the enjoyment of pleasure as "friends". Tom's narrative thus suggests that his enjoyment of spending time in the pool is linked to his appreciation of perceiving a simultaneous sense of community, liberation and retreat within this 'off-ward' space. By stressing how "we all just enjoy ourselves" when spending time there, Tom provides further evidence to the researcher to support his account of how swimming and social activity within this space can afford patients a mutual experience of pleasure.

Tom's appreciation of "variation" draws on wider public understanding that participating in varied activities and experiencing a variety of environments is expected in typical daily life and fundamental to health and well-being. Tom's remarks therefore emphasise how patients in secure settings have a restricted number of different spaces to inhabit and how patients' access to the environments that are available to them is generally controlled and limited. His comments also highlight how such access to space may be taken for granted by people in other circumstances, including the researcher, who is able to freely access multiple different spaces during the course of a typical day. Tom thus describes time spent in the pool as "cherished" and his references to time when describing the swimming sessions taking place twice weekly and for a "solid hour", presents an image of these as valued slots of time that form joyful highlights within his weekly routine. The image of activity being scheduled and divided up into hourly blocks that is evoked through the narrative also highlights how events, including visits from family and friends, are typically planned in advance. Reduced access to spontaneity and the impromptu activity associated with typical everyday home life may accordingly be contributing to patients' perceptions of an institutional environment and a lack of domesticity in secure settings.

5.2 Ingrid

Ingrid had been working as a healthcare assistant in a medium secure ward for several months at the time of meeting. She is enjoying the role and is also interested in working towards becoming a nurse. Throughout the narrative Ingrid presents herself as a caring and sociable person who enjoys spending time with patients and engaging together in different activities. She speaks about her commitment to nurturing therapeutic relationships and contributing to an overall sense of community amongst patients and staff on the ward.

Ingrid's narrative positions her within a very demanding environment and in a busy and varied role that she finds intense and challenging, but also enjoyable and

rewarding. Accounts of her work commonly refer to patient safety and Ingrid talks about different protocols and policies that staff and patients should follow to mitigate risk within the ward environment. The overall tone that underpins Ingrid's narrative is enthusiastic, optimistic and professional. However, she also highlights how staff resources are often limited and articulates her perceptions of how the daily experiences of patients and other staff members can be affected by staffing levels.

Ingrid is interested in how the ward environment is experienced by patients and staff and speaks about her appreciation of creative projects undertaken by patients to improve the appearance and ambience of the overall space with artwork. She also talks about her involvement with activities to help alleviate patients' experiences of boredom, particularly at weekends when sessions provided by the occupational therapy team are not typically scheduled.

Ingrid's storylines focus around: (1) appreciating efforts to improve the aesthetic environment and optimise space on the ward, (2) nurturing a sense of community amongst patients and staff, (3) valuing connections to external spaces and nature, (4) managing the intensity and challenges of the ward environment, (5) awareness of ward protocols and safety, (6) the impact of positive noise in the environment (e.g. the sound of patients talking and laughing together).

During the interview, Ingrid reflected on photographs of: (a) patient artwork and messages that form a 'welcome' collage on the ward entrance corridor wall, (b) wall artwork in entrance corridor, (c) paper fireplace attached to wall below the television unit, (d) decorations by patients on occupational therapy kitchen windows, (e) meeting room, (f) meeting room window decorations by patients, (g) occupational therapy room, (h) views through glazing to the external ward courtyard in the snow, (i) dining room and (j) exercise bike in quiet room.

The interpretation of Ingrid's narrative in the extracts that follow will focus on the third and fourth storylines, through which she highlights an awareness of the ward

as an intense and often stressful place where she typically works long shifts. In the following story, Ingrid reflects on her embodied experiences within this environment to consider how spatial interactions, both on and off the ward, might be associated with her affective experiences and perceptions of well-being at work.

Ingrid: Extract 1

1.	R:	[] So thinking about, about well-being, um
2.	I:	Yeah.
3.	R:	and um, your own well-being, um, are there any ways in, you think that, you
4.		think that your working environment might affect your sense of well-being at
5.		all?
6.	I:	(Slight pause) I think, er, not going outside is one, because I like to be outside,
7.		but then I always make sure that like what, if I do get my lunch break, I do go
8.		for an hour and half walk.
9.	R:	Mm.
10.	I:	I like to go outside and I think that's the one thing for me where it impacts
11.		because like, you don't leave the war-, the building, let a alone the ward a lot of
12.		the time
13.	R:	Mm.
14.	I:	especially if you don't get a break. Um, so I think that can be quite like negative
15.		on our own well-being.
16.	R:	Mm.
17.	I:	'cause I like to be outside, because like you come to work, especially in winter
18.		now.
19.	R:	Mm.
20.	I:	you come to work in the dark, you leave in the dark, you don't go outside, you
21.		don't see what it's like outside, you kind of like, your days just disappear.
22.	R:	Mm.
23.	I:	But, yeah, I think that's, that's the one thing, but, if we can try and get outside in
24.		the courtyard at least that's something
25.	R:	Mm.
26.	I:	but sometimes, if there's, if we're really short on staff, you, if you don't get a
27.		break, the chances are no one's going outside 'cause there's not enough staff to
28.		have people outside, someone outside and people in the day area as well as
29.		people doing all the other various different things.
30.	R:	Mm.
31.	I:	Um so, yeah. I think that's probably the one thing (<i>laugh</i>).

32. R :	Yeah. Yeah and, and i-, is, what is it about being outside that's?
33. I :	I don't know, I like the fresh air.
34. R :	Mm.
35. I:	And it helps me like think through things. I don't, I find it's just a good ki-, way
36.	of coping with things, like say we've had like a really hard morning, doing lots
37.	of restraints or
38. R:	Mm.
39. I:	you know, or someone got assaulted or something like that, you know, it's a lot
40.	to like process.
41. R :	Mm.
42. I :	So for me like going for a walk outside, it really helps me like work my way
43.	through it mentally, like just walking, I find it really therapeutic for me.
44. R :	Mm, mm.
45. I:	It's something I find really useful.
46. R:	Mm.
47. I:	It's like not having, sometimes when you don't have that option and the ward
48.	has been that hectic and there's no staff in, it can be quite detrimental.
49. R:	Mm.
50. I :	Like it just plays on your mind more, but.
51. R:	Yeah, yeah.
52. I :	Yeah.

Whilst Ingrid highlights how she values spending time outdoors during breaks when on shift, her description of typically walking outside, "if I do get my lunch break", suggests the possibility of not having the break to be a common occurrence. Taking a lunch break is thus presented to the researcher as being more of an exception than a given, such that opportunities for staff to spend time outside during a shift are not guaranteed. Ingrid's position is reinforced by her repeated assertions that "I like to be outside" and "I like to go outside", which are presented in contrast to her account of experiencing days when she does not leave the building, or indeed the ward itself. Having a lack of sensory connection to the external environment during the day is presented as a problematic experience that is especially poignant during winter months when it is dark outside at both ends of a shift. Through her narrative image of days that "just disappear", Ingrid conveys a sense of how the daytime cannot be fully appreciated when working inside the building and how a lack of access to external space affects her perceptions of time. Alongside this, the discomfort connected with not having access to the external environment is suggested to be further compounded by resourcing levels and their associated influence on the ability of staff to take breaks. Earlier in the interview, when discussing a photograph of the ward courtyard, Ingrid notes that her shifts are typically 12 hours long and mentions how she typically feels "energised" after spending time there during a shift.

Accordingly, Ingrid suggests here that access to the ward courtyard at least would be appreciated on the days when she is not able to leave the building. Her narrative implies, however, that when the ward is "really short on staff" and staff "don't get a break" then there are typically insufficient staff available to observe patients outside in the courtyard and also complete the other necessary duties on the ward. Ingrid summarises this situation and presents an image of the uncertainty associated with daily staffing levels through her observation that on these days, "the chances are no one's going outside". Her story therefore highlights similarities in the experiences of staff and patients and how staff availability affects the everyday activities of both groups. Correspondingly, staff resources are seen to be associated with, not only the availability of breaks for staff, but also with the daily experiences of patients and staff more broadly, including limits on access to outdoor space for both.

Figure 10 Ingrid's photograph of the ward courtyard taken on a snowy day



Following the researcher's question about what Ingrid appreciates when having opportunities to spend time outside, her initial response stating that she enjoys the "fresh air", positions the air quality of external space in contrast to the internal environmental conditions during a long shift. Ingrid's account also presents spending time walking outside as a valued activity that enables her to process the embodied experiences associated with troubling events taking place on the ward. When describing a "really hard morning" in which "lots of restraints" or assaults on staff might have taken place, Ingrid presents an image of an intense and demanding environment that she finds physically and emotionally challenging. Her depiction of aggression and the potential use of restrictive practice evokes imagery of an emotionally charged environment where staff and patients might experience distress personally or observe other people's distress during their everyday interactions on the ward. Ingrid's comments about her affective responses to being involved in restraining patients also reflects existing research findings which suggest that implementing restrictive measures is typically a significantly distressing and dissatisfying experience for nursing staff (Duxbury & Whittington, 2005; Moran et al., 2009). Ingrid's account of walking outside as being a "therapeutic" activity therefore acknowledges her everyday encounters with some events on the ward to be difficult experiences that require processing. By describing how she might work her way through troubling experiences by walking outside, Ingrid presents an image of this step-by-step movement through space and simultaneous processing of her experiences as an integrated mind-body process.

Her account of frequently not having the "option" to walk outside also conveys Ingrid's perceptions of experiencing a loss of choice associated with the days where there are reduced levels of staff. This story thus implies that staff and patients might typically share common everyday experiences and frustrations on the ward, including having restricted access to outdoor space and having the desire to spend time walking outside the hospital building as a therapeutic process. Ingrid's positioning of situations where she cannot go outside as being "quite detrimental" accordingly highlights associations between staff mental health and the availability of sensory connections to outdoor spaces during a shift. In reference to experiencing distressing events, Ingrid's concluding comment that, "it just plays on your mind more" implies that without opportunities to walk outside and process such experiences, they can become 'stuck' and continue to play over in her thoughts.

In the next extract, which follows directly from the first, Ingrid is prompted by the researcher's questioning to reflect on her use of indoor spaces to manage discomforting experiences at times when it is not possible to leave the building or access outdoor spaces.

Ingrid: Extract 2

1. R :	Are there places, are there places where you, you can go um, if you, you, if you
2.	if you don't have the opportunity to go outside, um?
3. I :	Um sometimes you can like ask for like, like half, like ten minutes, twenty
4.	minutes, just to like, I'd go sit out, um, just as you come out of the ward, there's
5.	like some chairs there.
6. R :	Mm.
7. I:	And you just go and sit there for a little bit, some, it's like even though you're
8.	not out the building, you're still off the ward
9. R:	Mm.
10. I:	and you're like away from everything, 'cause if you sit in the office, you're
11.	gonna get asked questions, like people will be like, oh, could you do this, could
12.	you do that, could you do this?
13. R:	Mm.
14. I:	And you don't really ever get a chance to like, even if you took your lunch break
15.	at, in the office, you're not on your lunch break
16. R:	Mm.
17. I:	'cause you're just, you'll just be running around, so yeah, so like just being able
18.	to like get off the ward in that little area is really nice.
19. R:	Yeah. Yeah and that's something that you might do?
20. I:	Yeah, I've done it a few times.
21. R :	Yeah.
22. I :	Like, on like when we hadn't had enough staff for like full breaks, just go for
23.	like ten minutes or something, just to get a bit of space (<i>laugh</i>). Go grab a cup of
24.	tea, sit down (Sound of air blowing out 'phew' sound and laugh).
25. R:	Mm.
26. I:	Relax as much as you can.
27. R :	Mm. Yeah, yeah.
28. I:	Yeah.

This account continues to present an image of staff being continually on the go in their duties at work and implies that perceiving a sense of disconnection and liberation from the intensity of the ward is necessary for taking a break and attempting to relax. Earlier Ingrid highlighted how she perceives open-air spaces outside the boundary of the hospital building to be optimal locations in which to walk and process her affective responses to difficult experiences at work. In instances when she is unable to go outside the building, working in the open-air courtyard on the ward was also presented as an alternative way to appreciate being in an outdoor setting whilst on shift. Ingrid's account here suggests, however, that even when she is not able to be outside in the open, it is helpful for her to perceive a sense of being *outside* the ward. Although the seating area she describes is situated directly outside the ward entrance, Ingrid's repeated positioning of this space as "off the ward" conveys her appreciation of the physical and psychological separation afforded by the doors that enclose the ward.

Ingrid thus presents the ward as an enclosed and seemingly 'airtight' space, in terms of both physical and atmospheric containment and she implies that an intense affective atmosphere associated with occupants' experiences of distress and prolonged confinement can build up over time. Ingrid's request for a short break outside conveys how her responses to the uncomfortable atmospheric 'pressure' on the ward include the urge to temporarily escape the ward environment. By highlighting how she is able to "get a bit of space" when taking a short break in the seating area, Ingrid expresses a sense of release in being outside the ward boundary, despite still being contained inside the hospital within an indoor space. This image of release is further accentuated by the 'phew' sound that Ingrid makes in conclusion, which expresses a sense of relief and evokes a symbolic image of her breath being held and then released during momentary respite from the intensity of the ward. It is noteworthy that Ingrid does not mention the designated staff room in the building that is referenced by other staff participants in the data set. Her account of short breaks on days when it is not possible to take a full break or leave the building, however, implies that there may not be sufficient time to make a meaningful visit to the staff room at these times.

Ingrid's portrayal of the ward as a hectic and unremitting environment is further expressed in her account of unsuccessful attempts to take a lunch break in the nursing office. The imagery of staff being constantly interrupted whilst on a break in there conveys a sense of exposure and indirectly alludes to how staff are very visible in

the nursing office due to the surrounding glazing. By describing being "away from everything" when taking a short break in the seating area outside the ward, Ingrid thus underscores her perception of how attempting to take a proper break is incompatible with remaining inside this highly intense environment. The account of how Ingrid would typically "go grab a cup of tea" and "sit down" in the 'off-ward' space, presents an image of fleeting activity, but also conveys her appreciation of a sense of restoration there and respite from the ward. This imagery implies that perceptions of comfort are provided by the simple pleasures of drinking tea and having a seat. Ingrid therefore indirectly conveys how mundane activities, including drinking or sitting down might be difficult to achieve in a busy ward environment, where staff are typically on their feet and continually moving or "running around".

Ingrid's experiences accordingly illustrate how spaces and objects are used by staff as resources to manage experiences of physical and psychological discomfort in the ward environment and enable restorative transitions away from troubling affective states. A sense of restoration is thus understood to be generated by physical movement through space and particularly by movement across the boundaries that shape the contrasting atmospheres between intense and pressured spaces on the ward and more open or relaxing 'off-ward' spaces. Ingrid's accounts can also be seen to support research suggesting that break spaces are most likely to be used when located in close proximity to nurses' working areas and how providing access to outdoor spaces can significantly improve the perceived restorative quality of staff break areas (Nejati et al., 2016). Furthermore, at times when she is unable to leave the building, Ingrid's use of the space directly outside the ward, or the courtyard illustrates how hospital staff might improvise to create environments for breaks that can afford restoration and increase a sense of well-being (Pink et al., 2020).

5.3 Patricia

Patricia is a middle-aged woman who had been a patient on a medium secure ward for a number of years at the time of meeting. She describes making progress in her journey through mental healthcare services and her hopes for moving on to a low secure ward, before leaving hospital to live in a place of her own. Patricia mentions having a love of reading and in particular talks about how she enjoys spending time reading novels and newspapers. As she does not currently have permission to leave the building for ground leave, Patricia's time is largely spent indoors, however, she describes her enjoyment of spending time within natural settings when she previously had access to the hospital grounds. Patricia did not include external spaces in her photographs but mentions walking around the ward's outdoor courtyard at times for exercise and talks hopefully about the possibility of having unescorted leave when she moves on to a low secure ward. Patricia also highlights her appreciation of keeping close connections to her family and friends, including her enjoyment of having visitors, or keeping in regular contact using the telephone booth on the ward.

Throughout the narrative, Patricia repeatedly articulates her perceptions of the ward overall as a generally noisy, testing and sometimes frightening environment. However, although many aspects of life on the ward are presented as being very difficult, the narrative suggests that Patricia is resigned to her circumstances and implies that she perceives some of the daily challenges in the environment to be an almost inevitable part of patient experience. The overall narrative conveys a sense of understated fortitude in relation to her situation and Patricia's general tone throughout is hopeful and accepting. Patricia mentions that patients on the ward are typically empathetic towards each other and she expresses appreciation for the positive and supportive relationships that she has developed in hospital with other patients and members of staff. Patricia's stories about her experiences are intertwined, but her main storylines focus around: (1) appreciation for spaces and objects that facilitate connection to other people and places, (2) awareness of associations between her personal possessions and perceptions of well-being, (3) affective responses to sound and the multi-sensory ward atmosphere and (4) enjoyment of spaces that afford connections to nature. Here the focus will be on the first, second and third storylines that Patricia returns to throughout the interview. These storylines have been selected to exemplify Patricia's affective responses to the ward environment and her ways of engaging with spaces and objects to mediate difficult and distressing psychosocial experiences.

Patricia discussed photographs of the following during the interview¹¹: (a) dining room (b) communal day room, (c) laundry room, (d) telephone booth, (e) armchair in dayroom, (f) meeting room also used for arts and crafts activities and mindfulness sessions, (g) quiet room, (h) bedroom corridor, (i) her bedroom and (j) en suite shower room.

In the following extract, Patricia reflects on her emotional responses to the day area, which is presented as an often noisy and unpredictable environment.

Patricia: Extract 1

1.	P:	Yes. That's one of the chairs that we sit on for the day, which gets a bit
2.		uncomfortable, but it's a good place for me because I spend a lot of time
3.		reading the newspapers to keep up with what goes on in life and also, I've got
4.		some good thriller books that I read, so that's usually a happy sort of place.
5.	R:	Yes.
6.	P:	Um it's a bit of a chaotic space in that there's a lot going on,
7.	R:	Mm.
8.	P:	people are often shouting. It can feel a little bit scary if they're kicking off and,
9.	R:	Mm.
10.	P:	and worried that you'd get hurt, but I haven't ever been hurt so that's good.

¹¹ It was necessary to review Patricia's photographs on the camera screen during the interview because prints could not be produced with the equipment available on the ward. As the images could not be transferred digitally, they are not available for inclusion.

- 11. **R**: Mm. That's good. Do you tend to sit in the same place, or do you?
- 12. **P:** Yes, I do.
- 13. **R**: You do, yes.
- 14. **P**: Yes, I've got my own little place and I've got my things around me.

Patricia's reflection on the day room centres on the armchair where she typically sits and her description of being seated there "for the day", implies a sense of routine and that her daily activity within this area is typically sedentary. Although Patricia acknowledges the discomfort of sitting for long periods of time in her chair, she emphasises the positive attributes of the seat as a "good place" where she enjoys spending time reading newspapers and books. By highlighting to the researcher how reading the papers in that setting is valued for maintaining connections to events in wider "life" beyond her own, Patricia also indirectly alludes to her position as a detained patient, who does not currently have physical access to the world outside the hospital building. Patricia juxtaposes her perceptions of her chair as a "happy sort of place", where she enjoys reading, with a contrasting account of the day area, which is presented as a hectic and sometimes stressful environment where "there's a lot going on". Although Patricia presents tensions perceived within the environment matter-offactly in this story, her description of the day room as a "chaotic space", where people are "often shouting" or "kicking off" and where she feels afraid for her own safety at times, presents an image of a volatile, noisy space imbued with potential aggression or threat.

Amidst imagery of a regularly unsettled atmospheric context, her chair is positioned as a discrete physical and psychological zone, which creates a personal territory within the communal lounge, that Patricia can perceive as "my own little place". By emphasising the space as being "my own", Patricia conveys an appreciation of perceiving a sense of ownership in this secure setting, where patients typically do not have access to their own furniture items and are generally restricted on the quantity or types of personal possessions they can have in hospital. Patricia's account of sitting in her armchair also positions it as a more grounded and stable space, when contrasted with imagery of a turbulent surrounding environmental atmosphere. This sense of stability is further emphasised by the physicality of the chair itself, which in line with the style of lounge seating typically specified in secure psychiatric settings, is large, robust and heavily weighted to hinder movement.

Patricia's confirmation to the researcher that she typically sits in the same chair each day reinforces an image of routine within her daily life and positions the armchair as a familiar space. Patricia's account of sitting habitually in the same seat can therefore be understood to be appreciated for the sense of stability and familiarity it might afford, in contrast to the ever-changing atmosphere and dynamics of the surrounding lounge area. Patricia's remark that "I've got my things around me", evokes imagery of her individual zone within the lounge being extended by her possessions, such that greater perceptions of stability and protection can be afforded in this unpredictable environment by having these objects around her. By creating a personal territory using her chair and personal possessions, Patricia can be understood to be using objects and space to generate a helpful boundary between conflicting circles of activity that are occurring within the same environment. Patricia can accordingly be seen to be enabling a greater sense of normality and domesticity through her creation of a personal territory that constitutes a world situated within the wider, volatile world of the overall day area.

Although the ward is an ostensibly therapeutic environment and a place that Patricia currently considers to be home, her exposure to the distress and agitation of others in this setting is such that at times she is concerned about her own physical safety. The narrative of feeling at risk of potential harm in this environment therefore contrasts with wider societal understandings of hospitals typically representing safe and restorative places for the provision of care. When clarifying to the researcher that she has never actually been hurt there, Patricia's concluding remark stating, "so that's

good", implies gratitude alongside providing a sense of reassurance about this point, to both the researcher and to herself. The researcher, in turn, is seen to be affirming this statement by repeating Patricia's own words, "that's good" back to her.

The volatility of the environment in many UK psychiatric wards has been highlighted in previous studies (e.g. Jones et al., 2010; Quirk et al., 2004, 2006), where some patients have similarly reported perceptions of aggression and being at risk of violence from other patients when in hospital, or in some cases actually experiencing violent behaviour. Jones et al. (2010) found that whilst many patients felt a sense of safety or support from staff and other patients within inpatient psychiatric settings, any threats to their perceptions of security, including exposure to intimidation, aggression or bullying, prompted patients to feel unsafe and experience distress.

In the following extract, Patricia returns later in the interview to concerns about disruption, noise and perceptions of safety and describes an awareness of her affective responses to the overall ward atmosphere and the acoustic environment.

Patricia: Extract 2

1.	R:	[] Just thinking about sort of environments and your sort of sense of
2.		well-being are there any ways you might describe ways in which your
3.		environment affects how, how you are feeling?
4.	P:	Oh, I think it does when there is noise. I mean you can feel quite afraid when
5.		people are shouting and sound as if they are going to perhaps hurt somebody.
6.	R:	Mm.
7.	P:	Um, that can be very unsettling.
8.	R:	Mm.
9.	P:	Um and sometimes you just want to have peace, you want to be able to read in
10.		peace without the background noise and also to watch the telly. When you are
11.		trying to watch the telly in here it is very hard with background shouting and
12.		things like that.
13.	R:	Of course.
14.	P:	So, it does,
15.	R:	Yes, yes.
16.	P:	the environment does affect us quite a lot.
17.	R:	Yes and are there ways in which you try to deal with that sense of noise?

- 18. P: Well, I try to do the distress tolerance and switch off to it and take myself to a
- 19. nicer place and just try to let the whole thing wash over me without getting
- 20. involved.
- 21. **R**: Yes.
- 22. **P:** I never answer back to the people either, just ignore them.

Here Patricia reiterates how noise produced by others is a significant contributor to her own feelings of distress experienced in the ward environment. Intrusive sounds and the vicarious experience of other patients' distress and agitation are presented as environmental stressors that Patricia cannot readily escape within a space that is shared with a number of other patients and staff members. Patricia's portrayal of the aggressive shouting as "unsettling" conveys an image of how this unwanted noise can have a destabilising effect on both her own affective state and on the ambience of the space overall. The term "unsettling" recurs throughout the interview and is used by Patricia to describe her perceptions of disruptive events or troubling situations experienced on the ward that influence her affective state. By contrast, the words "peace" and "peaceful" also repeat frequently throughout the interview. Patricia's assertions that, "sometimes you just want to have peace", or "read in peace" illustrate here examples of her over-arching narrative focus on seeking tranquillity and calm within a chaotic and unpredictable environment.

Background sound, including noise from other patients is therefore positioned as an obstruction to Patricia's full enjoyment of leisure activities in the space, which typically include reading or watching television. Later in the interview, when reflecting on perceptions of her affective experiences within her everyday environment, Patricia also notes that,

"[...] I am enjoying a book at the moment, so that takes me away from it, I feel in a different place [**R**: Yes] so that's good and sometimes I sleep in the lounge as well and that takes me somewhere different too (*laugh*), [**R**: Yes (*laugh*)] we're not supposed to (*laugh*)!"

This account positions both reading and dreaming as imaginative processes that allow Patricia to transition between affective states and enable her embodied experiences to be transported from the ward such that she can "feel in a different place". Patricia's use of her imagination when reading and engaging with television programmes, or when asleep dreaming, can thus be elaborated as a form of 'liminal affective technology' (Stenner, 2021; Stenner & Zittoun, 2020) that allows movement between worlds and enables her to experience a temporary sense of detachment from her current reality. In turn, these self-generated transitional processes of imagination can be understood to be helping Patricia manage the 'spontaneous' liminal experience of being in hospital or the experience psychological distress (Stenner, 2021). In other words, Patricia can be seen to be negotiating the discomfort of overarching and disruptive life transitions by using her imagination to generate liminal experiences that can facilitate transition between affective states. Patricia's story suggests, however, that background sounds on the ward, that may typically also convey distress, can disrupt her processes of affective transformation. Loud ambient noise is accordingly understood to be reducing the efficacy of her imaginative processes (e.g. those associated with reading or watching television), to facilitate transportation away from her current situation and enable supportive psychosocial connections to other places or people.

Patricia's appreciation of experiencing a sense of psychological detachment from the ward is also highlighted by her description of using the distress tolerance techniques she has learnt as a strategy to enable her to "switch off" from unwanted noise and take herself "to a nicer place". Alongside this, Patricia highlights later in the interview how she has learnt mindfulness meditation practices in hospital and uses these to take herself to a "safe place" psychologically, by focusing on her senses and embodied experiences. Throughout her overall narrative, Patricia's accounts repeatedly express how experiencing psychological safety as well as physical safety in hospital is significant to her overall sense of well-being. The predominant focus on

physical safety and risk mitigation in mental healthcare environments, however, is such that emotional or psychological safety is not seen to have the same parity of esteem (Veale et al., 2022). As Veale and colleagues argue, emotional safety is required for a person to maintain physical safety in psychiatric care and yet the prioritisation of physical safety in mental healthcare settings can result in higher levels of stress that may paradoxically increase physical risk.

Patricia's story also illustrates how patients might engage with liminal affective technologies (including mindfulness techniques) to negotiate their responses to witnessing other patients' distress and cope with the volatility of the ward environment. Through her concluding remark expressing how she avoids responding to provocation from other patients, Patricia also provides further testimony to the researcher of her desire to reduce stress by attempting to detach and disengage herself from an unsettled ward.

Whilst there is limited existing research examining occupants' experience of the acoustic environment in psychiatric inpatient settings, a body of evidence from studies in general hospitals suggests that excessive noise, or 'unwanted sound' can have significant detrimental effects on patients' physical and psychological health (Choiniere, 2010; Hsu et al., 2012; Ulrich et al., 2006). Westman and Walters' (1981) exploration of relationships between noise and stress also highlights how, in accordance with innate survival instincts, the meanings associated with sounds are key determinants of human responses, such that danger is signified by threatening sounds. Sounds can also take on specific meanings for individuals depending on their lived experiences and in secure psychiatric settings, where patients have commonly experienced difficult and distressing life events (Coid, 1992), perceptions of sound can be understood to be especially significant (S. D. Brown et al., 2019b).

As highlighted by Patricia's experiences, when sharing a hospital ward environment with a significant number of other people, patients also have minimal

control over ambient sound levels and have limited ability to avoid unwanted sound (Rice, 2003). Rice's (2003) ethnographic study of soundscapes in a general hospital explored how sounds are imposed on occupants and how limited sensory stimulation within the overall environment might contribute to heightened sensitivity to sound. Whilst sound interpretation enabled patients to navigate and understand routines in the environment, the study also found that it served as a continual reminder of their status as hospital patients. Although research exploring sound control in psychiatric settings is limited, seemingly simple interventions, such as the application of felt pads to furniture have been found to assist in reducing overall sound levels in mental healthcare environments (J. Brown et al., 2016). Furthermore, staff in Brown and colleagues' study perceived that the resulting decrease in noise contributed to a reduction in violent incidents observed on an older adult mental healthcare ward, as part of the overall effect of measures undertaken to reduce aggression in that environment.

Perceptions of sound and noise levels also contribute to staff experiences in hospital environments and a literature review undertaken by Ryherd et al. (2012) suggests that the effects of noise on hospital staff are generally negative. Sensory overload affects the central nervous system's capacity to process environmental stimuli and exposure to noise can diminish cognitive performance or induce psychophysiological stress responses, including activation of the sympathetic nervous system and raised blood pressure (Stansfeld & Matheson, 2003; Westman & Walters, 1981). Ryherd and colleagues' (2012) review findings accordingly imply that noise exposure can affect hospital staff members' health and levels of stress, in addition to their job satisfaction, performance and perceptions of the psychosocial environment at work. Anecdotally, in the present study, although interviews took place privately within closed rooms, loud noises, including slamming doors or distressed patients shouting were commonly heard in the background, including during the interview with

Patricia. In a few recordings, there were also instances where speech was momentarily inaudible due to the volume of the background noise.

Patricia also highlights her perception of noise and a sense of instability being prevalent in the bedroom corridor environment in the next extract when she reflects on her emotional responses to the bedroom corridor and her own bedroom space.

Patricia: Extract 3

1.	P:	That's the bedroom corridor which again can be quite noisy and a bit chaotic.
2.	R:	Mm.
3.	P:	And where we queue up in the morning in order to be let out, so that feels a bit
4.		strange (laugh).
5.	R:	Mm.
6.	P:	That's my bedroom, which at the moment hasn't got any of my possessions in
7.		so it's a bit sterile and a bit cold and obviously I do sleep there and relax there,
8.		but it doesn't feel very inviting right now.
9.	R:	No, how do you feel about um, not having your possessions?
10.	P:	Oh, no I don't like that at all. They are all under the bed and they are all
11.		higgledy piggledy and it's very unsettling.
12.	R:	Mm.
13.	P:	It's very hard to find outfits to wear. I can find my clothes, but I can't find
14.		outfits and it's very, very unsettling to do that.
15.	R:	Mm.
16.	P:	and I have to wear those very strange pyjamas, rip-proofs at night-time
17.	R:	Mm.
18.	P:	and again, that's rather uncomfortable and it feels a bit like a punishment,
19.		which of course, it is really.

Patricia's positioning of the bedroom corridor as "noisy" and "chaotic" echoes the earlier account of her perceptions of the day area and strengthens an image of the communal ward areas being loud spaces where unpredictable events might frequently take place. The imagery of patients queuing up there in the morning evokes a more carceral narrative and alludes to how power differences between staff and patients in secure settings are expressed and reinforced through spatial interactions and control of the environment. Whilst Patricia acknowledges that the experience of queuing there can feel "a bit strange", her short laugh following this also highlights a sense of understatement in this remark to the researcher. Patricia's story of being in a "chaotic" environment whilst waiting "to be let out" is unfamiliar in a residential context and evokes imagery that might be more commonly associated with animals awaiting release from an enclosure or with the experience of detention in prison settings. Through this story, Patricia also highlights how patients may experience various layers of physical or procedural restriction that typically occur at the thresholds between different processes or spaces within the secure boundary of the ward itself.

Patricia describes how her bedroom has been cleared of all her belongings including her clothes. It is assumed that this is intended for her physical safety, however, Patricia interprets this as a form of "punishment" for misbehaving and as a removal of privileges, rather than an expression of care in light of her self-harm or suicide risk. Patricia's depiction of the room without her possessions as "sterile", "cold" and uninviting creates imagery of a stark, clinical and non-domestic space. Interpreting the meaning of "sterile" as literally being free from life, the narrative presents an image of the room, when stripped of Patricia's belongings, as having also been stripped of any sense of her own life. As patients' bedrooms on a ward are typically repeated with an identical layout and design, the removal of Patricia's possessions can also be seen to be taking away the expression of her presence and individual identity from the space. Although the bedroom remains functional as a place to sleep and rest, in the absence of her belongings, the previously comforting qualities of the room as a supportive resource have been significantly diminished.

Patricia expresses how her distress is connected to being separated from her belongings and also by her feeling that these objects are disordered. Consistent with her earlier accounts of troubling experiences in the lounge, Patricia highlights again here her perceptions of "unsettling" embodied experiences. She articulates how these experiences are associated, not only with the loss of her possessions, but equally by her

awareness that they are in a state of disarray when locked away. When permitted access to her belongings, Patricia can find individual items of clothing, but a loss of choice can be inferred from her report of not being able to create outfits. In turn this can be interpreted as a loss of ability to express her identity fully using a selected ensemble of clothes. When considering her perceptions of disorder within the environment later in the interview, Patricia reiterates her experience of distress associated with her clothes being stowed "under the bed in a bit of a heap". She describes this experience as "very disheartening" and also asserts that "[...] I like my room ordered; I like to feel my things around me". This sense of the pleasure and reassurance gained from her belongings being associated not only with seeing them, but also with sensing their presence around her, resonates with Patricia's earlier comments about appreciating the feeling of having her things surrounding her in the day area.

Belk (1988) highlights the relationships between possessions and self-identity by building on William James' (1890, as cited in Belk, 1988) proposition that things which might be considered as 'mine', might also be perceived as 'me', to present a concept of the 'extended self'. Within the conceptual framework of the 'extended self', Belk argues that objects, people or places are all able to form part of an individual's sense of selfhood. In terms of forming an extension of the self, since clothes are worn directly on the body and are used to express a sense of identity, clothing can be interpreted as a form of possession with close links to selfhood. Patricia's account expresses her sense of both physical and psychological discomfort when sleeping in the rip-proof pyjamas and the requirement to wear these garments that have punitive characteristics is seen to contribute to her sense of being punished. Furthermore, her story evokes an image of the rip-proof pyjamas as a form of uniform (e.g. the standard issue clothing worn by prisoners) that in turn could be understood to be diminishing her sense of individuality. Her overall perceptions of having a lack of choice and control over her

everyday environment can thus be seen to be exacerbated by the removal of access to her belongings and her own clothing specifically. In the following extract, Patricia elaborates further on the affective significance of perceiving a sense of choice in her personal environment and having access to her possessions.

Patricia: Extract 4

1.	R:	[] Um, I was also, um, just thinking about, um, touching on it really thinking
2.		about your, your room, but um, wondering how important it is for you have
3.		choice when considering your personal environment?
4.	P:	Oh very much so. I like when I have got my own pictures around me, I put
5.		pictures in the wall of family and friends and oh, just have things like my books
6.		and toiletries, things personal to me, is important.
7.	R:	Yes.
8.	P:	And also to have my clothes organised so that I can feel on top of things.
9.	R:	Yes.
10.	P:	So yes, as it is at the minute, which is called a sterile room, it's really very
11.		unsettling.
12.	R:	It's absolutely sterile?
13.	P:	Yeah, it is sterile, that's what they call it. They clear everything out of it, all into
14.		the cupboards.
15.	R:	Mm.
16.	P:	But I'm hoping to get it back today (<i>laugh</i>).
17.	R:	Yes, yes. And are there any particular objects that you have, when, when those
18.		are, um, out in your room that you enjoy?
19.	P:	Oh yes, my photos.
20.	R:	Your photos.
21.	P:	My photos to remind me of everybody. They're the most important things to
22.		me.

When considering choice in her personal environment, Patricia's account of the significance of having her "own" pictures "around" her conveys a further sense of how personal possessions can be seen to represent an extension of her own selfhood and identity. The items she lists are "personal" to Patricia due to their psychosocial or physical closeness to her (e.g. as representations of family and friends, or as objects that she holds, such as books, or other items that come into contact with her body,

including toiletries and clothes). In contrast to her earlier account of the distress associated with her clothing being "higgledy-piggledy" and locked away under the bed, here Patricia clearly expresses her preference for experiencing a sense of order over disorder within her everyday environment. Patricia also underscores the psychological significance of being separated from her possessions in her account of this experience being "really very unsettling". The narrative accordingly implies there to be a direct relationship between the organisation and accessibility of her personal possessions and Patricia's own affective state. Whilst the experience of her clothes being unreachable and stored in disarray is described as distressing, in contrast, having an awareness of her clothing being orderly and organised enables Patricia to feel psychologically prepared and "on top of things". The removal and disordered storage of her belongings is thus seen to be having a disruptive and destabilising effect on Patricia by compromising her sense of agency and control over possessions that form part of her extended sense of self.

As an individual's most intimate belongings, including clothes, are typically stored in bedroom environments, the use of the term "sterile" to describe Patricia's room contrasts starkly with wider perceptions of bedrooms as highly personal and typically comfortable spaces in which to rest and relax. This description instead evokes imagery of an overtly clinical environment which surprises the researcher who queries if the space is "absolutely sterile". Patricia's clarification that the term is used by staff to describe the bedroom once her belongings have been stripped out thus presents an image of her possessions as a form of impurity or contamination within a spotless environment. This account also links back to Patricia's earlier description of her room as feeling "sterile", suggesting that her own experiences of the space may be reinforced by the term used by staff to refer to the room following the removal of her possessions.

Patricia's assertion that her photographs are the most important possession in her bedroom, however, highlights the significance of using these images to maintain

connections to family or friends and to events in the past. Her account of how the photographs "remind" her of everyone also evokes a sense that without them she might fear losing contact with people or forgetting them. Patricia's narrative highlights how there is a reassuring quality associated with having personal possessions in her room and being able to sense them around her. Whilst the removal of her access to these objects is reflective of predominant concerns about her physical safety, Patricia's account of being without her belongings suggests that this can also be destabilising and diminish her sense of psychological safety.

5.4 Concluding comments

The three narratives explored in this chapter have demonstrated how spaces and objects are used by patients and staff in secure settings to assist in navigating difficult emotions or facilitating transition between affective states. These accounts also provide illustrative examples of similar strategies expressed by participants across the overall data set. Drawing further on liminality as a theoretical framework through which to approach the interpretation of narratives, participants' accounts of managing difficult experiences have been considered in relation to processes used to self-produce liminal experience and assist transition between affective states (Stenner, 2020, 2021). Importantly, these self-generated, or 'devised' forms of liminal experience that can facilitate affective transformation (e.g. visualising future spaces), are also seen to be helping some participants mediate the raw or 'spontaneous' liminal experience associated with disturbance imposed by challenging and typically unforeseen circumstances (Stenner, 2017, 2021).

For patients, disruptions to their lives or challenges to their sense of selfhood could be understood to be associated with significant unplanned liminal transitions, including the experience of mental health crises or detention in hospital. Processes of self-producing liminal affective experience are accordingly seen to be helping some patient participants make sense of difficult or unfamiliar circumstances associated with
these significant transitions or ruptures within their lives. Staff participants' accounts similarly convey how processes of self-generating liminal experience to transition between affective states, such as physically moving between spaces, can help to mitigate unsettling experiences, or the sense of unease brought about by imposed occasions of spontaneous liminality. The narratives explored illustrate how incidents, for example, those involving the physical restraint of patients can disrupt staff members' routine flow of psychosocial experience. In circumstances such as these, members of staff are understood to be shifting from an enabling and caring role to an authoritarian or potentially punitively perceived position.

In order to produce 'devised' liminal experiences that enable affective transition, participants describe engaging with diverse forms of 'liminal affective technologies' or 'symbolic resources' (Stenner, 2017; Stenner & Zittoun, 2020). These experiences are typically used to enable transition from unsettling or distressing affective states to those that are more reassuring and supportive. Accordingly, patient accounts convey how imaginative processes can assist in creating a distraction from the difficult experiences associated with their current circumstances. These may include patients' unhelpful reflections about their individual situations when compared with ideal storylines about how their lives might have been envisaged previously. Imagination is thus seen to be engaged with for both supportive and motivational purposes and to assist in transforming experiences of troubling spaces and objects into comforting or incentivising imagery.

Such imaginative processes are also understood to offer a means for individuals to be psychologically transported to and from other places and across time. Spaces and objects are seen to facilitate participants' access to unlimited psychosocial connections between different people, places and events situated in past, present and future settings. Rich visual imagery to illustrate patient participants' future lives beyond hospital is accordingly constructed through detailed narrative accounts of the

envisaged spatial contexts in which these lives would be lived. Patient participants' processes of imagining and psychologically inhabiting future spaces are hence seen to provide a supportive and motivating sense of progressive transition from their current circumstances towards life beyond hospital.

Enacting physical movement through space via the use of visualisation is similarly understood to facilitate perceptions of forward progression that can enable participants to deviate from an unhelpful sense of being trapped between worlds or being suspended in an ongoing process of liminal transition. Actual physical movement through space and specifically, the transition between internal environments and external spaces is also seen to be mediating patient and staff experiences of psychological discomfort within the confines of restrictive indoor settings. This integration of mind-body activity produced by physical movement can therefore enable difficult or distressing experiences to be worked through by participants and processed by being 'walked through' during the transitional passage of moving through space.

Further examples of how liminal affective technologies are used to assist transition between emotional states are conveyed through participants' accounts of ordering and having control over space and objects, including personal possessions, using both physical and psychological processes. The narratives also highlight the significance of objects and personal possessions in helping to maintain and reinforce perceptions of self-identity. Objects and personal possessions are also seen to be helping participants create a sense of stability and construct a supportive 'worldwithin-a-world' that can provide a sense of protection from the often challenging and volatile ward environment.

Participants typically describe valuing positive social interaction and perceiving a shared sense of belonging or equality in the company of other people who are seen to be going through a passage of similar experiences. Correspondingly, spaces that are experienced as relatively less restricted than others are understood to afford

participants a greater sense of togetherness or equality and are accordingly seen to evoke a supportive sense of *communitas* as conceptualised by Turner (1969/1996). Experiences of transitional 'flow' between affective states or identity positions are thus portrayed as being smoother in less restricted spaces, where there are also typically fewer symbolic cues relating to power, status, risk or control perceived within the built environment. The narratives also highlight how spaces are appreciated and valued for their capacity to afford spontaneous and light-hearted social activity. Accordingly, it is noteworthy to contrast Tom's account of experiencing a sense of solidarity and *communitas* in 'off-ward' spaces that facilitate positive social interaction, with Patricia's attempts to disengage from other patients and her surroundings when she experiences the ward as unsettled and stressful.

Alongside this, narrative accounts of valued respite and relative liberation experienced when spending time in 'off-ward' spaces located beyond the physical threshold of the ward are frequently expressed and by patients and staff. Such experiences of movement through space are thus seen to symbolise a transition between the intensity of the world inside the ward and a contrasting world outside. In addition to actual physical movement, participants similarly describe using imaginative processes to psychologically inhabit less restrictive and more tranquil spaces. These imagined spaces are hence providing a sense of release from the challenges of the immediate ward surroundings and helping to generate feelings of safety or stability. Patient and staff participants' narratives here and throughout the data set particularly highlight the significance of external 'off-ward' spaces that can assist in maintaining a sense of the passing of time and providing valued access to the multi-sensorial aspects of nature, including natural light.

Chapter 6: Navigating relationships in secure spaces

The previous analytical chapters have explored participants' accounts that exemplify two overarching narratives expressed across the overall data set. Stories examined in the first analysis chapter illustrated how participants make sense of uncomfortable spaces within secure psychiatric environments, whilst narratives discussed in the second chapter focused on engagement with spaces and objects as strategies for coping with difficult affective experiences within those settings. The analysis presented in this chapter will focus on a third overarching narrative concerned with the stories told about how interpersonal relations between and amongst staff and patients might be associated with and be grounded within their everyday spatial experiences of the hospital environment.

The analysis draws on the narratives of four further participants, Mike, Nathan, Bradley and Estelle. Mike, Nathan and Estelle are members of staff working in medium and low secure ward environments and Bradley is a patient on a low secure ward. The stories explored here are concerned with participants' observations and experiences of interpersonal relationships and interactions amongst patients and staff within different spaces on the ward. Their accounts illustrate how perceptions of relational dynamics might be associated with everyday experiences of space and qualities of the environment within secure hospital settings.

6.1 Mike

Mike works on a medium secure ward and is an experienced occupational therapist and long-standing member of staff. He positions himself throughout the narrative as a creative and energetic person who is enthusiastic about his day-to-day work. Mike's account of his role suggests that he is especially passionate about initiatives developed with patients to visually enhance the hospital environment and facilitate creative expression. These projects include the creation of artworks and colourful murals in

some of the communal areas that patients have designed and produced during group sessions. Throughout the narrative Mike expresses his thoughts about how taking part in such activities might enrich patients' everyday experiences within the ward environment and enable the development of new skills. His stories imply that participation in collective projects, including painting and decorating, gardening, or cooking, can encourage increased communication and social interaction amongst patients who may have previously struggled to work in a team. Based on his own observations, Mike suggests that when patients feel invested in projects and take pride in the results, it can prompt overall improvements in the ward atmosphere, including potential reductions in incidents and levels of aggression.

Mike's overall tone is generally optimistic and throughout the narrative he expresses a strong sense of dedication to his role and appreciation of the time that he spends working with patients. He is also interested in how improvements might be made to the hospital environment and expresses disappointment in aspects of the ward design, including his perceptions of spaces often having bland and clinical qualities. His overall narrative is focused accordingly on the perceived benefits of functional or aesthetic adjustments to the environment, including the use of coloured paint on walls to enhance the ward and enable a degree of personalisation to spaces. He also talks about his belief in the therapeutic benefits to patients of spending time within outdoor spaces and feeling connected to the natural world. Whilst Mike has a busy and demanding role that is often challenging, his stories suggest that he gains much professional fulfilment from observing patients working together on projects and perceiving improvements in their self-confidence and mental health.

Mike's main storylines focus around: (1) how spending time in 'off ward' spaces might benefit patients, (2) how perceptions of space can promote or impede social activity, (3) aesthetics and symbolism within the physical ward environment, (4) the

benefits to patients of participating in collective and creative activities and (5) how spaces and objects mediate affective experience and distress.

In the interview Mike reflected on photographs of¹²: (a) on-site café courtyard, (b) dining room servery hatch, (c) mural artwork by patients on dining room wall, (d) patient lounge / day area, (e) bedroom corridor, (f) ward entrance corridor with welcome signage and mural artworks by patients and (g) therapy kitchen.

The following analysis will draw on all of these storylines to focus on Mike's perceptions of how staff and patient experiences of the hospital environment are connected to social interaction and relationships between and amongst both groups. In the following extract, Mike discusses a photograph of the communal lounge area on the ward and reflects on possible links between the design and ambience of the physical environment and the social interactions he observes and experiences in that space. Alongside his thoughts about how patients might make sense of their experiences there, Mike also talks about his own perceptions of the lounge area and experiences of spending time in the nursing office that is directly adjacent to it.

Mike: Extract 1

1.	M:	So that's the lounge.
----	----	-----------------------

- 2. **R:** Yeah.
- 3. M: Um, er, I just find it so depressing.
- 4. **R:** Yeah.
- M: (*Laugh*) er, it's and I, the thing is I, I'm just constantly, when I'm stuck here, I
 was gonna do one in the office, but I didn't get the chance to do it on the, the
 office, you know it, it's like a goldfish bowl, sitting in there as well, it's very
- 8. depressing, but just looking out on that, it's never really used a great deal.
- 9. **R:** Mm.
- 10. M: Um, it's just got furniture dotted around the outside, er, the TV stuck up there
- 11. and it's just, I think it's just so open and um, it's just nothing about it that gives
- 12. it any sense of, um, you know this is supposed to be their lounge, community,
- 13. this is where they're supposed to maybe interact and instead everyone

¹² It was not possible for Mike's images to be printed or digitally transferred and his photographs were reviewed on the camera screen during the interview.

- 14. congregates down the corridor where they are not supposed to be, where staff
 15. are supposed to be observing and this is supposed to be a place where they're,
 16. they can socialise and actually enjoy each other's company and you know you
- 17. got a few chairs there and there and it's just, doesn't work.
- 18. **R**: Mm.
- 19. M: I don't know if it's just too big, I don't know if we've got any clear plan on
- 20. how to improve it and make it a better environment, you know, we get table
- 21. tennis out here every now and then for the guys which is really good, but it
- 22. just, for me, just is, doesn't work.

In juxtaposition to wider societal understandings of modern domestic living rooms as typically comfortable and potentially psychologically comforting spaces, Mike's candid opening account of the lounge as "so depressing" evokes the language of mental distress to position the environment as dispiriting and actively lowering mood. This imagery is further reinforced by the repetition of "depressing" within Mike's description of spending time in the glazed nursing office that directly overlooks the lounge. Whilst Mike's stories generally focus on his perceptions of how patients might experience the ward, the brief reference to the nursing office as a "goldfish bowl" alludes to his own personal experiences and a sense of exposure when working in there. Later in the interview, Mike also elaborates on the unease and discomfort he perceives in both environments when considering the relationship between the two spaces in the following reflections.

"Um, I just, I just, I don't see it [*the lounge*] as being a sort of um, an area where I see much, much positivity, um, going on, um, you know, when, when you, normally in there you're in the office, you're normally ge-, the door's being banged on, um and I always, I seem to associate it really of just, where there's, um, where you're constantly being pressured for time. I'm on the computer, I'm looking out and it just, um, yeah, people, the only time people are in there is when they are waiting to go for their dinner, or um, waiting, or banging on the door for some, some, either medication or for, to go out for a session or, normally in a state of arousal, not in a pleasant, you know relaxed and sort of social mood." Patient activity in the day area is presented here as being typically transient. Mike extends his earlier dispirited account of the lounge by positioning it as a space without, "much positivity" that patients would generally only choose to occupy whilst speaking with staff or waiting for other events to take place. Mike observes that patients in the lounge area are not typically socially engaged there and focus instead on staff activity in the nursing station and on attempts to attract staff attention. Mike's report of the office door being, "banged on" contributes to images of tension and undertones of aggression being expressed within the environment. This imagery supports his later observation that patients encountered there are regularly in a state of agitation and not typically in a, "relaxed and sort of social mood". Mike's narrative thus positions the door and glazing as an intensely affective threshold between the two adjacent, yet very different worlds and contrasting circles of activity that comprise the lounge and nursing office. Patients may typically express frustration or distress along one side of this division, whilst staff on the other side may feel exposed, harassed and "pressured for time". Mike's report of patients, "banging on the door" also makes an indirect reference to the incidence of loud noise within the ward environment that may heighten the affective atmosphere and disturb, distress or annoy patients and staff (S. D. Brown et al., 2019b; Hsu et al., 2012; Mazuch & Stephen, 2007; Ryherd et al., 2012).

Whilst the glazing physically separates the office and lounge, it simultaneously provides a visual connection between the two spaces that highlights the differences in their intended functions and environmental characteristics. When observed through the glazing, each area can therefore be seen as a backdrop to and visual extension of the other space. An incongruous visual adjacency is hence created between the lounge, which is provided ostensibly as a leisure space with a TV and armchairs and the office, comprising an active workplace with desks, phones and computer equipment. Furthermore, since patients are not allowed inside the nursing office, the narrative highlights how the glazing not only presents a direct view of an office environment

from a residential lounge, but also offers up a space to patients that they cannot access. Whilst staff can be seen speaking together inside glazed nursing offices, previous research findings indicate that patients' inability to hear such conversations due to the glass barriers can lead to speculation about what is being said and concern that they are being talked about (Simonsen & Duff, 2021).

An earlier study by Edwards and Hults (1970) examined patient and staff perceptions of the ward milieu before and after the removal of glazing around the nursing station in a psychiatric ward. Without the glazed window, fewer patients perceived that they were disturbing staff when visiting the nursing station and fewer patients were concerned about seeing staff laughing or talking there. After the glazing was removed, patients spent less time visiting the nursing station and staff spent more time interacting with patients outside it. Patients also expressed their view that removing the glazing had in turn removed perceptions that staff needed physical protection from patients, which had been associated with the presence of the robust glass window. Mike's account of looking out to monitor the lounge whilst working on the computer in the office, highlights how via the glazing he is connected simultaneously to two separate worlds and attempting to interact with both.

Mike's earlier observation that the lounge is "never really used a great deal", implies that patients choose not to spend time there, whereas by contrast, a lack of choice and sense of confinement is implied in his own description of being, "stuck here" within the office. Mike's narrative can therefore be seen to echo the findings of Shattell, Andes and Thomas (2008) who examined perceptions of the environment in acute psychiatric care and reported that staff felt confined and "caged-in" (p.242) by the enclosed nursing station and desired greater interaction with patients. In the same acute psychiatric environment, Southard et al. (2012) and Shattell et al. (2015) explored staff and patient experiences of the nursing station before and after removal of the existing glazed enclosure. Although Southard and colleagues (2012) found no

statistically significant difference in staff and patient experiences of the therapeutic milieu before and after the adjustments, their perceptions of this did not worsen without the glazing. There was also no increase of aggression towards staff by patients associated with the open station, as had been predicted by some staff members. Additionally, there was a reported decrease of 26% in the use of restraint or seclusion in the year following the removal of the enclosure. A qualitative study undertaken in the same context by Shattell and colleagues (2015) found that patients unanimously preferred the open station. Patients described experiencing a sense of freedom and togetherness, alongside increased perceptions of safety, including the sense that staff were able to respond to emergencies more rapidly. Both staff and patients viewed the glazed enclosure as being a barrier to staff-patient interaction. Whilst staff perceived that the enclosed station had contributed to patient frustration, they reported that the open station instead assisted with patient de-escalation.

Mike's overall evaluation of the communal lounge area is unenthusiastic and underpinned with a tone of disapproval. In the first extract, his depiction of armchairs being, "dotted around" and the TV being, "stuck up there" imply a sense of randomness that contributes to his critical appraisal of the space planning and furniture placement. Mike's depiction of the lounge illustrates that, in line with typical practice to mitigate risk on secure wards, the television is housed in a protective casing and mounted high up on the wall. By implication, Mike's narrative suggests that the high-level TV placement is awkward and unlike conventional ergonomic relationships between the television and seating in an average domestic living room. Mike's view of the lounge as being "just so open" is also presented in contrast to wider sociocultural understandings of living areas as being typically enclosed or semi-enclosed rooms, with a domestic scale and a convivially spaced arrangement of furniture.

Whilst living rooms in other residential contexts might typically be viewed as private spaces, his depiction of the lounge that is overlooked by and directly connected

to several other spaces on the ward, positions it more as a public thoroughfare than a private dwelling space. Mike's disappointment with the lounge and perception of how its openness and layout might influence behaviour are further expressed through his unequivocal and exasperated assertion that there's, "just nothing about it that gives it any sense" of being a social space. The day area is accordingly presented as lacking the necessary character and material qualities required to produce an inviting or social environment as stereotypically characterised by a residential living room. Whilst research findings recommend the provision of ample space in communal areas to enable patients to regulate social interaction (Ulrich et al., 2018), Mike's remarks also imply that the degree of congruence between the scale and function of an environment can affect occupants' perceptions of comfort. His account of the large space therefore evokes a sense of discordance between its domestic intentions as a lounge for relaxing in and its over-scaled proportions that can be understood to be more reflective of a public foyer or reception space for transient occupants.

The narrative indicates that the social relationships and interactions which may contribute to creating a sense of "community" on the ward are not readily enabled in the lounge and may be actively hindered by its spatial design and atmospheric attributes. Mike's observations about patient activity in the day area are consequently pertinent to existing research examining relationships between the physical design of psychiatric wards and levels of social interaction. Holahan and Saegert's (1973) investigation into relationships between the ward environment and patient behaviour compared an extensively remodelled psychiatric admissions ward with an identical ward that was left unchanged. Compared to the existing ward with drab finishes and worn furniture, patients socialised significantly more in the refurbished ward, where adjustments included fresh paint colours and the addition of comfortable new furniture to the day areas and bedrooms. There was no difference found in levels of non-social active behaviour between the wards, however, there was a significant decrease in

isolated passive behaviour observed within the remodelled ward. Group interaction was also encouraged by arranging new chairs and tables into small social groupings in one of the day areas.

In one of the earliest studies examining associations between furniture layouts and social behaviour in psychiatric settings, Sommer and Ross (1958) observed social interaction amongst patients before and after adjusting the day room furniture on a geriatric ward. Their study highlights how typical 'shoulder to shoulder' seating arrangements in day areas, that are similar to those observed here by Mike, can make conversation both physically and psychologically uncomfortable. Accordingly, the authors also warn that such layouts can lead to patients becoming, "observers" and "silent individuals sitting eternally in a waiting-room for a train that never comes" (p.128). By contrast, social interactions almost doubled when the furniture layout was changed and chairs were arranged around tables to create 'sociopetal' groupings (i.e. those which promote social interaction, as opposed to 'sociofugal' arrangements that minimise social contact). Several subsequent studies have similarly found that arranging comfortable furniture in small and flexible groupings facilitates social interaction in psychiatric settings (Jovanović et al., 2019). The use of moveable furniture also comprises one of the ten key design interventions identified in Roger Ulrich and colleagues' conceptual model for promoting reduced aggression in psychiatric facilities by reducing stress within the environment (Lundin, 2021; Ulrich et al., 2018). Within this model it is argued that using moveable furniture in communal areas increases patients' ability to regulate personal space and thus reduces stress levels, which in turn can contribute to reductions in levels of aggression.

Within the first extract, Mike's repetition of, "supposed to" in relation to patient and staff activity in the day area and bedroom corridor emphasises how the expected behaviour of both groups and their use of space is determined by ward policies and protocols. His account of patients challenging the 'rules' by gathering in the corridor

area where they, "are not supposed to be", thereby alludes to a sense of patients' collective resistance to authority through the occupation of space. Mike's repeated assertion that it, "just doesn't work" emphasises his exasperation with the space and awareness that adjustments are required to increase levels of comfort or sociability in the lounge area. This statement also underscores Mike's belief that behaviour observed there is closely linked to the scale and layout of the lounge and its relationship to adjacent spaces on the ward, including the nursing office and bedroom corridor. By highlighting how sporadic standing-based activities such as table tennis have successfully provided a social focus to the space, Mike's narrative reinforces his perception that the sparse and inflexible layout of heavy furniture in the lounge contributes to its typically non-social character. Through his description of the social activity associated with the table tennis equipment, this part of Mike's narrative can also be seen to be bringing back the vitality and sense of purpose into the space in a way that is missing from the other parts of his account.

In the following extract, when discussing a photograph of the bedroom corridor, Mike reflects further about potential associations between the social interactions that he observes within both the corridor and lounge and the physical or atmospheric characteristics of these spaces.

Mike: Extract 2

1.	M:	So, er, bedroom corridor. Obviously, there're no patients in there, but that for
2.		me it, er, tends to be a room, a, an area where there's always conflict, always
3.		trouble, always pushing of boundaries and the environment, er, I just, I you
4.		look at it, the corridor and you always know that there's, you know, patients
5.		shouldn't be there, staff should be there observing, but, um, they're always
6.		obst-, obstructing the, you know, very regularly obstructing the corridor and so
7.		you can't see what's going on down there and you know, the lounge is where
8.		they should be but they're down there in that corridor, yeah, whenever I sort of
9.		see it you think, think sort of ah, yeah.
10.	R:	Mm. And why do you think, er, they tend to spend more time in the corridor?
11.	M:	It's strange, because, um, we have a thing called a man cave, er, doo, doo, doo,

12.	doo (looking at image), so this is, it would be just here (pointing on image).
13. R:	Yeah.
14. M :	And that's where there's like a little games room, and then so you have your
15.	staff who are supposed to be observing here and at the bottom and there's
16.	they're just there to observe, they're not supposed to really be socialising
17.	there, pa-, patients aren't supposed to be congregating there, supposed to be
18.	either in the lounge, games room or bedroom. Just bec- and um, because
19.	there's staff there patients tend to then automatically go there so they can
20.	interact and sort of talk to the um, staff and once one patient's there another
21.	one comes along, another one and then you know you've got staff who are
22.	supposed to be doing the observing and er, you've got 4 or 5 patients standing
23.	around and um, so I guess it's maybe that there's staff always there.
24. R :	Mm.
25. M :	But then they, it's just that maybe they feel comfortable and that there's a few
26.	few patients round there getting the chance to all have a bit of a social chat.
27. R :	Mm.
28. M :	Whereas again, you know, it's, it's more confined isn't it? You're actually able
29.	to talk near someone (<i>laugh</i>) and go to the lounge, it's, it's a big open space, it
30.	may be that if we had it set up as a, as an actual, you know, differently
31. R :	Mm.
32. M :	we might, it <u>might</u> help move those patients there, but if we've got a large room
33.	where there's just chairs are dotted around the outside wall, who'd want to go
34.	there?
35. R:	Mm. Mm.
36. M :	So maybe that's, maybe that's the way that we really need to, if we focus on
37.	improving the lounge, that will then stop the problems down the corridor.

Mike's earlier allusion to patients expressing a form of resistance to authority by gathering in the bedroom corridor is made more explicit here through his depiction of, "conflict", "trouble" and "pushing of boundaries" typically occurring in this area. The repetition of, "always" within Mike's observations of such activity also conveys his perception of friction being a constant and inevitable feature of the corridor area. His comments suggesting that patients, "shouldn't be there" and that staff are, "not supposed to really be socialising there" reinforce his earlier references to the expected behaviour of both groups being subject to the ward 'rules' and protocols. Mike's

spaces, however, further emphasises his observation that opposite behaviours are generally taking place. By suggesting that staff, "should be there observing" and not be engaging with patients, Mike highlights his perception of tensions between the institutional expectations of social behaviour in that space and the reality of what might typically occur there. The narrative therefore draws attention to the ironic circumstances of patients not spending time socially in the lounge despite being "supposed to" and yet gathering in the corridor where they are not "supposed" to be. Mike's observation that patients move "automatically" to interact with staff in the corridor creates an image of an instinctive pull that draws patients towards the members of staff in there. It also highlights how, in contrast to the lounge area where staff may often be separated from patients by the nursing office glazing, there is no physical barrier between staff and patients in the corridor.

Although the narrative presents the corridor as being imbued with tension and conflict, it is simultaneously described as a sociable place where Mike speculates that patients might "feel comfortable" and wish to spend time in the company of other patients and staff. Staff are continually present in the corridor and hence might be perceived as being more available there than in the lounge, where patients might typically need to attract the attention of staff working inside the office. Accordingly, Mike's narrative account of patients being drawn to staff also suggests that patients may desire attention from and interaction with staff, more so than with each other. Perceptions of having a closer connection or more equal relationship with staff can therefore be seen to be facilitated in spaces where staff must leave the office and be available for contact with patients. Associated with this, a sense of safety might also be perceived by patients in the corridor due to the constant presence of staff at close hand in the event of an emergency.

In contrast to the large, open lounge area, the narrative suggests that the corridor encourages social activity by bringing people closer together in a contained

environment and that a sense of comfort might be gained from the proximity of other patients and staff. Mike's report of, "4 or 5 patients standing around" in the corridor presents an image of informality and spontaneous activity being appreciated by patients in an environment typically characterised by rules and routine. Standing casually in the corridor and "getting the chance to all have a bit of a social chat" is thus juxtaposed with patients choosing not to sit in what Mike perceives to be a rigid and unsociable arrangement of furniture in the lounge. Mike's description of the armchairs, "dotted around the outside wall" echoes the disapproving tone of his earlier reflections about the sparse and awkward furniture layout. His sense of frustration with the social affordances of the day area is captured in his rhetorical question to the researcher asking, "who'd want to go there?". Mike's thinking aloud about how setting up the space, "differently" might encourage greater use of the lounge alludes to its nondomestic attributes and an aspiration to create a more homely and inviting living space. His concluding comments in this extract hence underscore his perceptions of a connection between social interaction and the physical and atmospheric characteristics of spaces on the ward. Mike's suggestion that changes to the lounge might help resolve issues in the corridor proposes that the behaviours observed in both spaces are intrinsically linked and acknowledges a need for spatial adjustments to the ward to potentially influence social behaviour.

6.2 Nathan

Nathan is currently based on a low secure psychiatric ward and has been working as a health care assistant at the hospital for several years. Throughout this time Nathan has worked with patients in different settings, including medium secure and dementia friendly wards. In recent months Nathan's ward has moved to its current location from another building within the hospital. Nathan highlights differences in the design and layout of these two environments and his perceptions of how spatial characteristics of

the current ward, such as its integrated living and dining area can influence social behaviour and relationships amongst patients and staff.

Nathan's overall narrative conveys a sense of caution in relation to his work and an underlying tone of worry associated with his perceptions of risk and the harm that could potentially occur to patients or staff members in spaces on the ward. When reflecting about his emotional responses to everyday spaces, Nathan frequently highlights his perceptions of how aspects of the physical environment, or objects within it, such as cutlery or pool balls, could present risk to patients or staff members. Nathan's accounts of distressing events such as violent assaults from patients that he has experienced on previous wards, also highlight how past experiences can be active in the present to shape his perceptions of spaces and how these environments are experienced day to day.

Throughout the narrative Nathan refers to the professional relationships established between colleagues on the ward and he highlights the reliance upon close teamwork and diligence amongst staff members to make sure the environment remains as safe as possible. Although Nathan's overall tone is positive, he expresses frustration about staffing levels and how the frequent secondment of staff members to assist on other wards depletes staff resources on his own ward. As he works in a low secure setting where patients are typically at the point of moving on to communitybased life beyond, Nathan expresses particular concern about the reduced availability of staff to support patients with regular daily activities outside the hospital.

Nathan's storylines are focused around: (1) being cautious and having concern about risk within the ward environment, (2) having a sense of responsibility for the well-being of others, (3) valuing positive relationships and teamwork with patients and staff colleagues, (4) how material and spatial attributes of the environment can affect social activity, (5) reflections on improving the ward design and aesthetics and (6) how staffing levels can influence everyday experiences for both patients and staff.

During the interview Nathan discussed photographs of: (a) ward clinic room, (b) small side room (also referred to as interview room or quiet room), (c) dining area half of the day room, (d) bedroom corridor, (e) nursing office (main area), (f) nursing office (back-office area).

The analysis will draw on the first five of these storylines and focus on Nathan's observations about links between the everyday spatial experiences of patients and staff on the ward and relationships between both groups. In the following extract, Nathan's reflections on his photograph of the communal lounge and dining area suggest that the physical and social milieu in this space can influence interpersonal relationships between patients and staff.

Nathan: Extract 1

1.	N:	Um and then obviously we've got, we've got the dining room side. Which is
2.		quite a social sort of area again. Um, where the guys just sit and chat
3.	R:	Mm.
4.	N:	and have a good laugh again.
5.	R:	Yeah and as a member of staff working in that, in that sort of space, um, how,
6.		how do you, how do you tend to feel when you're in this, in this area, um?
7.	N:	That's probably the room that we most, it's a difficult room 'cause you're really
8.		relaxed in there. Because it is that sort of social environment where people will
9.		sit and laugh and have a good joke, but it's also the area where probably like 70
10.		percent, 80 percent of attacks on the staff happen, is within the communal area
11.		like a lounge or something like that.
12.	R:	Mm.
13.	N:	Um, previously, on a different ward it was, I was attacked in the lounge area,
14.		um by a patient so, you know, but to me, like, it's a lot more of a positive area,
15.		because I see a lot more sort of good happening in that, um, it's just like that
16.		one, maybe two instances before, that sort of drags it down a little bit.
17.	R:	Mm. Mm. Yeah, yeah. And, um, do you have any sort of thoughts about, about
18.		the overall space at all and the, the sort of feeling of the um, of the
19.		environment?
20.	N:	Um, I think space-wise, like it's, it's OK, I mean we've, we've made changes and
21.		and things since we've moved here, with like the telly and things like that,
22.		lowering it so it's actually easier to watch, um, the pool table, the ping pong

23. table, they're all changes that we've made, 'cause when we first moved in it was 24. very clinical I mean, as you can see it's like white walls, it's like white carpet 25. and and red carpet and that's about it and they try to make it look a little bit 26. less clinical, there's posters, there's picture, big picture on the wall that 27. probably costs a lot of money that actually, men don't want to sit and look at 28. fields with lavender flowers and things like that, so I mean, to us it doesn't 29. necessarily feel like it's massively appropriate for our kind of people that we 30. work with, but we try and make it sort of (yawns) excuse me, fit to the best that 31. we can. 32. R: Mm. Mm. 33. N: Um, but I think regardless, it's still, a lot of the guys still see it as like a fairly 34. nice room to be in and it's like bright and (*slight pause*) cheerful. 35. R: Mm, mm. What sort of, could you give any examples of um, some of things 36. you've, you've sort of you've done to make it more appropriate or, or, or fit, 37. um? 38. N: Yeah, so the, the TV used to be up on the wall up here. 39. R: Mm. 40. N: Um and when you're sitting round, the chairs are quite low again because 41. they're all like, um, heavy duty furniture, so the chair's quite low, they're 42. difficult to move to move around and people weren't wanting to spend the time 43. in, in the lounge area, um, because you're talking of looking up at a TV that's 12, 44. 13 foot off the, off the floor, to stop people from being able to like yank it down 45. and things like that and so it 46. **R**: Mm. 47. N: it was a very anti-social space and we found that people were only coming 48. coming together when it was dinner time and, and lunch time and things like 49. that to sit at the dining tables, um, so we've got a new TV, we had to get all of 50. like the wiring put in for it, but we got that and that's lowered and now we're 51. actually finding that through most of the days and the evenings, like when 52. we're here, we're putting films on and we're getting sort of like 6, 7 people 53. watching films, I know like 6, 7 people doesn't sound like a lot but it's like, 70 54. percent of our ward will come out and sit and watch films or, or funny video 55. clips and things like that.

Nathan's reflections on this space, which comprises a communal lounge with open plan dining area, present an image of frequent social interaction amongst patients and between staff and patients in this space. His descriptions of patients having a "good laugh" and a "good joke" together evoke a sense of informality and a typically amiable and cheerful atmosphere. Nathan highlights his perception of paradoxical qualities within this space, however, by depicting it as both an enjoyable social environment where staff and patients can feel, "really relaxed", but also a place where staff must remain especially vigilant towards possible violence. The degree of potential risk within this environment is further emphasised to the researcher through Nathan's personal story of being assaulted in the day area on a previous ward. Recollections of difficult past experiences in similar settings are therefore understood to be present in his everyday experiences of spending time within the current lounge space.

Figure 11 Nathan's photograph of the communal lounge and dining area



Nathan talks about adjustments made to the ward that have aimed to address what he perceives to be the bland and "clinical" qualities of the design, including stark, white interior finishes. Alongside the addition of artwork to brighten up the space, attempts to soften the environment include the introduction of table tennis and pool tables to create a social focus of activity within the lounge. Elsewhere, Nathan also mentions how staff members who would, "historically sit in the office" now spend more time in

the lounge engaging with patients following the introduction of these communal games. The sociability afforded by the presence of the table tennis and pool tables is therefore understood to be promoting increased social interaction and influencing both patient and staff behaviour in this environment. As both these objects have a sizeable footprint and occupy three-dimensional space within the large volume of the day area, they can also be seen to be contributing to increased perceptions of closeness and a sense of conviviality amongst the people inhabiting that space. Furthermore, the addition of table tennis and pool tables has introduced forms of standing-based social activity to the lounge space, which create an alternative social dynamic when compared with more static seated interactions. Therefore, the fluidity of standing social encounters afforded in that space could also be argued to be encouraging the increased levels of informal and sociable interaction between patients and staff that Nathan observes there.

A largescale photographic artwork forms part of the adjustments made to brighten the day area, however, Nathan questions the suitability of the chosen imagery for the young male patient demographic on the ward. His comment suggesting that, "men don't want to sit and look at fields with lavender flowers and things like that" points to a disconnect between those specifying the artwork and their understandings of the individual patient groups who might inhabit that ward day to day. Elsewhere, Nathan also implies a sense of limited choice when referring to the ward artwork having been selected, "from a very bland catalogue of like generic, autumnal forestry scenes and things like that". Nathan's photograph of the space indicates that the image he refers to is applied to the wall of the day area as a large-scale mural. His speculation that this artwork, "probably costs a lot of money" also alludes to his impression of the mural as a well-intentioned, but not fully optimised investment for that space. Despite his assessment that the imagery is not well matched to the patient group, however,

Nathan observes that the overall day area, where some adjustments have been made to the environment, is generally perceived by patients as a "bright" and "cheerful" space.

Nathan's reflections on the nature-based content of the mural artwork can also be considered in reference to existing research exploring how different styles of art are experienced in healthcare settings (Nanda et al., 2008, 2011; Ulrich, 1991). A study by Nanda et al. (2011) undertaken in a psychiatric context examined patients' responses to different artworks displayed rotationally on the lounge walls within an assessment ward. A significant reduction in pro re nata (PRN) medicinal treatment for anxiety and agitation (i.e. medication dispensed as and when needed) was observed when artwork containing naturalistic landscape imagery was displayed, in comparison to an abstract image, or the control condition with no artwork. Earlier studies have also reported positive responses from patients to artwork containing natural images and negative reactions to surreal, abstract or ambiguous art, in both general hospital settings (Nanda et al., 2008) and psychiatric healthcare environments (Ulrich, 1991). Although research examining occupants' responses to artwork in mental healthcare settings is limited, the existing literature points towards potential psychological benefits of artwork containing nature-based imagery. Nathan's observations highlight, however, that artwork proposals should ideally be site-specific and be fully considered in relation to the intended environmental context, such that the content and imagery is tailored appropriately for the users of that space.

In a surprising contrast to its intended function as a communal lounge, where social interaction would typically be expected to take place, Nathan regards the day area before the adjustments as a, "very anti-social space". Based on his own observations, Nathan notes that patients did not typically spend time in the lounge previously and implies that the awkward relationship between the low seating and television positioned at high level was a contributing factor to this. By highlighting how the television has since been lowered to a more typical domestic height, Nathan argues

that this intervention has significantly influenced behaviour, such that patients now enjoy spending time there together, including watching films with other patients and staff. The adjustments made to the day area can accordingly be understood to have increased levels of perceived physical comfort in the space and in turn improved patients' perceptions of psychological comfort there. In the following quote, Nathan provides further elaboration on his thoughts about how patients might experience time spent sitting in that space and watching the television as a communal activity.

"[...] Um, we watch like funny clips and stuff like that 'cause we, we tend to find that, um, like funny [*name of channel*] videos and things actually increases the morale of everyone, because everyone sits around and laughs and jokes. I think sometimes, think like the patients, a lot of the guys sort of think, you know like, I'm, I'm here, but if actually I've not just fallen face first into a muddy puddle and they have a good laugh and joke about it and I think it can put, sort of things into perspective a little bit, but we watch a lot of different types of films with them."

Nathan's account suggests that the collective experience of enjoying films and funny videos in this space can lift spirits and encourage a shift in patients' sometimes downcast thinking about their individual circumstances, towards a more hopeful state of mind. The shared social interactions in this space, including film watching and playing games together can consequently be interpreted as communal forms of 'liminal affective technology', that are drawn upon to enable transition between affective states (Stenner, 2021; Stenner & Zittoun, 2020). These processes of affective transformation can also be argued to be helping patients cope with the 'spontaneous', or unforeseen liminal experience of 'crisis' brought about by psychological distress and involuntary detention in hospital (Stenner, 2017). This liminal experience is enduring, however, and patients' experience of being detained in hospital can thus be interpreted as a 'liminal hotspot' (Stenner et al., 2017), in which they are 'suspended' indefinitely in an indeterminate position between their past and future lives. Furthermore, as Turner (1969/1996) argued, the shared 'interstructural' experience of a liminal situation can

be socially equalising for those going through a transitional experience together. Consequently, collective liminal experiences may also give rise to a sense of *communitas*, or an intense feeling of fellowship amongst participants.

A sense of *communitas* being experienced in the lounge is evoked through Nathan's narrative and additionally, he implies that a sense of solidarity is heightened by the social activities, such as playing pool with staff, which enhance participants' collective perceptions of equality. It could hence be argued that the experience of group social activity amongst patients (and between patients and staff) is contributing to increased perceptions of community spirit and cohesion within that space. Nathan's story also implies that having shared perceptions of being in a better position than unfortunate people experiencing comic mishaps in film clips can increase patients' sense of togetherness and prompt alternative perspectives towards their own circumstances. Moreover, social interaction and the joy of shared humour can also be argued to be diminishing staff-patient status distinctions and hence enabling a greater sense of parity, which in turn might enhance relationships between both groups.

In the following short extract Nathan later reflects further on how aspects of the current ward design might contribute to improvements in the quality of relationships that he observes and experiences there, when contrasted with staff-patient interaction in the previous ward environment.

Nathan: Extract 2

- 1. **R**: Sure. Um, just thinking about your relationships with other people, whether
- 2. you feel there are any ways in which, um, they are affected by, by the
- 3. environment that you work in?
- N: Um (*slight pause*), no I think now that we're in the building that we are, I think
 the environment and the relationships
- 6. **R**: (*Coughs*). Excuse me.
- 7. **N:** with like the other patients in this environment are quite a lot better, mainly
- 8. because obviously like a lot of boundaries have gone like, just stupid things like
- 9. the dining room is now more like a lounge dining room, there's a lot of like our
- 10. boundaries that might cause issues with our relationship with patients, like,

- 11. 'no you can't leave the dining room because there's still cutlery', has gone, like
- 12. naturally through the fact that there's no walls or doors
- 13. **R**: Mm.
- 14. N: um and a lot of the ward now is, it's a lot more like less restrictive, that we can
- 15. take people into what would used to have been like a staff only area, like the
- 16. servery to go and wash up and just through things like that I think the
- 17. environment, the new environment we're in is helping relationships quite a lot
- 18. more between us and patients and us and other members of staff.

When compared with the previous ward, Nathan suggests that the reduced number of internal partitions and doors between spaces inside the current ward environment has had a positive influence on the interpersonal relationships experienced and observed there. The narrative presents such physical divisions within the ward as symbolic barriers that contribute to tensions experienced between staff and patients in a secure setting. Nathan's story of patients not being permitted to leave the dining room in the previous ward until all the cutlery had been counted in presents an image of routine, rules and an institutional environment imbued with perceived risk. Power imbalances between staff and patients are therefore understood to be emphasised at such threshold points between different parts of the ward, where staff are required to enforce rules and control patients' access to space. In contrast, however, the narrative implies that the open plan dining and lounge area in the current ward layout affords a smoother flow between spaces and accordingly reduces friction between staff and patients.

Elsewhere, Nathan notes that the use and return of cutlery by patients is still closely monitored within the open plan dining area on the current ward, however, his account suggests that the absence of a division between the lounge and dining spaces affords decreased perceptions of restriction for patients. The sense of increased autonomy associated with this may in turn contribute to reduced environmental tension and improved staff-patient relationships. Furthermore, the reduced quantity of physical barriers within the current ward can be seen to be helping diminish the

disparity in staff-patient status that is emphasised by the presence of spaces that are visible to patients but only accessible to staff. By contrast, Nathan's report of some patients clearing up in the servery with staff at mealtimes creates an image of trust and teamwork within a shared environment. Collaborative activity between patients and staff within communal spaces on the ward could thus be interpreted as being helpful to fostering a more collegiate and equal atmosphere between both groups.

In contrast to his experiences in the lounge, Nathan reflects in the following extract on his emotional responses to working in a side room on the ward where the distinctions in staff-patient status and power relations are overtly expressed. The room is used for multiple purposes and Nathan talks specifically about his experience of its use as a space where staff must "search" patients for security reasons when they reenter the ward after spending time outside the building.

Nathan: Extract 3

1. N :	And for me again, it's probably, it's, it's the room that we bring our patients in
2.	when they first come, um, back on the ward after they've been off the wards,
3.	so, um, in there like your risks are around a patient's brought a weapon in or
4.	something that they shouldn't have brought in and that's the room that you'd
5.	find it, um, so obviously there's a lot of anxiety around like and when I search
6.	there's something on them that could hurt me or the other people that I'm here
7.	to, to look after, both staff and patients.
8. R:	Mm.
9. N:	Um, so that's probably where a lot of the anxiety comes from in that area, I
10.	mean it's used for a lot of positive sort of things as well, there's guys that make
11.	make models, like have sessions on model making and things in there and
12.	music sessions, so it's not just sort like a, a search room as such, it's used for
13.	sort of general purposes, but I think the, the searching of people in that room is
14.	what carries a (<i>slight pause</i>), like a bit of anxiety around what are they doing,
15.	what are they bringing in? Are they trying to hide something that they don't
16.	want us to find if I pat them down and I've touched their pocket, they've got
17.	something in there, how are they going to react?
18. R:	Mm.
19. N :	Um and it's just all sort of things like that.
20. R :	Mm. Um, so typically, um, any patient who's been out of the ward during the

- 21. day, um, would, would come in here with staff
- 22. N: Yeah.

23. **R:** before entering the um, the main

24. N: Yeah.

25. R: area? Right, yeah. Yeah and do you think there's anything about the sort of, about that space, um, or um, you know, the, the, the sort of environmental conditions that um, affect your feelings when you're in there or, um?
28. N: Um, no I don't think there's anything like, specific, um, I think a lot of it is just around, um, again it's like fairly, it's not isolated area on the ward, like there's a lot of windows out on to the main corridor, so you can sort be seen by others, but it's off the main ward area, um (*slight pause*), like it's not, um, it's not like,

- 32. pa-, people don't pass so frequently through like the course of 5, 10 minutes
- 33. unlike the lounge or the bedroom corridor and things like that, so it is fairly34. sort of isolated.
- 35. **R**: Mm.
- 36. N: I think that with then with then what you're in that room trying to do, to search37. for things, just sort of heightens the anxiety around it a little bit.
- 38. R: Mm, mm. Sure. Um and in terms of the more positive, um, experiences, that,39. that take place in there, um, do you have any sort of reflections on um, how,
- 40. that space might be, um, perceived by people using, using it?
- 41. N: Yeah, I mean I think see it, obviously like, large, the vast majority of our42. patients have never sort of brought anything in or attempted to bring anything
- 43. in that they shouldn't have, one or two like you would expect one or two have44. tried, um, so I think, sort of, it is more of a positive room, er, like we go to the
- 45. kitchen or something and there's a music session going on and we can hear
- 46. them like singing and playing guitar and things like that, so it is quite a like47. relaxing room at times.
- 48. **R:** Mm.

49. N :	Um and they make like models and paint and, and do things in there as well,
50.	sort of, um, with OT staff, or if they wanna go on the, the laptop, a ward laptop,
51.	then they go in there so it's, it does have a lot of positive uses, but I think the,
52.	like the sort of, not negative, but that one sort of use of the room, um, sort of
53.	outweighs it for me.

The side room described here sits within the secure demise of the overall ward and is located off the entrance corridor that is passed through to access the main ward area. Nathan's repeated account of experiencing, "anxiety" as an emotional response to this room underscores a heightened awareness of safety and the risks associated with his role when working in this space. Nathan's concern is related not only to his own safety, however, as his description of, "other people that I'm here to, to look after" also conveys his sense of personal responsibility for ensuring the well-being of all his colleagues and the other patients on the ward. The narrative hence alludes to a link between Nathan's anxious state and his concern to be sure that 'contraband' items are intercepted, that is in turn associated with his sense of responsibility for the potentially serious consequences of something being missed. This image of possible danger and threat in the environment is underscored by Nathan's apprehension that a "weapon" might be brought onto the ward and his related concerns about potential physical harm to himself or others.

Figure 12

Nathan's photograph of the side room



The narrative accordingly conveys Nathan's typical state of unease in relation to searching patients and creates an image of him entering into the unknown within this room. A sense of unpredictability is further reinforced through his rhetorical questioning about patients' behaviour, including his concerns about, "what are they doing, what are they bringing in?" and if challenged, "[...] how are they going to react?". These questions build up an impression of Nathan's disquiet about this aspect of his work and highlight the mistrust towards patients that is implicit in the process of searching by staff. In contrast, the physical intimacy that is implied through Nathan's account of having to, "pat them down" during the search process would require patients to put their trust in staff. This may be especially difficult, however, for patients in secure settings who have commonly experienced distressing life events, including violent discord between parental figures and physical or sexual abuse (Coid, 1992).

Nathan's wariness towards patients when searching thus highlights how a balance of trust between patients and staff can be disrupted by the activity in this space. Nathan also mentions that he is especially mindful when working with patients in this room, as he perceives it to be more isolated than other areas of the ward. Some of his anxiety about working in this space may therefore also be associated with the implied consequence that colleagues might take longer to respond in the event of an emergency there. Furthermore, Nathan's account suggests that the specific nature of the searching activity taking place and its unpredictable outcomes, including the potential to encounter "weapons", can intensify his concerns about safety and sense of isolation from the rest of the ward when working there. As illustrated by the following alone with patients in other spaces on the ward, including the clinic room, which contains a number of objects that he perceives could potentially be used as weapons.

"[...] obviously there are a lot of like dangerous and risk items in there, like needles, glass bottles and obviously, (*slight pause*) drugs, like cabinets and, and medication trollies with drugs in um and particularly, one of the guys that we're having to observe to take his medication in there is quite risky at the moment."

Nathan's role also includes responsibilities for monitoring patients' physical health, which take place here and he describes how he may also spend time in the clinic with patients who have self-harmed and therefore may be at a "crisis point in their life". The small scale of the environment where Nathan perceives that, "you could probably only fit like, sort of three staff" to assist with managing highly distressed patients if required, plus its relative isolation are all factors which contribute to his sense of anxiety when working there.

Nathan's story about the side room also highlights how this space is typically used for multiple functions. Therefore, in addition to being a room where staff undertake searches and patients experience being searched, it is also a place where both groups might spend time in therapeutic sessions, including music and craft activities. The multi-purpose functionality of the room accordingly highlights its ambiguity as a space that houses activities with both carceral and therapeutic connotations for patients and staff. Both groups might potentially experience distress or concern associated with the search processes in this room, but it is also a space where patients and staff may be involved in therapeutic sessions. Affective responses to the room may therefore be influenced by both the activity currently taking place within it, or by previous activities experienced there. Nathan captures his own ambivalence towards the space by describing his perception that it can also sometimes be a "relaxing room" and his appreciation of overhearing patients' music sessions taking place there at other times. For Nathan, however, his affective experiences and concerns linked to the security-related function of the room are understood to overshadow his appreciation of its more "positive" and therapeutic connotations. Through this story, Nathan also draws attention to a sense of paradox inherent in the work of staff members in secure settings, whose professional identities can be seen to be suspended within a role with both carceral and caring responsibilities.

When used for security checking purposes, the room represents a transitional zone and a symbolic threshold between the 'outside' world beyond the building and the 'inside' world of the main ward area. The narrative thus implies a sense of this space sitting betwixt and between these different worlds and representing a form of 'noman's land' between the two. For patients, passage through this indeterminate space could also be interpreted as a liminal process of transition in which their perceptions of self-identity might shift from being members of the wider outside community, to reassuming the status of detained patients in a secure setting. Nathan's concern about unknown items being brought onto the ward also presents an image of possible 'contamination' and a sense in which potentially harmful objects from the external world might enter and disrupt the sealed and closely controlled internal environment.

6.3 Bradley

Bradley is a patient in a low secure psychiatric ward and mentions that he has been detained in hospital for several years. He talks about having a need to get out, as he feels that his time to move on from the hospital is now long overdue. Bradley's narrative tone throughout is generally upbeat and he is optimistic about the future. He describes thinking ahead to independent living in the community following discharge from hospital and potentially moving in with his girlfriend.

Bradley describes the enjoyment and sense of freedom he experiences when taking part in activities within spaces off the ward. In particular, he enjoys the sessions that take place in a vocational skills centre located on the hospital site. His narrative implies that spending time in this environment has a different quality to being on the ward. He feels an increased sense of independence when working there and perceives that he is less overlooked by staff in this setting. He also appreciates the various skills gained, including achieving proficiency in wood working, through his regular sessions there. Bradley especially values the staff members at the centre, who he describes as friendly, supportive and accommodating in terms of enabling him to work at a pace to suit his energy levels.

Bradley presents himself throughout the narrative as a down to earth and jovial person who enjoys social interaction and sharing a sense of humour with other patients and staff. He also appreciates the sense of connection and close bond with other patients that he feels is generated by living together in a communal environment.

Bradley's main storylines focus around: (1) how spaces affect social interaction and relationships with staff and other patients, (2) enjoyment of learning skills through activities and responsibilities, (3) how activities create a sense of focus and provide things to look forward to, (4) how his affective responses to environments are associated with the physical attributes of spaces and (5) the importance of having connections to nature and outside world.

Bradley reflected during the interview on photographs of the shrink-wrap machinery he uses during sessions in the vocational skills centre.

The analysis will draw on the first, third and fourth storylines to focus on Bradley's observations about connections between the physical layout and spatial affordances of the ward and the quality of social interaction between patients. In the following extract, following a question from the researcher, Bradley reflects on the overall ward environment and the communal day areas specifically.

Bradley: Extract 1

1. R: [] Um, and I wondered, you, um haven't included any of the, of th	e ward
---	--------

- 2. environment, I wondered was there a reason, um, for that, or do you have
- 3. any, any thoughts about the, space here at all?
- 4. **B:** The ward is really good.
- 5. **R:** Yeah.
- 6. B: It's really (*slight pause*), I don't know how to put it. It's really service user
 7. friendly.
- 8. **R:** Mm.
- 9. B: There's a TV in the lounge, there's a pool table, there's a table tennis table and10. what I like about it is, the fact that dining area is part of the lounge so it's not a

11.		separate room.
12.	R:	Yeah, yeah.
13.	B:	Yeah.
14.	R:	And, do you, how do think that impacts on how people, um, feel in the space,
15.	B:	People
16.	R:	having it open?
17.	B:	interact a lot better. Yeah. They interact.
18.	R:	Interact? Yeah. Yeah.
19.	B:	Because, for example, if you're in the dining area and there's someone in the
20.		lounge area, you can literally like call them over.
21.	R:	Yeah.
22.	B:	Instead of having to go round to their room or go to a separate room.

23. **R**: Yeah.

Whilst Bradley did not choose to take photographs of spaces on the ward, he expresses a general sense of satisfaction with the overall environment. His comments that the ward is "really good" and "really service user friendly" evoke positive imagery of a patient-centred environment and suggest that he perceives the design of spaces on the ward to be meeting patients' needs. To support his account of the ward as a "service user friendly" environment, Bradley lists various pieces of equipment (e.g. the television, pool table and table tennis table) that are on offer in the lounge to afford leisure activity and facilitate social interaction. Through this narrative, he draws on wider cultural understandings of playing games, or watching television with other people as typically sociable and entertaining pursuits. The affordance of positive social activity between patients is therefore understood to be a significant factor in Bradley's appraisal of the communal areas on the ward. He accordingly presents the list of available activities to the researcher as evidence of the ward being well-equipped and able to offer a variety of options for enjoyable pastimes that can promote social interaction.

Bradley's ward has recently moved from another low secure environment and his comments on, "what I like about it", in reference to the layout of the current ward are understood to be made in comparison to his experiences of the previous setting.

Bradley's account of the dining area being part of the patient lounge expresses his appreciation for the lack of division between the two spaces, whereby the social activities associated with each are combined into a single overall environment. His story also implies that the improvements in social interaction he has observed since moving to the current ward are linked to patients' embodied experiences in these more open and interconnected communal spaces. Improved social interaction amongst patients is therefore presented by Bradley as a beneficial attribute of the current ward. His observation that patients interact a lot "better" in this setting also presents an image of reduced friction in the environment and a sense that the spatial conditions are facilitating positive social interaction rather than working against it. Bradley's narrative thus implies that a greater sense of unity and esprit de corps is experienced by patients in this more integrated environment, where the flow of communication has been eased by increasing visual and acoustic connections and removing physical barriers. His account of the day area also evokes imagery of contemporary living and the common combining of dining and lounge areas in domestic open-plan settings to optimise living spaces and enhance a sense of connectivity. Whilst Bradley describes his appreciation for the layout of the communal areas within the current environment, later in the interview he also reflects on specific advantages he perceives in relation to aspects of the previous ward design.

"It felt more homely [*the previous ward*]. [**R**: Mm.] Yeah. [**R**: Yeah. Can you, can you think, can you describe any ways in which it, it, you know, it, um, how it felt more homely?] We had an upstairs, downstairs. So, upstairs was your bedrooms, downstairs was the lounge area, dining area, stuff like that."

The removal of the boundary between the dining and living areas on the current ward is positively regarded by Bradley, yet here he also expresses a preference for having a clear spatial distinction between the sleeping and living areas. These comments again generate imagery associated with more conventional residential environments, in

which the living areas are often located on lower floors, with the bedrooms on upper floors. Following Goffman's (1961/1991) observations that the borders between typical spheres of life, including sleep and leisure are generally taken away from patients in psychiatric institutions, this account implies that the spatial separation between floors has a role in maintaining a distinction between different types of activity. The staircase accordingly represents a transitional passageway that creates a sense of separation between the contrastingly 'public' and 'private' spheres of life associated with the communal living areas and bedrooms respectively. Elsewhere in the interview Bradley also describes how in the absence of another place, he currently considers the hospital environment to be his home. The differentiation between the sleeping and living areas created by the staircase on the previous ward can hence be interpreted as contributing to a valued sense of domesticity that might be experienced in more typical residential environments. Similarly, the sense of contemporary openplan living and togetherness implied in Bradley's account of the lounge-dining area on the current ward can be seen to be enhancing patients' perceptions of the environment, that are in turn improving experiences of social interaction.

It is noteworthy that some participants across the data set also comment on how the dining rooms on wards can typically be under-utilised between mealtimes. The incorporation of the lounge and dining areas as described by Bradley can consequently be seen to be optimising the space on the ward made available to patients for leisure activity. Bradley's account of being able to call across to patients, instead of "having" to go round to other rooms implies his awareness and appreciation of experiencing ease and efficiency within the current communal area layout. A sense of convenience in the environment may also be particularly significant for patients detained on secure wards, who frequently spend much of their time waiting for things to happen and whose ability to move freely through spaces is generally limited.

In the next story, Bradley provides further examples of ways in which he perceives that the material environment can shape the quality of social interaction and relationships between patients on the ward. Bradley contrasts his previous ward with the current environment to suggest that the spatial differences experienced in the latter can promote and enhance social bonding between patients.

Bradley: Extract 2

1.	R:	[] Um and in terms of sort of sharing the space here with other people, is
2.		that significant to your experiences here?
3.	B:	Yeah, everybody enjoys being in each other's company.
4.	R:	Right.
5.	B:	Yeah, I find that with this ward, that since we've been on this ward, the lads
6.		that are on it with me bonded really well.
7.	R:	Right, right.
8.	B:	Yeah.
9.	R:	Yeah. Do you, do you have any idea of why that, why that, why that was?
10.	B:	I would say it's because of the space, 'cause in (name of previous ward),
11.		when it was mealtimes, upstairs was open while the bedroom area was
12.		unlocked, no, was locked
13.	R:	Mm.
14.	B:	and but here it's unlocked. So instead of having, instead of thinking to
15.		yourself, right, 'I want to go to my room now',
16.	R:	Yeah.
17.	B:	we can go to our rooms any time.
18.	R:	Right. Yeah.
19.	B:	And, I reckon that has bigger impact because people don't want to be in their
20.		rooms all the time. 'Cause they're not forced to stay out of their room.
21.	R:	OK, so it's yeah.
22.	B:	Yeah.
23.	R:	They, they're wanting to spend more time.
24.	B:	Yeah.
25.	R:	Yeah, yeah, yeah. That's good. Yeah. So and are there any other sort of aspects
26.		about sharing, um, the space that, um, that's sort of, er, significant to you?
27.	B:	Yeah, we play pool, we play table tennis.
28.	R:	(<i>Coughs</i>). Excuse me.
29.	B:	Um, there's always somebody there to challenge. Which is good. (<i>Little laugh</i>).
30.	R:	Right. (<i>Little laugh</i>). A competitive challenge.
31. B :	Yeah.	
----------------	--	
32. R :	Yeah. (Little laugh and slight pause).	
33. B:	<u>Friendly</u> competitive challenge.	
34. R :	Of course! (Laughs).	
35. B:	(Laughs). Um, but there's no arguments, er and if two people have an	
36.	argument, then I leave them to it.	
37. R :	Mm.	
38. B :	I don't have <u>any</u> squabbles with anyone. I don't take anyone's side, it's up	

39. to them if they want to deal with it.

When responding to the researcher's question about his experiences of sharing a communal environment, Bradley presents positive imagery of a typically harmonious environment and a sense of mutual appreciation amongst patients, whereby "everybody" enjoys spending time together. His account also indicates that the positive bonding he has observed amongst patients since moving wards is linked to their spatial experiences in the current environment. Whilst patients could not access the bedroom area freely throughout the day on his previous ward, Bradley highlights how they can now visit their rooms at any time. His account suggests that through experiencing a sense of choice and not being denied access to their bedrooms, patients paradoxically no longer have such a strong desire to be in them and consequently prefer to spend time socialising together in the lounge. Bradley's story also indicates how patients' spatial experiences in one area of the ward can be intrinsically linked to the social milieu or behaviour observed in other spaces.

Bradley's account of there being people available in the day area to challenge to a game of pool or table tennis evokes imagery of a convivial and typically populated environment. In addition to inviting recreational play amongst patients, these items of table-based leisure equipment may also be encouraging people to stand around watching or to dwell socially in the space. Whilst Bradley's narrative presents patients' interactions as typically good-humoured, his account suggests that experiencing a sense of "friendly" competition or challenge is significant to the enjoyment of playing

games and contributes to the pleasure of interacting socially with other people. Bradley thus indicates how the presence of this equipment in the day area is helping to build a sense of community spirit that is lacking in his account of patients previously preferring to spend time alone in their rooms rather than socialise together. Through this story Bradley emphasises the importance of experiencing a close bond or sense of fellowship with other patients and he evidences his perception of typically harmonious relationships amongst patients through his account of infrequent arguments. Elsewhere in the interview Bradley mentions that he has lived with some of the same patients for several years and through his account of avoiding disputes with other people, he presents himself to the researcher as being typically friendly and laid-back.

6.4 Estelle

Estelle is a mental health nurse who works on a low secure psychiatric ward. She has also had several years' experience working in the hospital on rehabilitation wards before moving to a low secure environment. Estelle's overall narrative portrays her as a thoughtful and empathetic person who is concerned about quality of care and patients' everyday experiences in hospital. For example, stories of her daily activities typically highlight staff-patient interactions and include reflections from her own perspective about how patients might feel frustrated or uncomfortable when detained within a secure environment. Estelle reflects throughout the interview on the ways that the ambient or physical qualities of the environment may be associated with how patients or staff might feel when inhabiting different spaces on the ward. She considers especially how the direct adjacency of some spaces to others might be experienced by patients as uncomfortable or exposing and describes attempts to afford patients and their visitors the greatest possible sense of privacy.

Estelle's accounts of relationships and interactions with patients on the ward convey an underlying tone of professionalism and compassion. She also highlights her belief in the importance of establishing a mutual sense of trust between patients and

staff and the value of building therapeutic relationships. Whilst Estelle acknowledges it can be very difficult at times when patients vent frustrations about their circumstances onto staff members, she describes her efforts to avoid feeling hurt or taking these situations personally.

The overall tone that underpins Estelle's narrative is positive and whilst positioning her role as demanding and often highly stressful, she describes her work as typically fulfilling and enjoyable. Estelle's reflections on her working life convey a sense of a collegiate environment where staff on the ward work closely together to form a mutually supportive team. She also highlights concerns, however, about how the regular reallocation of staff members to assist on other wards affects resourcing on her own ward that can impact adversely on the everyday experiences of patients and staff. Her comments highlight that reduced staffing levels on the ward can affect the availability of escorted leave and contribute to increased frustration in patients that may in turn be taken out on staff. Furthermore, Estelle observes that patients' expressions of frustration associated with limited staff resourcing can have significant consequences in terms of their progression through the system, including being moved back to medium secure settings.

Estelle's storylines focus around: (1) the role of the environment in efforts to improve experiences for patients, staff and visitors, (2) appreciation of positive relationships with patients and staff colleagues, (3) awareness of how spaces can facilitate or hinder social interaction amongst staff and patients, (4) how individuals' mood states can influence the overall ward ambience and affective atmosphere (5) how staff resources affect the everyday experiences of patients and staff, including patients' transition from low secure to community living.

Estelle reflected in the interview on the photographs she had taken of: (a) side room/quiet room, (b) kitchen/dining room servery hatch, (c) kitchen servery

equipment, (d) clinic room, (e) handover room, (f) activities kitchen, (g) dining table in activities kitchen.

The analysis draws on storylines one to four which contribute to Estelle's perceptions of how relationships between patients and staff and the affective experiences of both groups might be associated with spatial experiences or physical attributes of the environment. In the extract below, Estelle reflects on her perceptions of the servery adjacent to the ward dining area and the interactions between patients and staff in this area.

Estelle: Extract 1

1. E:	OK, so this is the servery in the kitchen. Um (<i>slight pause</i>) that is, we have to
2.	give out the dinners now, um, which is a lot of work, er, it's quite challenging,
3.	'cause we have to wash them up and and it's, you know, ten patients, it's, it's a
4.	lot, um, but at the same time, um, it's a chance to speak to patients and they all
5.	hang around the hatch and have a laugh and a joke, um, you know and I find
6.	that time is, it's nice because, you know, once you've given out the dinners,
7.	they're all sitting at the tables and you're standing at the hatch, you know and
8.	they just, yeah, banter and it's a nice time
9. R:	Mm.
10. E:	for interaction, good interactions. Um and that's why I took that picture 'cause,
11.	it is, you interact loads with staff and patients at that time, um, yeah.
12. R :	Mm.
13. E:	And and some of them, you know, they tell you, they'll be standing at the hatch
14.	start talking about anything that's bothering them, um, yeah.
15. R:	Mm.
16. E:	So it's not just about having a laugh, it's you know, they do open up at that time,
17.	'cause, it's, it's an hour you know, just standing there and and talking.
18. R:	Mm, mm.
19. E:	So, I like that one.
20. R :	Mm. Yeah, yeah. Shall we?
21. E:	Yeah. (Slight pause) Yeah that was, sorry that was to, to prove that it was the
22.	kitchen.
23. R :	Yeah.
24. E:	But again
25. R :	Yeah.

26. E :	yeah, that's where we have our hot meals and dish them out to the patients and
27.	also staff are in the kitchen as well, again, it's a time where you're all
28.	interacting, you're all helping each other out, it's real-, like teamwork at that
29.	time, because we all know it's a lot of work, um, takes up to an hour, you know,
30.	with dishing out the food, washing, cleaning, everything.
31. R:	Mm.
32. E:	So you're all just helping each other, so again, in a stressful situation we will
33.	just come together and help each other. We have, er, a patient who comes in
34.	and cleans with us and um, then we have some that want to come and make
35.	their cheese toastie, so you know again, it's, it's like a big interaction for all of
36.	us, so that, it is quite a nice experience.

Estelle's initial reflections on her photo of the servery focus on the extent of work involved for staff in serving food to patients at mealtimes. Her description of how she and other staff members, "have to give out the dinners now" and "have to wash them up" presents these activities as both recent and required additions to their roles. Estelle also highlights the number of patients living on the ward as further evidence to the researcher about the amount of work entailed for staff within these additional duties. Whilst the narrative depicts this new aspect of her role as typically, "challenging" and "a lot of work" for staff, Estelle also conveys a sense of ambivalence by simultaneously positioning the associated interaction with patients and colleagues there as, "a nice time" and a rewarding experience. Alongside her expression of enjoyment in listening to patients' "banter" at the dining tables, Estelle's observation that patients will typically, "all hang round the hatch" to talk to staff contributes to an image of an informal and friendly atmosphere within that space. Patients can therefore be understood to be appreciating time spent with other patients there, but also to be enjoying having opportunities for social interaction with staff.

Figure 13 *Estelle's photograph of the servery hatch (with the roller shutter down)*



Figure 14 Estelle's photograph of the kitchen servery equipment



In contrast to some other areas of the ward, such as the nursing office, patients have immediate access to staff at the servery during mealtimes when staff are dedicated to an activity in which they are directly engaged with patients. The serving hatch also creates an opening between the servery and dining area to afford a direct physical connection between the two spaces and between staff and patients. The narrative therefore presents imagery of enjoyable social interaction and experiences of positive relationships between staff and patients. Experiencing the everyday domesticity of a space associated with serving food or washing up may also be a mediating factor in the positive social interactions between staff and patients described here. Furthermore, Estelle's depiction of patients, "standing at the hatch" and "standing there talking" implies that both the proximity to staff in that space and the informality of standing contribute to patients' ability to "open up" and confide in staff members who are also standing up in that space.

Estelle's description of the hour routinely allocated for serving and clearing up at mealtimes emphasises time-bound pressures for staff and she describes this hour as a period of intense activity. Despite portraying this activity in the servery as a "stressful situation", due to the volume of work required in a limited timescale, Estelle's account of mutual support and teamwork amongst colleagues working together presents an image of positive relationships and camaraderie in that space. Estelle's story also highlights how some patients also spend purposeful time in the servery on activities including helping to clear up after mealtimes or making "their cheese toastie". Instead of representing a 'staff-only' zone, the space can consequently be understood as an area of the ward that is shared by staff and patients. Estelle's description of the patient regularly cleaning up at mealtimes implies that a greater perception of parity with staff might be available to patients when working together there and a sense of satisfaction might be gained through contributing to the upkeep of the everyday environment. Through this story, Estelle also draws attention to the diversity of her own professional

role and how a variety of skills are required when working within different spaces on the ward during a typical day. Accordingly, within the following extract, Estelle describes the contrasting dynamics of nurse-patient interaction typically experienced when she is working in the clinic room on the ward.

Estelle: Extract 2

1. E :	Clinic. Um, that is, I took that photo because they, again, that's another time
2.	where you're being, you, you know, it's personal, um, you have to get to know
3.	the patients and and it's that trust that you know, they don't want to be on
4.	medication, some of them don't feel they should be on medication and having
5.	to dish it out to, to them and having them put their trust in you, that you've
6.	given them the right dose, you're not trying to kill them, you're not trying to
7.	poison em, um, and again they open up, they can open up in that situation and
8.	also, you know, some of them will get chest pains and think they're dying or
9.	and then you take their blood pressure and you know, it's quite a personal
10.	experience, um, urine samples, you know all those sort of intimate moments
11.	that you share with the patient.
12. R :	Mm.
13. E:	And you know, so I took that photo because I just think it's, you know, a real
14.	personal room
15. R:	Mm.
16. E:	and
17. R:	Mm.
18. E:	they have to trust you in that moment so
19. R:	Yeah, yeah.
20. E:	and that's when you're, the nursing, caring side comes out when it's in that, in
21.	those moments.
22. R :	Mm.
23. E:	The clinic area.
24. R :	Mm. Yeah, yeah.
25. E:	And trust, real trust. Because they have to strip off and you know, you have to
26.	be so non-judgemental and not make them feel you know, horrible in any way
27.	so, um
28. R:	Mm. Are there?
29. E:	that's very personal.

At the outset of her reflections on the photo of the clinic, Estelle's positioning of the room as a "personal" environment, presents an image of an intimate space on the ward

where she experiences a particular sense of closeness in her professional relationships with patients. Her photograph of this space also depicts an explicitly 'clinical' environment containing white metal medicine cabinets and medication trolley, alongside a pharmacy fridge and other pieces of medical equipment.

Figure 15

Estelle's photograph of the clinic room on the ward



In contrast to the sense of domesticity and parity between staff and patients evoked in Estelle's earlier account of working in the kitchen servery, the clinic is seen to be reinforcing unequal power dynamics in this space and the medicalisation of nursepatient relationships. The compact footprint and scale of the room can also be seen to be enhancing the sense of intimacy that Estelle experiences when working here with patients. Her account highlights how the activities taking place in this space are personal and private to each patient and therefore contribute to her perceptions of the clinic as a "personal" space.

Accordingly, the narrative conveys how the degree of trust perceived in relationships between staff and patients is highly significant to the affective

experiences of both groups within that space. Estelle's narrative also points to the relational tensions that medication might produce between staff and patients, to whom medicine might be legally given without consent under the conditions of their section. In noting that some patients express resistance and concerns about taking psychiatric medication, Estelle's story alludes to how members of staff may be required to enforce compliance. Interestingly, however, this is framed more in terms of a trust issue for patients within Estelle's observation that, "patients have to trust you in that moment". Her report of, "having to dish it out", further emphasises patients' lack of choice and conveys a sense of her professional obligation to ensure that medication is correctly dispensed. This account is also presented in contrast to her earlier story of "dishing out" food to patients at the servery and these illustrations of her activities in both spaces allude to a sense of repetition and the regular daily routine of mealtimes and medication times within institutional settings. The contrasts between these duties and Estelle's associated interactions with patients when working in both spaces again highlight the diversity of her professional role. In conjunction with her reflections on serving meals, Estelle's account of the "nursing, caring side" of her work being dominant here therefore alludes to how staff must transition between diverse aspects of their roles and professional identities when working in different spaces on the ward.

Estelle's account of how patients must place "real trust" in staff underscores her perception of how staff-patient relations are especially important in this space where patients may not typically feel at ease. The narrative illustrates how creating a sense of comfort and reassurance within a trusting relationship is a significant factor in enabling patients to "open up" and confide in staff about their health concerns. Estelle's reflections also highlight the varied range of responsibilities for staff, who must also monitor patients' physical health and undertake medical tests and procedures in the clinic room. Through her description of patients having "chest pains" and being concerned about "dying", Estelle presents an image of patients seeking reassurance

from staff and experiencing intense affectivity within the clinic. The narrative frames these concerns as somewhat irrational, yet experiences of chest pains and heart issues are not uncommon in people taking psychiatric medicine (De Hert et al., 2012; Mackin, 2008). Accordingly, patients' experiences of acute vulnerability as implied here might also be brought about by the medication itself. Estelle's description of medical processes and patients having to "strip off" in the clinic presents an image of physical intimacy between patients and staff being experienced within that space. Her account of being "non-judgemental" towards patients and concern not to let patients who have removed clothing to feel, "horrible in any way" conveys her awareness about patients' perceptions of self-esteem and alludes to how body image may be a concern for some patients in psychiatric settings. Indeed, patients taking psychotropic medication may commonly experience side effects, including weight gain that can significantly diminish self-confidence and self-worth (Every-Palmer et al., 2018; Waite et al., 2022). It is noteworthy, however, that Estelle's reflections do not explicitly link psychiatric medication use with patients' concerns in relation to body image, or low self-esteem.

6.5 Concluding comments

The stories discussed within this chapter have demonstrated varied ways in which interpersonal relations in secure settings may be linked to staff and patients' everyday experiences of space in the hospital environment. The narratives accordingly illustrate how participants' spatial experiences can affect interactions between staff and patients and relationships amongst both groups.

Throughout the narratives explored, participants' stories have again highlighted the affective significance of architectural elements, such as walls, windows and doors that divide or contain the hospital spaces. Interpretations of the narratives position these commonplace features of the built environment as symbolic thresholds that typically occur at transitional points between different 'worlds' or forms-of-process (Stenner, 2017, 2021). Physical passage between such worlds is generally controlled by

staff and participants' stories reveal how relational tensions within the ward are often heightened along these material borders between contrasting circles of staff and patient activity. Reports of intense affectivity experienced or observed by both groups on either side of these thresholds, including feelings of stress and frustration, are common across the narratives. Accordingly, participant accounts reveal that patients and staff may both experience psychological and embodied discomfort within spaces on the ward, which can affect interpersonal relations and social interaction amongst and between both groups.

Participants here and throughout the data set frequently reflect on their responses to windows or glazed partitions within the everyday environment and the contradictory nature of glass as a paradoxical material that can both connect and separate people simultaneously. Accounts hence illustrate that the visual link created by windows between spaces that are otherwise physically and acoustically separate, can exacerbate feelings of frustration that may contribute to discord or conflict amongst patients and staff. Participants' stories particularly highlight the connections created by glazing between areas that have distinct aesthetic and functional identities, such as the nursing office and patient lounge. Although these spaces remain physically separated, the glass allows their incongruent visual identities to be combined into the misleading appearance of a single world that comprises both lounge and office.

This sense of duality and the ambiguous status of the visually combined, but functionally separate worlds of the patient lounge and nursing office can therefore be seen to be intensifying the psychological unease frequently experienced there by patients and staff. Mike's observations about the lounge's lack of success as a place where patients might wish to spend time socialising or relaxing also points to one of the overarching tensions apparent within secure psychiatric facilities, whereby for patients, the ward environment constitutes both 'hospital' and 'home'. The narratives also suggest that internal glazing might contribute to discomfort experienced in spaces

on the ward by creating the perception of an exposing glass vitrine that frames the positions of both 'observer' and 'observed' on either side.

Participants' narratives allude to how material aspects of the ward may be experienced as divisions that physically separate staff from patients and symbolically emphasise their differing status positions and liberties. Stories explored in this chapter highlight how patients may be drawn towards and feel more comfortable in spaces that enable direct access to staff members and offer opportunities for staff-patient social interaction. The removal of spatial barriers is hence described as enabling an improved sense of 'flow' within the environment, which can offset perceptions of restriction and reduce relational tension by combining spaces into integrated worlds or circles of staff and patient activity.

The narratives indicate that a reduction in the friction that commonly occurs at the meeting point between contrasting worlds can in turn improve relationships and communication amongst occupants of the ward. Participants' experiences thus indirectly suggest that where fewer material boundaries occur within spaces, there is also an associated reduction in the prevalence of symbolic features, such as doors and locks, which can serve as unhelpful reminders of the unequal distribution of power between patients and staff. Estelle's experiences of the servery hatch accordingly reveal how the open physical connection between patients and staff there encourages perceptions of parity and closeness that can enhance communication and positive relationships between both groups. The absence of glass in the servery window can be seen to be connecting the spaces and activities on either side into a combined world. Without the physicality of a glass sheet to see or be seen through, it can also be understood that occupants' perceptions of surveillance may be diminished there, which might in turn contribute to greater perceptions of psychological comfort in that space. In contrast to the expression of heightened tension along the borders between different worlds that are both connected and separated by glass, participants'

narratives imply that improved social relations are perceived within physically connected worlds or spaces where staff and patient circles of activity are combined.

The narratives also convey how the multi-sensorial aesthetics and environmental ambience of spaces on the ward can influence occupants' interpersonal relationships and behaviour in those environments. Within communal areas that are frequently described as bland and clinical, the lack of atmospheric warmth and colour is portrayed as discomforting and unconducive to promoting relaxation and social activity amongst patients and between patients and staff. The scale and layouts of spaces, including furniture arrangements are also seen to affect participants' perceived levels of comfort, that can in turn influence their experiences of conviviality or a sense of community in different parts of the ward. The layout of furniture and ergonomic arrangements within spaces can thus limit or promote social interaction and the narratives imply that smaller scale environments can bring people closer together socially, whereas over-scaled spaces might push people apart. The volume of overly large communal environments as described here, plus their architectural finishes and the sparsity of objects within them may also contribute to the creation of an unsettling acoustic environment that can compromise the enjoyment of, or potential for social interaction within spaces. Physically and psychologically uncomfortable environments are therefore seen to discourage social interaction and may typically be regarded as transitory spaces as opposed to being experienced as inviting places to dwell.

By contrast, the narratives indicate that standing-based social activity can create an informal and flexible atmosphere that may enable a sense of ease within spaces on the ward. The stories told highlight how informal interaction amongst patients and between staff and patients can occur when standing around in areas including the bedroom corridor, day areas and the dining room servery. In contrast to accounts of the sense of routine and restriction imposed by the ward, the narratives indicate that spontaneous social interaction can be experienced and appreciated by

patients and staff when standing in these spaces. Nathan, Bradley and Mike's accounts of standing-based social activities within day areas and Estelle's observations of patients and staff standing together informally around the servery hatch highlight how standing is typically associated with a more flexible social dynamic than seated interaction.

Accordingly, it is suggested that occupants may experience a sense of comfort in spaces where the nature of social interaction is fluid and informal. Conversely, standing-based activity can also be seen to be less relaxed and more alert than sitting, such that it is possible to move more quickly from a standing position than when seated. Correspondingly, this could imply an association between perceptions of safety and a sense of psychological comfort for both patients and staff when standing in this setting. Estelle's experiences at the servery hatch reveal how meaningful and important ad hoc conversations between patients and staff can also be prompted when standing informally or moving around in spaces together. Mike's account of impromptu social activity within the bedroom corridor presents an image of social dynamics like those which may occur in a domestic house party, where guests might typically prefer standing together socially in smaller spaces to sitting less flexibly in larger rooms. The narratives hence indicate that fluid social dynamics within spaces on the ward can improve perceptions of flow and flexibility within the environment that can contribute to reduced friction and relational tension. Significantly, however, whilst perceptions of physical and psychological safety were key to enabling patients and staff to be able to feel relaxed, participants' accounts also highlighted how spaces on the ward could be experienced paradoxically as simultaneously relaxed and dangerous.

Participants also talk about how group activities within spaces on the ward can afford collective social experiences for patients and staff. Group activity that contributes to a sense of equality between staff and patients is described as able to enhance perceptions of community and togetherness in the ward environment. Spaces

are appreciated for enabling collective social interaction and facilitating shared experiences between staff and patients, including the simple pleasure of laughter, that can create a sense of parity and enhance relationships between both groups. Bradley's reflections on the communal areas of his ward suggest how the availability of leisure equipment (including table tennis and pool tables) actively encourages enjoyable social interaction and facilitates a sense of community on the ward.

Participants' stories suggest that movement through different spaces within the ward, or physical transitions between the hospital building and the world 'beyond' its boundaries may be associated with shifts in perceptions of self-concept or status positions for both patients and staff. For patients, physical transition between the different worlds within and beyond the building may be associated with shifting perceptions of self-identity and transition between the position of a detained patient in a secure institutional environment and a member of the wider community. Nathan's reflections about the process of searching patients within the side room accordingly position this space as a liminal zone which patients must navigate when physically and symbolically transitioning between the wider world outside the hospital and the inside world of the ward. The description of this in-between space thus implies a sense of distinction between these internal and external circles of activity and presents an image of institutional attempts to avoid 'contaminating' objects from the outside world entering the 'sterile' inside world of the ward.

Participants' observations reveal how relational tensions may exist within transitional or ambiguous spaces where the structures of different worlds can be seen to have been broken down. Nathan's account of his unease when searching patients in the side room, illustrates how activities within spaces that represent a transitional threshold between contrasting worlds, or identity positions, may be uncomfortable for both patients and staff. Stories such as this also point to inherent contradictions within the professional identity of staff in secure psychiatric settings who undertake a role

that has both caring and carceral responsibilities. Participants' accounts suggest that interactions taking place within different spatial and social contexts, both on and off the ward, may require staff to move between contrasting self-positions in their relationships with patients. This might include transitioning between an authoritarian role and more equitable 'peer-like' relations with patients when collaborating together or interacting informally in some spaces on the ward.

The nurse-patient relationship is a primary therapeutic tool within mental healthcare and a sense of trust is essential to building positive relationships between patients and staff (Gilburt et al., 2008). Participants describe complex negotiations around mutual trust and the narratives highlight how patients must engage in activities that require their trust in staff, yet staff members must also interact with patients on duties which convey a lack of trust. Nathan's account of socialising in the lounge with patients, despite his past experience of being attacked by a patient, illustrates how this is also the case from a staff perspective, whereby trust is required in a climate where there is mistrust. Participants' experiences of activities in spaces where perceptions of trust are at stake are thus described as being highly affective. The narratives also convey how spaces on the ward can be perceived contradictorily by staff as environments that are simultaneously both relaxed and dangerous.

The stories imply that awareness of risk and potential physical harm influences perceptions of the environment and the physical adjacency of spaces can contribute to staff concerns about safety and being isolated from colleagues' support in the event of an emergency. Mutual trust and a sense of teamwork amongst staff members are understood to be relied upon heavily to mitigate risk and help maintain safety within the ward environment. Teamwork experienced between patients and staff is also valued and can be afforded within spaces on the ward where both groups are able to collaborate on activities with a sense of equality. Spaces that afforded greater perceptions of equality and connectivity between patients and staff were therefore

understood to be assisting with building up a sense of trust between both groups that was perceived to be vital to the development of positive therapeutic relations (Gilburt et al., 2008).

Chapter 7: Discussion and conclusions

In the context of limited empirical exploration regarding the everyday experiences of patients and staff within the environments of secure psychiatric care, this thesis provides a close examination of staff and patients' lived experience, in its embodied and affective sense. Participants' experiences have been interpreted through a theoretical framework that integrates a psychosocial process account of experience with narrative psychology and the transdisciplinary concept of liminality. The primary research aim was to explore how participants make sense of their experiences in secure psychiatric settings through photographs and narratives and examine the role of space and environments in mediating affective experience. Through investigating how staff and patients inhabit the organisational spaces of institutional care, the research has also sought to examine how secure mental healthcare facilities can afford supposedly 'private' spaces inside their own 'public' spaces.

This visual-qualitative study employed semi-structured interviewing with 19 participants, in conjunction with participant-produced photographs, to invite patients and staff to talk about their everyday experiences of spaces in secure mental healthcare settings and the feelings or emotions associated with these environments. Together with the researcher, each participant constructed narratives around the spaces and objects presented in the photographs and offered reflections about their experiences in response to questions raised during the interview. The study aimed to ground patient and staff recollections of lived experiences in these material contexts, by focusing on images of physical spaces or objects within the everyday environment. Accordingly, it was envisaged that this approach would assist participants in recalling and reflecting on their embodied and affective experiences within specific physical settings, or recounting events that had taken place within them. Through this combination of visual images and narratives, staff and patients produced detailed and nuanced

accounts of the material, atmospheric and psychosocial characteristics of diverse spaces and of their own lived experiences within the hospital environment.

As discussed in the introductory chapter, there is renewed and increasing research interest in the intersection between architecture, design and health outcomes or lived experiences within healthcare environments. Accordingly, theories including those of 'therapeutic landscapes' (Gesler, 1992, 1993, 2005) and 'supportive design' (Ulrich, 1991), drawn respectively from geography and environmental psychology, alongside work in social psychology and architecture, have framed an expanding body of literature exploring connections between the physical, symbolic, social and atmospheric characteristics of mental healthcare environments and occupants' experiences (see Connellan et al., 2013; Curtis et al., 2007; Donald et al., 2015; Kanyeredzi et al., 2019; McGrath & Reavey, 2019; Mclaughlan et al., 2021; Reavey et al., 2019; Wood et al., 2013b). The thesis has correspondingly discussed varied ways in which spaces and objects mediated participants' lived experience and interpersonal relationships in secure psychiatric settings, through an exploration of how patients and staff made sense of their affective responses to the environment using narratives and photographs.

7.1 Summary of findings

It has been argued that the narratives explored within the thesis demonstrate an ongoing interplay between experience and space (McGrath & Reavey, 2019), such that the affective experiences of staff and patients in secure psychiatric settings are understood to be spatially distributed throughout the everyday environment. The findings presented in the three preceding empirical chapters have also drawn attention to similarities and differences in how patients and staff experience the everyday environment as co-occupants of secure mental healthcare spaces. Whilst the empirical chapters have been organised around examples of participants' stories that were felt best able to address the research questions, many of the storylines overlap within the

narratives explored in the analysis and across the overall data set. Alongside the narrative-based approach used to interpret the data, the concept of liminality has also provided a transdisciplinary perspective for exploring participants' accounts of their lived experiences within the framework of a relational process ontology (S. D. Brown & Stenner, 2009; Stenner, 2017).

In Chapter Four, patient and staff accounts of discomforting experiences in the hospital environment were interpreted in relation to the unease of being caught in a 'liminal hotspot' located between different 'worlds' or circles of activity (Greco & Stenner, 2017; Stenner, 2013). Participants were accordingly understood to be suspended in a variety of liminal zones situated between the contrasting 'forms-ofprocess' associated with: (a) patients' lives before and after hospital admission, (b) patient and staff status positions or spheres of activity, (c) the 'custodial' and 'caregiving' responsibilities of staff, (d) being 'inside' and 'outside' the hospital, (e) 'public' and 'private' spaces and (f) 'hospital' and 'home' environments.

Within Chapter Five, varied physical and psychosocial processes involving space as described by patients and staff were interpreted as forms of 'liminal affective technologies' (Stenner, 2017; Stenner & Zittoun, 2020) that were engaged with as means to self-produce liminal experiences that could enable transition between affective states. Such transitions typically involved movement away from distressing or difficult affective states towards preferred alternative states that provided reassurance and support. The technologies employed by patients and staff were associated with: (a) imaginative processes, (b) physical and psychological movement through space, (c) physical and psychological ordering of spaces and objects, (d) social interaction and (e) inhabiting spaces outside the ward.

The narratives examined in Chapter Six explored the role of spaces and objects in mediating interpersonal relationships and social interaction amongst patients and staff. Exploring participants' stories in process terms, the symbolic or material

divisions experienced by participants within the environment, or spatial features of the ward, were interpreted as thresholds situated at points of transition between contrasting 'worlds' or 'forms-of-process' (Stenner, 2017, 2021). Accordingly, participants' narratives highlighted a spatial relationship between interpersonal dynamics and the convergence of differing spheres of activity within the ward, whereby environmental and relational tensions were typically expressed at the meeting points between contrasting worlds. Participants' narratives included accounts of heightened affectivity experienced at the transitional thresholds situated between: (a) patient and staff spheres of activity, (b) contrasting patient self-identity positions, (c) different staff self-identity positions, (d) 'domestic' and 'clinical' perceptions of spaces and (e) worlds 'inside' and 'outside' the hospital.

7.2 Key insights

The introductory chapter explored a series of inherent tensions associated with secure mental healthcare facilities, including the paradoxical functionality of these settings as simultaneously restrictive places of carceral containment and therapeutic spaces to facilitate 'recovery' and rehabilitation. Participants' stories have illustrated ways in which these conflicting priorities are manifest environmentally and experienced by staff and patients through everyday interaction with the institutional spaces of secure psychiatric care. Key insights drawn from the analysis presented in the empirical chapters will be explored in the following sections and the implications of the research will then also be discussed.

7.2.1 The discomfort of feeling 'stuck'

One of the primary functions of secure psychiatric facilities is to support patients' journeys of 'recovery' and assist their preparation for reintegration into wider communities and the institution can itself be conceptualised in terms of a transitional or transformative process (McGrath, Mighetto, et al., 2021). Yet in these environments patients are typically detained for indefinite periods and can be seen to be caught in an ongoing liminal transition between their past and future lives. S.D. Brown and Reavey (2015) have referred to this state of suspension in terms of 'presenteeism', whereby through containment in a secure psychiatric setting, patients' previous lives and future aspirations may become displaced or disconnected through a predominant organisational focus on matters of the present. In the current study, the concept of liminality has similarly provided a perspective through which to interpret participants' stories of experiencing physical and psychological discomfort within the hospital environment, in the terms of unease associated with being caught in an 'in-between' or indeterminate state.

Due to patients' typically prolonged length of stay in hospital, secure psychiatric facilities also represent a 'home' environment (Brunt & Rask, 2005). However, as McKellar (2015) points out, whilst research findings imply a preference amongst patients and staff for familiar, deinstitutionalised and humane environments, psychiatric ward interiors generally reflect the dominant medical model of mental healthcare, giving rise to stark, clinical environments. Correspondingly, in the present research, participants' stories commonly conveyed a sense of the hospital wards as uncomfortable spaces, positioned awkwardly between conflicting aesthetic, functional and organisational priorities. Inherent tensions between operational concerns about safety or security on one hand and the provision of a therapeutic context on the other were revealed through interaction with the hospital environment and were seen to heighten participants' sense of discomfort or ambivalence towards spaces on the ward.

Loose furniture or joinery and architectural components, including door sets and windows comprise core components of interior spaces and the explicitly robust or safety/security-driven design and utilitarian detailing of these elements contributed to participants' experiences of a clinical or punitive ambience. These physical features of the environment were understood to be communicating directly with patients and

reinforcing unhelpful self-identity positions associated with 'deviancy' or 'abnormality', that were constraining and thus potentially limiting processes of transition or 'recovery'. Material or spatial attributes of the environment similarly 'spoke' to staff members and emphasised conflicts between carceral and caring responsibilities within their professional role that were frequently troubling and implicated in the dynamics of establishing or maintaining therapeutic staff-patient relationships. By contrast, spending time in more familiar or normative interior environments and outdoor spaces was found to reduce experiences of discomfort and enable transition between identity positions. Increased experiences of comfort were thus seen to enable patients to perceive a reduced sense of 'abnormality' and to optimise the caring role of staff, whilst promoting a sense of trust between both groups.

The physicality of the hospital architecture presented a material and symbolic boundary between the contrasting worlds and circles of activity inside and outside the institution. Experiencing connections between indoor and outdoor spaces and having direct access to the open air was seen to have a fundamental role in reducing the discomfort of an enclosed, restrictive environment for both patients and staff. Architectural features that connected patients visually to the outside world whilst simultaneously removing their connectivity to the multi-sensorial qualities of nature were seen to be both limiting the therapeutic potential of the environment and actively creating discomfort or distress. The scale and proportions of the hospital architecture and internal spaces were also frequently referenced and for some participants, the typically large scale of the physical environment was associated with experiences of affective discomfort. Features of the built environment, such as high courtyard walls or long corridors, were experienced as overbearing and embodying a sense of institutional dominance that was unhelpfully juxtaposed with patients' own experiences of having little power or control. Experiences of poor acoustic quality associated with the layout, scale and materiality of spaces were similarly described as

contributing to experiences of unease in the environment. Large, open day areas without clearly defined boundaries were unconducive to the affordance of comfort and were described as lacking a clear sense of spatial identity.

Spaces were seen to be incongruent with a residential context and correspondingly, it is argued that creating familiar environments which are reflective of 'normal life' is more significant to longer stay secure psychiatric contexts than traditional hospital settings for short term recovery (Olausson et al., 2021). Olausson and colleagues also draw on reflections from Malone (2003) to highlight how through hospital admission, individuals are displaced from the social contexts that shape identity, typically at times of acute vulnerability, to enter into an unfamiliar environment as a patient. Admission to a psychiatric hospital is accordingly a typically disruptive experience, which may in itself be frightening or confusing (Chevalier et al., 2018; Fenton et al., 2014; Riordan & Humphreys, 2007) and for some people may also be potentially traumatising (Murphy et al., 2017). Being disconnected from the familiarity and comfort of more 'normatively' designed everyday environments that exist outside secure institutions can also be seen to be compounding patients' experiences of restriction, or sense of being held in an awkwardly indeterminate position between worlds.

7.2.2 Creating comfort and becoming 'unstuck'

Participants described how ways of coping with discomforting everyday experiences in the hospital environment frequently included engaging physically and psychologically with spaces or objects to help facilitate affective transition, typically between troubled and more comfortable states of being. To gain a sense of respite from difficult circumstances, including stressful experiences associated with being on the ward itself, participants engaged physically with spaces and objects, or used imaginative processes to help create and inhabit psychologically safe or reassuring environments. These

processes were also seen to be helping patients and staff establish a sense of stability or be more grounded within disruptive or volatile situations, for example, at times when the ward was experienced as 'unsettled' by both groups. The varied processes used by participants to help mediate discomfort or distress by providing reassurance or a sense of stability, were interpreted through the data analysis as self-generated liminal experiences that were employed to assist affective transformation. Physical and psychological engagement with spaces and objects to facilitate affective transformation was also seen to be helping patients and staff mitigate the unsettling effects of experiencing imposed circumstances or events. For patients such events were seen to include the disruption of being detained in a secure hospital and the experience of acute mental distress. For staff these circumstances typically related to troubling aspects of their professional role, including involvement in restrictive practice and seclusion, or experiences of being physically assaulted at work.

Engagement with liminal affective processes involving space and objects in the environment was seen to be enabling patients and staff to move through difficult experiences and regain a sense of control over events taking places in their lives. Such processes were consequently understood to be helping patients and staff become 'unstuck' from a sense of unease or paralysis, by allowing them to move forward and to cope or be able to 'go on'. Experiencing a sense of progression by avoiding being caught in a position of discomfort was consequently seen to be significant in relation to experiences of well-being for both groups. For patients this sense of forward movement beyond their current circumstances could also be seen to be connected to 'recovery' and rehabilitation in a broader sense. Processes of psychologically planning or visualising and inhabiting alternative environments, including past or imagined future home spaces, were accordingly referred to as comforting or motivating and potentially healing experiences. Similarly, a sense of comfort was gained by patients

through physically arranging or organising and their bedroom space and from engaging with personal objects within it.

Within the empirical analysis chapters and across the data set, staff and patients also spoke frequently about their appreciation for spending time in spaces that were located off the ward, including outdoor areas. These settings were seen to facilitate a sense of respite from the intensity of the ward environment for both groups and were described as affording a more equal relational dynamic between patients and staff. The narratives also implied that the degree of physical distance from the ward was less significant than the experience of being spatially removed and separated from the ward environment. Staff typically observed that having an increased sense of freedom, comfort or privacy in spaces outside the ward could also help patients to confide in and 'open up' to staff, which was felt to be important therapeutically. Contrastingly, the narratives conveyed a sense of patients being more reticent and more constrained or held back in their emotional expression and ability to put their trust in staff within the restrictive or less private spaces of the ward.

Patient participant comments about being able to just "be ourselves" in areas such as the swimming pool, were echoed in staff narratives which suggested that patients were typically more at ease and able to express their individual personality more fully in areas off the ward. This was especially the case in spaces, such as kitchens or outdoor areas, that were experienced as more naturalistic and normative, or having a greater semblance of domesticity, with fewer aesthetic or procedural reminders of the secure context. Spending time in such spaces, typically when engaging in enjoyable and meaningful activity, including work, was seen to be enabling patients to get closer to the core of their own self-identity as individual people and be less influenced and constricted by labels such as 'offender' or 'patient'.

These findings can also be considered in relation to McQueen and Turner's (2012) study exploring views about work activity amongst people using forensic

mental health services in secure facilities and community settings. The theme of 'normalising my life', as identified by the authors, represented positive experiences of working and participants' observations implied that employment, including voluntary work, helped develop an independent sense of self that was important in building confidence or self-worth. A systematic review and narrative synthesis of qualitative research examining meanings of recovery for patients in secure psychiatric settings by Clarke et al. (2016) similarly found that experiences of 'connectedness' and 'a sense of self' were two prevalent facilitators of 'recovery' across the literature. Correspondingly, the findings of the current study suggest that spending time on meaningful activity in spaces containing fewer overt signifiers of security or risk management can helpfully contribute to rehabilitation processes by enabling patients to develop a more 'normalised' or familiar sense of selfhood.

Processes associated with movement through and within the secure environment were also relevant to participants' experiences of accessing spaces off the ward, including areas outside the hospital. The narratives implied that being outside the building, in natural settings or walking in open space was helpful in enabling transition between affective states for both patients and staff. Due to length of time required to leave or re-enter the building, however, staff were seen to forgo opportunities to access external space during shifts due to the limited time available on their breaks. Access to outdoor space for both groups was also mediated by the limitations of staff resources.

The narratives indicated how inpatient psychiatric settings can typically be intense and often volatile, or unsettling environments in which both patients and staff may feel concerned about their own safety or the safety of other people (Lundin, 2021; Mind, 2004, 2011; Quirk et al., 2004; Quirk & Lelliott, 2001). Staff and patient stories illustrated how experiences of physical and psychological safety for both groups are mediated by the material and atmospheric attributes of the hospital environment.

However, the removal of patients' personal possessions due to concerns about physical safety risk, can diminish patients' experiences of psychological and physical comfort at times when it may be most needed and illustrates how physical safety may typically be prioritised over emotional safety within these environments (Veale et al., 2022).

7.2.3 Integrating worlds

The stories explored in the analysis chapters and across the data set have highlighted varied ways in which staff and patient experiences of being in the physical environments of secure psychiatric services are closely intertwined with relational experience or social interaction in those spaces. The material or symbolic boundaries experienced by participants in the everyday environment were interpreted through the analysis as thresholds situated between contrasting 'worlds' or spheres of staff and patient activity. Environmental tensions were especially evident at the interface between different worlds and were revealed through staff and patient accounts of intense affective discomfort. These experiences of discomfort or distress were in turn seen to adversely influence communication processes and interpersonal relationships amongst both groups, often through prompting expressions of frustration. Spaces where the occupants of one world felt scrutinised by 'observers' located within another visually connected, but physically separate world were also associated with experiences of unease amongst patients and staff. By contrast, environments in which staff and patient activity was more fully integrated in the same physical space, were seen to enhance experiences of comfort that contributed to improved relations and communication between or amongst both groups.

Physical and symbolic thresholds in the environment were also seen to emphasise positional distinctions between patients and staff, including their differing experiences of status or power within the institution. Yet, in more integrated spaces, with fewer tangible divisions, participants reported a greater sense of parity between

staff and patients that could enhance social interaction and contribute to an increased sense of community. Spaces where a greater sense of equality was experienced between patients and staff in turn assisted in enhancing a sense of trust between both groups that is recognised as an essential aspect of positive therapeutic relationships (Gilburt et al., 2008). Participants' stories also suggested that a sense of congruence between the scale or proportion and the function of spaces is significant to experiences of physical and psychological comfort, or discomfort within the hospital environment. Correspondingly, the extent of social interaction in spaces on the ward was seen to be associated with the environments' active role in affording comfort and in bringing people together or conversely pushing them apart.

An increasing body of literature is focused on patient 'recovery' in secure psychiatric care (Clarke et al., 2016; Drennan et al., 2014; Livingston et al., 2012; McKeown et al., 2016; A. I. F. Simpson & Penney, 2011) and it is recognised that positive relationships experienced by patients in secure psychiatric settings, including therapeutic relationships with staff, are supportive to 'recovery' processes (Mezey et al., 2010). Correspondingly, having a sense of being respected, well-regarded and understood in therapeutic relationships with staff is significant to patient satisfaction in relation to secure mental healthcare services (Bressington et al., 2011; MacInnes et al., 2014). The quality of therapeutic relationships between staff and patients has hence been identified as a key determinant of satisfaction for patients detained within secure psychiatric settings (Coffey, 2006). Research findings also suggest that positive social environments and staff-patient interaction, alongside increased patient activity (i.e. engaging in social or therapeutic activity, rather than spending time alone) can improve clinical outcomes for patients in inpatient mental healthcare settings (Collins et al., 1985; Sharac et al., 2010). Sharac and colleagues' (2010) literature review examining staff and patient interaction and activity in psychiatric wards highlights, however, that nurses typically spend up to only half of their time engaging with

patients and a significant amount of patients' time is spent alone. Participants in Molin and colleagues' (2016) study exploring patients' experiences of everyday life and interactions in psychiatric wards emphasised the importance of familiar relationships with staff and the experience of ordinary social interactions together, such as sharing informal conversations and laughter (see also Reavey, Poole, et al., 2017). In the current study, participants' accounts similarly illustrated how patients valued spending time with staff members and generally wanted to interact together socially.

Participants' stories highlighted that although staff are not resident on the ward, the hospital constitutes a form of communal living because patients and staff continually co-occupy the same environment. The narratives illustrated, however, that an uneven distribution of everyday activity or duties within the environment emphasised the positional differences between patients and staff that diminished experiences of togetherness. Both groups suggested that for patients, having responsibility for the care or upkeep of spaces can promote perceptions of greater autonomy and provide a valued sense of having a more typical domestic life. Experiencing a sense of homeliness in the environment was linked to social interaction and the quality of relational experiences in spaces. Small spatial moves including repositioning the height of the television unit or introducing games equipment, such as table tennis or pool tables, were seen to bring a sense of domesticity and conviviality to spaces that encouraged greater social interaction. Accordingly, these modest adjustments to the physical environment contributed to participants' accounts of closer relationships and improved interaction between and amongst patients and staff, alongside an increased sense of community and camaraderie. Positive interpersonal relationships were also experienced in spaces that brought patients and staff together in everyday domestic activity and mundane tasks, such as clearing and washing up after a meal.

Correspondingly, Marshall and Adams' (2018) study examining staff relationships with patients in secure psychiatric settings, found that staff members shared a holistic view of 'recovery' that recognised the value of positive relationships and the need to create a sense of home within the institution. Staff acknowledged how patients typically viewed the hospital as their home due to length of stay and suggested that therapeutic relationships could be improved by creating more homely environments. Staff members' suggestions on how to improve relationships included incorporating small home-like touches to the ward milieu, in terms of physical changes, but also through homely actions or gestures, such as enabling staff to sit down and share a meal or have a coffee with patients. Nurse participants in Martin and Street's (2003) study of therapeutic relationships in forensic psychiatric nursing similarly observed that factors contributing to building positive nurse-patient relationships included sitting down informally with patients, or taking part in shared activities, such as sport, or going for a walk outside. Papoulias and colleagues' (2014) systematic review exploring the therapeutic potential of psychiatric ward design also found that home-like environments and the availability of private or semi-private spaces were associated with increased social interaction and improvements in patient well-being. Furthermore, studies examining modifications to enhance the ward environment have found significant reductions in the use of seclusion and restraint following minor physical adjustments, such as rearranging furniture to facilitate social interaction, replacing old furniture and introducing warm paint colours (Borckardt et al., 2011; Taxis, 2002).

7.3 Implications

The participants' stories explored in this study have illustrated diverse ways in which the inherent tensions between operational priorities within secure psychiatric settings are manifest in and mediated by the physical, social and symbolic environment. The

narratives have particularly highlighted ways in which tensions between safety and security requirements and the creation of an ostensibly therapeutic environment become apparent through participants' spatial interactions and affective experiences in relation to spaces and objects in the hospital.

In light of multiple legal, clinical and architectural considerations, designing and planning secure psychiatric facilities is a complex task (Seppänen et al., 2018). In recognition of the risk-related challenges associated with secure settings, a guiding principle in the design and development of forensic mental healthcare services is concerned with optimising physical, procedural and relational security (Department of Health, 2011a; Kennedy, 2002; Seppänen et al., 2018). Physical security is concerned with the material integrity of the secure environment, whilst procedural security relates to the effective implementation of robust operational policies regarding safety and risk management. Relational security has been defined as, "the knowledge and understanding staff have of a patient and of the environment, and of the translation of that information into appropriate responses and care" (Department of Health, 2011a p.10) and according to Kennedy (2002), it has utmost significance in maintaining therapeutic progress for patients. It is consequently argued that the therapeutic and safety, or security concerns of the environment in secure services should not be viewed in isolation, or as polarised ends of a spectrum, but instead be regarded and addressed as integrated concepts (Department of Health, 2011a; Seppänen et al., 2018). Correspondingly, A.I.F. Simpson and Penney (2011) point out that in order to align with a recovery paradigm within forensic mental healthcare, it is necessary to view, "[...] treatment and security as requirements of one another, rather than dichotomous goals" (p.304). The therapeutic alliance between staff and patients is hence understood to form a common ground between effective security and successful treatment in secure psychiatric services (A. I. F. Simpson & Penney, 2018). However, the creation of relational security is contingent on the physical environment (Seppänen et al., 2018)

and environmental design can also be regarded as an important therapeutic interface between security and treatment as integrated priorities.

Within the present research, the environmental tensions revealed through participants' affective experiences in secure psychiatric settings have been conceptualised in process terms. Experiences interpreted as forms of enduring liminality, or the sense of being stuck between 'worlds', were typically characterised by paradox and described as discomforting for both patients and staff. This sense of discomfort was frequently associated with participants' simultaneous experience of contradictory 'worlds' within the same physical environment. Within these hybrid spaces, there was generally a sense of imbalance. As an example, concerns regarding security, risk management and physical safety, were seen to dominate over comfort, familiarity and emotional safety in terms of the messaging conveyed to occupants through details in the environment (Veale et al., 2022).

As discussed earlier, however, an enduring indeterminate state may also give rise to novelty, through what Greco and Stenner (2017) have termed 'pattern shift', involving attempts to redefine paradoxical contradiction by embracing both 'worlds'. In secure services, pattern shift may then be brought about through design innovation and creativity to enable greater integration of staff and patient worlds, or the worlds 'outside' and 'inside' the hospital, alongside increased congruence between therapeutic and security or safety-driven requirements. Patients and staff talked about experiences of discomfort within the everyday environment, yet participants' stories also revealed diverse ways in which a sense of comfort could be created through engagement with spaces or objects, both physically and psychologically. Engaging with design as a creative process or technology to facilitate innovation could assist in addressing discomfort, whilst also helping to optimise the aspects of the environment that provide comfort or are vitalising, reassuring and supportive.

As discussed in the introductory chapter, the notion of 'designing in' comfort or familiarity into mental healthcare environments is not new. It represented a fundamental principle in 19th century asylum design, whereby environments were specifically designed to be cheerful, comfortable and homely as a means to offer a supportive surrogate family to patients (Boult, 2017). Through careful attention to detail at all scales, the built environment was understood to play an active role in rehabilitation, such that design was not regarded as neutral, but instead seen to have the capacity to either enhance or detract from the process of healing (Edginton, 1997, 2003). This philosophy and approach towards the creation of a therapeutic environment also attempted to provide a sense of familiarity in order to remove distinctions between the worlds 'outside' and 'inside' the institution (Edginton, 2003).

Despite more recent renewed recognition of the environment's role in mental healthcare and increased awareness of widespread preferences for familiar and nonclinical spaces, the design language of inpatient psychiatric care is typically lacking a congruent identity and remains situated within a medical design typology (McKellar, 2015). The findings of the current research can be seen to support the argument for creating familiar and non-sterile spaces as a key objective in mental healthcare design (Karlin & Zeiss, 2006; Shepley et al., 2016). However, it could also be argued that enabling 'pattern shift' in this context requires innovation and collaborative design thinking. This might include a shift in perspective from a position of attempting to make clinical or carceral environments more familiar or homely, to the creation of familiar and comfortable environments in which physical safety and security requirements are unobtrusively integrated. This would also involve raising aesthetic aspirations in relation to the overall built environment, including the design of interior products or architectural fixtures that contribute significantly to the clinical or carceral tone of spaces and 'speak' to occupants.

A similar point is made in relation to 'building in' emotional well-being into carceral spaces by Jewkes et al. (2019), who highlight that recent initiatives to provide trauma-informed care and practice in UK women's prisons are compromised by environments with, "hostile architecture, overt security paraphernalia, and dilapidated fixtures and fittings" (p.1). Such inherent and explicit aspects of the built environment are accordingly argued to be antithetical to trauma-informed approaches and may then trigger, exacerbate, or contribute to symptoms of trauma. A holistic and revolutionary approach is consequently called for, in which trauma-informed design is built into the environment, "from the ground up" (Jewkes et al., 2019 p.7). The authors also question if carceral settings could be designed for healing, by incorporating trauma-sensitive design principles and innovation from ground-breaking healthcare environments, such as the Maggie's centres for cancer support.

As non-residential sites located beside regional hospitals, each Maggie's centre is a unique space with a domestic architectural scale designed to optimise the therapeutic potential of the environment (D. Martin & Roe, 2022). Hopeful atmospheres and environments are accordingly manifest through high-quality architecture, connectedness to light and nature, alongside careful attention to detail and materiality, to create comfortable, non-clinical spaces that offset the anxieties which typically accompany cancer treatment (Butterfield & Martin, 2016; D. Martin et al., 2019). Whilst Jewkes et al. (2019) point out the challenge of reproducing the bespoke qualities of Maggie's centres in mainstream healthcare settings, the philosophy and design principles that underpin these environments are nonetheless aspirational.

Within the context of mental healthcare spaces, pioneering work by the Madlove collective blends research with creative practice to bring together people with diverse backgrounds and different mental health experiences in a project to reenvision psychiatric environments as 'designer asylums' by creating positive spaces to
experience mental distress (Zorwaska, 2019). This ongoing work integrates diverse perspectives, including experts by experience, non-experts, artists, designers, academics or mental health professionals and illustrates the importance of collaborative approaches, whilst demonstrating the need for creativity and innovation in approaches to mental healthcare design.

The *Design with People in Mind* booklet series published in the UK by the Design in Mental Health Network has also contributed to the sharing of key research evidence relating to environments and mental health across the literature, including nursing, psychiatry, psychology, architecture and design (Reavey et al., 2021; Reavey, Harding, et al., 2017; Reavey & Harding, 2018, 2019). By presenting summaries of key evidence in an accessible format, these publications aim to share research findings with diverse stakeholders, with the hope of benefiting those who design, live or work in mental healthcare environments.

The narratives explored within the current study have suggested that affording increased comfort to occupants by diminishing environmental contradictions and tensions has implications for both patient and staff well-being, in addition to the relationships between and amongst patients and staff. For patients, the reduction of environmental friction to promote greater ease and experiences of comfort in the everyday environment may then also have wider consequences for 'recovery' and rehabilitation.

7.4 Reflections on the research process

Reflections on the overall research process will be outlined in the following section, including thoughts about my experiences as a novice researcher undertaking this visual-qualitative study in the context of low and medium secure mental healthcare environments. This includes an awareness of my active role in the research and the influence of my own assumptions, lived experiences and personal characteristics on

the approach to the research, including the study design and the processes of collecting, interpreting and presenting the data.

The reflexive process recognises a distinction between, 'personal' and 'epistemological' reflexivity, as two reflexive modes that respectively acknowledge the researcher's role in shaping the research, as an individual person and through the theoretical and methodological approach to the research (Willig, 2013). Personal reflexivity, as outlined by Willig (2013), is concerned with the influence of individual characteristics on the research process, which may include the researcher's own values, beliefs, experiences, age, gender, or social identity. As discussed in the methodology and analysis chapters, my own personal characteristics and those of the participants, plus our social interaction together were accordingly understood to have a bearing on how the narratives were co-produced in the interviews and the content of the stories told (Silver, 2013).

Willig (2013) also details how epistemological reflexivity is concerned with the researcher's theoretical assumptions about the world and the production of knowledge. Epistemological reflexivity thus includes exploring the ways in which the study design or approaches to analysis contribute to how the data analysis is developed and presented. Correspondingly, in the context of narrative inquiry, Wong and Breheny (2018) point out how different analyses will be produced by different researchers as a result of their individual influence on the research at each stage of the process. This might include the questions that are asked during the interview, or the researcher's responses to participants' stories and the process of identifying and selecting the narratives that are felt best able to address the research questions. Reflexive processes recognise the researcher's own motivations or investments in the subject matter, although these can be viewed not as 'biases' to be removed from the research, but instead as conditions that frame the research process and have a part in shaping the findings (Willig, 2013).

Having a background as a practitioner within the discipline of design and architecture informed my interest in exploring experiences of everyday spatial interaction in secure mental healthcare environments. This perspective can also be seen to have played a part in shaping the overall study and my involvement throughout the research process. As an example, the inclusion of photography and visual research methods in the study design reflects the visual orientation and focus on creative expression within design practice. My interest in design and the built environment may have also influenced the interviews in terms of the emphasis given to certain topics, or the questions asked in response to participants' reflections on their photographs.

Reflexivity is also concerned with issues of ethics and undertaking the research in secure psychiatric environments presented a range of ethical considerations as discussed in Chapter Three, alongside practical considerations, including access and communication (Spiers et al., 2005). Consistent with the specific research focus on exploring experiences of the environment in secure psychiatric facilities, particular reflexive consideration is given to the research context and the experience of undertaking the study in these settings. These reflections recognise the significance of the active role played by the research environment itself in mediating the conduct of the study, including the process of data collection.

7.4.1 Research in secure settings

Following the initial contact made with patients during community meetings on the wards, it was necessary for the intervening liaison about meetings with patient participants to take place via staff members. Communication processes were consequently more removed with patient participants than staff members, with whom it was possible to liaise directly and increase a sense of rapport. Due to the nature of the secure environment, it was necessary for interviews with patients to be undertaken on the wards, typically in meeting rooms or side rooms and not within other areas of

the hospital. The participants' narratives highlighted, however, that patients might also typically spend time in these spaces for other meetings, including mental health tribunal panels and consultations or therapy sessions with psychiatrists and psychologists. Recollections of potentially difficult past experiences or events taking place there previously may have been significant in terms of how participants experienced those spaces during the interviews and the extent to which they felt at ease within them. Whilst some of the interviews with staff members took place in meeting rooms within office areas located elsewhere in the hospital building, the majority were also undertaken within side rooms or meeting spaces on each ward. This proximity to the ward and having an awareness of everyday ward activity taking place in the background may have influenced some staff members' ability to feel removed from their typical duties and responsibilities whilst participating in the interview. As an observable example of this, on one occasion it was necessary to pause the interview with a staff participant when they had to leave the room quickly to assist colleagues in responding to an incident on the ward.

Several interviews with staff and patients were also interrupted briefly by other members of staff knocking on the door, or simply entering the room. As highlighted in some participants' accounts, perceptions of privacy and potential intrusion within the environment may have played a mediating role in the degree to which people felt comfortable and able to talk freely about their experiences during the interview. From my own perspective, the experience of the interviews being disrupted in this way on several occasions was distracting and slightly unsettling in an unfamiliar environment where I was typically not fully at ease. It also revealed a sense of how functionality might typically be prioritised over privacy within the environment. It was necessary to relocate to another space partway through an interview on a couple of occasions, as the original room was subsequently required for other purposes. Despite the disruption, however, I found that there were also benefits to moving rooms twice

during the interview with one participant, as we spent time together inside several of the rooms discussed, rather than looking only at photographs of these spaces.

The typically difficult acoustic characteristics of the hospital environment, as discussed within some participants' narratives, were also significant to the experience of conducting the research and background noises from other parts of each ward were typically perceptible during the interviews. In some cases, the volume of individual sounds, such as doors slamming or people shouting, was so great that one or two words were rendered inaudible in some of the interview recordings. Instances of loud background noise during the interviews were therefore a distraction and potentially unsettling to participants, as they were at times to me as a researcher and a visitor to the ward. Noise disturbance is especially material to the interview context within psychiatric healthcare environments where people experiencing distress may commonly have a heightened sensitivity to sound (Stansfeld, 1992; Sutton & Nicholson, 2011) and some people may experience intrusive thoughts or voices. When visiting the wards, it was troubling to hear expressions of other people's distress or frustration, such as shouting and the adverse influence of these sounds on my own affective state was compounded by their volume and reverberation in the environment.

In the individual meetings with patient participants, it was typically necessary to wear a personal safety alarm which occasionally made unexpected beeping noises during the interviews. These sounds were disconcerting and implied that something might be amiss elsewhere in the ward. Although the beeping did not cause a significant disruption to the interviews, these sounds made the alarm's existence more visible in the research environment. In line with participants' comments about the symbolic significance of keys or radios worn by staff, the physical and audible presence of the safety alarm could be similarly understood to be unhelpfully emphasising positional differences between me as the researcher and patient participants. Correspondingly, as only the researcher wore a safety alarm during the interviews, this object may have

reinforced patients' discernment of having an 'offender' identity, or a sense of being feared or perceived as 'dangerous' by others (Jacob et al., 2009). I was consequently conscious of wearing the personal safety alarm in the meetings with patients and slightly unnerved by thoughts that I might inadvertently set it off. Alongside the CCTV cameras in the rooms used for the interviews, the alarm could also be seen to be accentuating wider discourses about safety, security and risk mitigation that underpin the management of secure psychiatric environments (Kennedy, 2002).

For reasons of patient protection, it is generally prohibited for cameras to be brought onto secure wards and a digital camera belonging to the multi-disciplinary team on each ward was typically made available for use in the study. In cases where a ward camera was not available, specific permission was obtained from the research setting to allow a camera to be taken onto the ward for research purposes. Each ward had different systems or limitations associated with printing or making participants' photographs available for use in the research and it was not possible to print out the images on some wards due to the equipment available there. In these instances, it was necessary to review participants' photographs on the camera screen rather than laying out prints on a table. Consequently, it was less easy for participants to visually crossrefer between photographs in these cases and the images were typically talked through in a more linear way. This also meant that participants often moved through the set of images more quickly.

The smaller size of images when compared with prints also meant that some details were not so readily appreciated when reviewing on a camera screen and it was less easy for both participant and researcher to have a clear view of each photograph simultaneously. Alongside this, in several instances where participants' photographs were taken using ward equipment, it was not possible for the digital images to be shared, so only paper prints, which were typically black and white, could be retained where available. As a result, not all participants' photographs could be reviewed again

for reference during the transcription and analysis processes. Additionally, this meant that some participants' photographs were not available for inclusion in the thesis and a few of those that have been included have a low image quality as it was necessary to reproduce them from paper prints.

The narratives revealed how participants frequently perceived a lack of control over the hospital environment and similarly, reflecting on the physical and acoustic intrusions in many of the interviews highlighted my own limited control as researcher over the spaces where these took place. The characteristics of the research context were thus typically contrasting with what might be considered 'ideal' conditions for undertaking qualitative research interviews, such as affording a peaceful environment with minimal distractions, for the benefit of both the participants and researcher. Additionally, I was mindful that the need for staff members to be aware that a patient was taking part in the study may have further affected patients' sense of privacy in relation to the research process in a way that did not apply to staff participants.

By regularly visiting wards and undertaking interviews there I observed general activity that provided valuable additional insights into the everyday experiences of patients and staff. Several of my own experiences as a visitor to the wards were parallel to those expressed in participants' narratives, including perceiving a lack of control over the environment as described above. Despite being at liberty to leave, I also experienced a sense of restriction and containment when visiting the wards. To exit the building, it was necessary for a staff member with key fob to escort me through the hospital to the main airlock, typically via a number of corridors and several sets of internal doors. Spending time visiting the wards enabled me to absorb a sense of the atmosphere and enhanced my understanding of how each of these slightly different environments operated. However, I was also conscious of not wishing to intrude in those spaces and typically felt a desire to have less visibility as a visitor to the ward. This was especially so within glazed nursing stations, where I felt

uncomfortable observing events through the glazing and experienced similar feelings of exposure in these spaces to those expressed by some of the staff participants. The experience of being in secure environments and undertaking research in this context was intense and quite difficult at times, in both practical and emotional terms. Participants' accounts of their experiences were often moving. In contrast to patients' experiences of detention, I was also mindful of my own experiences and relative freedom as a person living independently in the community. I was aware of researcher responsibility in relation to participants and conscious of ethical considerations or dilemmas, for example in writing about participants' lives, or how stories are interpreted in the analysis (Josselson, 2011). Throughout the project, it was important for my own well-being to have opportunities in supervisory meetings to discuss the research, including ethical considerations, at each stage.

7.5 Future directions and concluding comments

The aim of this thesis was to explore the role of space and objects in mediating the affective experiences of patients and staff within secure psychiatric settings and to understand how participants make sense of their experiences in these environments using narrative and photographs. Notwithstanding the logistical challenges already discussed, the use of photo-production as a visual method was a strength of the research and helped facilitate the generation of rich and spatially grounded narratives regarding participants' everyday experiences in secure mental healthcare settings. Accordingly, the study has afforded a close examination of secure psychiatric environments and afforded increased visibility to these very particular healthcare spaces that are not widely or easily accessed. It was envisaged that photo-production might be perceived as a creative and enjoyable activity and several participants remarked anecdotally that they had enjoyed taking the photographs and the process of choosing what to feature in the images. As the participants were based on six wards

located within a range of buildings across the hospital site, it was also possible to explore the experiences of staff and patients in several medium and low secure environments with varying layouts and designs. Whilst the overall sample size was determined by the availability of individuals volunteering to participate, the composition reflected an almost equal split between staff and patients, women and men and low and medium secure environments, which provided balance to the study.

People who use mental health services are becoming progressively more involved in the design and conduct of research within mental health settings, however, involvement is more limited in forensic mental healthcare environments (MacInnes et al., 2011). This may reflect a variety of reasons, including significant practical, access and communication issues involved in collaborating with patients in secure facilities, alongside the additional time and resources required to engage with people using forensic mental health services (Faulkner, 2007; Spiers et al., 2005; Völlm et al., 2017). These issues present challenges to research in secure psychiatric settings, however, collaborative studies can empower people who may typically be marginalised and enable the exploration and prioritisation of issues that are beyond clinical parameters. Whilst collaboration with experts by experience in secure settings would have similarly enhanced the design and development of the current research, a limitation of the study was that this was not achievable within the overall programme and available resources. Future co-produced research would be valuable to explore patient and staff perspectives and increase understanding about how the design of secure psychiatric spaces might optimise the therapeutic potential of the environment. The Experiencebased Co-design (EBCD) approach has been widely used in UK general healthcare as a framework for participatory action research in which patients and staff work together to share experiences of healthcare environments to help expose and prioritise areas of focus for service improvement (Bate & Robert, 2007; The Point of Care Foundation, 2016). Although its use to inform potential change within mental healthcare services is

relatively new, adapted forms of EBCD have also been developed to suit the complexities and ethical considerations of working with 'vulnerable' populations in mental healthcare contexts (Isobel et al., 2021; Larkin et al., 2015).

As mentioned in Chapter One, a second parallel study to explore the lived experiences of people using forensic mental health services and staff in high-support accommodation within the community was also developed and pursued originally as part of the current research project. The proposed study aimed to examine experiences of space within high-support housing settings and explore how the material, social or atmospheric features of such environments might mediate lived experience, including patients' experiences of transition between inpatient psychiatric settings and the community. For people discharged from secure hospital wards into high-support accommodation settings in the community, the latter represent a transitional zone between relative restriction and isolation experienced within secure inpatient environments and greater social re-integration into the wider community (Cherner et al., 2014). Consistent with secure inpatient settings, high-support housing facilities for people who use forensic mental health services similarly intersect the mental healthcare and criminal justice systems. Forms of surveillance and restriction are likewise apparent in this form of community-based shared accommodation, where the environment typically includes 24-hour staffing, CCTV monitoring, door access control and signing-in processes for visitors. Patients who have been detained in secure psychiatric facilities may typically be conditionally discharged into the community and failure to comply with prescribed conditions, such as engaging with mental health services or taking prescribed medication, may result in being recalled to hospital (Latham & Willliams, 2020).

There is, however, limited existing research exploring how supported accommodation settings in the community for people with mental health difficulties are experienced by occupants (Bengtsson-Tops et al., 2014; Krotofil et al., 2018), or

examining the efficacy of service provision (Chilvers et al., 2006; Killaspy et al., 2016; McPherson et al., 2018). Further empirical study would therefore usefully expand knowledge and understanding of how the environments that constitute high-support housing in the UK are experienced by people who use forensic mental health services and staff in those settings. Whilst the separate study proposed to address this could not be progressed within the current research programme, undertaking this research in the future would valuably build on findings from the present exploration of patient and staff lived experiences within secure inpatient facilities. Future research conducted in high-support accommodation could assist in increasing empirical understanding about how occupants make sense of their experiences within this form of residential setting, including the experience of transition between secure inpatient environments and the community (Coffey, 2012b).

Further study in this context could also help expand knowledge of how the concept of 'recovery' might be perceived by people who use forensic mental health services and staff respectively as a personal and organisational goal, in addition to furthering understandings of associations between 'recovery' and spatial experience (Borg et al., 2005; Borg & Davidson, 2008; Mezey et al., 2010; Milbourn et al., 2014). The discrimination perceived through labelling as both 'offender' and 'psychiatric patient' has been identified by patients in secure settings as an impeding factor in 'recovery', specifically in the context of discharge from hospital and independent community living (Mezey et al., 2010). Studies have also highlighted how housing quality and appropriateness of living conditions can be associated with social integration and community adjustment for people discharged from inpatient psychiatric settings (Baker & Douglas, 1990; Dorvil et al., 2005). The restrictions that may be imposed under conditions associated with hospital discharge can also limit the extent of social inclusion and choice that people who use forensic mental health services may experience in the community (Mezey & Eastman, 2009).

People who use mental health services in the community may typically prefer to live alone or with family (R. Warren & Bell, 2000), however, as with inpatient settings, people living in supported accommodation are not able to choose their coresidents. For people with severe and enduring mental health difficulties, improved perceptions of quality of life may also be associated with housing stability (Kyle & Dunn, 2008), or living in private accommodation as opposed to hostels or boarding houses (Browne & Courtney, 2004; Tanzman, 1993). Consequently, perceptions of housing and domestic space are pertinent to the 'recovery' experiences of people using forensic mental health services, who are typically marginalised and may commonly experience social exclusion or discrimination in relation to their mental health needs or offences (Adshead, 2000; Menditto, 2002; West et al., 2014; Williams et al., 2011). Reasons for this include widespread stereotyping and media portrayals of people with mental health difficulties as 'dangerous' or 'violent', alongside overlapping stereotypes about offenders (Perlin, 2008; West et al., 2014). It is envisaged that future research examining occupants' experiences of high-support housing settings could increase understandings of the environment's role as a material resource and how the design of spaces for supported living may be implicated in processes of rehabilitation (Chesters et al., 2005; Johnston et al., 2022).

In summary, this thesis has been undertaken in the context of limited research exploring patients' perspectives about the everyday experience of secure mental healthcare services (Coffey, 2006). Alongside this, studies examining the relationships between the built environment in psychiatric settings and patient outcomes are underrepresented within the literature (Liddicoat et al., 2020; Papoulias et al., 2014). Simonsen et al. (2022) also highlight that whilst the concept of 'healing architecture' has developed over recent years in relation to the spatial organisation of healthcare environments (see Frandsen et al., 2012; Lawson, 2010), there is at present no agreed definition and knowledge of how healing architecture can shape clinical and patient

outcomes is limited. There is also currently only a limited transdisciplinary evidence base informing the design of secure psychiatric facilities (Mclaughlan et al., 2021). Correspondingly, Shepley and Watson (2013) found that only 2% of 269 articles reviewed in an exploration of collaborative research relating to healthcare design represented collaborations between designers and healthcare professionals.

The present research has aimed to integrate the expert experience and knowledge of patients and staff in order to increase empirical understandings of how the everyday environment might mediate occupants' lived experiences in low and medium secure facilities. By exploring the narratives of staff and patients as cooccupants of secure psychiatric environments, it has also been possible to identify important similarities and differences in the lived experiences of both groups within these settings. The analysis of participants' stories highlights how spaces and objects within secure environments can be experienced by patients and staff members as supportive or challenging and sometimes both simultaneously. Increased understanding of how everyday spaces are experienced by staff, patients and people who use forensic mental healthcare services may help inform the development of service provision. Furthermore, research findings could assist in mitigating issues associated with the commonly conflicting expectations and priorities of patients and staff in relation to the environment in psychiatric settings (Curtis et al., 2013; Papoulias et al., 2014; Veale et al., 2022).

The stories explored in this study have drawn attention to the active role of the built environment in shaping occupants' everyday experiences in secure mental healthcare facilities. The data analysis highlights how the spatial and sensorial characteristics of the environment are interwoven with staff and patients' embodied psychosocial experiences in these settings. Participants' narratives point to the affective significance of detail within the environment, including the aesthetic and haptic qualities of architectural finishes or touchpoints such as fixtures and fittings, in

addition to the social and atmospheric attributes of spaces. Accordingly, the smaller scale details of spaces, including joinery elements, furniture and sanitaryware, are understood to be just as salient to occupants as the larger scale architectural aspects, such as the walls or volumes that form the overall environment. Because the internal spaces in secure psychiatric facilities are typically sparse, the architectural components and products that comprise the interior can be seen to have increased prominence in these settings than in more familiar, or less minimally furnished environments that have a more layered or textured ambience. Furthermore, the findings highlight the significance of imposed exposure to the environment's physical and atmospheric qualities, in these restrictive settings where staff typically work long shifts and patients spend long periods of unstructured time within a very limited number of spaces.

Correspondingly, throughout the data set, the environmental context was seen to be closely associated with staff and patients' experiences of both physical and psychological comfort and conversely, with their experiences of discomfort and distress. The narratives accordingly suggest that design and spatial planning opportunities to help mitigate experiences of discomfort could include paying careful attention to acoustics and materiality, alongside creating environments with a clear functional identity and a congruent match between the scale and function of the space. The findings additionally point to a significant need for aspirational and creative approaches to de-institutionalising the appearance and experiential qualities of the furniture and fittings typically produced for psychiatric settings, that were found to 'speak' unhelpfully to occupants through their form, detailing or materiality. The stories told by patients and staff also highlight varied ways in which the design of the environment mediates communication processes and social relationships within the institution. Both the spatial arrangement and the physical composition of the built environment are described as having the capacity to block or enable the spatial and communicative 'flows', that are in turn linked to the fluidity and quality of relationships

or social interaction between and amongst patients and staff. Participants' narratives suggest that material barriers such as glazed nursing station enclosures can impede such flows, whilst a greater sense of ease or improvements in occupants' relational experiences might be promoted in spaces where physical or symbolic barriers are diminished.

Whilst the insights gained contribute to the literature concerned with psychosocial studies of health and space, the study demonstrates a need for ongoing empirical examination of the complex relationships between the spatial, affective and relational dimensions of patient and staff experiences within secure mental healthcare environments. There is also a need for transdisciplinary collaborations as an important means to accelerate spatially informed and novel research regarding space and place in relation to health and care (Roxberg et al., 2020). Collaborative research and practice exploring the experiences of staff, patients and people who interact with mental healthcare services will continue to expand understandings of the interplay between health, space and environmental design.

References

- Abad, V. C., & Guilleminault, C. (2005). Sleep and psychiatry. *Dialogues in Clinical Neuroscience*, 7(4), 291–303.
- Abernethy, H. (2010). The assessment and treatment of sensory defensiveness in adult mental health: A literature review. *British Journal of Occupational Therapy*, *73*(5), 210–218. https://doi.org/10.4276/030802210X12734991664183
- Adshead, G. (2000). Care or custody? Ethical dilemmas in forensic psychiatry. *Journal of Medical Ethics*, *26*(5), 302–304. https://doi.org/10.1136/jme.26.5.302
- Adshead, G. (2011). The life sentence: Using a narrative approach in group psychotherapy with offenders. *Group Analysis*, 44(2), 175–195. https://doi.org/10.1177/0533316411400969
- Adshead, G. (2012). Their dark materials: Narratives and recovery in forensic practice. *Royal College of Psychiatrists Publication Archive*. www.rcpsych.ac.uk/pdf/Their Dark Materials. narratives and recovery in Forensic Practice Gwen Adshead.x.pdf (accessed 7 October 2015).
- Andes, M., & Shattell, M. M. (2006). An exploration of the meanings of space and place in acute psychiatric care. *Issues in Mental Health Nursing*, 27(6), 699–707. https://doi.org/10.1080/01612840600643057
- Anex, A., Dürrigl, M., Matthys, A., Felber, S., Medvedeva, T., Cleary, R., & Clesse, C. (2022). Guidelines, policies, and recommendations regarding the sexuality of individuals with severe mental disorders in psychiatric units, institutions, and supported housing across europe: A systematic review. *Archives of Sexual Behavior*. https://doi.org/10.1007/s10508-022-02430-4
- Askola, R., Nikkonen, M., Paavilainen, E., Soininen, P., Putkonen, H., & Louheranta, O. (2016). Forensic psychiatric patients' perspectives on their care: A narrative view. *Perspectives in Psychiatric Care*, *54*(1), 64–73. https://doi.org/10.1111/ppc.12201

Baker, F., & Douglas, C. (1990). Housing environments and community adjustment of

severely mentally ill persons. *Community Mental Health Journal*, *26*(6), 497–505. https://doi.org/10.1007/BF00752454

- Barton, J., Griffin, M., & Pretty, J. (2012). Exercise-, nature- and socially interactivebased initiatives improve mood and self-esteem in the clinical population. *Perspectives in Public Health*, 132(2), 89–96. https://doi.org/10.1177/1757913910393862
- Barton, J., & Pretty, J. (2010). What is the best dose of nature and green exercise for improving mental health? A multi-study analysis. *Environmental Science and Technology*, 44(10), 3947–3955. https://doi.org/10.1021/es903183r
- Bate, P., & Robert, G. (2007). Toward more user-centric OD: Lessons from the field of experience-based design and a case study. *Journal of Applied Behavioral Science*, 43(1), 41–66. https://doi.org/10.1177/0021886306297014
- Beauchemin, K. M., & Hays, P. (1996). Sunny hospital rooms expedite recovery from severe and refractory depressions. *Journal of Affective Disorders*, 40(1–2), 49–51. https://doi.org/10.1016/0165-0327(96)00040-7
- Beech, N. (2011). Liminality and the practices of identity reconstruction. *Human Relations*, 64(2), 285–302. https://doi.org/10.1177/0018726710371235
- Belk, R. W. (1988). Possessions and the extended self. *The Journal of Consumer Research*, *15*(2), 139–168.
- Benedetti, F., Colombo, C., Barbini, B., Campori, E., & Smeraldi, E. (2001). Morning sunlight reduces length of hospitalization in bipolar depression. *Journal of Affective Disorders*, 62(3), 221–223. https://doi.org/10.1016/S0165-0327(00)00149-X
- Bengtsson-Tops, A., Ericsson, U., & Ehliasson, K. (2014). Living in supportive housing for people with serious mental illness: A paradoxical everyday life. *International Journal of Mental Health Nursing*, 23(5), 409–418. https://doi.org/10.1111/inm.12072

Beresford, P. (2002). User involvement in research and evaluation: Liberation or regulation? *Social Policy and Society*, 1(02), 95–105. https://doi.org/10.1017/S1474746402000222

Berto, R. (2014). The role of nature in coping with psycho-physiological stress: A literature review on restorativeness. *Behavioral Sciences*, 4(4), 394–409. https://doi.org/10.3390/bs4040394

- Boden, Z., & Eatough, V. (2014). Understanding more fully: A multimodal hermeneuticphenomenological approach. *Qualitative Research in Psychology*, 11(2), 160–177. https://doi.org/10.1080/14780887.2013.853854
- Borckardt, J. J., Madan, A., Grubaugh, A. L., Danielson, C. K., Pelic, C. G., Hardesty, S. J.,
 Hanson, R., Herbert, J., Cooney, H., Benson, A., & Frueh, B. C. (2011). Systematic
 investigation of initiatives to reduce seclusion and restraint in a state psychiatric
 hospital. *Psychiatric Services*, *62*(5), 477–483.
 https://doi.org/10.1176/appi.ps.62.5.477
- Borg, M., & Davidson, L. (2008). The nature of recovery as lived in everyday experience. *Journal of Mental Health*, *17*(2), 129–140.

https://doi.org/10.1080/09638230701498382

Borg, M., Sells, D., Topor, A., Mezzina, R., Marin, I., & Davidson, L. (2005). What makes a house a home: The role of material resources in recovery from severe mental illness. *American Journal of Psychiatric Rehabilitation*, 8(3), 243–256. https://doi.org/10.1080/15487760500339394

Boult, M. (2017). Moral therapy and interior decor in Victorian lunatic asylums. Australian and New Zealand Society Medical History: Health , Medicine and Society: Challenge and Change , Melbourne , July 2017.

https://www.researchgate.net/publication/331561498_Moral_therapy_and_inter ior_decor_in_Victorian_lunatic_asylums

Bratman, G. N., Hamilton, J. P., & Daily, G. C. (2012). The impacts of nature experience

on human cognitive function and mental health. *Annals of the New York Academy of Sciences*, 1249(1), 118–136. https://doi.org/10.1111/j.1749-

6632.2011.06400.x

- Breheny, M., & Stephens, C. (2011). The bonds and burdens of family life: Using narrative analysis to understand difficult relationships. *Narrative Works*, 1(2), 34– 51.
- Bressington, D., Stewart, B., Beer, D., & MacInnes, D. (2011). Levels of service user satisfaction in secure settings A survey of the association between perceived social climate, perceived therapeutic relationship and satisfaction with forensic services. *International Journal of Nursing Studies*, *48*(11), 1349–1356. https://doi.org/10.1016/j.ijnurstu.2011.05.011
- Brooker, C., Sirdifield, C., & Gojkovic, D. (2007). *Mental health services and prisoners: An updated review*.

http://eprints.lincoln.ac.uk/id/eprint/2523/1/Systematic_Review_(FINAL).pdf

- Brown, D. J. (2009). Designing an effective nurses' station: Staff need both on- and offstage areas. *Behavioral Healthcare*, *29*(10), 22–24.
- Brown, J., Fawzi, W., Shah, A., Joyce, M., Holt, G., McCarthy, C., Stevenson, C., Marange, R., Shakes, J., & Solomon-Ayeh, K. (2016). Low stimulus environments: reducing noise levels in continuing care. *BMJ Quality Improvement Reports*, 5(1). https://doi.org/10.1136/bmjquality.u207447.w4214

Brown, S. D. (2012). Memory and mathesis: For a topological approach to psychology. *Theory, Culture & Society*, 29(4–5), 137–164. https://doi.org/10.1177/0263276412448830

Brown, S. D., Kanyeredzi, A., McGrath, L., Reavey, P., & Tucker, I. (2019a). Affect theory and the concept of atmosphere. *Distinktion: Journal of Social Theory*, 20(1), 5–24. https://doi.org/10.1080/1600910X.2019.1586740

Brown, S. D., Kanyeredzi, A., McGrath, L., Reavey, P., & Tucker, I. (2019b). Organizing

the sensory: Ear-work, panauralism and sonic agency on a forensic psychiatric unit. *Human Relations*, 1–26. https://doi.org/10.1177/0018726719874850

- Brown, S. D., & Reavey, P. (2014). Vital memories: Movements in and between affect, ethics and self. *Memory Studies*, 7(3), 328–338. https://doi.org/10.1177/1750698014530622
- Brown, S. D., & Reavey, P. (2015). *Vital memory and affect: Living with a difficult past.* Routledge.
- Brown, S. D., Reavey, P., Kanyeredzi, A., & Batty, R. (2014). Transformations of self and sexuality: Psychologically modified experiences in the context of forensic mental health. *Health*, *18*(3), 240–260. https://doi.org/10.1177/1363459313497606
- Brown, S. D., & Stenner, P. (2009). *Psychology without foundations: History, philosophy and psychosocial theory*. SAGE Publications Limited.
- Browne, G., & Courtney, M. (2004). Measuring the impact of housing on people with schizophrenia. *Nursing and Health Sciences*, 6(1), 37–44. https://doi.org/10.1111/j.1442-2018.2003.00172.x
- Brunt, D., & Rask, M. (2005). Patient and staff perceptions of the ward atmosphere in a
 Swedish maximum-security forensic psychiatric hospital. *Journal of Forensic Psychiatry and Psychology*, 16(2), 263–276.

https://doi.org/10.1080/1478994042000270238

Burr, V. (2003). Social Constructionism (2nd ed.). Routledge.

- Bury, M. (1982). Chronic illness as biographical disruption. Sociology of Health and Illness, 4(2), 167–182. https://doi.org/https://doi.org/10.1111/1467-9566.ep11339939
- Busch-Vishniac, I. J., West, J. E., Barnhill, C., Hunter, T., Orellana, D., & Chivukula, R. (2005). Noise levels in Johns Hopkins Hospital. *The Journal of the Acoustical Society of America*, *118*(6), 3629–3645. https://doi.org/10.1121/1.2118327

Butterfield, A., & Martin, D. (2016). Affective sanctuaries: understanding Maggie's as

therapeutic landscapes. *Landscape Research*, *41*(6), 695–706. https://doi.org/10.1080/01426397.2016.1197386

Champagne, T., & Stromberg, N. (2004). Sensory approaches in inpatient psychiatric settings: Innovative alternatives to seclusion and restraint. *Journal of Psychosocial Nursing*, 42(9), 34–44. https://doi.org/10.3928/02793695-20040901-06

Cherner, R., Aubry, T., Ecker, J., Kerman, N., & Nandlal, J. (2014). Transitioning into the community: Outcomes of a pilot housing program for forensic patients. *International Journal of Forensic Mental Health*, *13*(1), 62–74. https://doi.org/10.1080/14999013.2014.885472

- Chesters, J., Fletcher, M., & Jones, R. (2005). Mental illness recovery and place. *Australian E-Journal for the Advancement of Mental Health*, 4(2), 89–97. https://doi.org/10.5172/jamh.4.2.89
- Chevalier, A., Ntala, E., Fung, C., Priebe, S., & Bird, V. J. (2018). Exploring the initial experience of hospitalisation to an acute psychiatric ward. *PLoS ONE*, *13*(9), 1–17. https://doi.org/10.1371/journal.pone.0203457
- Chilvers, R., Macdonald, G., & Hayes, A. (2006). Supported housing for people with severe mental disorders. *Cochrane Database of Systematic Reviews*, 4. https://doi.org/10.1002/14651858.CD000453.pub2
- Choiniere, D. B. (2010). The effects of hospital noise. *Nursing Administration Quarterly*, 34(4), 327–333. https://doi.org/10.1097/NAQ.0b013e3181f563db
- Chow, W. S., & Priebe, S. (2016). How has the extent of institutional mental healthcare changed in Western Europe? Analysis of data since 1990. *BMJ Open*, 6(e010188), 1–7. https://doi.org/10.1136/bmjopen-2015-010188
- Clark, J. A., & Mishler, E. G. (1992). Attending to patients' stories: reframing the clinical task. *Sociology of Health & Illness*, *14*(3), 344–372. https://doi.org/10.1111/1467-9566.ep11357498

Clarke, C., Lumbard, D., Sambrook, S., & Kerr, K. (2016). What does recovery mean to a

forensic mental health patient? A systematic review and narrative synthesis of the qualitative literature. *Journal of Forensic Psychiatry and Psychology*, *27*(1), 38–54. https://doi.org/10.1080/14789949.2015.1102311

Clatworthy, J., Hinds, J., & Camic, P. M. (2013). Gardening as a mental health intervention: a review. *Mental Health Review Journal*, *18*(4), 214–225. https://doi.org/10.1108/MHRJ-02-2013-0007

Clough, P. T. (2008). The affective turn: Political economy, biomedia and bodies. *Theory, Culture & Society*, *25*(1), 1–22.

https://doi.org/10.1177/0263276407085156

- Coffey, M. (2006). Researching service user views in forensic mental health: A literature review. *Journal of Forensic Psychiatry & Psychology*, *17*(1), 73–107. https://doi.org/10.1080/14789940500431544
- Coffey, M. (2012a). A risk worth taking? Value differences and alternative risk constructions in accounts given by patients and their community workers following conditional discharge from forensic mental health services. *Health, Risk* & Society, 14(5), 465–482. https://doi.org/10.1080/13698575.2012.682976
- Coffey, M. (2012b). Negotiating identity transition when leaving forensic hospitals. *Health:*, *16*(5), 489–506. https://doi.org/10.1177/1363459311434649
- Coid, J. W. (1992). DSM-III diagnosis in criminal psychopaths: a way forward. *Criminal Behaviour and Mental Health*, *2*, 78–94.
- Coid, J. W., Kahtan, N., Gault, S., Cook, A., & Jarman, B. (2001). Medium secure forensic psychiatry services: Comparison of seven English health regions. *British Journal of Psychiatry*, 178(1), 55–61. https://doi.org/10.1192/bjp.178.1.55
- Collins, J., Ellsworth, R., Casey, N., Hyer, L., Hickey, R., Schoonover, R., Twemlow, S. W., & Nesselroade, J. R. (1985). Treatment characteristics of psychiatric programs that correlate with patient community adjustment. *Journal of Clincal Psychology*, 41(3), 299–308. https://doi.org/10.1002/1097-4679(198505)41:3<299::AID-</p>

JCLP2270410302>3.0.CO;2-A

- Connellan, K., Due, C., & Riggs, D. (2011). Gardens of the mind: Nature, power and design for mental health. *Diversity and Unity: Proceedings of IASDR2011, the 4th World Conference on Design Research*, 2–13.
- Connellan, K., Gaardboe, M., Riggs, D., Due, C., Reinschmidt, A., & Mustillo, L. (2013). Stressed spaces: Mental health and architecture. *HERD: Health Environments Research & Design Journal*, 6(4), 127–168. https://doi.org/10.1177/193758671300600408
- Connellan, K., Riggs, D. W., & Due, C. (2015). Light lies: How glass speaks. *Communication Design Quarterly*, 3(4), 15–24. https://doi.org/10.1145/2826972.2826974
- Costa, D. M., Morra, J., Solomon, D., Sabino, M., & Call, K. (2006). Snoezelen and sensorybased treatment for adults with psychiatric disorders. *OT Practice*, *11*(4), 19–23.
- Cromby, J., Harper, D. J., & Reavey, P. (2013). *Psychology, mental health and distress*. Palgrave Macmillan.
- Crossley, M. (2000a). *Introducing narrative psychology: Self, trauma and the construction of meaning*. Open University Press.
- Crossley, M. (2000b). Narrative psychology, trauma and the study of self/identity. *Theory & Psychology*, *10*(4), 527–546.

https://doi.org/10.1177/0959354300104005

- Csipke, E., Papoulias, C., Vitoratou, S., Williams, P., Rose, D., & Wykes, T. (2016). Design in mind: eliciting service user and frontline staff perspectives on psychiatric ward design through participatory methods. *Journal of Mental Health*, 25(2), 114–121. https://doi.org/10.3109/09638237.2016.1139061
- Cummings, K. S., Grandfield, S. A., & Coldwell, C. M. (2010). Caring with comfort rooms: Reducing seclusion and restraint use in psychiatric facilities. *Journal of Psychosocial Nursing and Mental Health Services*, 48(6), 26–30.

https://doi.org/10.3928/02793695-20100303-02

- Curtis, S. E., Gesler, W., Fabian, K., Francis, S., & Priebe, S. (2007). Therapeutic landscapes in hospital design: a qualitative assessment by staff and service users of the design of a new mental health inpatient unit. *Environment and Planning C: Government and Policy*, 25(4), 591–610. https://doi.org/10.1068/c1312r
- Curtis, S. E., Gesler, W., Wood, V., Spencer, I., Mason, J., Close, H., & Reilly, J. (2013).
 Compassionate containment? Balancing technical safety and therapy in the design of psychiatric wards. *Social Science and Medicine*, *97*, 201–209.
 https://doi.org/10.1016/j.socscimed.2013.06.015
- Dale, K., & Burrell, G. (2008). *The spaces of organization and the organization of space: Power, identity and materiality at work*. Palgrave Macmillan.
- Dalke, H., Little, J., Niemann, E., Camgoz, N., Steadman, G., Hill, S., & Stott, L. (2006).
 Colour and lighting in hospital design. *Optics and Laser Technology*, *38*(4–6), 343–365. https://doi.org/10.1016/j.optlastec.2005.06.040
- Dalke, H., Littlefair, P. J., Loe, D. L., & Camög, N. (2004). *Lighting and colour for hospital design: A report on an NHS estates funded research project*. The Stationery Office.
- De Hert, M., Detraux, J., van Winkel, R., Yu, W., & Correll, C. U. (2012). Metabolic and cardiovascular adverse effects associated with antipsychotic drugs. *Nature Reviews Endocrinology*, 8(2), 114–126. https://doi.org/10.1038/nrendo.2011.156
- de Niet, G. J., Tiemens, B. G., Lendemeijer, H. H. G. M., & Hutschemaekers, G. J. M. (2008). Perceived sleep quality of psychiatric patients. *Journal of Psychiatric and Mental Health Nursing*, *15*(6), 465–470. https://doi.org/10.1111/j.1365-2850.2008.01250.x
- Dein, K., & Williams, P. S. (2008). Relationships between residents in secure psychiatric units: are safety and sensitivity really incompatible? *Psychiatric Bulletin*, 32(8), 284–287. https://doi.org/10.1192/pb.bp.106.011478

Del Busso, L. (2011). Using photographs to explore the embodiment of pleasure in

everyday life. In P. Reavey (Ed.), *Visual methods in psychology* (pp. 43–54). Routledge.

- Department of Health. (2008). Improving the patient experience: Sharing success in mental health and learning disabilities: The King's Fund's Enhancing the Healing Environment programme 2004-2008. TSO.
- Department of Health. (2011a). Environmental Design Guide: Adult medium secure services. *Department of Health*.

http://www.dh.gov.uk/prod_consum_dh/groups/dh_digitalassets/documents/di gitalasset/dh_126177.pdf

- Department of Health. (2011b). *No health without mental health: A cross-government mental health outcomes strategy for people of all ages*. Department of Health. https://doi.org/10.1016/S0140-6736(07)61238-0
- Devlin, A. S., & Andrade, C. C. (2017). Quality of the hospital experience: Impact of the physical environment. In G. Fleury-Bahi, E. Pol, & O. Navarro (Eds.), *Handbook of environmental psychology and quality of life research* (pp. 421–440). Springer International Publishing. https://doi.org/10.1007/978-3-319-31416-7_23
- Devlin, A. S., & Arneill, A. B. (2003). Health care environments and patient outcomes: A review of the literature. *Environment and Behavior*, 35(5), 665–694. https://doi.org/10.1177/0013916503255102

Doise, W. (1986). Levels of explanation in social psychology. Cambridge University Press.

- Donald, F., Duff, C., Lee, S., Kroschel, J., & Kulkarni, J. (2015). Consumer perspectives on the therapeutic value of a psychiatric environment. *Journal of Mental Health*, *24*(2), 63–67. https://doi.org/10.3109/09638237.2014.954692
- Dorvil, H., Morin, P., Beaulieu, A., & Robert, D. (2005). Housing as a social integration factor for people classified as mentally ill. *Housing Studies*, *20*(3), 497–519. https://doi.org/10.1080/02673030500062525

Doucette, P. A. (2004). Walk and talk: An intervention for behaviorally challenged

youths. Adolescence, 39(154), 373-388.

- Doughty, K. (2013). Walking together: The embodied and mobile production of a therapeutic landscape. *Health and Place, 24,* 140–146. https://doi.org/10.1016/j.healthplace.2013.08.009
- Douglas, C. H., & Douglas, M. R. (2005). Patient-centred improvements in health-care built environments: perspectives and design indicators. *Health Expectations*, 8(3), 264–276. https://doi.org/10.1111/j.1369-7625.2005.00336.x
- Drennan, G., Wooldridge, J., Aiyegbusi, A., Alred, D., Ayres, J., Barker, R., Carr, S., Eunson,
 H., Lomas, H., Moore, E., Stanton, D., & Shepherd, G. (2014). Making Recovery a
 Reality in Forensic Settings. In *[Imroc Briefing paper 10]* (pp. 1–28). Cente for
 Mental Health and Mental Health Network NHS Confederation.
- du Plessis, R. (2012). The influence of moral therapy on the landscape design of lunatic asylums built in the nineteenth century. *De Arte, 47*(86), 19–38. https://doi.org/10.1080/00043389.2012.11877170
- Duque, M., Pink, S., Sumartojo, S., & Vaughan, L. (2019). Homeliness in health care: The role of everyday designing. *Home Cultures*, 16(3), 213–232. https://doi.org/10.1080/17406315.2020.1757381
- Durcan, G., Hoare, T., & Cumming, I. (2011). Pathways to unlocking secure mental health care. Centre for Mental Health. https://www.centreformentalhealth.org.uk/sites/default/files/2018-

09/Pathways_to_unlocking_secure_mental_health_care.pdf

- Duxbury, J., & Whittington, R. (2005). Causes and management of patient aggression and violence: staff and patient perspectives. *Journal of Advanced Nursing*, *50*(5), 469–478. https://doi.org/10.1111/j.1365-2648.2005.03426.x
- Dvoskin, J. A., Radomski, S. J., Bennett, C., Olin, J. A., Hawkins, R. L., Dotson, L. A., & Drewnicky, I. N. (2002). Architectural design of a secure forensic state psychiatric hospital. *Behavioral Sciences & the Law*, *20*(5), 481–493.

https://doi.org/10.1002/bsl.506

- Edginton, B. (1997). Moral architecture: the influence of the York Retreat on asylum design. *Health and Place*, *3*(2), 91–99. https://doi.org/10.1016/S1353-8292(97)00003-8
- Edginton, B. (2003). The design of moral architecture at the York Retreat. *Journal of Design History*, *16*(2), 103–117.
- Edwards, J., & Hults, M. S. (1970). Open nursing stations on psychiatric wards. *Perspectives in Psychiatric Care*, 8(5), 209–217. https://doi.org/10.1111/j.1744-6163.1970.tb01521.x
- Esin, C. (2011). Narrative analysis approaches. In N. Frost (Ed.), *Qualitative research methods in psychology: Combining core approaches* (pp. 92–117). Open University Press.
- Esin, C., Fathi, M., & Squire, C. (2014). Narrative analysis: The constructionist approach. In *The SAGE Handbook of Qualitative Data Analysis* (pp. 203–216). SAGE Publications Limited. https://doi.org/10.4135/9781446282243.n14
- Evans, G. W. (2003). The built environment and mental health. *Journal of Urban Health*, *80*(4), 536–555. https://doi.org/10.1093/jurban/jtg063
- Evans, K., Murray, C. D., Jellicoe-Jones, L., & Smith, I. (2012). Support staffs' experiences of relationship formation and development in secure mental health services. *British Journal of Forensic Practice*, *14*(2), 104–115. https://doi.org/10.1108/14636641211223666

Every-Palmer, S., Huthwaite, M. A., Elmslie, J. L., Grant, E., & Romans, S. E. (2018). Longterm psychiatric inpatients' perspectives on weight gain, body satisfaction, diet and physical activity: a mixed methods study. *BMC Psychiatry*, 18(300), 1–9. https://doi.org/10.1186/s12888-018-1878-5

Farnworth, L., Nikitin, L., & Fossey, E. (2004). Being in a secure forensic psychiatric unit: Every day is the same, killing time or making the most of it. *British Journal of*

Occupational Therapy, 67(10), 430–438.

https://doi.org/10.1177/030802260406701003

- Faulkner, A. (2007). Beyond Our Expectations: A report of involving service users in forensic mental health research. https://www.invo.org.uk/wpcontent/uploads/documents/Faulkner 2007 Beyond our expectations.pdf
- Fennelly, K. (2014). Out of sound, out of mind: noise control in early nineteenthcentury lunatic asylums in England and Ireland. *World Archaeology*, 46(3), 416– 430. https://doi.org/10.1080/00438243.2014.909098
- Fenton, K., Larkin, M., Boden, Z. V. R., Thompson, J., Hickman, G., & Newton, E. (2014). The experiential impact of hospitalisation in early psychosis: Service-user accounts of inpatient environments. *Health and Place*, *30*, 234–241. https://doi.org/10.1016/j.healthplace.2014.09.013
- Foley, R., & Kistemann, T. (2015). Blue space geographies: Enabling health in place. *Health and Place*, *35*, 157–165.

https://doi.org/10.1016/j.healthplace.2015.07.003

- Foster, R. G., & Wulff, K. (2005). The rhythm of rest and excess. *Nature Reviews Neuroscience*, 6(5), 407–414. https://doi.org/10.1038/nrn1670
- Foucault, M. (1991). *Discipline and punish: The birth of the prison*. (A. Sheridan, Trans.) Penguin (Original work published 1975).
- Frandsen, A. K., Gottlieb, S. C., & Harty, C. (2012). Spatial configurations of healthcare practices. In N. Thurairajah (Ed.), *Proceedings of the joint CIB International Conference: Management of construction: Research to practice.* (pp. 1062–1073).
 Birmingham School of the Built Environment, Birmingham City University.
- Frank, A. W. (1993). The rhetoric of self-change: Illness experience as narrative. *The Sociological Quarterly*, 34(1), 39–52. https://doi.org/10.1111/j.1533-8525.1993.tb00129.x

Frank, A. W. (1995). The wounded storyteller. Body, illness and ethics. The University of

Chicago Press.

- Frith, H., & Harcourt, D. (2007). Using photographs to capture women's experiences of chemotherapy: Reflecting on the method. *Qualitative Health Research*, 17(10), 1340–1350. https://doi.org/10.1177/1049732307308949
- Frost, N. (2009). "Do you know what I mean?": The use of a pluralistic narrative analysis approach in the interpretation of an interview. *Qualitative Research*, 9(1), 9–29. https://doi.org/10.1177/1468794108094867
- Galappathie, N., Khan, S. T., & Hussain, A. (2017). Civil and forensic patients in secure psychiatric settings: A comparison. *BJPsych Bulletin*, 41(3), 156–159. https://doi.org/10.1192/pb.bp.115.052910
- Gardner-Elahi, C., & Zamiri, S. (2015). Collective narrative practice in forensic mental health. *The Journal of Forensic Practice*, *17*(3), 204–218. https://doi.org/10.1108/JFP-10-2014-0034
- Gascon, M., Wilma, Z., Vert, C., White, M. P., & Nieuwenhuijsen, M. J. (2017). Outdoor blue spaces, human health and well-being: A systematic review of quantitative studies. *International Journal of Hygiene and Environmental Health*, 220(8), 1207– 1221. http://dx.doi.org/10.1016/j.ijheh.2017.08.004
- Gergen, K. J. (2004). Constructionism, social. In M. Lewis-Beck, A. Bryman, & T. Futing
 Liao (Eds.), *The SAGE Encyclopedia of Social Science Research Methods* (Vol. 1, pp. 183–185). SAGE Publications, Inc.

https://doi.org/10.4135/9781412950589.n164

- Gesler, W. (1992). Therapeutic landscapes: Medical issues in light of the new cultural geography. *Social Science and Medicine*, 34(7), 735–746. https://doi.org/10.1016/0277-9536(92)90360-3
- Gesler, W. (1993). Therapeutic landscapes: theory and a case study of Epidauros, Greece. *Environment and Planning D: Society and Space*, *11*(2), 171–189. https://doi.org/10.1068/d110171

- Gesler, W. (2005). Therapeutic landscapes: An evolving theme. *Health & Place*, *11*(4), 295–297. https://doi.org/10.1016/j.healthplace.2005.02.003
- Gesler, W., Bell, M., Curtis, S. E., Hubbard, P., & Francis, S. (2004). Therapy by design:
 evaluating the UK hospital building program. *Health & Place*, *10*(2), 117–128.
 https://doi.org/10.1016/S1353-8292(03)00052-2
- Gilburt, H., Rose, D., & Slade, M. (2008). The importance of relationships in mental health care: A qualitative study of service users' experiences of psychiatric hospital admission in the UK. *BMC Health Services Research*, 8(92). https://doi.org/10.1186/1472-6963-8-92
- Goffman, E. (1991). *Asylums: Essays on the social situation of mental patients and other inmates*. Penguin. (Original work published 1961).
- Golembiewski, J. A. (2013). Lost in space: the place of architectural milieu in the aetiology and treatment of schizophrenia. *Facilities*, *31*(9/10), 427–448. https://doi.org/10.1108/02632771311324981
- Golembiewski, J. A. (2015). Mental health facility design: The case for person-centred care. Australian and New Zealand Journal of Psychiatry, 49(3), 203–206. https://doi.org/10.1177/0004867414565477
- Goodings, L., & Tucker, I. (2019). Social media and mental health. In L. McGrath & P. Reavey (Eds.), *The handbook of mental health and space* (pp. 200–213). Routledge. https://doi.org/10.4324/9781315620312-13
- Greco, M., & Stenner, P. (2017). From paradox to pattern shift: Conceptualising liminal hotspots and their affective dynamics. *Theory and Psychology*, *27*(2), 147–166. https://doi.org/10.1177/0959354317693120
- Haigh, R., Harrison, T., Johnson, R., Paget, S., & Williams, S. (2012). Psychologically informed environments and the "Enabling Environments" initiative. *Housing, Care* and Support, 15(1), 34–42. https://doi.org/10.1108/14608791211238412

Hare Duke, L., Furtado, V., Guo, B., & Völlm, B. A. (2018). Long-stay in forensic-

psychiatric care in the UK. *Social Psychiatry and Psychiatric Epidemiology*, *53*(3), 313–321. https://doi.org/10.1007/s00127-017-1473-y

- Hartig, T., Mitchell, R., de Vries, S., & Frumkin, H. (2014). Nature and health. Annual Review of Public Health, 35, 207–228. https://doi.org/10.1146/annurevpublhealth-032013-182443
- Heerwagen, J. (2009). Biophilia, health and well-being. In L. Campbell & A. Weisen
 (Eds.), *Restorative commons: Creating health and well-being through urban landscapes. General technical report NRS-P-39*. USDA Forest Service.
- Hickman, C. (2009). Cheerful prospects and tranquil restoration: the visual experience of landscape as part of the therapeutic regime of the British asylum, 1800-60. *History of Psychiatry*, 20(4), 425–441.

https://doi.org/10.1177/0957154X08338335

- Hickman, C. (2014). Cheerfulness and tranquility: gardens in the Victorian asylum. *The Lancet Psychiatry*, 1(7), 506–507. https://doi.org/10.1016/S2215-0366(14)00098-4
- Hobday, R. A., & Dancer, S. J. (2013). Roles of sunlight and natural ventilation for controlling infection: historical and current perspectives. *Journal of Hospital Infection*, 84(4), 271–282. https://doi.org/10.1016/j.jhin.2013.04.011
- Holahan, C. J., & Saegert, S. (1973). Behavioral and attitudinal effects of large-scale variation in the physical environment of psychiatric wards. *Journal of Abnormal Psychology*, 82(3), 454–462. https://doi.org/10.1037/h0035391
- Holley, J., Weaver, T., & Völlm, B. (2020). The experience of long stay in high and medium secure psychiatric hospitals in England: qualitative study of the patient perspective. *International Journal of Mental Health Systems*, 14(1), 1–12. https://doi.org/10.1186/s13033-020-00358-7
- Holmberg, S. K., & Coon, S. (1999). Ambient sound levels in a state psychiatric hospital. *Archives of Psychiatric Nursing*, *13*(3), 117–126. https://doi.org/10.1016/S0883-

9417(99)80042-9

- Hsu, T., Ryherd, E. E., Waye, K. P., & Ackerman, J. (2012). Noise pollution in hospitals: Impact on patients. *Journal of Clinical Outcomes Management*, *19*(7), 301–309.
- Hunt, J. M., & Sine, D. M. (2017). Design guide for the built environment of behavioral health facilities. Facilities Guidelines Institute. https://www.fgiguidelines.org/
- Isobel, S., Foster, K., & Edwards, C. (2015). Developing family rooms in mental health inpatient units: an exploratory descriptive study. *BMC Health Services Research*, 15(238), 1–9. https://doi.org/10.1186/s12913-015-0914-0
- Isobel, S., Wilson, A., Gill, K., & Howe, D. (2021). 'What would a trauma-informed mental health service look like?' Perspectives of people who access services. *International Journal of Mental Health Nursing*, 30(2), 495–505. https://doi.org/10.1111/inm.12813
- Iyendo, T. O. (2016). Exploring the effect of sound and music on health in hospital settings: A narrative review. *International Journal of Nursing Studies*, 63, 82–100. https://doi.org/10.1016/j.ijnurstu.2016.08.008
- Iyendo, T. O., Uwajeh, P. C., & Ikenna, E. S. (2016). The therapeutic impacts of environmental design interventions on wellness in clinical settings: A narrative review. *Complementary Therapies in Clinical Practice*, 24, 174–188. https://doi.org/10.1016/j.ctcp.2016.06.008
- Jacob, J. D., Gagnon, M., & Holmes, D. (2009). Nursing so-called monsters: On the importance of abjection and fear in forensic psychiatric nursing. *Journal of Forensic Nursing*, 5(3), 153–161. https://doi.org/10.1111/j.1939-3938.2009.01048.x

Jansman-Hart, E. M., Seto, M. C., Crocker, A. G., Nicholls, T. L., & Côté, G. (2011). International trends in demand for forensic mental health services. *International Journal of Forensic Mental Health*, *10*(4), 326–336. https://doi.org/10.1080/14999013.2011.625591

- Jewkes, Y., Jordan, M., Wright, S., & Bendelow, G. (2019). Designing 'healthy' prisons for women: Incorporating trauma-informed care and practice (TICP) into prison planning and design. *International Journal of Environmental Research and Public Health*, 16(20), 1–15. https://doi.org/10.3390/ijerph16203818
- Johnson, K. (2011). Visualising mental health with an LGBT community group: Method, process, theory. In P. Reavey (Ed.), *Visual methods in psychology* (pp. 173–189). Routledge.
- Johnson, R., & Haigh, R. (2011). Social psychiatry and social policy for the twenty-first century: New concepts for new needs - the "Enabling Environments" initiative. *Mental Health and Social Inclusion*, 15(1), 17–23. https://doi.org/10.5042/mhsi.2011.0054

Johnston, A., Davidson, G., Webb, P., McCartan, N., McAllister, K., Broughton, R., Sutherland, D., & Kennedy, C. (2022). Physical design of supported accommodation for people with mental health problems and intellectual disabilities: A scoping review. *Architecture Media Politics Society Proceedings Series*. https://praxiscare.org/wp-content/uploads/2022/09/Design-of-Supported-Accomodation.pdf

Johnstone, A. (2004). Control and confinement: An archaeological review of modern mental health buildings. *Mental Health Review Journal*, 9(4), 29–31. https://doi.org/10.1108/13619322200400040

Jones, J., Nolan, P., Bowers, L., Simpson, A., Whittington, R., Hackney, D., & Bhui, K. (2010). Psychiatric wards: places of safety? *Journal of Psychiatric and Mental Health Nursing*, *17*(2), 124–130. https://doi.org/10.1111/j.1365-2850.2009.01482.x

Jordan, M. (2014). Moving beyond counselling and psychotherapy as it currently is – taking therapy outside. *European Journal of Psychotherapy and Counselling*, *16*(4), 361–375. https://doi.org/10.1080/13642537.2014.956773 Joseph, A. (2006). *The impact of light on outcomes in healthcare settings* (Issue August). The Center for Health Design. https://www.healthdesign.org/chd/knowledgerepository/impact-light-outcomes-healthcare-settings-1

Joseph, A., & Ulrich, R. (2007). Sound control for improved outcomes in healthcare settings. In *The Center for Health Design* (Issue 4, pp. 1–15). https://www.healthdesign.org/knowledge-repository/sound-control-improvedoutcomes-healthcare-settings

- Josselson, R. (2011). "Bet you think this song is about you": Whose narrative is it in narrative research? *Narrative Works: Issues, Investigations & Interventions, 1*(1), 33–51. https://id.erudit.org/iderudit/nm1_1art02
- Jovanović, N., Campbell, J., & Priebe, S. (2019). How to design psychiatric facilities to foster positive social interaction – A systematic review. *European Psychiatry*, *60*, 49–62. https://doi.org/10.1016/j.eurpsy.2019.04.005
- Kamphuis, J., Dijk, D. J., Spreen, M., & Lancel, M. (2014). The relation between poor sleep, impulsivity and aggression in forensic psychiatric patients. *Physiology and Behavior*, *123*, 168–173. https://doi.org/10.1016/j.physbeh.2013.10.015
- Kamphuis, J., Karsten, J., de Weerd, A., & Lancel, M. (2013). Sleep disturbances in a clinical forensic psychiatric population. *Sleep Medicine*, *14*(11), 1164–1169. https://doi.org/10.1016/j.sleep.2013.03.008
- Kamphuis, J., Meerlo, P., Koolhaas, J. M., & Lancel, M. (2012). Poor sleep as a potential causal factor in aggression and violence. *Sleep Medicine*, *13*(4), 327–334. https://doi.org/10.1016/j.sleep.2011.12.006

Kanyeredzi, A., Brown, S. D., McGrath, L., Reavey, P., & Tucker, I. (2019). The atmosphere of the ward: Attunements and attachments of everyday life for patients on a medium-secure forensic psychiatric unit. *The Sociological Review*, 67(2), 444–466. https://doi.org/10.1177/0038026119829751

Kaplan, R. (2001). The nature of the view from home: Psychological benefits.

Environment and Behavior, *33*(4), 507–542.

https://doi.org/10.1177/00139160121973115

- Kaplan, S. (1995). The restorative benefits of nature: Toward an integrative framework. *Journal of Environmental Psychology*, 15(3), 169–182. https://doi.org/10.1016/0272-4944(95)90001-2
- Karlin, B. E., & Zeiss, R. A. (2006). Environmental and therapeutic issues in psychiatric hospital design: Toward best practices. *Psychiatric Services*, 57(10), 1376–1378. https://doi.org/10.1176/ps.2006.57.10.1376
- Keats, P. (2009). Multiple text analysis in narrative research: visual, written, and spoken stories of experience. *Qualitative Research*, 9(2), 181–195. https://doi.org/10.1177/1468794108099320
- Kennedy, H. G. (2002). Therapeutic uses of security: mapping forensic mental health services by stratifying risk. *Advances in Psychiatric Treatment*, 8(6), 433–443. https://doi.org/10.1192/apt.8.6.433
- Killaspy, H. (2006). From the asylum to community care: learning from experience.British Medical Bulletin, 79–80(1), 245–258. https://doi.org/10.1093/bmb/ldl017
- Killaspy, H., Priebe, S., Bremner, S., McCrone, P., Dowling, S., Harrison, I., Krotofil, J., McPherson, P., Sandhu, S., Arbuthnott, M., Curtis, S. E., Leavey, G., Shepherd, G., Eldridge, S., & King, M. (2016). Quality of life, autonomy, satisfaction, and costs associated with mental health supported accommodation services in England: a national survey. *The Lancet Psychiatry*, *3*(12), 1129–1137. https://doi.org/10.1016/S2215-0366(16)30327-3
- Kirkman, M. (2002). What's the plot? Applying narrative theory to research in psychology. *Australian Psychologist*, 37(1), 30–38. https://doi.org/10.1080/00050060210001706646
- Kofoed, J., & Stenner, P. (2017). Suspended liminality: Vacillating affects in cyberbullying/research. *Theory and Psychology*, *27*(2), 167–182.

https://doi.org/10.1177/0959354317690455

- Koller, K., & Hantikainen, V. (2002). Privacy of patients in the forensic department of a psychiatric clinic: a phenomenological study. *Nursing Ethics*, 9(4), 347–360. https://doi.org/10.1191/0969733002ne520oa
- Krotofil, J., McPherson, P., & Killaspy, H. (2018). Service user experiences of specialist mental health supported accommodation: A systematic review of qualitative studies and narrative synthesis. *Health and Social Care in the Community*, 26(6), 787–800. https://doi.org/10.1111/hsc.12570
- Kumar, S., & Ng, B. (2001). Crowding and violence on psychiatric wards: Explanatory models. *Canadian Journal of Psychiatry*, 46(5), 433–437.
 https://doi.org/10.1177/070674370104600509
- Kyle, T., & Dunn, J. R. (2008). Effects of housing circumstances on health, quality of life and healthcare use for people with severe mental illness: a review. *Health & Social Care in the Community*, *16*(1), 1–15. https://doi.org/10.1111/j.1365-2524.2007.00723.x
- Laffey, P. (2003). Psychiatric therapy in Georgian Britain. *Psychological Medicine*, 33(7), 1285–1297. https://doi.org/10.1017/S0033291703008109
- Langellier, K., & Peterson, E. (2004). *Storytelling in daily life: Performing narrative*. Temple University Press.
- Larkin, M., Boden, Z. V. R., & Newton, E. (2015). On the brink of genuinely collaborative care: Experience-based co-design in mental health. *Qualitative Health Research*, *25*(11), 1463–1476. https://doi.org/10.1177/1049732315576494

Latham, R., & Willliams, H. K. (2020). Community forensic psychiatric services in England and Wales. CNS Spectrums, 25, 604–617. https://doi.org/10.1017/S1092852919001743

Lawson, B. (2010). Healing architecture. *Arts and Health*, *2*(2), 95–108. https://doi.org/10.1080/17533010903488517
Lawson, B., Phiri, M., & Wells-Thorpe, J. (2003). The architectural healthcare environment and its effects on patient health outcomes: A report on an NHS Estates Funded Research Project. The Stationery Office.

- LeBel, J., Champagne, T., Stromberg, N., & Coye, R. (2010). Integrating sensory and trauma-informed interventions: A Massachusetts state initiative, part 1. *Mental Health Specialist Interest Section Quarterly*, *33*(2), 1–5.
- Lewin, K. (1936). *Principles of topological psychology*. (F. Heider & G.M. Heider Trans.). McGraw-Hill Book Company Inc.
- Lewy, A. J., Bauer, V. K., Cutler, N. L., Sack, R. L., Ahmed, S., Thomas, K. H., Blood, M. L., & Latham Jackson, J. M. (1998). Morning vs evening light treatment of patients with winter depression. *Archives of General Psychiatry*, 55(10), 890–896. https://doi.org/10.1001/archpsyc.55.10.890
- Liddicoat, S., Badcock, P., & Killackey, E. (2020). Principles for designing the built environment of mental health services. *The Lancet Psychiatry*, *7*(10), 915–920. https://doi.org/10.1016/S2215-0366(20)30038-9
- Livingston, J. D., Nijdam-Jones, A., & Brink, J. (2012). A tale of two cultures: Examining patient-centered care in a forensic mental health hospital. *Journal of Forensic Psychiatry and Psychology*, *23*(3), 345–360.

https://doi.org/10.1080/14789949.2012.668214

Lloyd, C., King, R., & Machingura, T. (2014). An investigation into the effectiveness of sensory modulation in reducing seclusion within an acute mental health unit. *Advances in Mental Health*, *12*(2), 93–100. https://doi.org/10.1080/18374905.2014.11081887

Lundin, S. (2021). Can healing architecture increase safety in the design of psychiatric wards? *Health Environments Research and Design Journal*, *14*(1), 106–117. https://doi.org/10.1177/1937586720971814

MacInnes, D., Beer, D., Keeble, P., Rees, D., & Reid, L. (2011). Service-user involvement

in forensic mental health care research: Areas to consider when developing a collaborative study. *Journal of Mental Health*, *20*(5), 464–472. https://doi.org/10.3109/09638231003728109

- MacInnes, D., Courtney, H., Flanagan, T., Bressington, D., & Beer, D. (2014). A cross sectional survey examining the association between therapeutic relationships and service user satisfaction in forensic mental health settings. *BMC Research Notes*, 7(1), 657. https://doi.org/10.1186/1756-0500-7-657
- Mackin, P. (2008). Cardiac side effects of psychiatric drugs. *Human Psychopharmacology*, *23*, 3–14. https://doi.org/10.1002/hup.915
- Maller, C., Townsend, M., Pryor, A., Brown, P., & St Leger, L. (2006). Healthy nature healthy people: "Contact with nature" as an upstream health promotion intervention for populations. *Health Promotion International*, *21*(1), 45–54. https://doi.org/10.1093/heapro/dai032
- Malone, R. E. (2003). Distal nursing. *Social Science and Medicine*, *56*(11), 2317–2326. https://doi.org/10.1016/S0277-9536(02)00230-7
- Marshall, L. A., & Adams, E. A. (2018). Building from the ground up: exploring forensic mental health staff's relationships with patients. *Journal of Forensic Psychiatry and Psychology*, *29*(5), 744–761. https://doi.org/10.1080/14789949.2018.1508486
- Martin, D., Nettleton, S., & Buse, C. (2019). Affecting care: Maggie's Centres and the orchestration of architectural atmospheres. *Social Science and Medicine*, 240, 112563. https://doi.org/10.1016/j.socscimed.2019.112563

Martin, D., & Roe, J. (2022). Enabling care: Maggie's centres and the affordance of hope. *Health and Place*, *78*, 102758. https://doi.org/10.1016/j.healthplace.2022.102758

Martin, T., & Street, A. F. (2003). Exploring evidence of the therapeutic relationship in forensic psychiatric nursing. *Journal of Psychiatric and Mental Health Nursing*, *10*(5), 543–551. https://doi.org/10.1046/j.1365-2850.2003.00656.x

Mattingly, C., & Garro, L. C. (2000). Narrative as a construct and construction. In C.

Mattingly & L. C. Garro (Eds.), *Narrative and the cultural construction of illness and healing*. University of California Press.

- Mazuch, R., & Stephen, R. (2007). Creating healing environments: Humanistic architecture and therapeutic design. *Journal of Public Mental Health*, 4(4), 48–52. https://doi.org/10.1108/17465729200500031
- McAdams, D. P. (1993). *The stories we live by: Personal myths and the making of the self.* Morrow.
- McCann, E. (2010). Investigating mental health service user views regarding sexual and relationship issues. *Journal of Psychiatric and Mental Health Nursing*, 17(3), 251– 259. https://doi.org/10.1111/j.1365-2850.2009.01509.x
- McGrath, L., Brown, S. D., Kanyeredzi, A., Reavey, P., & Tucker, I. (2021). Peripheral recovery: 'Keeping safe' and 'keep progressing' as contradictory modes of ordering in a forensic psychiatric unit. *Environment and Planning D: Society and Space*, 39(4), 704–721. https://doi.org/10.1177/02637758211013032
- McGrath, L., Mighetto, I., Liebert, R. J., & Wakeling, B. (2021). Stuck in separation:
 Liminality, graffiti arts and the forensic institution as a failed rite of passage.
 Sociology of Health and Illness, 43(6), 1355–1371. https://doi.org/10.1111/1467-9566.13320
- McGrath, L., & Reavey, P. (2013). Heterotopias of control: Placing the material in experiences of mental health service use and community living. *Health & Place*, 22, 123–131. https://doi.org/10.1016/j.healthplace.2013.03.010
- McGrath, L., & Reavey, P. (2015). Seeking fluid possibility and solid ground: Space and movement in mental health service users' experiences of 'crisis'. *Social Science & Medicine*, *128*, 115–125. https://doi.org/10.1016/j.socscimed.2015.01.017
- McGrath, L., & Reavey, P. (2016). "Zip me up, and cool me down": Molar narratives and molecular intensities in 'helicopter' mental health services. *Health and Place*, 38, 61–69. https://doi.org/10.1016/j.healthplace.2015.12.005

- McGrath, L., & Reavey, P. (Eds.). (2019). *The handbook of mental health and space: Community and clinical applications*. Routledge.
- McGrath, L., Reavey, P., & Brown, S. D. (2008). The scenes and spaces of anxiety: Embodied expressions of distress in public and private fora. *Emotion, Space and Society*, 1(1), 56–64. https://doi.org/10.1016/j.emospa.2008.08.003
- McKellar, S. (2015). Contested spaces: The problem with modern psychiatric interiors.
 Interiors: Design, Architecture, Culture, 6(1), 21–39.
 https://doi.org/10.2752/204191115X14218559960150
- McKeown, M., Jones, F., Foy, P., Wright, K., Paxton, T., & Blackmon, M. (2016). Looking back, looking forward: Recovery journeys in a high secure hospital. *International Journal of Mental Health Nursing*, 25(3), 234–242. https://doi.org/10.1111/inm.12204
- Mclaughlan, R., Lyon, C., & Jaskolska, D. (2021). Architecture as change-agent? Looking for innovation in contemporary forensic psychiatric hospital design. *Medical Humanities*, 47(4). https://doi.org/10.1136/medhum-2020-011887
- McPherson, P., Krotofil, J., & Killaspy, H. (2018). Mental health supported accommodation services: A systematic review of mental health and psychosocial outcomes. *BMC Psychiatry*, 18(1), 1–15. https://doi.org/10.1186/s12888-018-1725-8
- McQueen, J. M., & Turner, J. (2012). Exploring forensic mental health service users' views on work: An interpretative phenomenological analysis. *British Journal of Forensic Practice*, *14*(3), 168–179. https://doi.org/10.1108/14636641211254897

Menditto, A. A. (2002). A social-learning approach to the rehabilitation of individuals with severe mental disorders who reside in forensic facilities. *Psychiatric Rehabilitation*, 6(1), 73–93.

https://doi.org/doi.org/10.1080/10973430208408423

Mezey, G. C., & Eastman, N. (2009). Choice and social inclusion in forensic psychiatry:

Acknowledging mixed messages and double think. *Journal of Forensic Psychiatry and Psychology*, *20*(4), 503–508. https://doi.org/10.1080/14789940903178023

- Mezey, G. C., Kavuma, M., Turton, P., Demetriou, A., & Wright, C. (2010). Perceptions, experiences and meanings of recovery in forensic psychiatric patients. *Journal of Forensic Psychiatry & Psychology*, 21(5), 683–696. https://doi.org/10.1080/14789949.2010.489953
- Milbourn, B. T., McNamara, B. A., & Buchanan, A. J. (2014). Do the everyday experiences of people with severe mental illness who are "hard to engage" reflect a journey of personal recovery? *Journal of Mental Health*, *23*(5), 241–245. https://doi.org/10.3109/09638237.2014.951485
- Mind. (2004). Ward Watch. Mind's campaign to improve hospital conditions for mental health patients: Report summary.

Mind. (2011). *Listening to Experience*. Mind.

Mind. (2013). Feel better outside, feel better inside: Ecotherapy for mental wellbeing, resilience and recovery.

Molin, J., Graneheim, U. H., & Lindgren, B.-M. (2016). Quality of interactions influences everyday life in psychiatric inpatient care - patients' perspectives. *International Journal of Qualitative Studies on Health and Well-Being*, 11. https://doi.org/10.3402/qhw.v11.29897

- Moran, A., Cocoman, A., Scott, P. A., Matthews, A., Staniuliene, V., & Valimaki, M. (2009).
 Restraint and seclusion: A distressing treatment option? *Journal of Psychiatric and Mental Health Nursing*, *16*(7), 599–605. https://doi.org/10.1111/j.1365-2850.2009.01419.x
- Muir-Cochrane, E., van der Merwe, M., Nijman, H., Haglund, K., Simpson, A., & Bowers, L. (2012). Investigation into the acceptability of door locking to staff, patients, and visitors on acute psychiatric wards. *International Journal of Mental Health Nursing*, *21*(1), 41–49. https://doi.org/10.1111/j.1447-0349.2011.00758.x

- Murphy, R., McGuinness, D., Bainbridge, E., Brosnan, L., Felzmann, H., Keys, M., Murphy,
 K., Hallahan, B., McDonald, C., & Higgins, A. (2017). Service users' experiences of
 involuntary hospital admission under the Mental Health Act 2001 in the Republic
 of Ireland. *Psychiatric Services*, 68(11), 1127–1135.
 https://doi.org/10.1176/appi.ps.201700008
- Murray, M. (2000). Levels of narrative analysis in health psychology. *Journal of Health Psychology*, *5*(3), 337–347. https://doi.org/10.1177/135910530000500305
- Murray, M. (2003). Narrative psychology and narrative analysis. In P. M. Camic, J. E. Rhodes, & L. Yardley (Eds.), *Qualitative research in psychology: Expanding perspectives in methodology and design* (pp. 95–112). https://doi.org/10.1037/10595-006
- Murray, M. (2008). Narrative psychology. In J. A. Smith (Ed.), *Qualitative Psychology a Practical Guide to Research Methods* (pp. 111–132). Sage Publications Limited.
- Nanda, U., Eisen, S. L., & Baladandayuthapani, V. (2008). Undertaking an art survey to compare patient versus student art preferences. *Environment and Behavior*, 40(2), 269–301. https://doi.org/10.1177/0013916507311552
- Nanda, U., Eisen, S., Zadeh, R. S., & Owen, D. (2011). Effect of visual art on patient anxiety and agitation in a mental health facility and implications for the business case. *Journal of Psychiatric and Mental Health Nursing*, *18*(5), 386–393. https://doi.org/10.1111/j.1365-2850.2010.01682.x

Naylor, C., Lincoln, J., & Goddard, N. (2008). Young people at risk of offending: Their views on a specialist mental health service in south east London. *Clinical Child Psychology and Psychiatry*, *13*(2), 277–286.
https://doi.org/10.1177/1359104507088347

Nejati, A., Shepley, M., Rodiek, S., Lee, C., & Varni, J. (2016). Restorative design features for hospital staff break areas: A multi-method study. *Health Environments Research and Design Journal*, 9(2), 16–35.

https://doi.org/10.1177/1937586715592632

- Ng, B., Kumar, S., Ranclaud, M., & Robinson, E. (2001). Ward crowding and incidents of violence on an acute psychiatric inpatient unit. *Psychiatric Services*, *52*(4), 521–525. https://doi.org/10.1176/appi.ps.52.4.521
- Nicholls, D., Kidd, K., Threader, J., & Hungerford, C. (2015). The value of purpose built mental health facilities: Use of the Ward Atmosphere Scale to gauge the link between milieu and physical environment. *International Journal of Mental Health Nursing*, *24*(4), 286–294. https://doi.org/10.1111/inm.12138
- Novak, C., Packer, E., Paterson, A., Roshi, A., Locke, R., Keown, P., Watson, S., &
 Anderson, K. N. (2020). Feasibility and utility of enhanced sleep management on
 in-patient psychiatry wards. *BJPsych Bulletin*, 44(6), 255–260.
 https://doi.org/10.1192/bjb.2020.30
- Nutsford, D., Pearson, A. L., Kingham, S., & Reitsma, F. (2016). Residential exposure to visible blue space (but not green space) associated with lower psychological distress in a capital city. *Health and Place, 39*, 70–78. https://doi.org/10.1016/j.healthplace.2016.03.002
- Nyrud, A. Q., Bringslimark, T., & Bysheim, K. (2014). Benefits from wood interior in a hospital room: A preference study. *Architectural Science Review*, *57*(2), 125–131. https://doi.org/10.1080/00038628.2013.816933
- O'Connell, M., Farnworth, L., & Hanson, E. C. (2010). Time use in forensic psychiatry: A naturalistic inquiry into two forensic patients in Australia. *International Journal of Forensic Mental Health*, *9*(2), 101–109.

https://doi.org/10.1080/14999013.2010.499558

O'Donahoo, J., & Simmonds, J. G. (2016). Forensic patients and forensic mental health in Victoria: Legal context, clinical pathways, and practice challenges. *Australian Social Work*, 69(2), 169–180. https://doi.org/10.1080/0312407X.2015.1126750
Office for Health Improvement and Disparities. (2022). *Working definition of trauma-*

informed practice. https://www.gov.uk/government/publications/workingdefinition-of-trauma-informed-practice/working-definition-of-trauma-informedpractice#other-professional-resources-and-tools

- Olausson, S., Danielson, E., Berglund Johansson, I., & Wijk, H. (2019). The meanings of place and space in forensic psychiatric care – A qualitative study reflecting patients' point of view. *International Journal of Mental Health Nursing*, *28*(2), 516– 526. https://doi.org/10.1111/inm.12557
- Olausson, S., Wijk, H., Johansson Berglund, I., Pihlgren, A., & Danielson, E. (2021).
 Patients' experiences of place and space after a relocation to evidence-based designed forensic psychiatric hospitals. *International Journal of Mental Health Nursing*, *30*(5), 1210–1220. https://doi.org/10.1111/inm.12871
- Papoulias, C., Csipke, E., Rose, D., McKellar, S., & Wykes, T. (2014). The psychiatric ward as a therapeutic space: Systematic review. *British Journal of Psychiatry*, 205(3), 171–176. https://doi.org/10.1192/bjp.bp.114.144873
- Payne, H., & May, D. (2009). Evaluation of a refurbishment scheme incorporating the King's Fund "Enhancing the Healing Environment" design principles. *Journal of Facilities Management*, 7(1), 74–89.

https://doi.org/10.1108/14725960910929583

- Pelaprat, E., & Cole, M. (2011). "Minding the gap": Imagination, creativity and human cognition. *Integrative Psychological and Behavioral Science*, 45(4), 397–418. https://doi.org/10.1007/s12124-011-9176-5
- Perlin, M. L. (2008). "Everybody is making love or else expecting rain": Considering the sexual autonomy rights of persons institutionalized because of mental disability in forensic hospitals and in Asia. Washington Law Review, 83(4), 481–512.
- Phibbs, S. (2008). Four dimensions of narrativity: Towards a narrative analysis of gender identity that is simultaneously personal, local and global. *New Zealand Sociology*, *23*(2), 47–60.

- Pilgrim, D., & Ramon, S. (2009). English mental health policy under New Labour. *Policy and Politics*, *37*(2), 273–288. https://doi.org/10.1332/030557309X411282
- Pink, S., Duque, M., Sumartojo, S., & Vaughan, L. (2020). Making spaces for staff breaks: A design anthropology approach. *Health Environments Research and Design Journal*, 13(2), 243–255. https://doi.org/10.1177/1937586719900954
- Polkinghorne, D. E. (1991). Narrative and self-concept. *Journal of Narrative and Life History*, 1(2–3), 135–153. https://doi.org/10.1075/jnlh.1.2-3.04nar

Procter, N., Ayling, B., Croft, L., DeGaris, P., Devine, M., Dimanic, A., Di Fiore, L., Eaton, H., Edwards, M., Ferguson, M., Lang, S., Rebellato, A., Shaw, K., & Sullivan, R. (2017). *Trauma-informed approaches in mental health: A practical resource for health professionals*. https://www.unisa.edu.au/siteassets/episerver-6files/global/health/sansom/documents/mhsa/trauma-informed-approaches-inforensic-mental-health-resource.pdf

- Quinn, C., & Happell, B. (2015a). Consumer sexual relationships in a Forensic mental health hospital: Perceptions of nurses and consumers. *International Journal of Mental Health Nursing*, 24(2), 121–129. https://doi.org/10.1111/inm.12112
- Quinn, C., & Happell, B. (2015b). Sex on show. Issues of privacy and dignity in a forensic mental health hospital: Nurse and patient views. *Journal of Clinical Nursing*, 24(15–16), 2268–2276. https://doi.org/10.1111/jocn.12860
- Quirk, A., & Lelliott, P. (2001). What do we know about life on acute psychiatric wards in the UK? A review of the research evidence. *Social Science and Medicine*, 53(12), 1565–1574. https://doi.org/10.1016/S0277-9536(00)00457-3

Quirk, A., Lelliott, P., & Seale, C. (2004). Service users' strategies for managing risk in the volatile environment of an acute psychiatric ward. *Social Science and Medicine*, 59(12), 2573–2583. https://doi.org/10.1016/j.socscimed.2004.04.005

Quirk, A., Lelliott, P., & Seale, C. (2006). The permeable institution: An ethnographic study of three acute psychiatric wards in London. *Social Science & Medicine*, *63*(8),

2105-2117. https://doi.org/10.1016/j.socscimed.2006.05.021

- Radley, A., & Billig, M. (1996). Accounts of health and illness: Dilemmas and representations. *Sociology of Health & Illness*, 18(2), 220–240. https://doi.org/10.1111/1467-9566.ep10934984
- Radley, A., & Taylor, D. (2003a). Images of recovery: A photo-elicitation study on the hospital ward. *Qualitative Health Research*, 13(1), 77–99. https://doi.org/10.1177/1049732302239412
- Radley, A., & Taylor, D. (2003b). Remembering one's stay in hospital: A study in photography, recovery and forgetting. *Health*, 7(2), 129–159.
 https://doi.org/10.1177/1363459303007002872
- Ray, J. L., & Smith, A. D. (2012). Using photographs to research organizations: Evidence, considerations, and application in a field study. *Organizational Research Methods*, 15(2), 288–315. https://doi.org/10.1177/1094428111431110
- Reavey, P. (Ed.). (2011). Visual methods in psychology: Using and interpreting images in qualitative research. Psychology Press.
- Reavey, P. (2017). Scenic memory: Experience through time-space. *Memory Studies*, *10*(2), 107–111. https://doi.org/10.1177/1750698016683844
- Reavey, P., Brown, S. D., Ciarlo, D., & Lazenby, K. (2021). *Design with People in Mind. Borders and Boundaries.* Design in Mental Health Network.
- Reavey, P., Brown, S. D., Kanyeredzi, A., McGrath, L., & Tucker, I. (2019). Agents and spectres: Life-space on a medium secure forensic psychiatric unit. *Social Science* and Medicine, 220, 273–282. https://doi.org/10.1016/j.socscimed.2018.11.012

Reavey, P., Brown, S. D., Ravenhill, J. P., Boden-Stuart, Z., & Ciarlo, D. (2022).
Choreographies of sexual safety and liminality: Forensic mental health and the limits of recovery. *SSM - Mental Health*, *2*, 100090.
https://doi.org/10.1016/j.ssmmh.2022.100090

Reavey, P., & Harding, K. (2018). Design with people in mind: The sound issue. Design in

Mental Health Network.

- Reavey, P., & Harding, K. (2019). *Design with people in mind: The nature issue*. Design in Mental Health Network.
- Reavey, P., Harding, K., & Bartle, J. (2017). *Design with people in mind*. Design in Mental Health Network.
- Reavey, P., & Johnson, K. (2017). Visual approaches: Using and interpreting images. In
 C. Willig & W. Stainton-Rogers (Eds.), *The SAGE handbook of qualitative research in psychology* (pp. 354–373). SAGE Publications Limited.
- Reavey, P., Poole, J., Corrigall, R., Zundel, T., Byford, S., Sarhane, M., Taylor, E., Ivens, J., & Ougrin, D. (2017). The ward as emotional ecology: Adolescent experiences of managing mental health and distress in psychiatric inpatient settings. *Health and Place*, 46, 210–218. https://doi.org/10.1016/j.healthplace.2017.05.008
- Revell, S., & McLeod, J. (2017). Therapists' experience of walk and talk therapy: A descriptive phenomenological study. *European Journal of Psychotherapy & Counselling*, 19(3), 267–289. https://doi.org/10.1080/13642537.2017.1348377
- Rice, T. (2003). Soundselves: An acoustemology of sound and self in the Edinburgh Royal Infirmary. *Anthropology Today*, *19*(4), 4–9.
- Richardson, C. R., Faulkner, G., McDevitt, J., Skrinar, G. S., Hutchinson, D. S., & Piette, J. D. (2005). Integrating physical activity into mental health services for persons with serious mental illness. *Psychiatric Services*, *56*(3), 324–331. https://doi.org/10.1176/appi.ps.56.3.324

Ricoeur, P. (1979). The human experience of time and narrative. *Research in Phenomenology*, 9(1), 17–34. https://www.jstor.org/stable/pdf/24654326.pdf
Riessman, C. K. (2008). *Narrative methods for the human sciences*. Sage.

Riordan, S., & Humphreys, M. (2007). Patient perceptions of medium secure care. *Medicine, Science and the Law*, 47(1), 20–26. https://doi.org/10.1258/rsmmsl.47.1.20 Rose, D., Carr, S., & Beresford, P. (2018). 'Widening cross-disciplinary research for mental health': What is missing from the Research Councils UK mental health agenda? *Disability and Society*, 33(3), 476–481. https://doi.org/10.1080/09687599.2018.1423907

Rose, G. (2014). On the relation between 'visual research methods' and contemporary visual culture. *The Sociological Review*, 62(1), 24–46. https://doi.org/10.1111/1467-954X.12109

- Rosenberg, S. D., Mueser, K. T., Friedman, M. J., Gorman, P. G., Drake, R. E., Vidaver, R. M., Torrey, W. C., & Jankowski, M. K. (2001). Developing effective treatments for posttraumatic disorders among people with severe mental illness. *Psychiatric Services*, 52(11), 1453–1461. https://doi.org/10.1176/appi.ps.52.11.1453
- Rossberg, J. I., & Friis, S. (2004). Patients' and staff's perceptions of the psychiatric ward environment. *Psychiatric Services*, 55(7), 798–803.
 https://doi.org/10.1176/appi.ps.55.7.798
- Roxberg, Å., Tryselius, K., Gren, M., Lindahl, B., Werkander Harstäde, C., Silverglow, A., Nolbeck, K., James, F., Carlsson, I. M., Olausson, S., Nordin, S., & Wijk, H. (2020). Space and place for health and care. *International Journal of Qualitative Studies on Health and Well-Being*, *15*(sup1), 1–13.

https://doi.org/10.1080/17482631.2020.1750263

Royal College of Psychiatrists. (2019). *Enabling environments standards 2019*. https://www.rcpsych.ac.uk/docs/default-source/improving-care/ccqi/qualitynetworks/enabling-environments-ee/ee-standards-2019.pdf?sfvrsn=337f7bf3_2

Rutherford, M., & Duggan, S. (2008). Forensic mental health services: Facts and figures on current provision. *British Journal of Forensic Practice*, *10*(4), 4–10. https://doi.org/10.1108/14636646200800020

Rutter, D., Manley, C., Weaver, T., Crawford, M. J., & Fulop, N. (2004). Patients or partners? Case studies of user involvement in the planning and delivery of adult

mental health services in London. *Social Science & Medicine*, *58*(10), 1973–1984. https://doi.org/10.1016/S0277-9536(03)00401-5

- Ryherd, E. E., Okcu, S., Ackerman, J., Zimring, C., & Waye, K. P. (2012). Noise pollution in hospitals: Impacts on staff. *Journal of Clinical Outcomes Management*, 19(11), 491– 500.
- Sakuragawa, S., Kaneko, T., & Miyazaki, Y. (2008). Effects of contact with wood on blood pressure and subjective evaluation. *Journal of Wood Science*, 54(2), 107–113. https://doi.org/10.1007/s10086-007-0915-7
- Scanlan, J. N., & Novak, T. (2015). Sensory approaches in mental health: A scoping review. Australian Occupational Therapy Journal, 62(5), 277–285. https://doi.org/10.1111/1440-1630.12224
- Scanlon, C., & Adlam, J. (2011). Who watches the watchers? Observing the dangerous liaisons between forensic patients and their carers in the perverse panopticon. *Organisational & Social Dynamics*, *11*(2), 175–195.
- Schweitzer, M., Gilpin, L., & Frampton, S. (2004). Healing spaces: Elements of environmental design that make an impact on health. *Journal of Alternative & Complementary Medicine*, 10(1), S-71-S-83.

https://doi.org/10.1089/acm.2004.10.S-71

Scott, J., Langsrud, K., Goulding, I. R., & Kallestad, H. (2021). Let there be blue-depleted light: In-patient dark therapy, circadian rhythms and length of stay. *BJPsych Advances*, *27*(2), 73–84. https://doi.org/10.1192/bja.2020.47

Seale, C., Chaplin, R., Lelliott, P., & Quirk, A. (2007). Antipsychotic medication, sedation and mental clouding: An observational study of psychiatric consultations. *Social Science and Medicine*, 65(4), 698–711.

https://doi.org/10.1016/j.socscimed.2007.03.047

Sendula-Jengić, V., Juretić, I., & Hodak, J. (2011). Psychiatric hospital-from asylums to centres for mind-body wellness. *Collegium Antropologicum*, *35*(4), 979–988.

http://www.ncbi.nlm.nih.gov/pubmed/22397227

Seppänen, A., Törmänen, I., Shaw, C., & Kennedy, H. (2018). Modern forensic psychiatric hospital design: Clinical, legal and structural aspects. *International Journal of Mental Health Systems*, 12(1), 1–12. https://doi.org/10.1186/s13033-018-0238-7

Shah, A., Waldron, G., Boast, N., Coid, J. W., & Ullrich, S. (2011). Factors associated with length of admission at a medium secure forensic psychiatric unit. *Journal of Forensic Psychiatry and Psychology*, 22(4), 496–512. https://doi.org/10.1080/14789949.2011.594902

- Sharac, J., McCrone, P., Sabes-Figuera, R., Csipke, E., Wood, A., & Wykes, T. (2010). Nurse and patient activities and interaction on psychiatric inpatients wards: A literature review. *International Journal of Nursing Studies*, 47(7), 909–917. https://doi.org/10.1016/j.ijnurstu.2010.03.012
- Sharp, N. L., Bye, R. A., & Cusick, A. (2018). Narrative analysis. In P. Liamputtong (Ed.), Handbook of research methods in health social sciences. Springer, Singapore. https://doi.org/10.1007/978-981-10-2779-6_106-1
- Shattell, M. M., Andes, M., & Thomas, S. P. (2008). How patients and nurses experience the acute care psychiatric environment. *Nursing Inquiry*, *15*(3), 242–250. https://doi.org/10.1111/j.1440-1800.2008.00397.x
- Shattell, M. M., Bartlett, R., Beres, K., Southard, K., Bell, C., Judge, C. A., & Duke, P. (2015).
 How patients and nurses experience an open versus an enclosed nursing station
 on an inpatient psychiatric unit. *Journal of the American Psychiatric Nurses Association*, 21(6), 398–405. https://doi.org/10.1177/1078390315617038
- Shen, G. C., & Snowden, L. R. (2014). Institutionalization of deinstitutionalization: A cross-national analysis of mental health system reform. *International Journal of Mental Health Systems*, 8(47). https://doi.org/10.1186/1752-4458-8-47
- Shepley, M. M., & Watson, A. (2013). Evidence-based design: medical and design researcher collaboration. *Evidence-Based Medicine*, *18*(1), 2–4.

https://doi.org/10.1136/eb-2012-100785

- Shepley, M. M., Watson, A., Pitts, F., Garrity, A., Spelman, E., Fronsman, A., & Kelkar, J. (2017). Mental and behavioral health settings: Importance & effectiveness of environmental qualities & features as perceived by staff. *Journal of Environmental Psychology*, *50*, 37–50. https://doi.org/10.1016/j.jenvp.2017.01.005
- Shepley, M. M., Watson, A., Pitts, F., Garrity, A., Spelman, E., Kelkar, J., & Fronsman, A. (2016). Mental and behavioral health environments: Critical considerations for facility design. *General Hospital Psychiatry*, 42, 15–21. https://doi.org/10.1016/j.genhosppsych.2016.06.003
- Shortt, H. (2015). Liminality, space and the importance of 'transitory dwelling places' at work. *Human Relations*, 68(4), 633–658. https://doi.org/10.1177/0018726714536938
- Silver, J. (2013). Narrative psychology. In C. Willig (Ed.), *Introducing qualitative research in psychology* (pp. 143–155). McGraw-Hill Education.
- Silver, J., & Reavey, P. (2010). "He's a good-looking chap aint he?": Narrative and visualisations of self in body dysmorphic disorder. *Social Science & Medicine*, 70(10), 1641–1647. https://doi.org/10.1016/j.socscimed.2009.11.042
- Simonsen, T. P., & Duff, C. (2020). Healing architecture and psychiatric practice: (re)ordering work and space in an in-patient ward in Denmark. *Sociology of Health and Illness*, 42(2), 379–392. https://doi.org/10.1111/1467-9566.13011
- Simonsen, T. P., & Duff, C. (2021). Mutual visibility and interaction: staff reactions to the 'healing architecture' of psychiatric inpatient wards in Denmark. *BioSocieties*, 16(2), 249–269. https://doi.org/10.1057/s41292-020-00195-4
- Simonsen, T. P., Sturge, J., & Duff, C. (2022). Healing architecture in healthcare: A scoping review. *Health Environments Research and Design Journal*, 1–14. https://doi.org/10.1177/19375867211072513

Simpson, A. I. F., & Penney, S. R. (2011). The recovery paradigm in forensic mental

health services. *Criminal Behaviour and Mental Health*, *21*(5), 299–306. https://doi.org/10.1002/cbm.823

- Simpson, A. I. F., & Penney, S. R. (2018). Recovery and forensic care: Recent advances and future directions. *Criminal Behaviour and Mental Health*, 28(5), 383–389. https://doi.org/10.1002/cbm.2090
- Simpson, E. L., & House, A. O. (2002). Involving users in the delivery and evaluation of mental health services : systematic review. *BMJ*, 325(1265), 1–5.
- Smith, B. (2016). Narrative analysis. In A. Coyle & E. Lyons (Eds.), *Analysing qualitative data in psychology* (2nd ed., pp. 202–221). SAGE Publications Limited.
- Somers, M. R. (1994). The narrative constitution of identity: A relational and network approach. *Theory and Society*, 23(5), 605–649. https://doi.org/10.1007/BF00992905
- Sommer, R., & Ross, H. (1958). Social interaction on a geriatrics ward. *International Journal of Social Psychiatry*, 4(2), 128–133.

https://doi.org/10.1177/002076405800400207

- Southard, K., Jarrell, A., Shattell, M. M., McCoy, T. P., Bartlett, R., & Judge, C. A. (2012).
 Enclosed versus open nursing stations in adult acute care psychiatric settings:
 Does the design affect the therapeutic milieu? *Journal of Psychosocial Nursing*, *50*(5), 28–34. https://doi.org/10.3928/02793695-20120410-04
- Southwell, M. T., & Wistow, G. (1995). Sleep in hospitals at night: Are patients' needs being met? *Journal of Advanced Nursing*, *21*(6), 1101–1109.
- Spiers, S., Harney, K., & Chilvers, C. (2005). Service user involvement in forensic mental health: Can it work? *Journal of Forensic Psychiatry and Psychology*, 16(2), 211–220. https://doi.org/10.1080/14789940500098137
- Staniszewska, S., Mockford, C., Chadburn, G., Fenton, S.-J., Bhui, K., Larkin, M., Newton,
 E., Crepaz-Keay, D., Griffiths, F., & Weich, S. (2019). Experiences of in-patient
 mental health services: Systematic review. *British Journal of Psychiatry*, 214(6),

329-338. https://doi.org/10.1192/bjp.2019.22

Stansfeld, S. A. (1992). Noise, noise sensitivity and psychiatric disorder: Epidemiological and psychophysiological studies. *Psychological Monograph Supplement*, 22, 1–44. https://doi.org/10.1017/S0264180100001119

Stansfeld, S. A., & Matheson, M. P. (2003). Noise pollution: Non-auditory effects on health. *British Medical Bulletin*, 68, 243–257. https://doi.org/10.1093/bmb/ldg033

Stenner, P. (2008). A.N. Whitehead and subjectivity. *Subjectivity*, *22*(1), 90–109. https://doi.org/10.1057/sub.2008.4

- Stenner, P. (2013). Affectivity, liminality, and psychology without foundations. In J.
 Straub, E. Sørensen, P. Chakkarath, & G. Rebane (Eds.), *Cultural Psychology, Aesthetics and Postmodernity*. Psychosozial Verlag. https://www.researchgate.net
- Stenner, P. (2014a). Psychosocial: qu'est-ce que c'est? Journal of Psycho-Social Studies, 8(1), 205–216.
- Stenner, P. (2014b). Transdiciplinarity. In T. Teo (Ed.), *Encyclopaedia of critical psychology* (pp. 1987–1993). Springer. https://doi.org/10.1007/978-1-4614-5583-7_317
- Stenner, P. (2017). *Liminality and experience: A transdisciplinary approach to the psychosocial*. Palgrave Macmillan.
- Stenner, P. (2020). Affect: On the turn. In B. Bösel & S. Wiemer (Eds.), Affective Transformations: Politics- Algorithms-Media (pp. 19–39). Meson Press. https://doi.org/10.14619/1655

Stenner, P. (2021). Theorising liminality between art and life: The liminal sources of cultural experience. In B. Wagoner & T. Zittoun (Eds.), *Experience on the Edge: Theorizing Liminality* (pp. 3–42). Springer. https://doi.org/10.1007/978-3-030-83171-4

Stenner, P., Greco, M., & Motzkau, J. F. (2017). Introduction to the special issue on

liminal hotspots. *Theory and Psychology*, 27(2), 141–146.

https://doi.org/10.1177/0959354316687867

- Stenner, P., & Moreno-Gabriel, E. (2013). Liminality and affectivity: The case of deceased organ donation. *Subjectivity*, 6(3), 229–253. https://doi.org/10.1057/sub.2013.9
- Stenner, P., & Taylor, D. (2008). Psychosocial welfare: Reflections on an emerging field. *Critical Social Policy*, 28(4), 415–437.

https://doi.org/10.1177/0261018308095278

- Stenner, P., & Zittoun, T. (2020). On taking a leap of faith: Art, imagination, and liminal experiences. *Journal of Theoretical and Philosophical Psychology*, 1–24. https://doi.org/10.1037/teo0000148
- Stephens, C. (2011). Narrative analysis in health psychology research: Personal, dialogical and social stories of health. *Health Psychology Review*, 5(1), 62–78. https://doi.org/10.1080/17437199.2010.543385
- Stephens, C., & Breheny, M. (2013). Narrative analysis in psychological research: An integrated approach to interpreting stories. *Qualitative Research in Psychology*, 10(1), 14–27. https://doi.org/10.1080/14780887.2011.586103
- Stichler, J. F. (2008). Healing by design. *Journal of Nursing Administration*, *38*(12), 505–509. https://doi.org/10.1097/NNA.0b013e31818ebfa6
- Sutton, D., & Nicholson, E. (2011). *Sensory modulation in acute mental health wards: A qualitative study of staff and service user perspectives*. The National Centre of Mental Health Research, Information and Workforce Development. http://www.tepou.co.nz/library/tepou/sensory-modulation-in-acute-mental-health-wards-a-qualitative-study-of-staff-and-service-user-perspectives-
- Sutton, D., Wilson, M., Van Kessel, K., & Vanderpyl, J. (2013). Optimizing arousal to manage aggression: A pilot study of sensory modulation. *International Journal of Mental Health Nursing*, 22(6), 500–511. https://doi.org/10.1111/inm.12010

- Szakolczai, Á. (2009). Liminality and experience: Structuring transitory situations and transformative events. *International Political Anthropology*, *2*(1), 141–172.
- Szakolczai, Á. (2017). Permanent (trickster) liminality: The reasons of the heart and of the mind. *Theory and Psychology*, 27(2), 231–248. https://doi.org/10.1177/0959354317694095
- Tanzman, B. (1993). An overview of surveys of mental health consumers' preferences for housing and support services. *Hospital and Community Psychiatry*, 44(5), 450–455. https://doi.org/10.1176/ps.44.5.450
- Tapp, J., Warren, F., Fife-Schaw, C., Perkins, D., & Moore, E. (2013). What do the experts by experience tell us about 'what works' in high secure forensic inpatient hospital services? *The Journal of Forensic Psychiatry & Psychology*, 24(2), 160–178. https://doi.org/10.1080/14789949.2012.760642
- Taxis, J. C. (2002). Ethics and praxis: Alternative strategies to physical restraint and seclusion in a psychiatric setting. *Issues in Mental Health Nursing*, 23(2), 157–170. https://doi.org/10.1080/016128402753542785
- The Point of Care Foundation. (2016). *Experience-based co-design toolkit*. https://www.pointofcarefoundation.org.uk/resource/experience-based-codesign-ebcd-toolkit/
- Thomassen, B. (2009). The uses and meanings of liminality. *International Political Anthropology*, *2*(1), 5–27.
- Thomassen, B. (2016). *Liminality and the modern: Living through the in-between*. Routledge.
- Thornicroft, G., & Bebbington, P. (1989). Deinstitutionalisation--from hospital closure to service development. *The British Journal of Psychiatry*, 155(6), 739–753. https://doi.org/10.1192/bjp.155.6.739
- Tomlin, J., Egan, V., Bartlett, P., & Völlm, B. (2020). What do patients find restrictive about forensic mental health services? A qualitative study. *International Journal of*

Forensic Mental Health, 19(1), 44–56.

https://doi.org/10.1080/14999013.2019.1623955

- Tucker, I. (2010a). Everyday spaces of mental distress: The spatial habituation of home. *Environment and Planning D: Society and Space*, *28*(3), 526–538. https://doi.org/10.1068/d14808
- Tucker, I. (2010b). Mental health service user territories: Enacting 'safe spaces' in the community. *Health:*, *14*(4), 434–448.

https://doi.org/10.1177/1363459309357485

- Tucker, I., Brown, S. D., Kanyeredzi, A., McGrath, L., & Reavey, P. (2019). Living 'in between' outside and inside: The forensic psychiatric unit as an impermanent assemblage. *Health and Place*, 55(2019), 29–36. https://doi.org/10.1016/j.healthplace.2018.10.009
- Tucker, I., & Smith, L.-A. (2014). Topology and mental distress: Self-care in the life spaces of home. *Journal of Health Psychology*, 19(1), 176–183. https://doi.org/10.1177/1359105313500260
- Turner, T. (2004). The history of deinstitutionalization and reinstitutionalization. *Psychiatry*, *3*(9), 1–4. https://doi.org/10.1383/psyt.3.9.1.50257
- Turner, V. (1967). Betwixt and between: The liminal period in rites de passage. In *The forest of symbols: Aspects of Ndembu ritual* (pp. 93–111). Cornell University Press.
- Turner, V. (1982). *From ritual to theatre: The human seriousness of play*. PAJ Publications.
- Turner, V. (1996). *The ritual process: Structure and anti-structure*. Taylor and Francis Group (Original work published 1969).
- Turner, W. R., Nakamura, T., & Dinetti, M. (2004). Global urbanization and the separation of humans from nature. *BioScience*, 54(6), 585. https://doi.org/10.1641/0006-3568(2004)054[0585:guatso]2.0.co;2
- Tyson, G. A., Lambert, G., & Beattie, L. (2002). The impact of ward design on the

behaviour, occupational satisfaction and well-being of psychiatric nurses. *International Journal of Mental Health Nursing*, *11*(2), 94–102. https://doi.org/10.1046/j.1440-0979.2002.00232.x

- Ulrich, R. S. (1983). Aesthetic and affective response to natural environment. In I. Altman & J. F. Wohlwill (Eds.), *Human behavior and environment, vol. 6: Behavior and natural environment* (pp. 85–125). Plenum. https://doi.org/10.1007/978-1-4613-3539-9
- Ulrich, R. S. (1984). View through a window may influence recovery from surgery. *Science*, *224*, 420–421. https://doi.org/10.1126/science.6143402
- Ulrich, R. S. (1991). Effects of interior design on wellness: Theory and recent scientific research. *Journal of Healthcare Interior Design*, *3*(1), 97–109.
- Ulrich, R. S., Bogren, L., Gardiner, S. K., & Lundin, S. (2018). Psychiatric ward design can reduce aggressive behavior. *Journal of Environmental Psychology*, *57*, 53–66. https://doi.org/10.1016/j.jenvp.2018.05.002
- Ulrich, R. S., Bogren, L., & Lundin, S. (2012). Towards a design theory for reducing aggression in psychiatric facilities. *ARCH 12: ARCHITECTURE / RESEARCH / CARE / HEALTH*.
- Ulrich, R. S., Zimring, C., Quan, X., & Joseph, A. (2006). The environment's impact on stress. In S. O. Marberry (Ed.), *Improving Healthcare with Better Building Design* (Issue January, pp. 37–61). ACHE Management Series/Health Administration Press.

Ulrich, R. S., Zimring, C., Zhu, X., DuBose, J., Seo, H.-B., Choi, Y.-S., Quan, X., & Joseph, A. (2008). A review of the research literature on evidence-based design. *Health Environments Research and Design Journal*, 1(3), 61–125. https://doi.org/10.1177/193758670800100306

Unsworth, C. (1993). Law and lunacy in psychiatry's 'golden age.' *Oxford Journal of Legal Studies*, *13*(4), 479–507. https://doi.org/10.1093/ojls/13.4.479

van der Kolk, B. (2015). The body keeps the score. Penguin Random House UK.

- van Gennep, A. (1960). *The Rites of Passage*. (M.B. Vizedom and G.L. Caffee Trans.) University of Chicago Press. (Original work published 1909).
- Veale, D. (2019). Against the stream: Intermittent nurse observations of in-patients at night serve no purpose and cause sleep deprivation. *BJPsych Bulletin*, 43(4), 174– 176. https://doi.org/10.1192/bjb.2018.116
- Veale, D., Robins, E., Thomson, A. B., & Gilbert, P. (2022). No safety without emotional safety. *The Lancet Psychiatry*. https://doi.org/10.1016/S2215-0366(22)00373-X
- Vethe, D., Scott, J., Engstrøm, M., Salvesen, Ø., Sand, T., Olsen, A., Morken, G., Heglum, H.
 S., Kjørstad, K., Faaland, P. M., Vestergaard, C. L., Langsrud, K., & Kallestad, H.
 (2021). The evening light environment in hospitals can be designed to produce less disruptive effects on the circadian system and improve sleep. *Sleep*, *44*(3), 1–12. https://doi.org/10.1093/sleep/zsaa194
- Völker, S., & Kistemann, T. (2011). The impact of blue space on human health and wellbeing - Salutogenetic health effects of inland surface waters: A review. *International Journal of Hygiene and Environmental Health*, 214(6), 449–460.
 https://doi.org/10.1016/j.ijheh.2011.05.001
- Völlm, B., Foster, S., Bates, P., & Huband, N. (2017). How best to engage users of forensic services in research: Literature review and recommendations. *International Journal of Forensic Mental Health*, *16*(2), 183–195. https://doi.org/10.1080/14999013.2016.1255282

Vygotsky, L. S. (2004). Imagination and creativity in childhood. *Journal of Russian & East European Psychology*, 42(1), 7–97. https://doi.org/10.1080/10610405.2004.11059210

Waite, F., Langman, A., Mulhall, S., Glogowska, M., Hartmann-Boyce, J., Aveyard, P.,
Lennox, B., Oxford Cognitive Approaches to Psychosis Patient Advisory Group,
Kabir, T., & Freeman, D. (2022). The psychological journey of weight gain in

psychosis. *Psychology and Psychotherapy: Theory, Research and Practice, 95*(2), 525–540. https://doi.org/10.1111/papt.12386

Ward Thompson, C. (2011). Linking landscape and health: The recurring theme. *Landscape and Urban Planning*, 99(3–4), 187–195. https://doi.org/10.1016/j.landurbplan.2010.10.006

Ward Thompson, C., Roe, J., Aspinall, P., Mitchell, R., Clow, A., & Miller, D. (2012). More green space is linked to less stress in deprived communities: Evidence from salivary cortisol patterns. *Landscape and Urban Planning*, *105*(3), 221–229. https://doi.org/10.1016/j.landurbplan.2011.12.015

- Warner, J., Pitts, N., Crawford, M. J., Serfaty, M., Prabhakaran, P., & Rizkar, A. (2004).
 Sexual activity among patients in psychiatric hospital wards. *Journal of the Royal Society of Medicine*, *97*, 477–479. https://doi.org/10.1177/0141076809701005
- Warren, R., & Bell, P. (2000). An exploratory investigation into the housing preferences of consumers of mental health services. *The Australian and New Zealand Journal of Mental Health Nursing*, 9(4), 195–202. https://doi.org/10.1046/j.1440-0979.2000.00172.x

Warren, S. (2005). Photography and voice in critical qualitative management research. Accounting, Auditing and Accountability Journal, 18(6), 861–882. https://doi.org/10.1108/09513570510627748

Watson, W. (1998). Designed to cure: The clinician led development of England's regional secure units. *Journal of Forensic Psychiatry*, 9(3), 519–531.
https://doi.org/10.1080/09585189808405371

West, M. L., Yanos, P. T., & Mulay, A. L. (2014). Triple stigma of forensic psychiatric patients: Mental illness, race, and criminal history. *International Journal of Forensic Mental Health*, *13*(1), 75–90. https://doi.org/10.1080/14999013.2014.885471

Westman, J. C., & Walters, J. R. (1981). Noise and stress: A comprehensive approach.

Environmental Health Perspectives, Vol. 41(October), 291–309. https://doi.org/10.1289/ehp.8141291

- Williams, A., Moore, E., Adshead, G., McDowell, A., & Tapp, J. (2011). Including the excluded: high security hospital user perspectives on stigma, discrimination, and recovery. *The British Journal of Forensic Practice*, *13*(3), 197–204. https://doi.org/10.1108/14636641111157841
- Willig, C. (2013). *Introducing qualitative research in psychology*. Open University Press McGraw Hill.
- Wilson, E. O. (1984). Biophilia. Harvard University Press.
- Wong, G., & Breheny, M. (2018). Narrative analysis in health psychology: A guide for analysis. *Health Psychology and Behavioral Medicine*, 6(1), 245–261.
 https://doi.org/10.1080/21642850.2018.1515017
- Wood, V. J., Curtis, S. E., Gesler, W., Spencer, I. H., Close, H. J., Mason, J. M., & Reilly, J. G.
 (2013a). Spaces for smoking in a psychiatric hospital: Social capital, resistance to control, and significance for "therapeutic landscapes." *Social Science and Medicine*, 97, 104–111. https://doi.org/10.1016/j.socscimed.2013.08.009
- Wood, V. J., Curtis, S. E., Gesler, W., Spencer, I. H., Close, H. J., Mason, J., & Reilly, J. G.
 (2013b). Creating 'therapeutic landscapes' for mental health carers in inpatient settings: A dynamic perspective on permeability and inclusivity. *Social Science & Medicine*, *91*, 122–129. https://doi.org/10.1016/j.socscimed.2012.09.045
- Zhang, X., Lian, Z., & Wu, Y. (2017). Human physiological responses to wooden indoor environment. *Physiology and Behavior*, 174, 27–34. https://doi.org/10.1016/j.physbeh.2017.02.043
- Zittoun, T. (2007a). Symbolic resources and responsibility in transitions. *Young*, *15*(2), 193–211. https://doi.org/10.1177/110330880701500205
- Zittoun, T. (2007b). The role of symbolic resources in human lives. In J. Valsiner & A. Rosa (Eds.), *The Cambridge Handbook of Sociocultural Psychology 4* (pp. 343–361).

Cambridge University Press. https://doi.org/10.1017/CB09780511611162

- Zittoun, T., & Cerchia, F. (2013). Imagination as expansion of experience. *Integrative Psychological and Behavioral Science*, 47(3), 305–324.
 https://doi.org/10.1007/s12124-013-9234-2
- Zorwaska, A. (2019). Madlove: a designer asylum. In L. McGrath & P. Reavey (Eds.), *The handbook of mental health and space: Community and clinical applications* (pp. 50–54). Routledge.

Appendices

London South Bank University

Service User Participant Information Sheet (Study One - Group One and Group Two)

Public and private space within forensic mental health accommodation

You are being invited to take part in a research study. This study will contribute to my PhD studies in Psychology at London South Bank University. Taking part is entirely up to you and before you decide it is important for you to understand why the research is being done and what it will involve. This sheet explains the purpose of the study and provides information about what would be involved in taking part. I will read through this sheet with you and will answer any questions you may have. Please ask if anything is unclear.

This research aims to explore the everyday experiences of service users and staff in forensic mental health residential settings and I would be very interested to hear your thoughts about the place where you live. The study is looking to explore the ways in which your personal environment might affect or reflect how you feel day to day. The research will include other service users living here and in supported housing in the community. Members of staff who work here and in supported housing in the community will also be asked to participate. In total, approximately 30 people will be included in the study. The study is open to people over the age of 18 and research will be taking place here during the coming weeks.

Do I have to take part?

No, you are under no obligation and if you do not wish to participate it will have no impact on your current care.

What's involved?

You are being invited to take part in a study exploring the everyday experiences of forensic mental health service users and staff in residential environments. You are being invited to participate as you have personal experience of living here and I am interested in hearing about your experience. If you decide to participate, your care team will be made aware that you are taking part in a research project. If you'd like to take part, you will be asked to produce up to ten photos of the place where you currently live. These should reflect your typical everyday experiences and should capture the range of feelings you may hold towards the places, spaces or objects which you encounter in daily life. You should avoid taking photos of other people and photos including other people who could be recognised will be deleted. A digital camera will be provided for use in this study and I will print your images before we meet for an interview at a convenient time for you.

When we meet I will ask you to talk about the photos and I will ask you some questions in an interview which may last about 45 minutes. It is not anticipated that you will be at any disadvantage from participating. I will not ask you any direct questions about your mental health or why you live here. Some questions may invite you to reflect on personal and possibly sensitive subjects, however you can decline to answer any questions raised and you may also ask to stop the interview at any time. Afterwards I will also give you a short debrief about the study. Should anything arise during the research to cause you to feel concerned or upset, contact details for independent support organisations that you could contact for information or to speak with someone in confidence are provided at the end of this sheet. You should also speak with the staff where you live or your care coordinator if you feel affected by the research.

There's no financial benefit, but you might find it interesting to contribute your experiences to research which aims to expand knowledge of the experience of forensic mental health accommodation, or to have the opportunity to discuss this topic with a receptive listener. At any point until completion of the PhD thesis, if you wish to withdraw entirely from the

Appendix A: Service user participant information sheet (page 2 of 2)

research you may do so without giving a reason and your care will not be affected. You may also request that any answers given in the interview are removed from the transcribed data or request that any images supplied are deleted or are not reproduced in publications of the research.

Confidentiality & data protection

Interviews will be recorded and transcribed before being studied as part of a research project. Only the researcher will listen to the recording of the interview. All information provided will be treated confidentially, with the exception that you disclose a serious or previously unknown crime, or suggest that future harm to others or self-harm may occur. Such disclosures will be reported to your care team. All data will be kept confidential and anonymous in any future publication of the research and all names will be changed to ensure anonymity. Any other identifying details including place names will also be changed or omitted and any reference to you will be coded.

Any data collected will be kept in a locked filing cabinet and any electronic data will be stored on a password protected computer in an environment which is locked when not occupied. Data collected will be destroyed 10 years after the project is complete. The anonymised interview transcript may be shared with other researchers to optimise the use of good quality research data and to ensure peer scrutiny.

Consent process

Please take time to decide if you would like to participate in the research. If you decide to take part I will ask you to provide written consent using a consent form which I will read through with you. You will be given a copy of the signed consent form and participant information sheet to take away and a copy of these will also be retained in your personal file.

Research ethics

All research in the NHS is looked at by an independent group of people, called a Research Ethics Committee, to protect your interests. This study has been reviewed and given favourable opinion by the Greater Manchester East Research Ethics Committee.

Thank you for your interest and for considering taking part in the research.

Researcher
Katharine Harding
Division of Psychology,
London South Bank University
103 Borough Road, London SE1 0AA

Research supervisor Professor Paula Reavey Division of Psychology, London South Bank University 103 Borough Road, London SE1 0AA

If you have a concern about any aspect of this study, you should ask to speak to the researcher who will do their best to answer your questions. If you remain unhappy and wish to complain formally, you can do this by contacting Professor Nicola Thomas whose contact details are noted below. Email: nicola.thomas@lsbu.ac.uk Tel: 020 7815 8045

Support organisations

Samaritans website: www.samaritans.org.uk Tel: 116 123 Provides support to all experiencing difficulties or crisis. Mind website: www.mind.org.uk Tel: 0300 123 3393 Provides support to all experiencing mental health problems or distress.

Appendix B: Staff participant information sheet (page 1 of 2)

London South Bank University

Staff Participant Information Sheet (Study One and Study Two)

Public and private space within forensic mental health accommodation

You are being invited to take part in a research study. This study will contribute to my PhD studies in Psychology at London South Bank University. Taking part is entirely up to you and before you decide it is important for you to understand why the research is being done and what it will involve. This sheet explains the purpose of the study and provides information about what would be involved in taking part. I will read through this sheet with you and will answer any questions you may have. Please ask if anything is unclear.

This research aims to explore the everyday experiences of service users and staff in forensic mental health residential settings and I would be very interested to hear your thoughts about the place where you work. The study is looking to explore the ways in which your personal environment might affect or reflect how you feel day to day. The research will include other staff working in forensic mental health inpatient settings and supported housing in the community. Service users who live in these settings will also be asked to participate. In total, approximately 30 people will be included in the study. The study is open to people over the age of 18 and research will be taking place here during the coming weeks.

Do I have to take part?

No, you are under no obligation to participate.

What's involved?

You are being invited to take part in a study exploring the everyday experiences of forensic mental health service users and staff in residential environments. You are being invited to participate as you have personal experience of working in these environments and I am interested in hearing about your experience. If you'd like to take part, you will be asked to produce up to ten photos of the place where you currently work. These should reflect your typical everyday experiences and should capture the range of feelings you may hold towards the places, spaces or objects which you encounter in daily life. You should avoid taking photos of other people and photos including other people who could be recognised will be deleted. A digital camera will be provided for use in this study and I will print your images before we meet for an interview at a convenient time for you.

When we meet I will ask you to talk about the photos and I will ask you some questions in an interview which may last about 45 minutes. It is not anticipated that you will be at any disadvantage from participating. Some questions may invite you to reflect on personal and possibly sensitive subjects, however you can decline to answer any questions raised and you may also ask to stop the interview at any time. Afterwards I will also give you a short debrief about the study. Should anything arise during the research to cause you to feel concerned or upset, contact details for independent support organisations that you could contact for information or to speak with someone in confidence are provided at the end of this sheet.

There's no financial benefit, but you might find it interesting to contribute your experiences to research which aims to expand knowledge of the experience of forensic mental health accommodation, or to have the opportunity to discuss this topic with a receptive listener. At any point until completion of the PhD thesis, if you wish to withdraw entirely from the research you may do so without giving a reason. You may also request that any answers given in the interview are removed from the transcribed data or request that any images supplied are deleted or are not reproduced in publications of the research.

Appendix B: Staff participant information sheet (page 2 of 2)

Confidentiality & data protection

Interviews will be recorded and transcribed before being studied as part of a research project. Only the researcher will listen to the recording of the interview. All information provided will be treated confidentially, with the exception that you disclose a serious or previously unknown crime, or suggest that future harm to others or self-harm may occur. All data will be kept confidential and anonymous in any future publication of the research and all names will be changed to ensure anonymity. Any other identifying details including place names will also be changed or omitted and any reference to you will be coded.

Any data collected will be kept in a locked filing cabinet and any electronic data will be stored on a password protected computer in an environment which is locked when not occupied. Data collected will be destroyed 10 years after the project is complete. The anonymised interview transcript may be shared with other researchers to optimise the use of good quality research data and to ensure peer scrutiny.

Consent process

Please take time to decide if you would like to participate in the research. If you decide to take part I will ask you to provide written consent using a consent form which I will read through with you. You will be given a copy of the signed consent form and participant information sheet to take away.

Research ethics

All research in the NHS is looked at by an independent group of people, called a Research Ethics Committee, to protect your interests. This study has been reviewed and given favourable opinion by the Greater Manchester East Research Ethics Committee.

Thank you for your interest and for considering taking part in the research.

Researcher Katharine Harding Division of Psychology, London South Bank University 103 Borough Road, London SE1 0AA Research supervisor Professor Paula Reavey Division of Psychology, London South Bank University 103 Borough Road, London SE1 0AA

If you have a concern about any aspect of this study, you should ask to speak to the researcher who will do their best to answer your questions. If you remain unhappy and wish to complain formally, you can do this by contacting Professor Nicola Thomas whose contact details are noted below. Email: nicola.thomas@lsbu.ac.uk Tel: 020 7815 8045

Support organisations Samaritans website: www.samaritans.org.uk Tel: 116 123 Provides support to all experiencing difficulties or crisis. Mind website: www.mind.org.uk Tel: 0300 123 3393 Provides support to all experiencing mental health problems or distress.

Appendix C: Consent form

London South Bank University

Service User and Staff Consent Form (Study One and Study Two)

Public and private space within forensic mental health accommodation

Name of Researcher: Katharine Harding

			Please	
			initial box	
1.	I confirm that I have read the information sheet dated (version) for the above study. I have had the opportunity to consider the information, ask questions and ha had these answered satisfactorily.			
2.	I understand that my participation is voluntary and that I may refuse to answer any questions and I am free to withdraw at any time before the research is published, without giving any reason, without my medical care or legal rights being affected.			
3.	I have been informed about what the data collected will be used for, to whom it may be disclosed and for how long it will be retained.			
4.	I agree for the interview to be recorded digitally and transcribed.			
5.	I agree to having anonymised direct quotations from my interview used in any publications or presentations of this research.			
6.	I understand that the interview being carried out is confidential, unless I disclose a serious or previously unknown crime or indicate that harm may occur to myself or others.			
7.	I agree that the photographs or images which I supply may be reproduced and included in any publications or presentations of this research.			
8.	I agree to take part in th	e above study.		
Name of Participant		Date	Signature	
Name of Person		Date	Signature	

taking consent

When completed: 1 copy for participant; 1 copy for researcher site file; 1 (original) to be kept in medical notes

Version 2: 14.03.16

Appendix D: Service user interview schedule (Study one - group one)

Service User Interview Schedule (Study One - Group One)

Public and private space within forensic mental health accommodation

Introduction

- Could you begin by telling me a little bit about yourself? You can include any information that you'd like. (If the participant's age is not mentioned, ask if they would mind sharing).
- Looking at the set of images which you've produced, could you take me through them individually and tell me about each one? It would be interesting to hear about why you chose to make each image and any feelings you hold about them.

The interviews will be led primarily by the participant and will focus on the photographs produced as a means by which to elicit rich narratives from each service user around their experiences of space and place. The indicative list below is a prompt for the researcher from which certain questions may be drawn and threaded into the course of the dialogue as and when appropriate. This approach is supported by the research supervisor's previous experience of interviews with mental health service users and acknowledges that for reasons including medication, some participants may have limited capacity to maintain concentration.

Home

- Could you describe what you would consider to be a definition of home?
 - Could you describe whether you experience any sense of enjoyment or pleasure in the
- physical qualities of the place where you are living or from any of the objects within it?
 How important is choice to you in considering the details of your personal environment?
- How important is choice to you in considering the details of your personal environment?

Well-being

- Could you describe any ways in which your personal environment affects your well-being?
- At times when you might have experienced distress, could you describe any ways in which thinking about other places where you have lived, or places in the community has been reassuring?
- Are there any ways in which you feel that an awareness of order or disorder within the place where you live has been relevant to your experiences of well-being?

Transition

- How do you feel in your personal environment at the moment?
- Could you tell me about any expectations or concerns you have in thinking about where you might live in the future?
- Could you tell me about past experiences of moving between places you've lived before?

Privacy

- Could you describe any spaces here which you consider to be private?
- How important is privacy to your sense of well-being and for what reasons?
- Could you describe whether there are ways in which your relationships with other people are affected by the place where you live?
- Are there ways in which sharing the place where you live with other people is significant to your experience of home?

Conclusion

Finally, could you tell me something about why you were interested in taking part in this type of research and how did you find the interview?

Version 1: 25.01.16

Appendix E: Service user interview schedule (Study one - Group two)

Service User Interview Schedule (Study One - Group Two)

Public and private space within forensic mental health accommodation

Introduction

- Could you begin by telling me a little bit about yourself? You can include any information that you'd like. (If the participant's age is not mentioned, ask if they would mind sharing).
- Looking at the set of images which you've produced, could you take me through them individually and tell me about each one? It would be interesting to hear about why you chose to make each image and any feelings you hold about them.

The interviews will be led primarily by the participant and will focus on the photographs produced as a means by which to elicit rich narratives from each service user around their experiences of space and place. The indicative list below is a prompt for the researcher from which certain questions may be drawn and threaded into the course of the dialogue as and when appropriate. This approach is supported by the research supervisor's previous experience of interviews with mental health service users and acknowledges that for reasons including medication, some participants may have limited capacity to maintain concentration.

Home

- Could you describe what you would consider to be a definition of home?
 - Could you describe whether you experience any sense of enjoyment or pleasure in the
- physical qualities of the place where you are living or from any of the objects within it?
- How important is choice to you in considering the details of your personal environment?

Well-being

- Could you describe any ways in which your personal environment affects your well-being?
- At times when you might have experienced distress, could you describe any ways in which thinking about other places where you have lived, or places in the community has been reassuring?
- Are there any ways in which you feel that an awareness of order or disorder within the place where you live has been relevant to your experiences of well-being?

Transition

- How do you feel in your personal environment at the moment?
- Could you tell me about any expectations or concerns you have in thinking about where you might live in the future?
- Could you tell me about past experiences of moving between places you've lived before?

Privacy

- Could you describe any spaces here which you consider to be private?
- How important is privacy to your sense of well-being and for what reasons?
- Could you describe whether there are ways in which your relationships with other people are affected by the place where you live?
- Are there ways in which sharing the place where you live with other people is significant to your experience of home?

Conclusion

Finally, could you tell me something about why you were interested in taking part in this type of research and how did you find the interview?

Version 1: 25.01.16

Appendix F: Staff interview schedule

Staff Interview Schedule (Study One and Study Two)

Public and private space within forensic mental health accommodation

Introduction

- Could you begin by telling me a little bit about yourself? You can include any information that you'd like. (If the participant's age is not mentioned, ask if they would mind sharing).
- Looking at the set of images which you've produced, could you take me through them individually and tell me about each one? It would be interesting to hear about why you chose to make each image and any feelings you hold about them.

The interviews will be led primarily by the participant and will focus on the photographs produced as a means by which to elicit rich narratives from each member of staff around their experiences of space and place at work. The indicative list below contains questions which may be threaded into the course of the dialogue as and when appropriate.

Workplace

- Could you describe whether you experience any sense of enjoyment or pleasure in the physical qualities of the place where you work or from any of the objects within it?
- How important is a sense of choice to you in considering the details of your work environment?

Well-being

- Could you describe any ways in which your working environment affects your well-being?
- Are there any ways in which you feel that an awareness of order or disorder within the place where you work has been relevant to your experiences of well-being?

Privacy

- Could you describe any spaces here which you consider to be private?
- How important do you think privacy is to the experience of well-being here and for what reasons?
- Could you describe any ways in which you feel that your relationships with other people are affected by the place where you work?

Transition

- Could you describe any ways in which you feel that spaces within the housing project/inpatient setting and in the local community might assist or hinder service users' experience of transition e.g. transition between different secure inpatient settings, transition between inpatient settings and the community, or transition between high support and lower support accommodation in the community?

Conclusion

- Finally, could you tell me something about why you were interested in taking part in this type of research and how did you find the interview?

Version 1: 25.01.16

Appendix G: Service user participant debrief

London South Bank University

Service User Participant Debrief (Study One and Study Two)

Public and private space within forensic mental health accommodation

Thank you for taking the time to be interviewed for this research and I hope you may have found it interesting to take part. As mentioned in the information sheet, I am studying Psychology at London South Bank University. The interview and images you have provided will contribute to my PhD research project and possible future academic publications or conference presentations.

The interview will be transcribed and analysed in the context of previous research in this area. The current study aims to provide further understanding of the everyday experiences of UK forensic mental health service users and staff within inpatient and community residential settings. The research may potentially inform the management and development of services and increase understanding of how residential environments may affect the experience of well-being and the recovery process.

All the information you have provided will be made anonymous to ensure complete confidentiality. Some sections of the anonymised interview transcript may be quoted directly within publications of the research. If you wish to withdraw entirely from the research you may do so without giving a reason, at any point until completion of the PhD project, however, after that time, it would be impossible for the researcher to comply. Should you wish to withdraw from the research, your care will not be affected. You may also request that any answers given in the interview are removed from the transcribed data or request that any images supplied are deleted or are not reproduced in publications of the research.

Because the material discussed was personal to you, you may have started thinking about subjects or feelings which came up in the interview. Should anything arise during the research to cause you to feel concerned or upset, contact details for independent support organisations that you could contact for information or to speak with someone in confidence are provided at the end of this sheet. You should also speak with the staff where you live or your care coordinator if you feel affected by the research.

Thank you again for your participation in this research.

Researcher

Katharine Harding Division of Psychology, London South Bank University 103 Borough Road, London SE1 0AA Research supervisor Professor Paula Reavey Division of Psychology, London South Bank University 103 Borough Road, London SE1 0AA

If you have a concern about any aspect of this study, you should ask to speak to the researcher who will do their best to answer your questions. If you remain unhappy and wish to complain formally, you can do this by contacting Professor Nicola Thomas whose contact details are noted below. Email: nicola.thomas@lsbu.ac.uk Tel: 020 7815 8045

Support organisations

Samaritans website: www.samaritans.org.uk Tel: 116 123 Provides support to all experiencing difficulties or crisis. Mind website: www.mind.org.uk Tel: 0300 123 3393 Provides support to all experiencing mental health problems or distress.

Appendix H: Staff participant debrief

London South Bank University

Staff Participant Debrief (Study One and Study Two)

Public and private space within forensic mental health accommodation

Thank you for taking the time to be interviewed for this research and I hope you may have found it interesting to take part. As mentioned in the information sheet, I am studying Psychology at London South Bank University. The interview and images you have provided will contribute to my PhD research project and possible future academic publications or conference presentations.

The interview will be transcribed and analysed in the context of previous research in this area. The current study aims to provide further understanding of the everyday experiences of UK forensic mental health service users and staff within inpatient and community residential settings. The research may potentially inform the management and development of services and increase understanding of how residential environments may affect the experience of well-being and the recovery process.

All the information you have provided will be made anonymous to ensure complete confidentiality. Some sections of the anonymised interview transcript may be quoted directly within publications of the research. If you wish to withdraw entirely from the research you may do so without giving a reason, at any point until completion of the PhD thesis, however, after that time, it would be impossible for the researcher to comply. You may also request that any answers given in the interview are removed from the transcribed data or request that any images supplied are deleted or are not reproduced in publications of the research.

Because the material discussed was personal to you, you may have started thinking about subjects or feelings which came up in the interview. Should anything arise during the research to cause you to feel concerned or upset, contact details for independent support organisations that you could contact for information or to speak with someone in confidence are provided at the end of this sheet.

Thank you again for your participation in this research.

Researcher

Katharine Harding Division of Psychology, London South Bank University 103 Borough Road, London SE1 0AA Research supervisor Professor Paula Reavey Division of Psychology, London South Bank University 103 Borough Road, London SE1 0AA

If you have a concern about any aspect of this study, you should ask to speak to the researcher who will do their best to answer your questions. If you remain unhappy and wish to complain formally, you can do this by contacting Professor Nicola Thomas whose contact details are noted below. Email: nicola.thomas@lsbu.ac.uk Tel: 020 7815 8045

Support organisations

Samaritans website: www.samaritans.org.uk Tel: 116 123 Provides support to all experiencing difficulties or crisis. Mind website: www.mind.org.uk Tel: 0300 123 3393 Provides support to all experiencing mental health problems or distress.

Appendix I: Transcription conventions

Adapted from Coffey (2012b)

[??]	Inaudible word or section of speech.
(Laughs)	Description of participant or researcher action or body language
	during interview (e.g. laughter).
<u>Underlined</u>	Underlined text denotes the speaker's emphasis on a word or
	part of a word.
Um	Sounds uttered by the researcher or participants during the
	course of speech transcribed phonetically (e.g. 'Um' or 'err').
[]	Omission of part of the transcript.
[Name of ward]	Text in square brackets indicates information omitted from the
	transcript, or a note adding clarification to the meaning or
	context of speech.
Wor-	Hyphen appended to an incomplete word indicates a cut-off
	word or sound.
R:	Denotes researcher and other letters are used to indicate the
	first letter of the pseudonym for each participant.

Appendix J: Alice

Alice was a patient in a low secure environment at the time of the interview. Alice mentions being quite young when she was first admitted to hospital and prior to moving to her current ward, she also received care in another low secure environment. Alice is close to her family and whilst she has not been able to go back to the family home since being admitted to hospital, she is excited because her first home visit is due to take place soon. Alice's family live a significant distance away from the hospital and she expresses appreciation for the regular visits that she receives from her parents, despite the long distances that they need to travel.

Alice describes her enjoyment of cooking and baking. With staff support she shops for ingredients and uses the patient kitchen to cook meals and bake on a regular basis. Alice talks enthusiastically about her fondness for animals and how she values having the opportunity to spend time with therapeutic dogs that sometimes visit the ward. Alongside listening to music on the radio, Alice enjoys socialising in small groups and through the narrative she positions herself as a person who cherishes moments of fun enjoyed with other people. Although she describes having had a very difficult journey to this point, Alice brings humour into the stories she tells about the challenging times and events she has experienced. Her overall narrative has an underlying tone of positivity and she is enthusiastic about life beyond hospital, including her plans for studying towards a future career.

Alice's main storylines focus around: (1) links between space and her emotions, (2) enjoyment of spaces that facilitate social interaction, (3) enjoyment of outdoor spaces, (4) associations between spaces, possessions and selfhood, (5) negotiating relationships on the ward and (6) visualising future spaces.

In the interview Alice reflected on photographs of: (a) her bedroom, (b) quiet room, (c) visitors' room, (d) patient kitchen, (e) day area, (f) ward corridor, (g) crafts room, (h) patient lockers, (i) meeting room and (j) day area mural.

Appendix K: Darren

Darren is a patient on a medium secure psychiatric ward and he mentions having also spent time in prison previously. He is reserved when speaking and the overall tone of his narrative is quite flat throughout. Darren describes having had no belongings at the time of his admission to hospital and how he appreciates having since acquired some possessions, including a stereo. Although aspects of the ward design remind Darren of prison, he perceives the hospital to be a less institutional setting overall. In contrast to prison, he talks about how having his own bedroom key affords a valued sense of having freedom to come and go within the ward environment.

Darren describes his enjoyment of socialising with other patients during activities including cooking sessions in the therapy kitchen, or gardening in the hospital café courtyard. He also mentions how spending time in outdoor spaces, such as the café courtyard or a smoking point on the perimeter of the grounds is especially valued and offers a form of distraction from his current situation. Woodworking sessions in the off-ward workshop are described as enjoyable and able to provide an escape from boredom. The narrative also conveys a sense of Darren's pride in particular items that he has handcrafted and the skills gained in the workshop.

Darren's main storylines focus around: (1) spaces that provide a retreat and sense of sanctuary, (2) how possessions and objects are linked to self-identity and a sense of personal investment, (3) having an awareness of carceral symbolism within the physical environment, (4) appreciation of spaces that facilitate social interaction, (5) using space as a distraction or form of escape and (6) enjoyment of outdoor spaces and spaces outside the hospital building.

Darren talked about photographs taken of: (a) his bedroom, (b) bedroom corridor, (c) therapy kitchen, (d) off-ward communal café courtyard, (e) off-ward woodworking workshop, (f) door to medication room, (g) dining room servery hatch and (h) dining room.

Appendix L: Felicity

Felicity is a trainee occupational therapist working on a medium secure ward. She is enthusiastic about her role and her overall narrative tone is positive. Felicity describes experiencing a sense of enjoyment in observing patients gain confidence and skills through participation in therapeutic activities. She expresses appreciation for the items created by patients, particularly the artworks displayed on walls throughout the ward.

Felicity's accounts convey her sense of how the scale, temperature, atmosphere and appearance of the everyday ward environment can affect the quality of therapeutic relationships when working with patients in groups or individually. She especially values the non-institutional attributes of the therapy kitchen and perceives this space on the ward to have the homely appearance and realistic functionality of a typical domestic kitchen. Felicity highlights how the informal and impromptu conversations with patients that arise during individual cooking sessions in this space provide useful opportunities for patients to confide in staff about their worries or concerns. She also believes that patients benefit from spending time in spaces that are outside the ward itself, including the craft room. Felicity observes that 'off-ward' areas are an important resource for patients and feels that these spaces are particularly valued by people who don't have ground leave or access to the community.

Felicity's main storylines focus around: (1) appreciation for patient artwork, (2) therapeutic benefits of home-like and non-institutional spaces, (3) associations between spatial attributes and staff-patient relationships, (4) the importance of maintaining staff-patient boundaries and (5) how the scale or atmospheric conditions of spaces affect therapeutic work (6) benefits of off-ward spaces for patients and staff.

During the interview Felicity presented photographs of: (a) ward entrance with patients' handmade welcome signage above the door, (b) therapy kitchen, (c) small group room including patients' artwork on the walls, (d) off-ward therapy room for arts and crafts projects.

Appendix M: Gavin

Gavin is a trainee occupational therapist on a medium secure ward. It is his first experience of working in a mental healthcare environment and he speaks positively about his overall experiences in this setting. The tone of his narrative is upbeat and he expresses appreciation for the strong sense of teamwork and lack of hierarchy he perceives amongst colleagues within the multi-disciplinary team.

In his reflections about the hospital spaces, Gavin remarks on the large overall scale and proportions of the building and his sense that a spacious environment is important for patients, especially for people who are not able to leave the ward. Whilst noting that generous spaces can benefit patient and staff safety in medium secure settings, he also perceives that some parts of the hospital environment are overly large or disproportionate in relation to their function and some spaces have poor acoustics.

Gavin highlights the physical and psychological health benefits of spending time in outdoor spaces and feels it is positive that patients have access to external spaces via the dedicated courtyard and garden on the ward. He generally stays inside the building during his breaks, however, due to the effort of entering or leaving the building via the security airlock and the time that this takes.

Gavin's main storylines focus around: (1) appreciating positive relationships with colleagues and patients, (2) the proportions and scale of the building and interior spaces (3), having awareness of carceral symbolism within the built environment, (4) how sense of openness in the environment is linked to staff and patient experiences (5) the benefits of off-ward spaces for patient and staff well-being and (6) associations between the layout of spaces on the ward and relationships between and amongst patients and staff.

Gavin discussed photographs taken of: (a) café inside the hospital building, (b) dining room including food servery hatch, (c) ward courtyard, (d) patient lounge/ day area.

Appendix N: Jo

Jo is a clinical nurse leader on a low secure ward. She has worked with the same patients in an older building previously and notes how it took her time to adjust to the increased number of doors and locks in the current environment when compared with the previous ward. Whilst she perceives the original ward to have been a less carceral environment overall, Jo highlights how it was harder to monitor patient safety there due to the ward layout which presented a number of blind spots.

Jo appreciates the current ward for its spacious attributes, including large bedrooms, generous day areas and an outdoor garden. Throughout the interview she empathises with patients' experiences and reflects on how people might feel in spaces on the ward, including the lounge, which she feels to be like a waiting room with perimeter seating and the television positioned at high level. Jo highlights the value of having 'off-ward' spaces in which staff can take short breaks from the stresses of the ward. Accordingly, she notes that staff, patients and visitors typically enjoy spending time in a side room, which affords a sense of privacy and separation from the main ward. Jo's narrative indicates that she is dedicated to improving patient experiences and she comments on the importance of consulting with nurses when new hospital spaces are designed, as nursing staff will spend each day on the ward.

Jo's main storylines focus around: (1) how the generosity of space on the ward benefits patients, (2) the influence of aesthetics on mood, (3) comparisons between the current ward environment and an older ward, (4) ways in which the ward layout affects safety and risk management, (5) attempting to optimise patient privacy, (6) negotiating uncomfortable spaces, (7) benefits of off-ward spaces for patients, staff and visitors, (8) ward rules and safety protocols (9) and creating a home-like environment.

Jo reflected on the following photographs in the interview: (a) larger patient day area, (b) smaller patient day area, (c) ward courtyard garden, (d) seclusion room, (e) patient bedroom corridor and (f) circulation area outside the nursing office.

Appendix O: Karen

Karen is a healthcare assistant based on a low secure ward. She is a long-standing member of staff and has worked previously in both medium and low secure environments. Karen's accounts indicate that she is dedicated to her role and she is particularly enthusiastic about helping patients to prepare for independent living. Whilst she alludes to her perception that healthcare assistants' perspectives are not especially influential within the overall institution, she describes feeling valued by being invited to share her experiences in the research.

Karen's narrative conveys a strong sense of camaraderie amongst staff and she notes how all colleagues in the building support each other as a team when responding to incidents. She refers frequently to risk and safety and highlights how environmental features, including the use of electronic fobs instead of keys enable staff to respond quickly. Karen often works outside the ward and feels that patients typically confide in staff more in 'off-ward' spaces, particularly outdoor spaces, or places in the community.

Karen's main storylines focus around: (1) awareness of risk and ward safety, (2) how spatial experiences influence staff-patient relationships, (3) the importance of connections to nature and outdoor spaces, (4) having emotional responses to spaces and objects, (5) how physical features of the environment help staff and patients, (6) optimising spaces and resources in the environment to benefit patients and (7) perceptions of her professional role.

Karen discussed the following photographs: (a) CCTV monitors above seclusion room door, (b) seclusion room door, (c) seclusion room door hatch, (d) seclusion room interior, (e) extra care suite courtyard, (f) digital panel to alert staff to emergencies on other wards, (g) doors to unused kitchen in the building, (h) panel to enable staff to open locked doors inside the building using an electronic fob, (i) unused IT suite within the building, (j) sports hall within the building, (k) glass wall in off-ward circulation area, (l) fire extinguishers inside secure cabinets.

Appendix P: Sal

Sal is currently a patient in a medium secure ward environment. He has been detained in hospital for several years and mentions that he has also previously spent time in prison. Sal describes having been on a very difficult journey and talks about how he is excited to be moving on imminently to a low secure setting.

The overall tone of Sal's narrative is optimistic about the future and his eventual discharge into the community. Although he is looking forward to moving to a low secure ward in another hospital, he hasn't yet seen the place he'll be moving to and is waiting for confirmation on the timing. Sal describes how he will miss the current ward and his close friends but feels that he has been there too long and become too settled. A sense of him having been detained for several years within institutional environments is also conveyed through his very detailed accounts of the ward rules and protocols which recur throughout the interview.

Sal presents himself as a sociable person and he expresses his enjoyment of spending time with other patients in off-ward spaces, including playing football, cooking and gardening in the hospital courtyard garden or an allotment in the grounds.

Sal's main storylines focus around: (1) moving forward and looking ahead to his life beyond hospital, (2) the importance of social interaction and engaging in meaningful activities, (3) the enjoyment and satisfaction of perceiving a sense of choice and independence, (4) how spaces and activities can enable psychosocial transportation away from his current circumstances, (5) being aware of and respecting rules and routines on the ward and (6) the importance of experiencing a clean and tidy everyday environment.

In the interview Sal reflected on photographs of: (a) ward entrance corridor mural, (b) therapy kitchen, (c) off-ward communal café courtyard, (d) off-ward communal café, (e) dining room, (f) dining room servery hatch and (g) bedroom corridor.

Appendix Q: Vicky

Vicky is an assistant psychologist and currently working with patients on a medium secure ward. Through the narrative she presents herself as being dedicated to her role and she is enthusiastic about working therapeutically with patients. Vicky describes her experience of the nursing station as like being in a goldfish bowl and she is concerned that patients can become distressed or perceive that their needs are not being met when staff are busy on other tasks there. Managing her own stress can be difficult in a medium secure environment and she appreciates having access to an office area off the ward which affords a sense of respite from the ward when required and a space to complete administrative work without interruption.

Vicky is also mindful of how the ward environment is experienced by patients. She suggests that the atmospheric qualities of meeting rooms, including light levels, temperature and affordance of privacy are significant to patient experiences of participating in therapeutic sessions in those spaces. Vicky highlights how patients benefit from spending time in outdoor spaces and she uses the ward courtyard for mindfulness sessions, in addition to providing mindful walks in the hospital grounds for patients with ground leave. Whilst she went outside regularly during her breaks when based in a smaller building, in her current workplace Vicky now finds it easier to stay indoors due to the scale of the building and time needed to go through the airlocks.

Vicky's main storylines focus around: (1) the clinical qualities of the overall environment, (2) how the scale, layout or atmospheric attributes of spaces affect therapeutic work, (3) the value of off-ward spaces for staff and patients and (4) the therapeutic benefits of green spaces and spending time outdoors.

Vicky reflected on the following photographs in the interview: (a) her desk in an off-ward office area, (b) large meeting room on the ward, (c) smaller meeting room on the ward, (d) small office within the nursing station, (e) main nursing station area.

Appendix R: Sample participant storyline summary

Wendy: Main storylines

(1) Managing professional challenges and stress

- Feeling stressed by aspects of her professional role e.g.

- Sharing responsibility for several wards' staff + patients, plus her own, when holding bleep.
- Adjusting to working in a setting with wards split across several separate buildings.
- Dealing with the stress of coordinating staffing when limited staff resources are available.
- Managing patients' expectations on her own ward when staff are seconded to other wards.

(2) Negotiating uncomfortable spaces

- Experiencing affective discomfort in spaces on the ward e.g.
- Feeling exposed when working within the glazed nursing office.
- Having awareness of the ward atmosphere building up and becoming uncomfortably potent.
- Feeling a need to 'escape' from the enclosed ward to 'reset' during a long shift.
- Being disconnected from nature and natural light within the ward environment.
- Managing conflicting emotions associated with use of the seclusion room.

(3) Managing her own and colleagues' emotions within her professional role

- Managing her own emotions to maintain calm amongst colleagues during ward incidents.
- Concealing emotions to maintain a professional position amongst colleagues + patients.
- Managing emotions to express authority to patients e.g. during aggressive incidents.
- Having a 'prison guard' feeling associated with seclusion use affects nursing self-identity.

(4) Having emotional responses to spaces and objects

- Experiencing anxiety associated with certain objects on the ward e.g.
- The bleep equipment prompts a sense of unease in anticipation of challenging events.
- Clinic and medical equipment are reminder of a previous experience of patient emergency.
- Metal detector wand that can pick up contraband / weapons that might constitute a threat.

(5) Navigating boundaries and relationships with patients

- Role involves building patient relationships, but environment reinforces power differences.
- Awareness of power being expressed through physical environment + patient interactions.
- Needing to maintain professional boundaries and not become complacent amongst patients.
- Patient relationships involve both caring nursing and authoritarian 'policing' roles.